

2000 PNC PLAZA 500 WEST JEFFERSON STREET LOUISVILLE, KY 40202-2828 MAIN: (502) 333-6000 FAX: (502) 333-6099

RECEIVED

JUN 06 2012

PUBLIC SERVICE COMMISSION

J. WADE HENDRICKS DIRECT DIAL: (502)560-4227 DIRECT FAX: (502) 627-8727 wade.hendricks@skofirm.com

June 6, 2012

RECEIVED

VIA HAND-DELIVERY

JUN 06 2012

PUBLIC SERVICE

Mr. Jeff DeRouen Executive Director Kentucky Public Service Commission 211 Sower Boulevard P.O. Box 615 Frankfort, KY 40602-0615

Re: Application of Kentucky Utilities Company for an Order Authorizing the Issuance of Securities and the Assumption of Obligations

Dear Mr. DeRouen:

Enclosed for filing, please find the original and ten (10) copies of the Application of Kentucky Utilities Company for an Order Authorizing the Issuance of Securities and the Assumption of Obligations. Also enclosed are an original and ten (10) copies of Applicant's Petition for Confidential Protection and Motion for Deviation from Commission Rules. The confidential material which is the subject of the Petition and Motion is being filed under seal. An extra copy of both the Application and the Petition and Motion are enclosed to be filed stamped and returned to the undersigned.

Please do not hesitate to contact me should you have any questions or require additional information.

Very truly yours,

In Underton

J. Wade Hendricks

JWH/dvg Enclosures

cc: Dennis G. Howard, II

400001.143117/830208.1

COMMONWEALTH OF KENTUCKY

BEFORE THE PUBLIC SERVICE COMMISSION

)

)

)

)

)

JUN 06 2012

RECEIVED

PUBLIC SERVICE COMMISSION

In The Matter Of:

THE APPLICATION OF KENTUCKY UTILITIES COMPANY FOR AN ORDER AUTHORIZING THE ISSUANCE OF SECURITIES AND THE ASSUMPTION OF OBLIGATIONS

CASE NO. 2012-

APPLICATION

Kentucky Utilities Company ("KU" or the "Company") hereby requests, pursuant to KRS 278.300, that the Commission authorize KU to incur debt in the form of First Mortgage Bonds in a principal amount not to exceed \$300,000,000. KU further requests authority to increase the amount of its multi-year revolving line of credit by up to an additional \$100,000,000. In support of its Application, KU states as follows:

1. The Company's full name is Kentucky Utilities Company. The post office address of the Company is One Quality Street, Lexington, KY 40507. KU is a Kentucky and Virginia corporation, a utility as defined by KRS 278.010(3)(a) and, as of March 31, 2012, provides retail electric service to approximately 509,000 customers in seventy-seven counties in Kentucky. A description of KU's properties is set out in Exhibit 1 to this Application. A certified copy of the Company's Articles of Incorporation was filed with the Commission in Case No. 2010-00204 and is incorporated by reference herein pursuant to 807 KAR 5:001, Section 8(3).

2. KU obtains financing through numerous sources of capital, including the forms of debt that are the subject of this Application. KU does not assign specific financing to any particular project or use, and does not project-finance capital projects. All components of KU's

capital structure are used to fund capital expenditures. Thus, the uses cited below are general reasons for KU's need for debt financing, rather than projects to which specific financing will be assigned.

FIRST MORTGAGE BOND DEBT

3. KU requests, pursuant to KRS 278.300, that the Commission authorize it to incur additional long-term debt in the form of First Mortgage Bonds in a principal amount not to exceed \$300,000,000.

4. During 2012 and 2013, KU anticipates incurring up to approximately \$1.45 billion in construction costs. In Case No. 2011-00375¹ by Order dated May 3, 2012, the Commission granted KU a Certificate of Public Convenience and Necessity and a Site Compatibility Certificate to construct a new 640 MW combined cycle combustion turbine unit at the Cane Run Generating Station in Jefferson County, and to purchase three turbines in LaGrange, Kentucky. By Order dated December 15, 2011, in Case No. 2011-00161,² the Commission approved KU's 2011 Amended Environmental Compliance Plan and granted KU Certificates of Public Convenience and Necessity to Construct Particulate Matter Control Systems at the Company's Brown Unit 3 and at Ghent Units 1, 2, 3 and 4. By order dated December 23, 2009, in Case No. 2009-00197,³ the Commission among other actions, granted KU a Certificate of Public Convenience and Necessity to construct a new landfill at the Company's Ghent Station in Carroll County, Kentucky. The facilities which were the subjects of Case Nos. 2011-00375, 2011-00161 and 2009-00197 are described in greater detail in Exhibit 4

¹ Joint Application of Louisville Gas and Electric Company and Kentucky Utilities Company for a Certificate of Public Convenience and Necessity and Site Compatibility Certificate for the Construction of a Combined cycle Combustion Turbine of the Cane Run Generating Station and the Purchase of Existing Simple Cycle Combustion Turbine Facilities From Bluegrass Generation Company, LLC in Lexington, Kentucky.

² Application of Kentucky Utilities Company for Certificates of Public Convenience and Necessity and Approval of its 2011 Compliance Plan for Recovery by Environmental Surcharge.

³ Application of Kentucky Utilities Company for Certificates of Public Convenience and Necessity and Approval of Its 2009 Compliance Plan for Recovery by Environmental Surcharge.

and in the records of those cases. The costs of these projects are reflected in KU's capital budget which is attached as Exhibit 2.

5. The Company's Mortgage Indenture (the "Indenture") authorizes it to issue, from time to time, bonds ("First Mortgage Bonds") of one or more series, with each series having such date, maturity date(s), interest rate(s), and other terms as may be established by a supplemental indenture executed by the Company in connection with such series. All bonds issued under the Indenture would be equally and ratably secured by a first mortgage lien on substantially all of the Company's permanently fixed properties in Kentucky. A copy of the form of the Indenture has been previously filed with the Commission in Case No. 2010-00206.⁴

6. The First Mortgage Bonds may be sold at various times through the remainder of 2012 and 2013 in one or more underwritten public offerings, negotiated sales, or private placement transactions utilizing the proper documentation.

7. The First Mortgage Bonds of each series would be issued and secured by the Indenture as to be further supplemented and amended by a supplemental indenture creating the bonds of such series. Such supplemental indenture would set forth the terms and provisions of such series, including without limitation, the maturity date(s), interest rate(s), redemption provisions and other applicable terms. The First Mortgage Bonds of each series may be sold in one or more underwritten public offerings, negotiated sales, or private placement transactions utilizing the proper documentation. The price, maturity date(s), interest rate(s), and the redemption provisions, and other terms and provisions of each series of First Mortgage Bonds (including, in the event all or a portion of the First Mortgage Bonds bear a variable rate of interest, the method for determining the interest rates), would be determined on the basis of negotiations among KU and the underwriters or other purchasers of such First Mortgage Bonds.

⁴ Application of Kentucky Utilities Company for An Order Authorizing the Restructure and Refinancing of Unsecured Debt and the Assumption of Obligations and for Amendment of Existing Authority.

The amount of compensation to be paid to underwriters or purchasers for their services would not exceed one percent (1%) of the principal amount of the First Mortgage Bonds of the series to be sold. Based upon past experience with similar financings, KU estimates that issuance costs, excluding underwriting fees, would be approximately \$600,000.

8. Because of the historical spread between long-term fixed interest rates and shortterm rates, all or a portion of the bonds could be issued with an interest rate that fluctuates on a quarterly or semi-annual basis.

9. In connection with the issuance of First Mortgage Bonds, KU may enter into one or more interest rate hedging agreements (including an interest rate cap, swap, collar, or similar agreement, collectively, the "Hedging Facility") with a PPL affiliate company, or a bank or financial institution (the "Counterparty"). The Hedging Facility would be an interest rate agreement designed to allow KU to actively manage and limit its exposure to variable interest rates or to lower its overall borrowing costs on any fixed rate First Mortgage Bond. The estimated cost of the financing does not include the costs of any Hedging Facility which would be determined at the time of the hedge. However, based on current market conditions, the cost of a three-year hedge would be approximately 17 basis points (.17%). The Hedging Facility could also be used to lock in interest rates in advance of a debt issuance. Once again based on current market conditions, the Company could lock in current rates for six months for a cost of approximately 15 basis points (.15%).

10. The terms of each Hedging Facility will be negotiated by KU with the respective Counterparty and would be the most favorable terms that can be negotiated by the Company.

INCREASE IN AMOUNT OF MULTI-YEAR REVOLVING LINE OF CREDIT

11. KU requests authority to increase the amount of its multi-year revolving line of credit by up to an additional \$100,000,000 or, in the alternative, to enter into similar additional facilities not to exceed that amount.

12. By Order dated September 30, 2010, in Case No. 2010-00206, the Commission authorized KU to enter into one or more multi-year revolving credit facilities (the "Revolving Line of Credit") with one or more financial institutions in an aggregate amount not to exceed \$400,000,000. The Revolving Line of Credit replaced similar revolving credit facilities, originally authorized in Case No. 2007-00233, which by Order dated August 2, 2007, the Commission had found would, by allowing KU the ongoing ability to incur short-term debt from time to time, alleviate the time and cost of annually negotiating and renewing short-term debt arrangements.

13. KU subsequently entered into its Revolving Line of Credit with a termination date of December 31, 2014. Subsequently, in Case No. 2011-00307 KU requested, and by Order dated October 10, 2011, the Commission authorized, KU to extend the term of its revolving credit facilities through December 31, 2016. By doing so, KU was able to pay lower on-going commitment fees immediately and incur a lower upfront fee to extend the existing facility than if it waited until the term expired in 2014. The facility was amended in October 2011, and now matures on October 19, 2016. Amending the existing facility also significantly reduced the credit spread the Company pays when it borrows under the line of credit, and the commitment fee paid under the facility for undrawn amounts.

14. The Revolving Line of Credit provides KU with the opportunity to request that the maximum debt allowed under the credit facility be increased by \$100,000,000 to \$500,000,000. While the lenders are not obligated to increase the limit under the Revolving Line of Credit, KU believes that it is likely that the lenders will agree to do so. However, in the event

that the current lenders are unable or unwilling to increase the credit limit, KU proposes to obtain an additional revolving credit facility in the amount of \$100,000,000. KU anticipates that any new, additional revolving credit facility would be on similar terms as its current Revolving Line of Credit, including a term not to exceed five (5) years. Exhibit 3 shows the anticipated additional cost of increasing the credit limit under the existing credit facility, and alternatively, of entering into an additional revolving credit facility with a credit limit of \$100,000,000.

15. The additional credit under the revolving credit facility would be available for the same purposes for which existing credit is currently available. Loan proceeds could be used to provide short-term financing for KU's general funding needs, for example, general costs of operation, costs of KU's various construction programs or other general business purposes, until permanent or long-term financing can be arranged. In addition, the additional funds could be used to provide new or expanded liquidity or credit support for KU's other debt. For example, credit could be used to ensure that KU has readily available funds with which to repay commercial paper borrowings at their maturity.

16. No contracts have been made for the disposition of any of the securities which KU proposes to issue.

17. A redacted copy of a contract with respect to the projects noted in Paragraph 4 is attached as Exhibit 4. Filed concurrently with this Application is KU's Motion for Confidential Protection and for Deviation from Commission Rules.

18. KU shall, as soon as reasonably practicable after the issuance of each series of First Mortgage Bonds referred to herein, file with the Commission a statement setting forth the date or dates of issuance of the First Mortgage Bonds, the proceeds of such series of Bonds, the interest rates, costs or gains with any Hedging Facility and all fees and expenses associated with such issuance.

19. As soon as reasonably practicable after amendment of KU's Revolving Line of Credit to increase the debt allowed under such instrument, or, alternatively, entrance into a new and additional line of credit, KU shall file with the Commission a statement setting forth the date or dates of such action, as well as all fees and expenses.

20. Exhibit 5 to this Application contains the financial exhibit required by 807 KAR 5:001, Section 11(2)(a) and described by 807 KAR 5:001, Section 6. It also contains information required by 807 KAR 5:001, Section 11(2)(b).

21. Exhibit 6 to this Application is a certified copy of KU's Board of Directors' Resolution authorizing the issuance of the First Mortgage Bonds, the increase in the debt limit under the revolving credit facility, and the transactions related thereto as discussed in this Application.

22. Other requirements of the Commission's regulations concerning this Application, 807 KAR 5:001, Section 11, including (1)(b) regarding the amount and kind of bonds, etc., and 1(c) regarding the use to be made of the proceeds, have been supplied in the discussion above in Paragraphs 2 through 17 of this Application.

THEREFORE, Kentucky Utilities Company respectfully requests that the Commission enter its Order authorizing KU to issue securities in the form of First Mortgage Bonds in a total amount not to exceed \$300,000,000 and to increase the credit limit under its existing revolving line of credit by up to an additional \$100,000,000, or in the alternative, to enter into one or more new revolving credit facilities in addition to its existing credit facility provided that the total revolving credit limit increase, whether under the existing line of credit, under the new revolving credit facilities or in combination of both, shall not exceed an additional \$100,000,000 all as set forth in this Application. KU further requests that the Order of the Commission specifically include provisions stating: 1. KU is authorized to issue long-term debt in the form of First Mortgage Bonds in one or more series at one or more times during the remainder of 2012 through 2013, in an aggregate principal amount not to exceed \$300,000,000 in the manner set forth in its Application.

2. KU is authorized to execute an amendment to its existing revolving line of credit increasing the total principal amount of debt under said line or in the alternative, to enter into one or more new additional revolving credit facilities with a term not to exceed five (5) years, such increases in the existing line of credit and/or new facilities to not exceed, in total a combined additional aggregate principal amount of \$100,000,000.

3. KU is authorized to execute, deliver and perform the obligations of KU under all such agreements and documents as set out in its Application, and to perform the transactions contemplated by such agreements.

Dated: **forme 6**, 2012.

Respectfully submitted,

to know

Kendrick R. Riggs John Wade Hendricks Barry L. Dunn Stoll Keenon Ogden PLLC 2000 PNC Plaza 500 West Jefferson Street Louisville, Kentucky 40202 (502) 333-6000

Allyson K. Sturgeon Senior Corporate Attorney LG&E and KU Energy LLC 220 West Main Street Louisville, KY 40202 (502) 627-2088 Counsel for Kentucky Utilities Company

VERIFICATION

COMMONWEALTH OF KENTUCKY

COUNTY OF JEFFERSON

Daniel K. Arbough, being first duly sworn, deposes and says that he is Treasurer for Kentucky Utilities Company, that he has read the foregoing Application and knows the contents thereof, and that the same is true of his own knowledge, except as to matters which are therein stated on information or belief, and that as to these matters, he believes them to be true.

Daniel K. ARBOUGH

Subscribed and sworn before me this <u>for</u> day of <u>June</u>, 2012.

My Commission Expires: August 31, 2015

Kinde J. anduron NOTARY PUBLIC, STATE A

400001.143116/819186.4

KENTUCKY UTILITIES COMPANY (807 KAR 5:001, Section 11, Item 1 (a))

A DESCRIPTION OF APPLICANT'S PROPERTY, INCLUDING A STATEMENT OF THE NET ORIGINAL COST OF THE PROPERTY AND THE COST THEREOF TO APPLICANT

March 31, 2012

The applicant's generating, transmission and distribution systems described herein are calculated annually. As of December 31, 2011, the applicant owned 13 and operated 11 coal fired steam electric generating units having a total capacity of 3,355 Mw; a hydroelectric generating station having a total capacity of 24 Mw; and 17 gas/oil peaking units having a total capacity of 1,454 Mw.

The applicant's owned electric transmission system included 133 substations (55 of which are shared with the distribution system) with a total capacity of 13 million kVA and 4,078 miles of lines. The electric distribution system included 478 substations (55 of which are shared with the transmission system) with a transformer capacity of 7 million kVA, 14,112 miles of overhead lines, and 2,265 miles of underground conduit.

KU's service area includes an additional 11 miles of gas transmission pipeline providing gas supply to natural gas combustion turbine electrical generating units.

Other properties include office buildings, service centers, warehouses, garages and other structures and equipment.

The net original cost of the property and cost thereof to the applicant at March 31, 2012, was:

		Utility Plant	
Original Cost			
Production Plant	\$	4,189,773,098	
Distribution Plant		1,434,552,256	
Transmission Plant		667,945,984	
General Plant		140,094,552	
Intangible Plant		60,204,133	
Construction Work in Progress		345,238,438	
Total Plant at Original Cost	\$	6,837,808,461	
Less Reserve for Depreciation	<u></u>	2,052,590,028	*
Net Original Cost		4,785,218,433	

* Excludes \$367,106,725 related to cost of removal reserves that is not included in the reserve in the Financial Statements and Additional Information, but instead is included as a regulatory liability.

		KU								
	(\$ millions) Projected									
	2	012	2	013	2	014	2	015	2	016
Construction expenditures (a) Generating facilities (b)	\$	129	\$	177	\$	217	\$	173	\$	65
Distribution facilities	³ 129 78		•	95		86	-	103	•	100
Transmission facilities (c)		57		49		53		43		40
Environmental		379		453		411		233		51
Other		13		21		21		24		22
Total Construction Expenditures	\$	656	\$	795	\$	788	\$	576	\$	278

(a) Construction expenditures include AFUDC, which is not expected to be significant for the years 2012 through 2016.

١.,

(b) Includes approximately \$500 million of currently estimable costs related to replacement generation units due to EPA regulations not recoverable through the ECR mechanism. KU expects to recover these costs over a period equivalent to the related depreciable lives of the assets through future rate proceedings.

(c) Includes approximately \$30 million of currently estimable transmission costs related to replacement generation units. KU expects to recover these costs over a period equivalent to the related depreciable lives of the assets through future rate proceedings.

.

Estimated Cost of Increasing Current Credit Facility by \$100 Million

승규

125,000 50,000 ŝ ŝ 50,000 67,268 66,721 Total Cash Outlay (Aftertax) Present Value 66,178 250,168 မ ဗ ဗ ዓ ዓ 50,000 67,820 67,820 67,820 ••••• (57,180) (57,180) (57,180) Cost of Borrowing Amortization Upfront Cost Taxes \$ 50,000 Assumes Lenders are paid 5bp upfront fee to commit ფფ (125,000) (16,667) (141,667) 57,180 (84,486) 16,667 16,667 16,667 Undrawn ь θ 60 60 Undrawn ъ I Annual Utilization Fee: \$100 million x 0.125% აფ 125,000 125,000 125,000 40.363% 0.82% **ភ្**ភ្ Income Statement View Estimated Tax Rate Estimated Discount Rate Income Before Taxes Interest Expense Amortization Upfront fees: Net Income Taxes Year 2 Year 3 Year 0 Year 1 Costs: NΡV

Exhibit 3 Page 1 of 2

0 ~ N @

Estimated Cost of Additional New \$100 Million Credit Facility

Costs:

Assumes one time upfront fee of 25bps.	Estimated Legal fees	Total estimated upfront fees
Upfront fees:	-	

\$ 250,000
50,000
\$ 300,000

Ь

\$ 125,000

Annual Utilization Fee: \$100 million x 0.125%

Estimated Tax Rate 40.363 Estimated Discount Rate 0.82

3% 2%

-	
sent Value 300,000	33,906 33,631 33,357 400,894
Pres \$	ა ა ა ა
Cash Outlay (Aftertax) F 300,000	\$ 34,184 \$ 33,906 \$ 34,184 \$ 33,631 \$ 34,184 \$ 33,357 \$ 400,894
∫otal \$	የ የ የ
	\$ (90,816) \$ (90,816) \$ (90,816)
Upfront Cost Taxes \$ 300,000	
rtization	100,000 100,000 100,000
wn Amo	ម ម ម
Undrawn t of Borrowing Amortization	125,000 125,000 125,000
Cost o	ម ម ម
NPV	Year U Year 1 Year 2 Year 3

Income Statement View		
		Undrawn
Interest Exnense	ម	(125,000)
Amortization	ម	(100,000)
Income Refore Taxes	÷	(225,000)
Taxes	φ	90,816
Net Income	φ	(134,184)

.¢.х

Engineering, Procurement and Construction Agreement

Between Kentucky Utilities Company, as Owner

and

TIC – The Industrial Company, as Contractor

12/29/11

|2|30|11 Gr

TABLE OF CONTENTS

ARTIC	LE 1 DEFINITIONS	
1.1	Definitions	
1.2	Contract Interpretation	16
ARTIC	LE 2 EFFECTIVENESS	18
2.1	Effectiveness	18
2.2	Commitments Prior to Mobilization.	
	LE 3 GENERAL PROVISIONS	19
3.1	Intent of Contract Documents	19
3.1	Independent Contractor	19
3.3	Subcontracting	19
3.3 3.4	Specified Equipment Subcontractors	20
3.5	Certain Provisions in Subcontracts	20
3.6	Assignment of Subcontracts	
3.0 3.7	Subcontracts	
3.8	Inclusion; Order of Precedence	
	-	
	LE 4 CONTRACTOR'S RESPONSIBILITIES	21
4.1	The Work	
4.2	Sufficient Personnel	
4.3	Labor Matters	
4.4	Discipline and Protection	23
4.5	Supervision	23
4.6	Contractor's Key Personnel	
4.7	Design and Engineering	24
4.8	Quality Control	24
4.9	Training	25
4.10	Certain Contractor-Provided Items	
4.11	Equipment and Materials Subcontractor Presence	25
4.12	Current Records; As-Built Drawings	25
4.13	Transportation Costs	25
4.14	Operating and Maintenance Manuals	25 DE
4.15	Control of Work	20 26
4.16	Emergency Response	20
4.17	Local Conditions	
4.18	Job Site Conditions	
4.19	Witnessing and Inspection Rights	
4.20	Use of Site	
4.21	Compliance With Applicable Law Permits and Approvals	
4.22	Periodic Reports & Meetings	30
4.23	Ciences	31
4.24	SignageSpare Parts	31
4.25	Interference with Traffic	
4.26	Source and Expedition	30
4.27	Supplies and Facilities	11/
		2 martin A
	i	NU
		`

12/30/11 M

	ii (2/2A)	
		1
		11
7.6	Limitations of Liability	
7.5	Consequential Damages	
7.4	Payment	
7.3	Work Scheduling and Liquidated Damages	
7.2	Commercial Operation Delay; Liquidated Damages	
7.1	Unit Outage Day Liquidated Damages	
ARTIC	LE 7 LIQUIDATED DAMAGES & LIABILITY LIMITATIONS	
6.11	Contractor Requested Outage	
6.10	Maintenance Outage	
6.9	Final Completion	
6,8	Possession and Control	
6.7	Commercial Operation	
6.6	Re-Setting of Ghent CCR Transport System 42	
6.5	Performance Testing	
6.4	Project Schedule Update	
6.3	Tie-in and Mechanical Completion	
6.2	Turnover of the Systems	
6.1	Commencement	
ARTIC	LE 6 PROJECT SCHEDULE	
5.11	Contractor's Personnel	
5.10	System Needs	
5.9	Operating Personnel	
5.8	Rights Not Limited	
5.7	Right to Carry Out the Work	
5.6	Owner Permits	
5.5	Access and Inspection	
5.4	Right to Apply Monies	
5.3	Owner-Provided Items	
5.2	Owner's Review	
5.1	Key Personnel	
	LE 5 OWNER RIGHTS, DUTIES & OBLIGATIONS	
	*	
4.39	Documentation Requirements or Standards	
4.38	NERC Critical Infrastructure Security Management Controls	
4.38	No First of a Kind	
4.37	Administrative Facilities	
4.36	Covering	
4.35	Release	
4.34	Artifacts and Other Valuable Items	
4.33	Deliveries by Truck	
4.32	Permit Assistance	
4.31	Water Control	
4.30	Waste Disposal	
4.29	Cleaning Up	
4.28	Cutting and Patching	

ARTIC	LE 8 CONTRACTOR'S COMPENSATION	. 47
8.1	Contract Price	. 47
8.2	Taxes	. 47
8.3	Progress Payments	
8.4	Application For Payment	. 49
8.5	Payment of Substantiated Amount	. 49
8.6	Supporting Documentation	. 50
8.7	Withholding to Protect Owner from Loss	. 50
8.8	Final Payment	51
8.9	Disputed Applications For Payment	. 52
8.10	Payment of Subcontractors	
8.11	Retainage and Withholding	
8.12	Release of Retainage	
8.13	Överpayment	
8.14	Tax and Accounting Information	53
ARTIC	LE 9 EXCUSABLE EVENTS	53
9.1	Notice	
9.2	Rights Limited	
ADTIC	LE 10 CONTRACT CHANGES	
10.1	Owner-Initiated Changes	
10.1	Contractor Change Notice.	
10.2	Emergencies	
10.3	Resolution	
10.4	Contractor Caused Delays.	
	•	
	LE 11 TEST AND INSPECTIONS	
11.1	Testing	
11.2	Witnessing Tests and Inspection.	
11.3	Defects During Performance Guarantee Tests	
11.4	Retests	39
ARTICI	LE 12 CORRECTION OF WORK	
12.1	Correction of Work	59
12.2	Urgent Repairs	
12.3	Damage to Existing Facilities	60
ARTICI	LE 13 WARRANTY	Ġ0
13.1	Work Warranty	
13.2	Warranty Period	
13.3	Defects	
13.4	Subcontractor Warranties	
13.5	Primary Liability	
13.6	Defect Limitations	
13.7	Warranty Assistance	
13.8	Reasonable Access	
13.9	Root Cause Repairs	
13.10	Exclusivity of Warranties and Remedies	

12/30/11

2124/1n

ARTICI	LE 14 PROTECTION OF PERSONS AND PROPERTY	63
14.1	Safety Programs	63
14.2	Applicable Law	
14.3	Safety Precautions	63
14.4	Community Relations	
14.5	Security	
14.6	Dangerous Materials	
14.7	Safety Personnel	
14.8	Loading	
14.9	Notices to Owner	
14.10	Code of Business Conduct	
14.11	Hazards and Training	
14.12	Drug and Alcohol	
	LE 15 SEPARATE CONTRACTORS AND ACTIVITIES BY OWNER	
15.1	Separate Work	
15.2	Integration	
15.3	Coordination	
15.4	Use of Job Site	
15.5	Deficiency in Work of Owner and Separate Contractors	. 66
	E 16 INTELLECTUAL PROPERTY	
16.1	Ownership/License	
16.2	Indemnity Against Intellectual Property Infringement	
16.3	Contractor's Responsibility for Litigation	
16.4	Assistance	. 68
16.5	Injunction	
16.6	Contractor's Continuing Obligation	. 68
16.7	Limitations and Conditions	. 68
16.8	Availability of Intellectual Property	
ARTICL	E 17 REPRESENTATIONS AND WARRANTIES	. 69
17.1	Contractor	
17.2	Owner	
ADTICI	E 18 CONFIDENTIAL INFORMATION	
	Confidential Information	
18.2		
	Import or Export of Technical Data	
18.3	Third Party Proprietary Information	
18.4		
h	E 19 HAZARDOUS SUBSTANCES	
19.1	Hazaidous Substances	72
ARTICL	E 20 INDEMNIFICATION	76
20.1	Contractor's Indemnity	76
20.2	Owner's Indemnity	
20.3	Defense of Claims or Actions	
	Liens	
		I A I "
		2/2/1
		i IL
1	iv)

i

ARTIC	LE 21 INSURANCE	. 80
ARTIC	LE 22 TITLE & RISK OF LOSS	. 80
22.1	Transfer of Title	. 80
22.2	Title Warranty	
22.3	Risk of Loss	. 80
22.4	Contractor Tools	. 81
ARTIC	LE 23 DISPUTE RESOLUTION	. 81
23.1	Resolution	
23.2	Arbitration Proceedings	. 81
23.3	Continuation of Work	. 82
ARTICI	LE 24 TERMINATION	. 83
24.1	Termination for Convenience	. 83
24.2	Termination by Owner for Cause	. 84
24.3	Termination by Contractor for Cause	. 85
24.4	Suspension of the Work	. 86
ARTICI	LE 25 MISCELLANEOUS PROVISIONS	87
25.1	Governing Law	
25.2	Entire Agreement	. 87
25.3	Successors and Assigns	87
25.4	Contractual Relationship	
25.5	Notices	88
25.6	Rights Cumulative	
25.7	Incorporation by Reference	
25.8	No Waiver	
25.9	Agreed Rate	
25.10	Financing Cooperation	
25.11	Audit	
25.12	Survival	
25.13	No Third Party Beneficiaries	
25.14	Non-Recourse	•
25.15	Parent Guarantee	
25.16	Provisions Required by Law	
25.17	Severability	
25.18	Joint Effort	
25.19	Counterparts	
25.20	Approvals Not To Relieve Contractor	
25.21	Consultants	
25.22	Equal Employment Opportunity	93
25.23	Minority, Women, Disadvantaged and Local Business Enterprises	
25.24	Local Involvement; Reporting	
25.25	Patent Limitations	94

1,21,20/11/2

EXHIBITS

- Exhibit A Technical Specification 172150.70.0100 dated June 7, 2011
- Exhibit B Contract Price and Milestone Payment Schedule
- Exhibit C Basis of Bid
- Exhibit D Project Schedule and Key Dates
- Exhibit E Acceptable Equipment and Materials Suppliers
- Exhibit F Forms
 - F-1 Applications For Payment
 - F-2 Certificates
 - F-3 Change Orders
 - F-4 Integrated Project Schedule
 - F-5 Lien Waivers and General Release
 - F-6 NOT USED
 - F-7 Station Forms
 - F-8 Parent Guarantee
 - F-9 Second Tier Procurement Program
 - F-10 Letter of Credit
- Exhibit G Guarantees and Performance Guarantee Test Protocol
- Exhibit H Health and Safety Requirements; Passport Program
- Exhibit I Insurance
- Exhibit J NOT USED
- Exhibit K Key Personnel
- Exhibit L NOT USED
- Exhibit M Meetings and Progress Reports
- Exhibit N NOT USED
- Exhibit O Owner Provided Items
- Exhibit P Permits
- Exhibit Q Reference Drawings
- Exhibit R Rates
- Exhibit S Site
- Exhibit T Options
- Exhibit U Training
- Exhibit V Operating and Maintenance Manuals
- Exhibit W Work Breakdown Structure
- Exhibit X Submittals, Review, and Hold Points
- Exhibit Y Ghent CCR Transport Equipment Contracts
- Exhibit Z Assignment and Assumption Agreements

12/30/11

ENGINEERING, PROCUREMENT AND CONSTRUCTION AGREEMENT

This Engineering, Procurement and Construction Agreement ("Agreement") is entered into as of the 30th day of December 2011 ("Effective Date") by and between Kentucky Utilities Company, a Kentucky corporation ("Owner"), and TIC – The Industrial Company, a Delaware corporation ("Contractor").

RECITALS

WHEREAS, Owner desires to enter into an agreement with a qualified contractor to design, engineer, procure, construct, start-up, commission, and test the Ghent CCR Transport System (as hereinafter defined) on a lump-sum turnkey basis as set forth in this Agreement;

WHEREAS, Contractor represents that it is qualified to design, engineer, procure, construct, start-up, commission and test the Ghent CCR Transport System and desires to perform all work and services in connection therewith on a lump-sum turnkey basis in accordance with the requirements and provisions of this Agreement; and

WHEREAS, Contractor agrees to perform the Work as more specifically described in this Agreement.

NOW, THEREFORE, in consideration of the foregoing and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties, intending to be legally bound, do hereby agree as follows:

ARTICLE 1

DEFINITIONS

1.1 Definitions. The following terms have the respective meanings specified in this Article 1 when capitalized and used in this Agreement or in any notice delivered under or in respect of this Agreement.

"Affiliate" means any Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, Owner or Contractor.

"Agreed Rate" has the meaning set forth in Section 25.9.

"Agreement" has the meaning set forth in the first paragraph hereof and includes the body of this Agreement (the "Body of this Agreement") and all Exhibits herein referenced, as amended, modified or supplemented from time to time.

"Applicable Credits" has the meaning set forth in Section 25.11.3.

"Applicable Law" means any applicable statute, law (including common law), rule, treaty, regulation, code, ordinance, approval, interpretation, injunction, judgment, decree, writ, order or the like, when issued, enacted, or promulgated by a Governmental Authority and interpretations thereof by a Governmental Authority.

121321M

"Application For Payment" means a written request by Contractor to Owner for payment, completed in the form required by Exhibit F-1 together with the documentation required by Section 8.6.

"Arbitration Range" has the meaning set forth in Section 23.1.

"As-Built Drawings" means: (i) all drawings prepared in the performance of the Work that are "issued for construction" by Contractor and (ii) all of the drawings specified by Owner in Exhibit Q, in each case as modified and updated to accurately show the final actual design and construction of the Work upon Final Completion.

"Business Day" means any Day other than a Saturday, Sunday, or a holiday observed by the United States federal government, the Commonwealth of Kentucky, or the Owner.

"CCR" means Coal Combustion Residuals.

"Certificate" means the applicable Certificate of Model Demonstration, Tie-in, Mechanical Completion, Commercial Operation, or Final Completion, as the case may be.

"Certificate of Commercial Operation" will be in the form set forth in Exhibit F-2.

"Certificate of Final Completion" will be in the form set forth in Exhibit F-2.

"Certificate of Mechanical Completion" will be in the form set forth in Exhibit F-2.

"Certificate of Model Demonstration" will be in the form set forth in Exhibit F-2.

"Certificate of Tie-in" will be in the form set forth in Exhibit F-2.

"Change in Law" means an amendment, modification, or change of Applicable Law enacted, adopted, or promulgated after the Effective Date by a Governmental Authority. A change in Applicable Law with respect to any of the following will not constitute a Change in Law hereunder: (i) taxes or levies assessed on the basis of Contractor's net income or net profits; (ii) Contractor Taxes; or (iii) taxes, levies or withholdings that vary the compensation, benefits or amounts to be paid to, on behalf of or on account of Contractor's or its Subcontractors' employees; or (iv) changes in Applicable Law enacted, published or issued before the Effective Date whether or not such changes became effective after the Effective Date of this Agreement.

"Change Order" has the meaning set forth in Section 10.1.3.

"Change Order Request" has the meaning set forth in Section 10.1.1.

"Change(s)" has the meaning set forth in Section 10.1.1.

"Claim Notice" has the meaning set forth in Section 20.3.1.

"Claims" has the meaning set forth in Section 20.1.

, 2/2/1"

"Climatic Conditions" means hurricane, fifty (50) year or greater flood and other severe climatic conditions that are similarly unusual.

"Codes and Standards" means the most recent edition of the codes, standards, and guidelines applicable to the Work, including those listed in Exhibit A.

"Commercial Operation" will have been achieved when all of the following have occurred: (i) Mechanical Completion has occurred; (ii) the Work is complete except Punch List Items; (iii) all Performance Guarantee Tests have been successfully completed in accordance with this Agreement, and Contractor has completed making system adjustments identified during the start-up and testing process which are necessary to permit the Ghent CCR Transport System to operate as specified in the Technical Specification; (iv) all Performance Guarantees have been simultaneously achieved in a single Performance Guarantee Test; (v) all obligations of Contractor expressly required to have been performed as of the Commercial Operation Date have been properly discharged; (vi) the Ghent CCR Transport System and all Units are capable of being operated in the normal course of business up to individual full unit generating capacity; (vii) Owner has received from Contractor all Permits (including Permits (other than Owner Permits) necessary to allow transfer of care, custody, and control of the Ghent CCR Transport System to Owner), all of which shall be valid and in full force and effect; (viii) levels of Consumables associated with the Ghent CCR Transport System are fully charged; (ix) final versions of the Operating and Maintenance Manuals approved by Owner have been delivered; (x) currently marked drawings showing as-built conditions or the As-Built Drawings (current as of Commercial Operation) have been delivered; (xi) other submittals required to be submitted prior to or as of Commercial Operation; (xii) all Special Tools have been delivered to Owner; and (xiii) Owner has executed the Certificate of Commercial Operation.

"Commercial Operation Date" means the date on which Commercial Operation is achieved.

"Component" means any and all Systems, subsystems, subassemblies, Equipment, Materials, spare parts, and every item of whatever nature, including all documentation related thereto, connected with the Work performed or provided by Contractor or its Subcontractors under this Agreement that is permanently incorporated into the Ghent CCR Transport System, provided as part of the Work, and retained by Owner following Final Completion.

"Computer Program(s)" means a sequence of instructions, data, or equations in any form, and explanations thereof, intended to cause a computer, a control data processor or the like to perform any kind of operation, which is provided by Contractor or its Subcontractors under this Agreement.

"Confidential Information" has the meaning set forth in Section 18.1.

"Connect" means the provision by Contractor of all labor, Materials, Consumables, and Construction Aids to effect the connection of the Ghent CCR Transport System at the Terminal Points.

"Construction Aids" means all equipment (including construction equipment), apparatus, tools, supplies, construction tools, support services, field office equipment, supplies, $\sum_{n=1}^{\infty}$

12/30/11 a

structures, apparatus, form lumber, protective fencing, and other goods and items that are required to construct, clean, commission, or test the Ghent CCR Transport System, but which are not incorporated into the Work and retained by Owner.

"Consumables" means items such as compressed air or gases, chemicals, oils, lubricants, cleaning materials, demineralized water, valve packing, lamps, light bulbs, gaskets, fuel filters and comparable items which, by normal industry practices, are considered consumables and are replaced on a regular basis, required for cleaning, preparing, or completing the Work or are required for the proper operation of the Equipment.

"Contract Price" has the meaning set forth in Section 8.1.

"Contractor" has the meaning set forth in the preamble of this Agreement.

"Contractor Change Notice" has the meaning set forth in Section 10.2.

"Contractor Default" has the meaning set forth in Section 24.2.1.

"Contractor Indemnitees" has the meaning set forth in Section 20.2.

"Contractor Response" has the meaning set forth in Section 10.1.2.

"Contractor Taxes" has the meaning set forth in Section 8.2.1,

"Contractor's Representative" means the individual designated by Contractor pursuant to Section 4.6.

"Day" means a calendar day, including Saturdays, Sundays and legal holidays, except that, in the event that a payment obligation to be performed under this Agreement falls due on a calendar day that is not a Business Day, the payment obligation shall be deemed due on the next Business Day thereafter.

"DBE" means Disadvantaged Business Enterprise.

"Defect" (and derivative forms thereof, e.g., "Defective") has the meaning set forth in Section 13.3.

"Design" (or any derivation thereof) means all design, calculation, and engineering products or services and the conduct thereof, including all preliminary and detailed design of and associated with the Work (including the manner in which the Ghent CCR Transport System is integrated into the Units).

"Design Documents" has the meaning set forth in Section 4.7.

"Dispute" has the meaning specified in Section 23.1.

"Effective Date" has the meaning set forth in the first paragraph of this Agreement.

(212A/1)

"Emergency Notification List" means a list of Owner personnel, with associated contact information, that sets forth the individuals to be notified first in the event of an emergency involving public health or safety, including environmental harm, or material damage to property, as such list is amended by Owner and provided to Contractor from time to time.

"Environmental Action" has the meaning set forth in Section 19.1.1(ii).

"Equipment" means any product, including without limitation any product or material supplied by a Glient CCR Transport Equipment Supplier, that: (i) is to be incorporated into the Ghent CCR Transport System; (ii) is an assembly of operational and/or non-operational parts, whether motorized or manually operated; and (iii) requires service connections, such as wiring, piping, or other process connections.

"Excusable Events" means the following events to the extent that such event materially and adversely (x) affects the critical path of the Work, (y) results in an increase in Contractor's cost of performing the Work, or (z) otherwise materially and adversely affects Contractor's performance hereunder:



12/30/11 52



"Exhibits" means all of the exhibits referenced in this Agreement.

"Existing Facilities" means structures, installations, roadways, walkways, natural features, and the Units, and auxiliary and support facilities located on the Ghent Generating Station Site, including all existing facilities used in connection with the generation of electricity; the transportation, handling or storage of Fuel; or the transmission of electricity.

"Existing Facility Requirements" has the meaning set forth in Section 4.20.

"Final Completion" will be deemed to have occurred when all of the following have occurred in respect of the Work: (i) Commercial Operation has been achieved; (ii) all items identified on the Punch List have been completed; (iii) all As-Built Drawings, Information, Design Documents, Operating and Maintenance Manuals, submittals and other documents required to be delivered to Owner hereunder have been delivered; (iv) all liquidated damages for which Contractor is liable pursuant to Article 7 and other amounts owed by Contractor to Owner under this Agreement, if any, have been paid to Owner; (v) all Work has been completed other than Work and other obligations that require future performance (e.g., warranty Work); and (vi) Owner has issued the Certificate of Final Completion.

"Final Completion Date" means the date on which Contractor successfully achieves Final Completion.

"Financing" means any form of construction, interim, long-term debt, lease, tax-exempt, recourse, non-recourse, equity or other form of funding, or refinancing that Owner or an Affiliate of Owner obtains, utilizes or attempts to obtain or utilize in connection with such Owner's obligations hereunder.

"Financing Parties" means any Person that provides Financing, or credit support for Financing, any trustee(s) acting in connection therewith, and their respective successors and assigns.

"Fly Ash Tie-in" means, for each Unit, the Tie-In of the Fly Ash System to that Unit.

"Fly Ash Tie-in Outage" means, for each Unit, the scheduled outage for Fly Ash Tie-In for that Unit, during which time Contractor must complete Tie-in (the time and date for such outage and the period during which Contractor may perform Work is set forth in Exhibit D as such time and date may be adjusted pursuant to this Agreement).

"Fly Ash Tie-in Start Date" means the date set forth in Exhibit D when Contractor may commence the Work to achieve Fly Ash Tie-in for each of the Units as such date may be adjusted pursuant to this Agreement.

"Force Majeure" means any condition, event, or circumstance, including the examples set forth below, but only if, and to the extent: (i) such condition, event, or circumstance is not within the reasonable control of the Party affected; (ii) such condition, event or circumstance, despite the exercise of reasonable diligence, could not be prevented, avoided or removed by such Party; (iii) such condition, event, or circumstance has a material adverse effect on the ability of the affected Party to fulfill its obligations under this Agreement; (iv) the affected Party has taken all reasonable precautions, due care and reasonable alternative measures in order to avoid the effect of such condition, event, or circumstance on the affected Party's ability to fulfill its obligations under this Agreement and to mitigate the consequences thereof; and (v) such condition, event, or circumstance is not the result of any failure of such Party to perform any of its obligations under this Agreement. By way of example, such events, conditions and circumstances shall include war, rebellion, sabotage, civil strife, insurrection, public disorder, Climatic Conditions, earthquake, quarantine, acts of terrorism, industry-wide or national strikes,

12/30/, Si

2 2 à

and Changes in Law. Notwithstanding the foregoing, Force Majeure shall not include the following events, conditions or circumstances:

- (i) late delivery of Equipment, Materials, Consumables, or Construction Aids required for the Work whether caused by congestion at a Subcontractor's plant or elsewhere, oversold market conditions, inefficiencies, transportation delays or other similar circumstances, except where such late delivery is caused by a Force Majeure event;
- (ii) shortages of supervisors, labor, Equipment, Materials, Consumables, or Construction Aids, except where such shortage is caused by a Force Majeure event;
- (iii) late performance as a consequence of any violation of Applicable Law or decisions of a Governmental Authority related to the conduct of Contractor's or any Subcontractor's business, including insolvency or any delay related to the economic, commercial or labor circumstances of Contractor, any Subcontractor or other business conducted by Contractor or any Subcontractor;
- (iv) breakdown, loss, or damage to or theft of Equipment, Materials, Consumables, or Construction Aids except where such breakdown, loss, or damage is directly due to the occurrence of an independent condition, event, or circumstance described in and meeting the conditions of the first full paragraph of this definition;
- (v) failure of a Party to pay amounts due and owing under this Agreement;
- (vi) strikes or other labor disturbances affecting Contractor or any of its Subcontractors, except as expressly set forth in the examples described in the first full paragraph of this definition and except with respect to strikes at the Ghent Generating Station that are not against Contractor or any of its Subcontractors but which result in Contractor or any of its Subcontractors being physically blocked from access to the Job Site;
- (vii) increased costs of the Work, general economic or industry conditions; or
- (viii) weather conditions other than Climatic Conditions.

"Fuel" means coal, fuel oil, or natural gas, as applicable.

"Ghent CCR Transport Equipment Contracts" means the agreements between Owner and each of the individual Ghent CCR Transport Equipment Supplier as listed on Exhibit Y.

2/2/1

"Ghent CCR Transport Equipment Supplier" means the equipment provider for one or more of the following four (4) major equipment packages:

- FGD Dewatering System
- Pipe Conveyor
- Bottom Ash Dewatering System
- Fly Ash Conveying System.

"Ghent CCR Transport System" means the Ghent Coal Combustion Residual Transport System as specified in Exhibit A.

"Ghent Generating Station Site" means the site, as more particularly described in Exhibit S.

"Governmental Authority" means any federal, state, county, regional, city, parish or local government body, agency, authority, branch, department, arbitrator, court or any subdivision, instrumentality or agency thereof, having, or claiming, a regulatory interest in, or jurisdiction over, the Work (or any portion thereof), the Ghent CCR Transport System, the Units, the Job Site, this Agreement or one or more of the Parties.

"Guaranteed Dust Emissions" has the meaning assigned in Exhibit G.

"Guaranteed Commercial Operation Date" means which date may be adjusted pursuant to the terms and conditions of this Agreement.

"Guaranteed Final Completion Date" means one hundred and eighty-three (183) Days following the Guaranteed Commercial Operation Date.

"Guaranteed Sound Emissions" has the meaning assigned in Exhibit G.

"Guarantor" means TIC Holdings, Inc.

"Hazardous Substance" means: (i) any chemicals, materials, substances, or wastes which are now or hereafter defined as or included in the definition of "hazardous substance," "hazardous material," "hazardous waste," "solid waste," "toxic substance," "extremely hazardous substance," "pollutant," "contaminant," or words of similar import under any Applicable Law; (ii) any petroleum, petroleum products (including crude oil or any fraction thereof), natural gas, natural gas liquids, liquefied natural gas, or synthetic gas useable for fuel (or mixtures of natural gas and such synthetic gas), or oil and gas exploration or production waste, polychlorinated biphenyls, asbestos-containing materials, mercury, urea formaldehyde insulation, radioactivity, and lead-based paints, or any other substance that has been contaminated, polluted or made toxic; and (iii) any other chemical, material, substances, waste, or mixture thereof which is prohibited, limited, or regulated pursuant to, or that could reasonably be expected to give rise to liability under, any Applicable Law.

"Hazardous Substance Management Plan" has the meaning set forth in Section 4.20(vii).

2221/1

12/30/11

"Hold Point" means an inspection point with respect to which Contractor or its Subcontractors may not proceed further with respect to the Work unless and until Owner has either (i) inspected the applicable Work and authorized Contractor (including its Subcontractors) to proceed, or (ii) waived such inspection in writing. Hold Points include those points designated in Exhibit X, and those reasonably designated by Owner from time to time by formal written notice to Contractor.

"Improvements" means inventions, ideas, designs, concepts, discoveries, techniques, works, processes, formulas, or modifications (whether or not patentable, copyrightable, or otherwise protectable as intellectual property) conceived, reduced to practice, developed or acquired by any Party during the term of this Agreement and related directly to the Ghent CCR Transport System. Improvements also include modifications and derivative works of any copyrightable works related to any of the foregoing.

"Indemnified Parties" has the meaning set forth in Section 20.3.1.

"Indemnifying Parties" has the meaning set forth in Section 20.3.1.

"Information" means all drawings; documents; manuals; training materials; Computer Programs; operating, maintenance, and other guidelines and procedures; and Design, Design Documents and other data, trade secrets, and information used or supplied by Contractor, whether directly itself or indirectly through Subcontractors, whether paper or electronic media or format, in performance of this Agreement which would be reasonably useful or necessary in Owner's operation, maintenance, repair, personnel training, modification, or use of the Ghent CCR Transport System.

"Insolvency Event" means, with respect to the Person: (i) the Person's (a) failure to generally pay its debts as they become due, (b) admission in writing of its inability to pay its debts as they become due or (c) making a general assignment for the benefit of creditors; (ii) any proceeding being instituted by or against the Person seeking: (a) to adjudicate it as bankrupt or insolvent, (b) liquidation, winding up, reorganization, arrangement, adjustment, protection, relief or composition of it or its debts under any Applicable Law relating to bankruptcy, insolvency, reorganization or relief of debtors or (c) the entry of an order for relief or the appointment of a receiver, trustee, custodian or other similar official for it or for any substantial part of its property and, in the case of any such proceeding instituted against the person, either such proceeding remains undismissed for a period of thirty (30) Days or any of the actions sought in such proceeding occur; or (iii) the Person's taking any action to authorize any of the actions set forth above in this definition.

"Intellectual Property" means all patents, patent applications, copyrights, trade secrets and all other intellectual property rights.

"Job Site" means that portion of the Ghent Generating Station Site on which the Ghent CCR Transport System will be constructed, including areas for parking, storage, laydown, and administrative facilities, as more particularly described in Exhibit S hereto.

"Landfill Site" means the proposed location for Ghent Generating Station byproduct disposal. The location as set forth in Exhibit S.

2 rts

"LC" has the meaning set forth in Section 25.23.

"Liabilities" has the meaning set forth in Section 20.1.

"Lien Indemnitees" has the meaning set forth in Section 20.4.

"Liens" has the meaning specified in Section 20.4.

"Local Hires" means workers hired as a part of Contractor's direct craft work force whose primary residences are located within: (i) Carroll County, (ii) the Commonwealth of Kentucky, or (iii) the Standard Metropolitan Statistical Area (as defined by the U.S. Census Bureau) for each of Louisville, Kentucky; Evansville, Indiana; or Cincinnati, Ohio.

"Local Spend" means expenditures with firms whose primary operations are within: (i) Carroll County, (ii) the Commonwealth of Kentucky, or (iii) the Standard Metropolitan Statistical Area (as defined by the U.S. Census Bureau) for each of Louisville, Kentucky; Evansville, Indiana; or Cincinnati, Ohio.

"Major Subcontractor" means a Subcontractor providing labor, Materials and/or Equipment in relation to the Work under this Agreement which has a value

"Marshaling Requirement" has the meaning set forth in Section 6.3.2.

"Materials" means any products, supplies, bulks, materials, logic, or Computer Programs that are to be incorporated into the Ghent CCR Transport System or the Unit as part of the Work, whether or not substantially shaped, cut, worked, mixed, finished, refined or otherwise fabricated or processed, and which are not items of Equipment or Consumables.

"MBE" means Minority Business Enterprise.

"MBE Spend" means expenditures with MBEs certified by a recognized certification entity, including National Minority Supplier Development Council (NMSDC and their regional affiliates), Louisville Metropolitan Sewer District, Louisville Metro Human Relations Commission, Department of Transportation, Small Business Administration, and other states' agencies as approved by Owner.

"Mechanical Completion" means when, with respect to the Work, all of the following have occurred: (i) all Equipment and Materials has been furnished and installed in accordance with Exhibit A and manufacturers' requirements (and in a manner that does not void any warranty) and the terms of this Agreement and, as appropriate, checked for alignment, lubrication, rotation, and hydrostatic and pneumatic pressure integrity; (ii) Turnover Acknowledgement of all Systems has been achieved; (iii) all Systems have been installed, cleaned and statically tested and cold commissioning has been completed; (iv) all initial tests have been completed and all instruments have been calibrated; (v) all Tie-ins have occurred; (vi) and the Units as modified by the Ghent CCR Transport System are capable of safe operation; (vii) the Ghent CCR Transport System has been properly integrated into individual Units (physically and electronically) (viii) the Ghent CCR Transport System is ready to commence

12/30/11

commissioning, testing and integrated operations without the use of temporary equipment or installations; (ix) an initial Punch List has been established and mutually agreed upon by Owner and Contractor; (x) the Performance Guarantee Test Procedures have been submitted to and accepted by Owner; (xi) the training required by Section 4.9 has been completed; and (xii) Owner has executed the Certificate of Mechanical Completion.

"Milestone Payment Schedule" means the schedule of milestone values as set forth in **Exhibit B**, as it may be adjusted in accordance with this Agreement.

"Modification" means: (i) a written amendment to this Agreement signed by all Parties; (ii) a Change Order; or (iii) an Owner Authorization.

"MSDSs" has the meaning set forth in Section 19.1.2.

"NERC" means the North American Electric Reliability Corporation, a reliability organization responsible for the oversight of the regional reliability organizations established to ensure the reliability and stability of the regions.

"NERC Requirements" has the meaning set forth in Section 4.39.

"Notice" has the meaning set forth in Section 23.2.1.

"Owner" has the meaning set forth in the first paragraph of this Agreement.

"Owner Authorization" has the meaning set forth in Section 10.1.4.

"Owner Default" has the meaning set forth in Section 24.3.1.

"Owner Engineer" means a third party engineering firm selected by Owner.

"Owner Indemnitees" has the meaning set forth in Section 20.1.

"Owner Permits" means Permits for which Owner is responsible hereunder.

"Owner Representative" has the meaning set forth in Section 5.1.

"Owner Response Period" has the meaning set forth in Section 10.1.3,

"Owner Review Period" has the meaning set forth in Section 6.2.2.

"Parent Guarantee" means a guarantee in the form set forth in Exhibit F-8.

"Party" or "Parties" means either Owner, Contractor or both of them, as the context or the usage of such term may require.

"Patent Rights" mean any and all forms of patents issued or granted anywhere in the world arising from or related to the Work, all renewals and extensions thereof, and all applications for such patents which have not been abandoned or expired.

229/11

"Performance Guarantees" means the Guaranteed transfer rates for materials associated with the Ghent CCR Transport System, and the Guaranteed Sound Emissions and the Guaranteed Dust Emissions, all as set forth in Exhibit G.

"Performance Guarantee Test Procedures" means the final detailed procedures for conducting the Performance Guarantee Tests prepared by Contractor and approved, in writing, by Owner in accordance with Exhibit G.

"Performance Guarantee Tests" means the Guaranteed transfer rates for materials associated with the Ghent CCR Transport System and the Sound Emissions Tests, in each case, to determine whether the Performance Guarantees have been achieved.

"Permits" means any waiver, exemption, variance, franchise, permit, authorization, approval, identification number, inspection, certification, license, clearance or similar order, filing, registration, application of, from or to any Governmental Authority required to be obtained or maintained to perform the Work (including, for Owner's occupancy of the Ghent CCR Transport System or any portion thereof), including those set forth in Exhibit P.

"Person" means any individual, company, corporation, firm, joint venture, partnership, association, limited liability entity, organization, trust, Governmental Authority or similar entity.

"Pre-Existing Hazardous Substance" means a Hazardous Substance existing on the Ghent Generating Station Site as of the Effective Date, which Hazardous Substance is of a nature or exists in an amount that requires special handling, disposal or remediation under Applicable Law. Hazardous Substances that are used or useful in the operation of the Existing Facilities that are labeled and in appropriate containers are not considered Pre-Existing Hazardous Substances under this Agreement.

"Prime Rate" means the per annum (365 or 366 Days, as appropriate) prime rate as published from time-to-time in the "Money Rates" table of The Wall Street Journal; provided, however, if more than one such prime rate is published, the average shall be used for purposes of this Agreement, unless an equivalent bank rate is agreed to by the Parties.

"Professional Standards" means those standards and practices used by, and the degree of skill and judgment exercised by, recognized United States national engineering and/or construction firms when performing first class quality services under similar conditions in connection with performing work related to operating coal-fired generation facilities similar to the Ghent Generating Station."Project Requirements" means with respect to the Work or the Ghent CCR Transport System or any portion of either or both: (i) Applicable Law; (ii) Codes and Standards; (iii) the provisions of this Agreement; (iv) the requirements and warranties of Subcontractors, including Equipment and Materials Subcontractors; (v) the requirements of insurers providing insurance pursuant to Article 21; (vi) the Operating and Maintenance Manuals; and (vii) Professional Standards.

"Project Schedule" means the critical path method project schedule, including key dates and milestones for completion of the Work established in accordance with and as set forth in 12/29/11 Exhibit D, as such Project Schedule may be adjusted pursuant to this Agreement.

12/30/11

"Prudent Utility Practices" means the practices, methods, materials, supplies, equipment, and standards of safety, performance, and service that are commonly applied in the electric utility industry in the United States to operate and maintain generating facilities similar to the Units, including the use of, and adherence to, equipment, practices and methods, applicable industry codes, standards, and regulations that in the exercise of reasonable judgment and in light of the facts and circumstances known at the time the decision was made would be reasonably expected to accomplish the desired result while protecting the Ghent CCR Transport System , the Existing Facilities, the interconnection facilities, individuals, and the environment from damage, loss or injury. Prudent Utility Practices are not intended to be limited to the optimum practice or method to the exclusion of all others, but rather to be a spectrum of possible, but reasonable practices and methods having due regard for vendor warranty requirements, Applicable Law, and the operating and maintenance procedures of co-located generation and associated facilities.

"Punch List" or "Punch List Items" means and refers to a comprehensive list initially prepared prior to certification of Mechanical Completion which may be supplemented thereafter to identify those minor defects or deficiencies in the Work that require repair, completion, correction or re-execution, the non-completion of which do not interfere with the occupancy, use, commercial operation, safety, or reliability of the Ghent CCR Transport System.

"Records" has the meaning set forth in Section 25.11.1.

"Response" has the meaning set forth in Section 23.2.2.

"Retainage" has the meaning set forth in Section 8,11.

"Retainage Amount" has the meaning set forth in Section 8.11.

"Sales Taxes" has the meaning set forth in Section 8.2.1.

"Scheduled Mechanical Completion Date" means November 16, 2013, which date may be adjusted pursuant to the terms and conditions of this Agreement.

"Senior Officer" has the meaning set forth in Section 23.1.

"Special Tools" means tools that are described in the Technical Specification or are provided by a Subcontractor for the installation, checking, inspection, operation, repair, or maintenance of Equipment.

"Specialty Supplier" means any Subcontractor that provides Equipment or Materials that are to be made part of the Work and for which there are no replacements and/or services for all or any part of such Equipment or Materials that are readily available from multiple sources other than that Subcontractor.

"Start-Up" following the commencement of an outage (of any kind), means the time that the Unit is released to station operations.

222/11

12/30/11
"Subcontractor" means and refers to a Person (at any tier other than Contractor) which has a contract, agreement or other arrangement to perform a portion of the Work, including the supply of services, Equipment, Materials, Consumables, and Construction Aids in connection with the Work.

"Supplier Diversity Policy" has the meaning set forth in Section 25.23.

"System" means a system or subsystem of the Ghent CCR Transport System set forth in the plan established pursuant to Section 6.2.

"System Turnover Package" means the collection of enumerated items of Information including diagnostic equipment tests that comprise a complete description of a System and its operating requirements in form and substance reasonably acceptable to Owner and established pursuant to Section 6.2.

"Technical Specification" means and refers to Exhibit A attached hereto, and documents specified therein that define generally the requirements and the conceptual Design, scope, and purpose of the Ghent CCR Transport System.

"Terminal Point(s)" means the points of connection including the interfaces and terminal points between the Ghent CCR Transport System and the Units specified in the Technical Specification.

"Tie-in" shall mean when the applicable Ghent CCR Transport System component is Connected to the applicable terminal points of the Ghent Generating Station; the following are the list of Tie-ins and when they will be deemed to have occurred:

Bottom Ash System will be deemed to be Tied-in when it is Connected with the Ghent Generating Station's bottom ash sluicing system;

Gypsum System will be deemed to be Tied-in when it is Connected with the Ghent Generating Station's 0-1 and 0-2 Gypsum Storage Tanks.

Each of the Fly Ash Unit subsystems will be deemed to be Tied-in when it is Connected to the outlet hopper of the electrostatic precipitator for the applicable Unit.

"Tie-in Date" means for each Tie-in, the date on which that Tie-in occurs.

"Turnover Acknowledgment" has the meaning set forth in Section 6.2.2.

"Unit Outage Day" has the meaning set forth in Section 7.1.

"Unit" means one of the four existing coal-fired units operating on the Ghent Generating Station Site that are commonly identified as Units 1, 2, 3, and 4. Unless the context indicates otherwise, the term "Units" refers to all four Units.

"Warranty Period" means the period described in Section 13.2, as it may be extended in accordance with this Agreement.

2/2/1

"WBE" means Women Business Enterprise.

"WBE Spend" means expenditures with WBEs certified by recognized certification entity, including Women's Business Enterprise National Council (WBENC and their regional affiliates), Louisville Metropolitan Sewer District, Louisville Metro Human Relations Commission, Department of Transportation, National Women Business Owners Corporation (NWBOC) (affiliate of NAWBO), and other states' agencies as approved by Owner.

"Work" means all of the work, services, Equipment, Materials, Consumables, and Construction Aids to be performed, provided, or installed by Contractor in accordance with this Agreement, including all Job Site preparation, Design, reconditioning, procurement, transportation, expediting, construction, training, start-up, commissioning, testing, clean-up and waste disposal, and other services or items that are necessary or appropriate to complete the Ghent CCR Transport System, effect Tie-in and Connect the Ghent CCR Transport System, achieve Final Completion and fulfill Contractor's obligations during the Warranty Period in accordance with this Agreement. Work specifically includes any options exercised by Owner in accordance with Exhibit T.

1.2 Contract Interpretation. In this Agreement, unless the context otherwise requires:

- (i) the singular includes the plural and vice versa;
- (ii) reference to any Person includes such Person's successors and assigns, to the extent that such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity;
- (iii) reference to any agreement (including this Agreement), document, insurance policy or instrument means such agreement, document or instrument as amended or modified and in effect from time to time in accordance with the terms thereof;
- (iv) any accounting term used and not otherwise defined in the Agreement has the meaning assigned to such term in accordance with generally accepted accounting principles consistently applied;
- (v) "including" (and "include") means: (a) including without limiting the generality of any description preceding such term and (b) with respect to any description following such term, means "including, without limitation" or "including, but not limited to";
- (vi) reference to Applicable Law means Applicable Law as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time;
- (vii) when applied to Equipment and Materials or Systems, "furnish" "provide" or words of similar import means to secure, pay for, deliver to the Job Site

2.129/11

(or other portions of the Ghent Generating Station Site, as appropriate), unload, inspect and uncrate, store per manufacturer's recommendations and any other services or activities appropriate to that portion of the Work;

- (viii) when applied to Equipment, Materials, or Systems, "install" or "installation" or words of similar import mean to assemble, place in position, incorporate, adjust, clean, make fit for use and any other services or activities appropriate to that portion of the Work;
- (ix) unless the context specifically requires otherwise, the terms "approval," "consent," "accept," "acceptance," "authorization," and terms of similar import shall be deemed to be followed by the phrase "which shall not be unreasonably withheld, unreasonably conditioned or unreasonably delayed";
- (x) the words "shall" and "will" have equal force and effect; the words "herein," "hereof," or "hereunder" or similar terms refer to this Agreement as a whole and not to any specific section or article;
- (xi) the table of contents and article, section and exhibit titles and similar headings are inserted for convenience only and are not be used for the purposes of construing or interpreting this Agreement;
- (xii) the Work is intended to be a term that encompasses all of the necessary performance obligations of Contractor. Any listing of types of work such as "construct," "erect," "check" or "align" is not meant to be exclusive in the context of this Agreement or exclude similar or other services or activities appropriate to that portion of the Work;
- (xiii) words and abbreviations not defined in this Agreement which have well known technical or design, engineering or construction industry meanings are used in this Agreement in accordance with such recognized meanings;
- (xiv) all documentation to be supplied under this Agreement shall be provided in the English language;
- (xv) all dimensions must be specified in the U.S. customary system;
- (xvi) this Agreement and all terms hereunder (including terms related to the reasonableness of the Owner's actions) shall be construed taking into consideration the Existing Facility Requirements; and
- (xvii) payments under this Agreement which are due must be made exclusively in United States dollars.

2/20/11

12/30/11

ARTICLE 2

EFFECTIVENESS

2.1 Effectiveness.

2.1.1 <u>Effectiveness</u>. This Agreement shall be effective and the Parties shall be bound by the conditions applicable to their conduct upon execution of this Agreement, Contractor shall commence the Work immediately upon the effectiveness of this Agreement.

2.1.2 <u>Assignment</u>. Contemporaneously with the execution of this Agreement, the Parties shall execute assignment agreements in the forms attached hereto as **Exhibit** Z under which, effective as of the Effective Date, Owner shall assign to Contractor all of its right, title, and interest in and to the Ghent CCR Transport Equipment Contracts and Contractor shall assume all of Owner's obligations under the Ghent CCR Transport Equipment Contracts. Owner hereby represents and warrants the following to Contractor, on and as of the Effective Date, which representations and warranties will survive the execution of this Agreement:

- (i) prior to the Effective Date, there has been no breach or default of any provision of Ghent CCR Transport Equipment Contracts for which it is responsible;
- (ii) prior to the Effective Date, to the best of Owner's knowledge, there has been no breach or default of any provision of any of the Ghent CCR Transport Equipment Contracts by any of the Ghent CCR Transport Equipment Suppliers;
- (iii) prior to the Effective Date, no amounts have become due under the terms of the Ghent CCR Transport Equipment Contracts that Owner has not paid;
- (iv) prior to the Effective Date, it has provided Contractor with a true and correct copy of the Ghent CCR Transport Equipment Contracts together with any amendment and any change order to the Ghent CCR Transport Equipment Contracts; and
- (v) prior to the Effective Date, it has not received from the counterparty to the Ghent CCR Transport Equipment Contracts any letter of credit, parent company guarantee or other security instrument required to be provided by such counterparty under the Ghent CCR Transport Equipment Contracts that has not been assigned to Contractor.

From and after the Effective Date, Owner has no authority to authorize the counterparty to the Ghent CCR Transport Equipment Contracts to make a change to or under the Ghent CCR Transport Equipment Contracts. Notwithstanding the foregoing, Contractor will not amend or otherwise restrict any right granted to "Owner" under the Ghent CCR

12/30/11 Al

Transport Equipment Contracts and Owner may exercise its rights thereunder, including any required enforcement thereof. Owner shall use reasonable efforts to coordinate the exercise of its rights with Contractor.

2.2 Commitments Prior to Mobilization.

- (i) Contractor hereby confirms that as of the execution of this Agreement there exist no grounds on which a claim by Contractor pursuant to Article
 9 or Section 10.2 may be based;
- (ii) Simultaneously with the execution of this Agreement, Contractor shall deliver the Parent Guarantee (as applicable) required by Section 25.15;
- (iii) Prior to on-site mobilization, Contractor shall provide to Owner copies of insurance policies and certificates required to be obtained by Contractor in accordance with Article 21;
- (iv) Prior to on-site mobilization, Contractor shall confirm through receipt of notice from Owner that Owner shall have received all Owner Permits, set forth in Exhibit P, as are necessary to commence construction, on terms and conditions acceptable to Owner, which permits, approvals or licenses have become final and are not subject to rehearing or appeal;
- (v) Prior to on-site mobilization, Contractor shall deliver the Project Schedule in no less than a Level 3 Primavera® 5.0 format, with a fully logic-tied schedule showing engineering, procurement, and construction developed and resource-loaded;
- (vi) Prior to on-site mobilization, Contractor shall provide Owner with all necessary information from to allow Owner to have submitted all necessary filings to Governmental Authorities that are required prior to the commencement of construction; and

ARTICLE 3

GENERAL PROVISIONS

3.1 Intent of Contract Documents. It is the intent of the Parties that this Agreement be a lump-sum, turnkey contract with a fixed Contract Price and Project Schedule which will not be increased or lengthened, except in accordance with Article 10.

3.2 Independent Contractor. Contractor will perform and execute the provisions of this Agreement as an independent contractor to Owner. Contractor is not and may not act as an agent of Owner for any purpose or reason whatsoever.

3.3 Subcontracting. Subject to Section 3.4, Contractor may have portions of the Work performed by Subcontractors, including entities related to or affiliated with Contractor; provided, however, Contractor may not subcontract all or substantially all of the Work. No

12/30/11

contractual relationship will exist between Owner and any Subcontractor with respect to the Work to be performed hereunder, except pursuant to Sections 3.5, 3.6 and 13.4. Notwithstanding whether any provision of this Agreement specifically refers to Contractor's Subcontractors, Contractor will be fully responsible for all acts, omissions, failures or faults of any Subcontractor as fully as if they were the acts, omissions, failures or faults of Contractor and will require its Subcontractors to provide or perform their portion of the Work in a manner applicable to the performance of Contractor's obligations under this Agreement. If any Subcontractor fails to perform any portion of the Work as such Work is required to be performed in accordance with this Agreement, Contractor is responsible therefor and hereby binds itself to promptly and diligently correct such failure in accordance with this Agreement, at no cost or expense to Owner. The exercise of the right to subcontract will not in any way increase the cost, expense, or liability of Owner hereunder. Contractor shall require any Subcontractor to perform its portion of the Work: (i) under Contractor's supervision and (ii) in accordance with the requirements of this Agreement to the extent applicable to Subcontractor's scope of Work.

Specified Equipment Subcontractors. For those items of Equipment specified 3.4 in Exhibit E. Contractor will only use the services of, or procure Equipment from, those Subcontractors listed under the headings specified in Exhibit E. Contractor will be responsible for the negotiation of the terms and conditions of any purchase orders or subcontracts entered into with such identified Subcontractors (including cost, performance guarantees and equipment warrantees), and, subject to Section 8.2.4, it will enter into written purchase orders and/or subcontracts in its own name directly with such Subcontractors, it being the intent of the Parties that such Subcontractors will be subcontractors of Contractor and not of Owner. Operability, maintainability, reliability, quality, and compatibility with Equipment, Materials, and Consumables utilized in the Existing Facilities must be material selection factors in Contractor's procurement decisions. Contractor will undertake to include MBEs, WBEs, DBEs, and LCs to the maximum practicable extent on bid lists for portions of the Work. Contractor agrees that before submitting specifications for bids to Major Subcontractors it will give Owner a reasonable opportunity to comment on such specifications and Owner agrees that if they wish to comment, Owner will submit such comments within the time period specified for that item in Exhibit X and if not so specified, within fifteen (15) Business Days after delivery of the specifications to Owner. Notwithstanding Owner's right to review and comment in accordance with this Section 3.4, no such review or comment will relieve Contractor or otherwise serve as a defense to Contractor's full performance of its obligations hereunder.

Certain Provisions in Subcontracts. All subcontracts or other arrangements 3.5with a Major Subcontractor must contain provisions, which Contractor may not waive, release, modify, or impair: (i) giving Contractor an unrestricted right, without the consent of the Subcontractor, to assign and thereafter reassign the relevant subcontract and any or all benefits, interests, rights and causes of action arising under it to Owner and/or its designees (and such assignment right will be assigned as part of such assignment); (ii) complying with the provisions of Section 20.4; (iii) authorizing either Owner or Contractor to enforce guarantees and warranties: (iv) requiring Subcontractors that will have a presence on the Ghent Generating Station Site to comply with the plan provided for in Section 14.1 of this Agreement; (v) indemnifying Owner on the terms and conditions set forth in Section 20.1; (vi) incorporating Section 16.3; and (vii) causing Subcontractors of Equipment or Materials, upon the request of Owner, to segregate such Equipment or Materials at their fabrication facilities and identify 12/2/11

12/30/11 ar

Owner's property as such in a manner acceptable to Owner. As a condition of subcontracting with a Specialty Supplier, Contractor shall require each Specialty Supplier to sign the LG&E and KU Services Company standard General Services Agreement for work, equipment, or materials related to the Ghent CCR Transport System but not part of the Work. Contractor shall notify Owner when it enters into any subcontract(s) or other arrangements with a Major Subcontractor and shall promptly provide Owner with an electronic copy (redacted as to price) of such subcontracts and all change orders and amendments thereto. Copies (redacted as to price) of other subcontracts shall be available to Owner upon request.

3.6 Assignment of Subcontracts. Contractor shall, if so requested by Owner after termination of Contractor pursuant to Article 24 or the expiration of the Warranty Period as it may be extended, assign: (i) any subcontract with a Subcontractor and/or (ii) the benefits of any remaining Subcontractor warranty to Owner, its Affiliates, or an operator of the Ghent CCR Transport System, which assignment must not require the consent of the Subcontractor.

3.7 Subcontracts. All subcontracts with Subcontractors shall provide that title will be transferred in the name of Owner in accordance with Section 22.1. Each such subcontract shall limit recourse exclusively to Contractor, except upon the assignment of such subcontract in accordance with Section 3.6. Subcontracts shall allow for suspension of all or a portion of the Work to be performed thereunder and shall be terminable for convenience pursuant to a termination payment schedule.

Inclusion; Order of Precedence. The Body of this Agreement and the Exhibits 3.8 hereto are to be considered complementary and what is required by one will be binding as if required by all. Contractor has included within the Contract Price the cost to complete the Work in its entirety and to fulfill its other obligations hereunder. In addition, the Work includes all that should be properly included and all that would be customarily included within the general scope and magnitude of the Work in order to achieve the Performance Guarantees. The failure to specifically list a requirement in one document, once such requirement is specifically listed in another, does not imply the inapplicability of such requirement and Contractor shall provide or perform, as appropriate, as part of this Agreement all Work or items required to conform to the Technical Specification, the other Exhibits and the standards herein contained. In the event of a conflict between the Body of this Agreement and the Exhibits, the Body of this Agreement governs. In the event of a conflict in an Exhibit or between or among Exhibits, the requirement most favorable to Owner will take precedence, except as may be otherwise determined by Owner, Each Modification will take precedence over that part of this Agreement (including, as applicable, any prior Modification) which it supersedes.

ARTICLE 4

CONTRACTOR'S RESPONSIBILITIES

2/211

4.1 The Work.

4.1.1 <u>Performance of the Work</u>. Contractor hereby covenants and agrees that it shall continuously and diligently provide, perform, install, and complete the Work and its other obligations hereunder in accordance with Project Requirements and the Project

430/11

Schedule. Contractor further covenants and agrees that it shall provide and pay for all Equipment, Materials, Consumables (other than Consumables to be provided by Owner pursuant to Section 5,3), Construction Aids, and other items or services necessary for the proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated into the Work, including all installation, construction services, Design, drafting and procurement, all administration. reporting, expediting, inspection, testing, training, scheduling, and coordination; all startup, commissioning and testing services; all labor (skilled and unskilled); all insurance (as set forth in Exhibit I); Permits and inspections required to be supplied by Contractor under this Agreement; all tools, machinery, storage, and transportation and all other facilities and services necessary to provide and complete the retrofitting of the Unit with the Ghent CCR Transport System in accordance with Exhibit A and this Agreement, it being understood that to the extent **Exhibit** A does not expressly delineate an aspect of the Work, the Parties intend that the Ghent CCR Transport System be designed, equipped, and integrated into the Unit consistent with Professional Standards. Contractor shall order, expedite, receive, furnish, handle, inspect, store, maintain and install Equipment, Materials, and Consumables in accordance with vendor/manufacturer requirements and, in the absence thereof, in accordance with Professional Standards. Contractor will perform all managerial, supervisory, and administrative services that may be necessary to ensure the proper and timely completion of all such activities which form a part of the Work. As such, Contractor agrees that Contractor shall be ultimately responsible for the proper and timely completion of the entirety of the Work in accordance with this Agreement, whether performed by Contractor or by any Subcontractor.

4.2 Sufficient Personnel. Contractor shall, at all times during the term of this Agreement, employ a sufficient number of qualified supervisory personnel, craft persons, and other persons, so that the Work and the other obligations to be performed by Contractor hereunder are completed in an efficient, prompt, economical, and professional manner. Contractor shall be responsible for all overtime or other premium time Work, except as expressly approved by Owner pursuant to Article 10. Without in any way limiting the foregoing, such personnel must include sufficient qualified buyers, inspectors, and expediters necessary to provide Equipment, Materials, and Consumables in a timely manner consistent with the Project Schedule. Whenever required by Applicable Law, licensed (in the applicable jurisdiction) professional engineers will perform the Design services required to perform the Work. Other portions of the Work shall be performed by properly licensed personnel as required by Applicable Law and Professional Standards. Contractor shall also provide all construction and technical services, supervision, and craft personnel as required for system adjustments during start-up, commissioning and testing. During start-up, testing or the operation of the Work until Commercial Operation, Contractor shall maintain qualified personnel on the Job Site twenty-four (24) hours a Day to advise Owner's operators regarding operation and maintenance of the Ghent CCR Transport System. Design Documents must be stamped by a Kentucky Registered Professional Engineer as required by Applicable Law.

12/29/11

12/30/11 en:

4.3 Labor Matters.

4.3.1 Labor Peace. With respect to its and its Subcontractors' employees, Contractor shall be responsible for labor peace on the Job Site and for maintaining good labor relations with local labor organizations. Contractor shall at all times exert its best efforts and judgment as an experienced contractor to adopt and implement policies and practices designed to avoid work stoppages, slowdowns, disputes, or strikes when reasonably possible and practical under the circumstances. Contractor shall advise Owner promptly, in writing, of any actual, anticipated, or threatened labor dispute that might affect the completion of the Work by Contractor or by any of its Subcontractors in accordance with the Project Schedule.

4.3.2 <u>Verification of Employment Eligibility</u>. Contractor shall comply with the Immigration Reform and Control Act of 1986.

4.4 Discipline and Protection. Contractor shall enforce strict discipline and good order among Contractor's employees, Subcontractors' employees, and any other Persons carrying out portions of the Work on the Ghent Generating Station Site and provide for the protection and maintenance of the Work and of all Persons and property related thereto. Contractor shall at all times take all necessary precautions to prevent any unlawful or disorderly conduct by or amongst its employees and those employees of Subcontractors and for the preservation of peace and protection of individuals and property at, or in the vicinity of, the Ghent Generating Station Site. Contractor shall not permit the employment of unfit individuals or individuals not skilled in tasks assigned to them. Contractor understands that importance of maintaining good relations with the community in which the Existing Facilities are located and shall emphasize the importance of good community relations to its employees and other Persons under its supervision on the Job Site.

4.5 Supervision. Contractor shall supervise, coordinate, and direct the Work, using Contractor's best skill, judgment, and attention. Contractor is responsible for the conduct of Persons under its supervision.

4.6 Contractor's Key Personnel. Exhibit K contains a list of Contractor's key personnel who will be responsible for supervising the performance of Contractor's obligations hereunder. Such list includes the designation of Contractor's principal representative (the "Contractor's Representative"), who will be Contractor's authorized representative having the responsibility and authority to direct and manage the Work, administer this Agreement, serve as Contractor's primary point of contact from and with Owner and be authorized to make decisions related to the Work and bind Contractor. Contractor recognizes that a good working relationship must exist between its key personnel and Owner. Owner has the right to request replacement of key personnel upon reasonable notice after having given Contractor a reasonable opportunity (not to exceed ten (10) Days) to rectify the situation leading to such request. Upon the expiration of such period, if Owner remains unsatisfied, Contractor shall promptly effect such replacement. Any replacement of key personnel will be subject to the prior written approval of Owner.

1221111

12/30/11 ar.

4.7 Design and Engineering.

4.7.1 Design Requirements. As engineer of record, Contractor has full Design responsibility for the performance of the Work. Contractor shall engage all supervisors, engineers, designers, draftsmen, Subcontractors, and others necessary for the Design of the Ghent CCR Transport System (including modification of the drawings of Existing Facilities as appropriate) and the preparation of all drawings, specifications, calculations, plans, reports and other Design documentation (including all media) for the Work, setting forth in detail the requirements for the construction of the Ghent CCR Transport System in accordance with this Agreement (collectively the "Design Documents"). Contractor shall Design the Ghent CCR Transport System in accordance with Project Requirements. During performance of the Work, Contractor shall upload to and maintain the Design Documents on a web-based database as and when such Design Documents (or iterations thereof) are completed or revised and Owner shall have unlimited access (including download capability) to the Design Documents on such web-based database. Contractor will provide hard-copy Design Document upon request.

4.7.2 Reviews and Holds. Design Documents will be available to Owner electronically in an agreed format and in hard copy. Owner will be entitled, but not obligated, to review and comment upon the Design Documents and other required submittals in accordance with Exhibit X prior to Contractor commencing with any subsequent phase of the Work related to such Design Documents or submittals. Contractor shall give due consideration to Owner's comments in the final version of the Design Documents or submittals. Design Documents that are required to be certified or under seal must be certified or sealed by professional engineers, licensed and qualified to perform engineering services in the applicable jurisdictions. In no event will any review, comment, or approval of Owner, or acceptance or acknowledgment of any of the Work, in any way, relieve Contractor of any of its guarantees or obligations hereunder, including its full responsibility for errors and omissions, confirming all quantities, selection of fabrication processes, construction techniques, the accuracy of the dimensions, details and the quality of its instruments of service prepared in connection with the Ghent CCR Transport System as well as its responsibility for the quality, integrity, safety, and timely performance of the Work.

4.8 Quality Control. Contractor shall develop, implement and maintain a quality assurance plan for the Work which must include: Ghent CCR Transport System quality assurance; management and control of the Design, engineering, construction, procurement, and supply services; and management and control of Subcontractors and their subcontracts. Such plan must be designed to meet Project Requirements, include procedures for effective implementation, and must be submitted to Owner within thirty (30) Days of the Effective Date for Owner's review and approval. Contractor shall diligently revise the plan as necessary to obtain Owner's approval. Contractor shall also require Major Subcontractors to establish, implement, and maintain comparable quality control and safety programs with respect to their respective portions of the Work. Compliance with the quality assurance plan will not relieve Contractor of any duty, obligation or responsibility under this Agreement.

12/29/111

12/30/11 si

4.9 Training. Contractor shall develop and implement a program to adequately instruct and train personnel made available by Owner in accordance with the provisions of **Exhibit U**. Notwithstanding any other provision of this Agreement, training materials do not constitute Confidential Information. Contractor grants Owner the right to record all training sessions and replay or otherwise provide such recordings for retraining or training of others. Contractor shall provide technical assistance to Owner's operating personnel in connection with the development of training procedures.

4.10 Certain Contractor-Provided Items. Contractor shall provide and pay for all utilities and associated fees (not provided or paid for by Owner pursuant to Section 5.3) required prior to Commercial Operation, including telephone service, Internet service, water, Consumables, sanitary facilities, power, and waste and sewage disposal, including sanitary sewage, and wastewater disposal. Contractor's responsibility for payment shall include all rental, consultation, removal, usage, and other costs or fees. Contractor shall provide its own information technology and telecommunications, cable, or satellite communications. Contractor shall be responsible for providing a first fill of all Consumables as well as refills and replacements during the period prior to Commercial Operation. Contractor shall "top off" consumables, as appropriate, promptly after Performance Guarantee Tests that occur prior to or as a condition of Commercial Operation. Contractor shall use its best efforts to use the same products for Consumables that are used by Owner in the Existing Facilities.

4.11 Equipment and Materials Subcontractor Presence. Contractor shall be responsible for notifying and paying for any Equipment Subcontractor representative that it deems necessary to be present: (i) at any of the training sessions; (ii) for erection supervision; (iii) for commissioning; or (iv) during the Performance Guarantee Tests.

4.12 Current Records; As-Built Drawings. Contractor (and its Subcontractors) shall maintain in good order at the Job Site at least one (1) record copy of the Design Documents, marked currently to record changes made during construction, a copy of this Agreement and Modifications thereto, one record copy of approved shop drawings, product dates, samples and other submittals required by Contractor, all of which will be available to Owner for inspection and use at all times. Prior to and as a condition of Final Completion, the As-Built Drawings shall be delivered by Contractor to Owner, as well as a set of reproducible record drawings (in native and other formats requested by Owner) showing all changes made during construction to the Design Documents and to the drawings of the Existing Facilities. From and after Contractor's mobilization to the Job Site, Contractor shall prepare and display for Owner's benefit on the Job Site a visual depiction, in a storyboard format reasonably acceptable to Owner, of the Work to be performed during the current month and the following month.

4.13 Transportation Costs. Contractor shall arrange and pay for all transportation, freight, storage, and transfer costs (including duties and similar charges) of every kind and nature in connection with the Work.

4.14 Operating and Maintenance Manuals. Contractor shall prepare and provide to Owner the Operating and Maintenance Manuals in accordance with the requirements of Exhibit V.

2229/11

12/30/11

4.15 Control of Work. Consistent with the terms of this Agreement, Contractor shall be responsible for all Design and construction means, methods, techniques, sequences, procedures, quality assurance, and quality control programs in connection with the performance of the Work.

4.16 Emergency Response. Contractor is responsible for developing an emergency response plan for use in connection with emergency situations that may occur on the Ghent Generating Station Site and arise out of its performance of (or failure to perform) the Work. If such an emergency situation occurs, Contractor may not rely on the services of emergency response teams of Owner and shall put its emergency response plan into effect and take such other actions as are necessary to stabilize and ameliorate the situation. Immediately upon the occurrence of an emergency endangering public health or safety, including environmental harm, or material damage to property, Contractor shall: (i) implement its emergency response plan; (ii) without limiting its other reporting obligations hereunder, immediately orally notify an individual on the Emergency Notification List; and (iii) cooperate with Owner by providing information, documentation, and reports as may be appropriate such that both Parties can (a) fulfill all reporting obligations required by Applicable Law and (b) implement procedures appropriate to avoiding a repetition of such occurrence.

Local Conditions. Information on the Ghent Generating Station Site, Job Site and 4.17 for all Units furnished by Owner in specifications, drawings or otherwise is made without representation or warranty of any nature by Owner, is not guaranteed by Owner, and is furnished solely for the convenience of the Contractor. Contractor represents that it has taken steps necessary to examine and ascertain the nature and location of all conditions relevant to the Work and its surroundings which a prudent contractor should have examined or ascertained upon reasonable investigation, and that it has investigated and satisfied itself as to the general and local conditions that can affect the Job Site, the Ghent Generating Station Site, the performance of the Work, or the construction and operation of the Ghent CCR Transport System, including: (i) conditions bearing upon access, egress, transportation, waste, and water disposal, handling, laydown, parking, and storage; (ii) the availability, nature, and conditions of labor, water, electric power, the Internet, other utilities, and roads; (iii) uncertainties of weather or other physical conditions at the Job Site, the Ghent Generating Station Site and the proximate area; (iv) the character of Construction Aids, Equipment and Materials or other facilities needed preliminary to and during the performance of the Work; (v) the condition of the Existing Facilities; and (vi) the proximity of the Existing Facilities, local residences and businesses. Contractor has conducted a transportation survey to assess the circumstances affecting delivery of Equipment, Materials, Consumables, and Construction Aids to the Job Site. Contractor acknowledges that craft and other individuals that are to be present on the Ghent Generating Station Site for the performance of all or any portion of the Work will be required to park at a remote location on the Ghent Generating Station Site. Contractor shall provide transportation within the Ghent Generating Station Site for such individuals as necessary. Contractor assumes the risk of such conditions and the conditions of the Job Site and other portions of the Ghent Generating Station Site on which Work is to be performed as more particularly described in Section 4.18. Any failure by Contractor to take the actions described in this Section 4.17 or Section 4.18 will not relieve Contractor from responsibility for estimating properly the difficulty, time and cost of successfully performing the Work, or for proceeding to successfully perform the Work, without additional expense to Owner, Contractor shall only be entitled to adjustments to Contract Price

12/20/11

12/30/11 an

and Schedule to the extent provided with respect to Excusable Events and Contract Changes as provided in Articles 9 and 10.

Job Site Conditions. Contractor has inspected the Job Site, the other portions of 4.18 the Ghent Generating Station Site on which Work is to occur, the Existing Facilities and surrounding locations, has had an opportunity to conduct such tests as it may desire, and has reviewed information provided by or on behalf of Owner, copies of which are attached hereto in Exhibits S and Q, relating to both the surface and subsurface conditions, to the extent it deems necessary for its purposes, and is familiar with the conditions thereof related to the performance of the Work. Contractor is aware of the karstic character of Ghent Generating Station Site and the Job Site. Contractor further understands the scope of construction activity that is and will be undertaken on the Ghent Generating Station Site during the term of this Agreement and has taken the inherent constraints associated therewith into account in the Project Schedule. Information provided to Contractor concerning the Job Site, other portions of the Ghent Generating Station Site, the Existing Facilities or surrounding areas, including the information provided in Exhibit S and Q, specifications, Design documentation or otherwise is made without representation or warranty of any kind or nature. Such information is not warranted by Owner to be accurate, complete, or otherwise suitable or sufficient for Contractor's purposes and is provided solely as a convenience to Contractor. Any reliance thereon by Contractor is at its sole risk. Geotechnical, topographical, geological, seismic, hydrographical, hydrological, or other natural physical conditions related to the Ghent Generating Station Site or the Job Site, including conditions relating to foundation Design or construction, Job Site preparation, Design, construction or any other portion of the Work will neither be deemed nor constitute the basis of an Excusable Event except to the extent actual subsurface conditions are encountered by Contractor that deviate from the information or descriptions of the Ghent Generating Station Site or Job Site provided by Owner in Exhibit S and Q, the observations of Contractor, or the results of any investigation, test result, or information and conditions reasonably inferable therefrom. Notwithstanding the foregoing, the discovery of Pre-Existing Hazardous Substances at the Job Site will be handled as provided in Article 19.

Witnessing and Inspection Rights. Owner and its employees, agents, 4.19 representatives, and invitees are hereby granted access by Contractor to the Work (including the Design Documents) at all times so as to enable them to witness and inspect the Work, including inspection at the point of fabrication, sub-assembly, preparation for shipment or elsewhere. Contractor shall cooperate with Owner in scheduling visits to Subcontractor factories or subassembly locations for such Persons for purposes of inspecting the Work. Owner has the right to reject any portion of the Work that does not comply with Project Requirements. Contractor shall also afford Owner with the opportunity to witness the events set forth in Exhibit A, Exhibit X (Submittals Reviews and Hold Points), and such other events as to which Owner gives Contractor notice. Contractor shall provide written notice to Owner of all events Owner is entitled to witness pursuant to the terms of its agreements with Major Subcontractors promptly upon execution of such agreements. Contractor shall provide Owner with reasonable notice (but not less than fifteen (15) Days in each instance) of the schedule of the occurrence of all such events and with reasonable advance notice of any rescheduling of all such events. Contractor shall cause such events to appear on the Project Schedule.

222/11

12/30/11

Contractor will have access to the Job Site as provided in 4.20 Use of Site. Section 5.5.1. Access to other portions of the Ghent Generating Station Site on which Work is to be performed, will be on an as-needed basis as requested by Contractor sufficiently in advance of such needs to allow Owner to schedule such activities without adversely impacting the operations of any of the Existing Facilities. Contractor shall be entitled to utilize the internal roadways of the Ghent Generating Station Site as indicated in Exhibit S. Such use is not exclusive and must not interfere with the ongoing operations of the Existing Facilities or other construction activities affecting Existing Facilities other than the Unit. Any use of such roadways that could result in such interference shall be subject to the prior written approval of Owner. The Existing Facility is a baseload generating station serving the needs of ratepayers and other customers and supporting the electrical transmission grid. It is critical that the operation of the Existing Facility not be interfered with or otherwise impaired or put at risk except to the extent absolutely necessary for the performance of the Work (the preceding two sentences outline the "Existing Facility Requirements"). Even if such interference, impairment or risk is absolutely necessary, Contractor shall be required to coordinate its applicable activities with Owner so that Owner can act to anticipate difficulties that may thereby arise (and Contractor shall comply with Owners reasonable instruction such as requiring such activities take place at such times and in such a manner so as to reduce the impact and/or risk). Contractor shall confine its operations at the Job Site to areas permitted by Applicable Law and this Agreement and, with respect to the Existing Facilities or other areas of the Ghent Generating Station Site where Work is to be performed, the reasonable instructions of Owner. Notwithstanding anything in this Agreement to the contrary, Contractor may not interfere with the conduct of the Existing Facilities (including deliveries thereto) or any business operating adjacent to or in close proximity to the Job Site or the Ghent Generating Station Site, except to the extent necessary to Contractor shall coordinate the performance of the Work with the perform the Work. requirements and business operations of the Existing Facilities and construction activities related thereto. Contractor shall prepare a Job Site coordination plan to be delivered to Owner no later than fifteen (15) Days prior to Contractor's mobilization to the Job Site, setting forth the procedures and guidelines to be implemented by Contractor, its Subcontractors, and Owner to maximize site coordination and minimize the likelihood of interference and any adverse effect therefrom on the operations of the Existing Facilities. Owner will be entitled to review and comment on such plan and any such comments shall be incorporated into the final version of such plan. Contractor shall also implement and enforce rules necessary for safe, efficient, and proper prosecution of the Work. At a minimum, Contractor, its Subcontractors, and their respective employees and invitees shall be subject to and shall strictly comply with the health, safety, and environmental protection procedures and regulations established by Owner, including the procedures and regulations set forth in Exhibit H, as they may be reasonably changed from time to time in the sole discretion of Owner. Failure to strictly comply with the requirements of Exhibit H shall be grounds for removal from the Job Site, the Ghent Generating Station Site or the Existing Facilities, at the sole discretion of Owner, and any adverse consequence thereof shall be borne by Contractor. In addition, Contractor shall:

> use, and shall cause all Subcontractors and their respective employees to (i) use, only such gate(s) for access to the Job Site, as identified in Exhibit S, except as otherwise designated by Owner. All Equipment, Materials, Consumables, and Construction Aids must be received, stored and routed, and all waste and demolition debris shall be routed and stockpiled in strict 12/20/2

12/30/11

accordance with the Job Site coordination plan prepared by Contractor and approved by Owner, as described above;

- be responsible for the security of the Work and the Job Site until (ii)Commercial Operation, it being acknowledged by Contractor that neither Contractor nor its Subcontractors are entitled to rely on any security measures or procedures in place at the Ghent Generating Station Site for the protection of individuals or property;
- at least fifteen (15) Days prior to Contractor's mobilization to the Job Site, (iii) Contractor shall develop and provide to Owner a temporary facilities plan. Within fifteen (15) Days of its receipt of the plan, Owner will provide its comments to Contractor. Contractor shall incorporate Owner's comments into a final version of such plan and revise the plan, as necessary, and issue the final version within fifteen (15) Days of receipt of Owner's comments. Thereafter, Contractor shall implement the provisions of the plan, confine temporary structures, machinery, and other property of Contractor (and its Subcontractors), and the storage of Equipment, Materials, Consumables, and Construction Aids to the laydown or other areas more specifically identified in Exhibit S, or otherwise provided in the plan, permitted by Applicable Law, by this Agreement and by Owner, Temporary structures shall be neat in appearance, must not constitute a fire or any other safety hazard and must be properly maintained;
- obtain approval from the Owner Representative prior to performing any (iv)Work on the Ghent Generating Station Site that is not wholly within the Job Site. Contractor understands that it must consult with the Owner Representative to assure that operation of the Existing Facilities will not be interrupted by the Work to be performed and that decisions made by the Owner Representative hereunder will be based upon the operating and maintenance requirements of the Existing Facilities. Contractor further understands that certain activities associated with the Work will require advance notice or approval of Owner, including submission and approval of forms completed by Contractor. The forms to be submitted by Contractor are set forth in Exhibit F-7 and must be completed and submitted to the Owner Representative a reasonable time prior to the time Contractor needs to conduct the subject activity;
- (v)except as expressly permitted with the prior written consent of Owner, not deliver any Equipment and Materials or perform any Work that would be considered "heavy construction" except during the hours of 6 a.m. to 7 p.m. Business Days. Contractor will not schedule deliveries during hours that school bus transportation is scheduled to utilize roads in the general vicinity of the Ghent Generating Station Site. Compliance with all applicable sound regulations or restrictions imposed by Applicable Law to which the construction activity is subject shall be strictly followed by 1, 1, Re/x, Contractor;

12/30/11 Se

- (vi) at least fifteen (15) Days prior to Contractor's mobilization to the Job Site, Contractor shall provide to Owner a construction environmental control plan in connection with the Work for the purpose of properly managing the sediment control, erosion, fugitive dust, storm water runoff, noise, and other adverse environmental impacts of construction, all in conformance with the Project Requirements. Within fifteen (15) Business Days of receipt of the plan, Owner shall provide comments to Contractor and Contractor shall incorporate Owner's comments into the final version of such plan. Thereafter, Contractor shall implement the provisions of the plan;
- (vii) within sixty (60) Days of Contractor's mobilization to the Job Site (but in any case at least fifteen (15) Days prior to bringing any Hazardous Substances onto the Job Site), Contractor shall provide to Owner a Hazardous Substances management plan ("Hazardous Substances Management Plan") that it will vigilantly implement during performance of the Work on any portion of the Ghent Generating Station Site. Such plan, at a minimum, must incorporate the Ghent Generating Station policies and procedures and require Contractor to cooperate and coordinate with Owner. Within fifteen (15) Business Days of receipt of the plan, Owner shall provide comments to Contractor and Contractor shall incorporate Owner's comments into the final version of such plan, Thereafter, Contractor shall implement the provisions of the plan;

4.21 Compliance With Applicable Law. Contractor shall comply, and shall cause its Subcontractors to comply, with Applicable Law in effect from time to time relating to the Work, and shall give all applicable notices pertaining thereto. Contractor shall ensure that the Ghent CCR Transport System, as designed, engineered, and constructed, complies and, when fully integrated into the Unit and operated in accordance with Prudent Utility Practices, will be capable of complying with Applicable Law, except with regard to any emissions or noise limitations, as to which Contractor's sole obligation shall be to comply with the Performance Guarantees set forth in Exhibit G.

4.22 Permits and Approvals. Contractor shall secure and pay for any and all Permits necessary for the proper execution and completion of the Work, except Owner Permits. Contractor shall maintain and comply with all Permits and monitor and keep accurate records of the status of all Permits. Permits obtained by Contractor shall be maintained on the Job Site and copies shall be available to Owner on request.

4.23 Periodic Reports & Meetings.

4.23.1 <u>Monthly Status Report</u>. Within ten (10) Days after the end of each calendar month after the Effective Date, Contractor shall prepare and submit to Owner a status report, covering the previous calendar month, that will be prepared in a manner, level of detail, and format acceptable to Owner and that must include: (i) a detailed description of the progress of the Work, including a critical path chart illustrating the progress which has been made together with a comparison against the Project Schedule;

(ii) a statement of any significant issues and trends, including Change Orders that remain unresolved and Contractor's recommendations for resolution of the same; (iii) an updated report as to Contractor's adherence to the Project Schedule; (iv) a summary of any significant events that are scheduled or expected to occur in connection with the Ghent CCR Transport System during the following thirty (30) Days; (v) a report on safety program performance and results for the previous month, including statistics, lost time accidents, OSHA recordables and near misses, as well as root causes and corrective actions; (vi) the monthly report required pursuant to Section 25.24.3, and (vii) such additional information reasonably requested by Owner. Contractor shall also comply with the requirements of Exhibit M,

4,23.2 Attendance and Participation. Until Final Completion, Contractor shall attend and participate in meetings on a daily, weekly, and monthly basis between Contractor and Owner for the purpose of discussing the status of the Work and anticipating and resolving problems. Such meetings may also include other Persons at the invitation of Owner, such as consultants of Owner; provided such Persons execute reasonable and customary agreements to maintain Work-related information on a confidential basis. Other Persons shall be permitted to attend with the prior consent of Owner, Contractor shall prepare and distribute notes of monthly meetings. Publication or distribution of notes of such meetings shall neither constitute a notice pursuant to Section 25.5 for any purpose under this Agreement nor a permitted basis to assert claims under this Agreement. No implication whatsoever shall be drawn as consequence of a failure by any Party to comment upon or object to meeting notes prepared or distributed by the other Party.

4.24 Signage. Contractor shall not display, install, erect, or maintain any advertising or other signage at the Ghent Generating Station Site without Owner's prior written approval, except as may be required by Applicable Law.

4.25 Spare Parts. Contractor shall provide and properly store and maintain in strict accordance with manufacturer's requirements all spare parts, including start-up and commissioning spare parts prior to Commercial Operation, in accordance with Project Requirements. Spare parts must be equivalent or better to and interchangeable with the original parts they are intended to replace. Such spare parts must be of the same material, of identical manufacture, and must present the same properties as the corresponding parts of the Equipment (except to the extent upgraded by the manufacturer thereof). Spare parts must be properly treated and packed for prolonged storage, considering Job Site ambient conditions. All boxes and packing must be labeled, marked and numbered for identification and a detailed packing list shall be provided by Contractor. Spare parts are to be delivered to storage locations specified by Owner. Contractor shall implement all necessary precautions for proper storage. Contractor shall provide spare parts information in a manner fully compatible for downloading into the spare parts monitoring software maintained by Owner. Contractor shall give Owner the right to purchase, at no more than Contractor's out-of-pocket cost, any surplus items on the Job Site upon Final Completion. Contractor shall be entitled to use any spare parts that are acquired by Owner and are then available on the Job Site; provided that Contractor shall place an order to replace the spare parts it uses immediately and any such parts shall be replaced DDP Job Site (Incoterms 2000) as soon as possible at Contractor's expense. Contractor must provide to Owner 12/29/11

12/30/11 51

a list of strategic and other spare parts that Contractor and its Subcontractors recommend be purchased to maintain reliable Ghent CCR Transport System operations. The spare parts list shall identify the price of each such part (which shall be valid for no less than twelve (12) months), the OEM and other vendors, the OEM's part name and the OEM's part number, expected useful life and typical delivery lead times.

agrees to purchase additional spare parts requested by Owner that are not included in the Technical Specification. The actual cost of such spare parts will be charged to Owner. Contractor will cooperate with Owner to determine the best pricing for obtaining spare parts.

4.26 Interference with Traffic. Contractor shall carry out the Work so as not to damage or interfere with access to, use, or occupation of, public or private roads, footpaths, or other properties, whether in the possession of Owner or of any other Person. Contractor shall communicate with, and ascertain the requirements of, all Governmental Authorities in relation to access to and egress from the Ghent Generating Station Site (and the Job Site) and shall comply with those requirements. Contractor shall be deemed to have satisfied itself as to and shall be fully responsible for the routing for deliveries of Equipment, Materials, Consumables, and Construction Aids, including delivery of heavy, large, or oversize loads to the Ghent Generating Station Site or the Job Site, as appropriate.

4.27 Supplies and Facilities. Contractor shall make available on the Job Site (i) an adequate supply of potable water and (ii) sanitary facilities, for the benefit of those working on the Job Site.

4.28 Cutting and Patching. Contractor shall be responsible for all cutting, repairing, fitting, or patching which is required to complete or repair the Work or to make its parts fit together properly. It is the intent of this Agreement that all areas requiring cutting, fitting, repairing, or patching will be restored to a completely finished equivalent-to-new condition.

Cleaning Up. Contractor shall, at all times during the term of this Agreement, 4.29 keep the Job Site, other portions of the Ghent Generating Station Site and surrounding streets (whether public or private), properties, waterways, sidewalks, and other areas free from accumulations of waste materials, rubbish, dirt, debris and other garbage, liquid and non-liquid materials whether spilled, dropped, left behind, discharged, blown out, or leaked during performance of the Work. Contractor shall maintain its working, storage, laydown, and parking areas in a clean and non-hazardous condition, and shall employ adequate dust control measures. Contractor must provide adequate tire washing facilities for trucks leaving the Job Site. Contractor shall properly maintain Equipment and Systems containing any Hazardous Substances. Contractor shall repair (if necessary) roads and other infrastructure (internal, public or private) on or in the vicinity of the Job Site and the Ghent Generating Station Site that are adversely affected by Contractor (and its Subcontractors) and their employees, agent and representatives' construction activities or traffic related to the Work, as needed. Hazardous Substances, including chemicals used by Contractor or its Subcontractors, must be properly handled and must be properly disposed of off the Ghent Generating Station Site. Prior to Final Completion with respect to the entire Job Site (including laydown, parking, and construction areas) Contractor shall remove all tools, trailers, surplus, waste materials, and rubbish, and shall clean all glass (inside and out), remove all paint spots and other smears, stains or scuff marks,

2/29/1M

12/30/11

clean all plumbing and lighting fixtures, wash all concrete, tile, and finished floors, and otherwise leave the Job Site and the Ghent Generating Station Site where Work was performed neat and clean. If Contractor fails to take the actions required by this Section 4.29, Owner may do so (or cause it to be done) and the cost thereof will be charged to Contractor.

4.30 Waste Disposal. Contractor is responsible for disposal of all wastes generated by it or its Subcontractors during the performance of the Work, including Hazardous Substances, waste water, sanitary wastes, demolition debris, construction debris, spoil, surplus excavation material, office wastes, and wastes related to preparation, commissioning, testing, and start-up of Systems or Equipment. All such wastes must be handled, stored, or disposed of in accordance with Applicable Law in a suitable off-site location. All costs related to such waste disposal are the responsibility of Contractor.

4.31 Water Control. Contractor shall carry out all dewatering, storm water control drainage, punping, and disposal required to keep the Work dry during performance of the Work. Contractor shall not: (i) place Hazardous Substances (including site soils or water that may be contaminated), waste materials, trash, or rubbish; or (ii) discharge contaminated water, into any area that will pollute a natural stream or body of water. No discharge of contaminated water may be permitted to cause soil contamination (whether documented or undocumented) in violation of any Applicable Law.

4.32 Permit Assistance. Contractor, at its expense, will reasonably assist and support Owner's efforts to obtain Owner Permits, or to otherwise satisfy the conditions thereof. By way of example, such assistance includes providing customary information and documentation, and Design data. Contractor shall provide all drawings, documents and other information, stamped as appropriate, that are necessary for Owner to properly and timely complete and defend all Permit-related hearings and actions.

4.33 Deliveries by Truck. Contractor shall provide Owner with reasonable advance notice of any delivery of Equipment or Materials that is to arrive: (i) by truck and is oversized or is extraordinary in any other material way or (ii) by rail.

4.34 Artifacts and Other Valuable Items. If Contractor should discover any artifact, fossil or other items of historical, religious, monetary, geological, archeological or other value, it shall immediately cease Work in the immediate vicinity thereof, notify Owner and take reasonable actions to preserve and protect such items from damage or theft. All such items are the property of Owner and Owner shall provide Contractor with direction on how to proceed.

4.35 Release. Except to the extent expressly provided otherwise in this Agreement: (i) Contractor is responsible for making all investigations and determinations necessary or desirable for it to enter into this Agreement, (ii) Contractor hereby releases Owner from any and all liability in any way arising out of any information, document, statement or report related to the Existing Facilities, the Job Site or the Ghent Generating Station Site, and (iii) Contractor expressly disclaims any right to any Change Order or other adjustment of the terms of this Agreement based on its reliance on information, actions or omissions or Owner with respect to any information, document, statement or report related to the Existing Facilities, the Job Site or the or report related to the Existing Facilities, the Job Site or the or the factor of the terms of this Agreement based on its reliance on information, actions or omissions or Owner with respect to any information, document, statement or report related to the Existing Facilities, the Job Site or the Ghent Generating Station Site provided by Owner.

12/29/14

12/30/11 in

4.36 Covering. No portion of the Work (including foundations) may be covered or otherwise be made inaccessible for inspection without (i) conducting (and passing) any test or inspection required pursuant to this Agreement and (ii) allowing Owner to inspect such Work if a hold point has been established in relation to such Work as required by Exhibit X. If any Work is covered or made inaccessible for inspection in violation of the previous sentence, Contractor shall uncover (or make accessible) such Work and pay the costs resulting therefrom, including recovery or reinstallation costs. There will be no adjustment to the Project Schedule in connection therewith. If a portion of the Work has been covered or made inaccessible for inspection that Owner has not specifically requested to observe, Owner may request to see such Work and Contractor shall uncover it or make it accessible for inspection. If such Work is in accordance with this Agreement, costs resulting therefrom, including recovering or reinstallation costs, will be borne by Owner, and, if applicable, the Project Schedule will be equitably adjusted. If such Work is not in accordance with this Agreement, Contractor shall pay all such costs and will not be permitted any adjustment to the Project Schedule.

4.37 Administrative Facilities. Contractor shall provide adequate furnished office facilities for Owner's personnel in Contractor's offices in Lenexa, Kansas during the Design and procurement phase of the Work. Contractor shall provide temporary office facilities for itself and the Ghent CCR Transport Equipment Supplier on the Ghent Generating Station Site. Such facilities, together with office equipment, sanitary facilities, and communications services are to be provided in accordance with the Technical Specification.

4.38 No First of a Kind. Without Owner's consent, Equipment will not include prototype equipment, meaning any equipment or component thereof that is at that stage in the evolutionary developmental process where the design, functionality, environmental suitability, material composition, scale, reliability, maintainability, and other operational characteristics of the equipment or any component thereof has not been substantiated by at least two (2) full operating years of commercial experience.

12/30/11 qr.



ARTICLE 5

OWNER RIGHTS, DUTIES & OBLIGATIONS

Key Personnel. Owner shall designate, from time to time, one or more 5.1 individuals who will act on Owner's behalf, in connection with the Work, together with the scope of their authority. Among such designees there must be appointed a principal representative of Owner (the "Owner Representative"), who will be Owner's authorized representative, and who will receive and initiate all communications from and with Contractor 12/20/11 and who will be authorized to render decisions related to the Work and bind Owner.

12/30/11 Gr

5.2 Owner's Review. Owner will be entitled to review, comment on, evaluate, or approve the Design Documents and other submittals as provided in Exhibit X. Contractor shall consider Owner's comments in good faith, provided, however, Owner will not have any responsibility or liability for the accuracy or completeness of such documents, for any defects, deficiencies, or inadequacies therein or for any failure of such documents to comply with the requirements set forth in this Agreement, the responsibility for all of the foregoing matters being the sole obligation of Contractor. Contractor shall advise Owner in writing of the disposition of each of the comments.

5.3 Owner-Provided Items. Owner, at its expense, shall provide: (i) a location for Contractor to connect to and obtain construction power. Items to be provided by Owner prior to Commercial Operation, including items necessary for start-up, are specifically identified, limited in quantity, and scheduled as set forth in Exhibit O. Additional quantities shall be made available by Owner, subject to availability, at Contractor's expense.

5.4 **Right to Apply Monies.** Owner shall have the right to deduct from any funds or monies due or to become due to Contractor (or draw on the Retainage) any amounts actually due to Owner from Contractor as a result of any losses, expenses, damages, obligations, or liabilities for which Contractor is responsible pursuant to the terms and provisions of this Agreement.

5.5 Access and Inspection.

5.5.1 <u>Access to Site</u>. Owner shall provide reasonable non-exclusive access to the Job Site (as shown in **Exhibit S**), other portions of the Ghent Generating Station Site, and the Existing Facilities on which Work is to be performed to Contractor and its Subcontractors, subject to Applicable Law, applicable Ghent Generating Station Site regulations, and the terms of this Agreement.

5.5.2 <u>No Relief</u>. No inspection or review of, or failure to inspect or review, the Work or any portion thereof by Owner or any other Person on behalf of Owner will relieve Contractor of its obligation to properly execute and complete the Work. Inspection by Owner will not be deemed to be supervision or direction by Owner, and is only for the purpose of attempting to confirm for Owner's purposes that the Work conforms to the requirements of this Agreement.

5.6 Owner Permits. Owner is responsible for obtaining Owner Permits.

5.7 Right to Carry Out the Work. If Contractor defaults under this Agreement or neglects to carry out the Work in accordance with this Agreement and fails within a seven (7) Day period after receipt of written notice from Owner to take steps to commence and continue curing such default or neglect with diligence and promptness, Owner, without prejudice to any other rights or remedies Owner may have under this Agreement, including declaring Contractor in default, and with or without terminating this Agreement, may correct such deficiencies at Contractor's expense (including Owner's internal, general, and administrative expenses) and Owner shall have the right to: (i) deduct an amount equal to the amounts incurred by Owner in so doing from amounts due or to become due to Contractor; (ii) draw such amounts on the Retainage; and/or (iii) obtain reimbursement of such amounts from Contractor.

12/29/11

12 (30/11

5.8 Rights Not Limited. The rights and remedies provided in this Article 5 are in addition to, and not in limitation of, any other rights or remedies otherwise available to Owner under this Agreement.

Operating Personnel. At such time as Contractor has achieved Mechanical 5.9 Completion (except for the Tie-ins), it may commence the Tie-ins (in the order designated in Article 6). Upon the first Tie-in, the commissioning of the Ghent CCR Transport System shall commence and shall continue until commissioning is completed after the final Tie-in. Owner shall operate the Units (as improved by the Ghent CCR Transport System) from and after Mechanical Completion with its normal complement of personnel. Contractor shall communicate with supervisory personnel identified by Owner to convey any directions with which it wants Owner to comply with respect to the operation of the Ghent CCR Transport System during commissioning, hot testing, and the Performance Guarantee Tests. Owner shall undertake to operate Ghent CCR Transport System in accordance with such directions to the extent such directions are consistent with Good Utility Practices and comply with Applicable Law. If Contractor directs Owner's operation of the Ghent CCR Transport System and such direction is responsible for damage to any of the Units or the Ghent CCR Transport System, Contractor shall; (i) reimburse Owner on demand for correction of such damage within any of the Units (subject to Section 12.3 as to damage to the Units and as to those portions of the Ghent CCR Transport System to which care, custody, and control and risk of loss has passed to Owner); (ii) correct damage within the Ghent CCR Transport System (subject to Section 12.3 as to damage to those portions of the Ghent CCR Transport System to which care, custody and control and risk of loss has not passed to Owner); and (iii) pay liquidated damages for each Unit Outage Day in accordance with Section 7.1. Contractor understands that each of the Units is a critical generating resource for Owner and consequently Contractor will exert maximum efforts to give directions that are consistent with Owner's requirement to maximize Unit operations on an unrestricted basis. If Contractor desires or needs to have additional technical personnel and/or special monitoring or measuring equipment for any tests, then Contractor shall furnish such personnel or equipment at no cost to Owner.

5.10 System Needs. Each of the Units is a critical generating resource for Owner and must be continually available for full operation, including after each Tie-in commences (excluding, with respect to the Fly Ash Tie-in for each Unit, during the scheduled time for the Fly Ash Tie-in Outage for that Unit) but prior to Commercial Operation. Thus, Contractor shall plan and implement its Work in such a fashion to ensure that the Units can be so continually operated and will prepare (and, if necessary, implement) contingency arrangements so that the Units can be so operated and the CCR from the Units can be properly transported to their permanent disposal location on the Ghent Generating Station Site.

5.11 Contractor's Personnel. Owner has the right to object to any representative or Person employed or engaged by Contractor that engages in misconduct, is believed by Owner to lack competence for the tasks assigned, or is negligent while on the Job Site or the Ghent Generating Station Site. Contractor shall remove such Person from the Job Site or the Ghent Generating Station Site, as applicable upon receipt of Owner's notice. Any cost for replacement Persons will be at Contractor's expense. The rights of Owner under this Section 5.11 are in addition to the rights of Owner with respect to the personnel pursuant to Section 4.6.

12/29/11

12/30/11 Si

ARTICLE 6

PROJECT SCHEDULE

6.1 Commencement. Contractor shall immediately commence performance of the Work in its entirety in accordance with the requirements of this Agreement and continuously and diligently fulfill its obligations under this Agreement.

Contractor shall provide a detailed written turnover and start-up plan to Owner for its review and comment. Such plan must include: a complete listing of the Systems along with a scheduled turnover date for each such System and a schedule of documents to be included in the System Turnover Package that will accompany the System being turned over. Each System Turnover Package must be properly completed and include sufficient checkout and operation information to clearly indicate that the System has been completely checked out, including: instrumentation checkout and calibration data sheets, hydrostatic test reports, factory test reports, chemical cleaning and lubrication records, non-destructive testing records, operating manuals, marked-up P&IDs reflecting as-built conditions, and electrical test data sheets, including megger test reports and vendor field reports. The data provided in each System Turnover Package must be complete and compatible for review such plan insertion into Owner's hold card system. and provide written comments to Contractor. Contractor will promptly address such comments and resubmit the plan until such time as Owner approves the plan. Owner and Contractor will mutually agree on the final turnover and start-up plan prior to its implementation. Thereafter, Contractor's turnover of Systems will be accomplished as follows:

6.2.1 <u>Ready for Turnover</u>. When Contractor deems that it has achieved completion of construction of a System or, in the case of any System that is a modification, expansion or improvement of a portion of the Unit, completion of such modification, expansion, or improvement, in accordance with this Agreement, including static integrity tests, alignment, electrical continuity tests, lubrication, and demonstration of readiness for operation as appropriate (but exclusive of Punch List Items, final vendor reports, manuals, and other information not necessary for start-up), Contractor shall notify the Owner Representative, in writing, that the System is ready for turnover to Owner, which notice must be accompanied by the applicable System Turnover Package meeting the requirements of this Agreement.

6.2.2 <u>Turnover Acknowledgment</u>. Owner will agree that the System is ready for turnover in writing ("Turnover Acknowledgment") by Owner Representative of notification from Contractor (the "Owner Review Period"), unless Owner reasonably believes that: (i) the System Turnover Package does not comply with the requirements of this Agreement or (ii) the System: (a) contains deficiencies, defects, or non-conformities that preclude safe testing, safe commissioning, or safe operation; (b) has not been prepared, flushed, or cleaned as necessary or appropriate; (c) requires Work that has not been completed and which does not constitute a Punch List

12/29/11

12/30/11

Item; or (d) materially differs from the System required under this Agreement. Owner and Contractor must develop a Punch List and endeavor to ensure that the Punch List contains all deficiencies and incomplete items of Work with respect to the System submitted for Turnover Acknowledgment. Notwithstanding the foregoing, if the turnover of a System by Contractor is not made in accordance with the agreed upon turnover Project Schedule and as a result Owner is unable to accept or reject such System within the Owner Review Period, Owner and Contractor will mutually agree upon a reasonable extension of the Owner Review Period for the affected Systems, which extension will reflect the nature of the affected System, Turnover Acknowledgment and the provision of operating personnel neither constitutes acceptance of any System, Owner's acceptance of care, custody, or control of such System, nor any transfer of control to Owner, including responsibility for commissioning, start-up, testing, operations, or maintenance.

6.2.3 Deficiencies. If Owner reasonably believes that any of the circumstances set forth in clauses (i) and (ii) of Section 6.2.2 exist with respect to a System submitted for Turnover Acknowledgment, Owner will so notify Contractor in writing during the Owner Review Period, stating the deficiencies noticed or the incomplete items of Work, as applicable. When Contractor deems it has remedied such deficiencies or completed such items of Work, as applicable, Contractor will then again notify Owner Representative as provided in Section 6.2.1. This procedure will be repeated until Owner issues the Turnover Acknowledgment for such System. Nothing contained herein will prevent Owner from identifying any Defects, deficiencies, incomplete Work, or Punch List Items if discovered after Turnover Acknowledgment of any System.

6.3 Tie-in and Mechanical Completion.

6.3.1 Sequence. Contractor shall not commence making any Tie-in until (i) the entire Ghent CCR Transport System has achieved Mechanical Completion and is ready to commence commissioning (except for Work required to be performed as part of the Tieins), (ii) Contractor has given Owner a certificate to that effect, and (iii) Owner has signed a certificate that Tie-ins may commence. Contractor will then perform the Tie-ins in the following sequence: (i) Bottom Ash System (which Tie-in will be performed without a Unit outage), (ii) Gypsum System (which Tie-in will be performed without a Unit outage), and (iii) the four Fly Ash Unit subsystems during the respective Unit Fly Ash Tie-in Outages for such Tie-ins.

6.3.1 Achievement. Contractor shall achieve each of the Fly Ash Tie-Ins prior to the scheduled end of the applicable Fly Ash Tie-in Outage.

If timely completion of the Marshalling Requirement by 6.3.2 Tie-in. Contractor for a Tie-in is in jeopardy in the reasonable judgment of Contractor or Owner, Contractor shall promptly prepare and implement, in good faith, a detailed recovery plan such that the performance of the Work can progress in accordance with the Project Schedule. Owner is obligated to issue a Certificate of Tie-in

receipt of Contractor's notice of the completion of a Tie-in if Contractor has satisfied the requirements of Tie-in; provided, however, that if such certificate is given, the Tie-in completion date shall relate back to the date of submission of Contractor's notice of 12/29/11

12/30/11 51

completion for which the certificate is given. If Contractor fails to complete a Fly Ash Tie-in within the Fly Ash Tie-in Outage and the affected Unit cannot be placed in service and operate as a consequence of Contractor's failure to complete the Tie-in, liquidated damages will be assessed against Contractor, which liquidated damages will be calculated as if they had been incurred under Section 7.1, for each Day, or part thereof (without proration), from and after the first Day following the Fly Ash Tie-in Outage until the Day that the Fly Ash Tie-in has been completed. Contractor's Tie-in Work must, however, be coordinated with the Work to be performed by Owner during such outage. If Contractor requires additional time to complete the Tie-in Work and requires that the Unit be out of service in order to complete such Tie-in Work, at Contractor's request, Owner will schedule the required number of Days for Contractor to complete the Work and Contractor will be assessed liquidated damages for each such Day, as if such liquidated damages were incurred under Section 7.1. Contractor may not commence Work to accomplish Tie-in unless its workforces, its Subcontractors' workforces, Equipment, Materials, Consumables, and Construction Aids necessary for the performance of such Work, have been fully marshaled and are present at the Job Site prior to the commencement of the outage during which such Work will be performed (the "Marshaling Requirement"). Contractor will withdraw from the Unit so as not to delay Start-up as planned by Owner. Contractor is obligated to achieve the Tie-in notwithstanding the occurrence of an event of Force Majeure (other than an event of Force Majeure occurring after it has met the Marshaling Requirement) and without adjustment of the terms of this Agreement on account thereof. Contractor recognizes that the date of the scheduled Fly Ash Tie-in Outage is not subject to change except in Owner's sole discretion.

If timely achievement of Mechanical 6.3.3 Mechanical Completion. Completion is in jeopardy in the reasonable judgment of Contractor or Owner or if it is not timely achieved, Contractor shall promptly prepare and implement, in good faith, a detailed recovery plan and accelerate the Work such that the performance of the Work can progress in accordance with the Project Schedule. after receipt by Owner of written notice from Contractor certifying that Contractor has satisfied the requirements for Mechanical Completion (other than issuance of the Certificate therefore by Owner), Owner shall determine whether the applicable requirements have been achieved and shall either issue the Certificate of Mechanical Completion or give notice to Contractor in writing of Defects in the Work (other than Punch List Items), of which Owner then has knowledge. Owner is obligated to issue the Certificate of Mechanical Completion if Contractor has satisfied the requirements of Mechanical Completion (other than issuance of the Certificate therefor by Owner). Upon receipt of a written notice of the reasons why Contractor has not achieved Mechanical Completion from Owner, Contractor shall promptly perform corrective measures to eliminate any Defects or deficiencies in the Work and shall thereafter provide another written notice to Owner containing the applicable certification set forth above. Owner shall,

required, determine whether the applicable requirements have been achieved and either issue the Certificate of Mechanical Completion or advise Contractor of any Defects or

12/29/11

12/30/11

deficiencies (other than Punch List Items) remaining in the Work that prevent achievement of Mechanical Completion.

6.4 Project Schedule Update. Without altering, revising, or otherwise changing the scheduled Tie-in Dates, the Scheduled Mechanical Completion Date, the Guaranteed Commercial Operation Date, or the Guaranteed Final Completion Date, Contractor shall submit, on a monthly basis, with the monthly status report submitted pursuant to Section 4.23(i), electronic and updated hard copy of a total project integrated Project Schedule, including critical path activities interconnected by schedule logistics, in Level 3 Primavera® 5.0 format (fully logic-tied and resource loaded) and meeting the requirements set forth in Exhibit F-4, to Owner for Owner's review and comment. All extensions of the Project Schedule to which Contractor is entitled shall be determined by the impact of the delay giving rise to the change in Project Schedule on the critical path. The difference or "margin" between the planned date for Commercial Operation and the Guaranteed Commercial Operation Date belongs to Contractor. Contractor shall "own the float" to the extent of retaining its right to finish early.

6.5 **Performance Testing.**

6.5.1 <u>Performance Guarantee Tests</u>. At such time as Mechanical Completion and all Tie-ins have been achieved and Contractor believes that the Ghent CCR Transport System is ready for the performance of the Performance Guarantee Tests, Contractor shall so notify Owner in writing of the date on which it desires to have the Performance Guarantee Tests conducted,

Contractor will be entitled to reschedule commencement of the Performance Guarantee Tests

6.5.2 <u>Performance Guarantee Tests</u>. Performance Guarantee Tests shall be scheduled with due regard for minimizing the operating expenses of and maximizing revenues to be derived from Unit operations during such tests. The Performance Guarantee Tests must be conducted by a third party contractor as required by Exhibit G and passed simultaneously while the Units, as improved by the Ghent CCR Transport System, are in compliance with the terms of this Agreement and producing CCR at or near their maximum rates (the Performance Tests will utilize additional CCR held in the Existing Facility and/or stored in the Ghent CCR Transport System to achieve the design limits for the Ghent CCR Transport System as set forth in this Agreement). Such tests must be conducted: (i) in the presence of Owner; (ii) in accordance with Section 5.9, Section 6.5.4 and Article 11; and (iii) in accordance with the requirements of Exhibit G and the Performance Guarantee Test Procedures.

6.5.3 <u>Equipment</u>. No auxiliary, standby, or temporary Equipment may be used during the performance of the Performance Guarantee Tests, unless otherwise approved in writing by Owner. All Equipment and Systems of the Ghent CCR Transport System must be operational.

6.5.4 <u>Testing</u>. Contractor's directions issued to Owner pursuant to Section 5.9 must be consistent with Prudent Utility Practices and Applicable Law, including operation of all Systems and Equipment within the manufacturers' specifications,

12/29/2

12/30/11

recommendations, and warranty requirements, without over-stressing or overpressurizing any such systems. Test technicians collecting data and providing other testing-related services may not operate any Equipment during any Performance Guarantee Test.

If Contractor fails to successfully perform any of the 6.5.5 Retesting. Performance Guarantee Tests, the defects, deficiencies, and other conditions which so prevent performing such tests successfully must be immediately thereafter corrected or remedied in accordance with Section 12.1. Upon completion of such corrective or remedial actions, Contractor may cause the Performance Guarantee Tests to be reprior written notice to Owner: performed provided, that if Contractor completes such corrective or remedial actions promptly after the end of the failed Performance Guarantee Tests, Contractor may request a reperformance on notice to Owner and Owner shall not unreasonably deny such a request (e.g., Owner may deny the request if Owner's relevant personnel and advisors cannot reasonably be available to observe the testing through the projected conclusion of the re-performance) so long as Contractor establishes to Owner's reasonable satisfaction that Contractor has successfully determined the defects, deficiencies, and other conditions which prevented performing such tests successfully and has completed the necessary corrective or remedial actions. The Performance Guarantee Tests will be repeated until the Performance Guarantee Tests have been successfully conducted by Contractor but will in no way excuse Contractor from the timely achievement of the Guaranteed Commercial Operation Date. The results of all inspections and tests will be provided to Owner together with gross and reduced data and other information reasonably requested by Owner. The Performance Guarantee Tests will be conducted in accordance with this Section 6.5, Section 5.9, and Exhibit G for the purpose of demonstrating the achievement of Commercial Operation or Final Completion, as the case may be.

6.5.6 Excused Delays. If the Performance Guarantee Tests (or retest) cannot commence as of the date established pursuant to this Section 6.5 (or if the Performance Tests or retest are interrupted) because Owner cannot operate the Units at the capacity required for the Performance Tests (unless such inability is caused by the Work, the Contractor or a Subcontractor) (a "Unit Derate or Outage Delay"), then the Guaranteed Commercial Operation Date will be extended by one day for each day until the end of the Unit Derate or Outage Delay. If after Mechanical Completion in connection with the Performance Tests (or retest) Contractor requires and requests an outage pursuant to (i) of the last sentence of Section 6.11 and Owner delays providing such outage beyond a reasonable period of time, then the Guaranteed Commercial Operation Date will be extended by one day for such delay.

6.6 Re-Setting of Ghent CCR Transport System. During those periods after Commercial Operation, Contractor shall perform whatever Work is necessary, including resetting of Equipment and repairs of damage or modifications caused by testing, to return the Ghent CCR Transport System or relevant portion thereof to the normal operating control settings and configurations; provided, however, Contractor is not entitled to require that the Unit be taken out of service or operate on a restricted basis.

12/30/11 an

6.7 Commercial Operation. Contractor shall successfully achieve Commercial Operation on or before the Guaranteed Commercial Operation Date. At such time as Contractor, in good faith, believes that it has completed the requirements necessary for the achievement of Commercial Operation, including the passage of the Performance Guarantee Tests required therefor, it shall give written notice of its belief to Owner. Such notice shall be accompanied by a preliminary Performance Guarantee Test report as soon as reasonably practicable but in no event

or as

soon thereafter as such reports are first available to Contractor) providing a summary of the Performance Guarantee Tests on which it is based and including all raw data taken during such Performance Guarantee Tests. A final Performance Guarantee Test report must be delivered to and such report will Owner otherwise be in accordance with Exhibit G. Contractor shall include sufficient results of testing in the preliminary Performance Guarantee Test report to allow Owner to reasonably determine that there is a high probability that the final Performance Guarantee Test results will confirm that the applicable conditions of Commercial Operation have been achieved. As soon as practicable, after receipt of notice by Owner, Owner shall but in any event either issue the Certificate of Commercial Operation or reject Contractor's notice, in which case it shall state its reasons for rejection. Such procedure shall be repeated until Contractor submits a notice demonstrating that it has fulfilled the requirements necessary to achieve Commercial Operation. Owner is obligated to issue the Certificate Commercial Operation

period set forth above if Contractor has satisfied the requirements of Commercial Operation. If such certificate is given, the actual date of Commercial Operation shall relate back to the date Contractor submitted its written notice of achievement of Commercial Operation to Owner for which the certificate is given.

Possession and Control. On Commercial Operation, Owner shall take and 6.8 thereafter be responsible for the care, custody, control, operation, and maintenance of the Ghent CCR Transport System. Following transfer of possession and control of the Ghent CCR Transport System to Owner, Contractor (and its Subcontractors) will have reasonable access to the Ghent CCR Transport System to complete any Work still remaining to be performed hereunder; provided, however, Contractor, Subcontractors and Owner will be required to complete any and all Work required to achieve Commercial Operation or Final Completion in a manner consistent with the operational requirements of the Units (as improved by the Ghent CCR Transport System) as directed by Owner. In no event will Owner be required to take any of the Units out of service or otherwise adversely affect any of the Units' operations, except as set forth in this Section 6.8.

Final Completion. Contractor shall successfully perform all of the Work and 6.9 obligations (except obligations requiring future performance, e.g., warranty obligations) and shall achieve Final Completion on or before the Guaranteed Final Completion Date. At such time as Contractor, in good faith, believes that the requirements of Final Completion have been met, Contractor shall give notice to Owner, together with reasonable substantiating documentation thereof, including all gross and reduced data from Performance Guarantee Tests conducted by Contractor. Final laboratory results are not required to be provided with such of such notice from Contractor, Owner will notice. determine whether the requirements of Final Completion have been achieved, subject to confirmation by reference to the final laboratory results. If Owner agrees with Contractor's 21211°

12/30/11 a

notice, upon receint of final laboratory results that confirm that Final Completion has been achieved, Owner, of the later of receipt of Contractor's notice or such final laboratory results, shall issue the Certificate of Final Completion. If, however, Owner believes that Final Completion has not been achieved, Owner shall give notice to Contractor to that effect, with reasons for such belief, period, and, upon receipt of such notice from Owner, Contractor shall promptly take corrective action to fulfill the requirements of Final Completion. Contractor shall thereafter provide another notice to Owner when it believes that Final Completion has been achieved and the above-described procedures will be repeated until Final Completion is achieved. Owner is obligated to issue the Certificate of receipt of Contractor's notice if Contractor of Final Completion has satisfied the requirements of Final Completion. Nothing in this Section 6.9 shall relieve Contractor of its obligation to achieve Final Completion by the Guaranteed Final Completion Date.

6.10 Maintenance Outage. Owner will inform Contractor as soon as practical if a maintenance outage is scheduled for the Unit. With Owner's permission, Contractor may use the period established by Owner for such maintenance outage to perform Work, including Tie-in. Owner may change the schedule for commencement of a maintenance outage without notice. If Contractor causes Owner to extend a maintenance outage beyond the scheduled duration, then each Day of such extension will constitute a Unit Outage Day and Contractor shall compensate Owner for any such extension at the rate established for liquidated damages pursuant to Section 7.1.

6.11 Contractor Requested Outage. In addition to an outage scheduled as part of the Tie-in Work, Contractor may request permission to perform Work during a forced outage of the Unit. If the Unit experiences a forced outage, Contractor may give Notice that it wishes to use a portion of such forced outage to perform Work. Owner, to the extent practicable, will specify the dates, times, and restrictions on Contractor's access to the Unit for the performance of such Work. Contractor may also request access to Existing Facilities off the Job Site at other times. Such request for access at other times shall be made in writing and provide

Existing Facilities. Owner will specify in writing the dates, times, and restrictions on <u>Contractor's access prior to providing such access to Contractor</u>. Not later

to the start of any such additional outage, Owner has the additional right to change the dates and times during which Contractor will have such access. If (i) Contractor's request for additional outage time cannot be accommodated during a scheduled or forced outage and the Unit must be taken out of service or operated on a restricted basis or (ii) Contractor fails to properly withdraw from the outage on schedule and thereby causes a delay in Start-Up, as applicable, then liquidated damages pursuant to Section 7.1 will apply.

2/22/1/1

ARTICLE 7

LIQUIDATED DAMAGES & LIABILITY LIMITATIONS

that

Unit Outage Day Liquidated Damages. The Parties agree that it would be 7.1 extremely difficult and impracticable under the presently known and anticipated facts and circumstances to ascertain and fix the actual damages that Owner would incur should Contractor fail to achieve a Fly Ash Tie-in during the applicable Fly Ash Tie-in Outage or, during the performance of the Work, cause a Unit to remain (or be taken or forced) out of service or operate on a restricted basis for all or a portion of a Day (each a "Unit Outage Day," for example, if two Units are forced out of service for all of one Day, that would constitute two Unit Outage Days). Accordingly, the Parties hereby agree that if Contractor fails to commence and complete the Ticin during a Fly Ash Tie-in Outage, extends an outage as set forth in Section 5.9, or otherwise is liable for liquidated damages pursuant to this Section 7.1, including as set forth in Section 6.10, Section 6.11, Section 7.3 or this Section 7.1, then Contractor shall pay Owner, as Owner's sole and exclusive remedy and Contractor's sole and exclusive liability for such failure, as liquidated for each damages, and not as a penalty, whole Unit Outage Day (pro rated for partial Unit Outage Days and for operations on a restricted basis). Contractor shall not be liable for liquidated damages under this Section 7.1 to the extent that any applicable Unit Outage Day (or portion thereof) results from Owner's failure to exercise commercially reasonable efforts (but not including the performance of any Work to be performed by Contractor) in accordance with Prudent Utility Practices (i) to return the applicable Unit to service as soon as practicable and (ii) to maintain it in service,

7.2 Commercial Operation Delay; Liquidated Damages. The Parties agree that it would be extremely difficult and impracticable under the presently known and anticipated facts and circumstances to ascertain and fix the actual damages that Owner would incur should Contractor fail to achieve Commercial Operation by the Guaranteed Commercial Operation Date. Accordingly, the Parties hereby agree that if Contractor fails to achieve Commercial Operation Date, then Contractor shall pay to Owner, as Owner's sole and exclusive remedy and Contractor's sole and exclusive liability for such delay, as liquidated damages, and not as a penalty,

(pro rated for partial Days)

Commercial Operation is delayed beyond the Guaranteed Commercial Operation Date, and for each whole Day (pro rated for partial Days) thereafter

until Commercial Operation is achieved. It is agreed by the Parties that the liquidated damages set forth in this Section 7.2 relate solely to Contractor's failure to achieve Commercial Operation by the Guaranteed Commercial Operation Date.

7.3 Work Scheduling and Liquidated Damages. Contractor acknowledges that time is of the essence in this Agreement and that Owner intends to place the Work in commercial service on or prior to the Guaranteed Commercial Operation Date. Accordingly, upon achievement of Commercial Operation, Contractor shall be required to complete any and all Work required to achieve Final Completion, in a manner consistent with the operational requirements of the Units as directed by Owner. Contractor is not entitled to require Owner to take any of the Units (or all or any portion of the Ghent CCR Transport System) out of service or

230/1

otherwise adversely affect any of the Unit operations. Accordingly, Contractor shall schedule and coordinate with Owner any Work, including performance testing, required to achieve Final Completion to avoid adverse impacts on Owner's ability to operate any of the Units. If the Units or any portion thereof must remain or be taken out of service or the nature of the Work to be performed will otherwise prevent Owner from operating any of the Units as it requires, Owner must first agree to take the Units out of service or to permit such Work to be performed and Contractor shall be required to pay to Owner, as liquidated damages, the sum established in Section 7.1 for each Day of portion thereof for which any of the Units remain or are taken out of service or operated on a restricted basis after the Guaranteed Commercial Operation Date. If the Ghent CCR Transport System or any portion thereof must remain or be taken out of service or the nature of the Work to be performed will otherwise prevent Owner from operating any portion of the Ghent CCR Transport System as it requires, Owner must first agree to take the portion of the Ghent CCR Transport System out of service or to permit such Work to be performed and Contractor shall be required to plan and implement its Work in such a fashion to ensure that the Units can be continually operated on an unrestricted basis and will prepare (and, if necessary, implement) contingency arrangements so that the Units can be so operated and the CCR from the Units can be properly transported to their permanent disposal location on the Ghent Generating Station Site.

7.4 Payment. The liquidated damages specified in this Agreement are due and payable by Contractor to Owner on the first Business Day of the month following the month in which they were incurred.

7.5 Consequential Damages. Neither Owner nor Contractor (or any Subcontractor) will be liable for any consequential, special, incidental, or indirect damages, sustained by the other Party or any of such other Party's respective parents or Affiliates.

7.6 Limitations of Liability.

7.6.1 <u>Liquidated Damages Limit</u>. Contractor's maximum aggregate liability for liquidated damages arising under this Agreement will in no event

7.6.2 <u>Overall Limit</u>. Contractor's overall cumulative liability for damages to Owner arising under or in relation to this Agreement will in no event exceed an amount of the Contract Price; provided, however, such limitation of liability shall not apply to, and no credit shall be issued against such limitations for:

- (i) Contractor's indemnity obligations as set forth in Sections 8.8.2 and 22.2 and Articles 16 and 20 (provided, that for Article 20, only third party claims shall not apply to such limitation); and
- (ii) Claims which arise or result from fraudulent acts, violations of Applicable Law, gross negligence or willful misconduct of Contractor, its Subcontractors or others for whom Contractor is responsible.

12/29/111

(iii) any cost or expense arising from Contractor's failure to achieve Mechanical Completion.

ARTICLE 8

CONTRACTOR'S COMPENSATION

8.1 Contract Price. Owner shall pay Contractor for the due, proper, and complete performance of the Work as required hereunder and for the due performance of all other obligations and duties imposed upon Contractor pursuant to this Agreement

subject to (i) a reduction by the aggregate amounts, if any, paid by the Owner under the Ghent CCR Equipment Contracts, (ii) adjustments in the event of any options exercised by Owner in accordance with Exhibit T, and (iii) other additions and deductions by Change Order as provided in this Agreement (the "Contract Price"). Contractor shall receive the Contract Price as full compensation for the performance of the Work and its other obligations hereunder whether or not Contractor has properly estimated or anticipated the costs required to fully perform its obligations under this Agreement. The Parties agree that as of the Effective Date, Owner will have made progress payments to the Ghent CCR Transport Equipment Suppliers under the Ghent CCR Transport Equipment Contracts to Contractor, will have been made for the benefit of Contractor. Accordingly, notwithstanding anything to the contrary herein, Owner shall be entitled to deduct the aggregate amount of its payments to the Ghent CCR Transport Equipment Contracts to the Ghent CCR Transport Equipment Contracts to the Ghent CCR Transport Equipment of the Ghent CCR Transport Equipment contracts from the first amounts due and owing to the Contractor hereunder.

8.2 Taxes.

Tax Cooperation and Information. Contractor shall pay all payroll and 8.2.1 other related employment compensation taxes for Contractor's employees; federal, state and other taxes that may be assessed on Contractor's net income, net worth, license, privilege, or gross receipts (other than taxes that are designated as sales or use taxes) arising from the Agreement; taxes, duties, excise fees, and other costs associated with the importation or exportation of Equipment, Materials, Consumables, Construction Aids, or services; other fees, royalties and assessments of any nature in connection with the Work, contributions and taxes for which Contractor is liable under Sections 8.2.3 and 8.2.6, and engineering and business license costs (collectively, the "Contractor Taxes"). Sales or use taxes required under Applicable Laws to be (i) collected by Contractor from Owner, or (ii) paid by Contractor or any Subcontractor (and for which no exemption is available) in connection with purchases or sales of Equipment, Materials or Consumables required for performance of the Work, (collectively "Sales Taxes") including sales or use taxes on Equipment, Materials, and Consumables provided by Contractor, shall be administered by Contractor in accordance with Sections 8.2.4 below and 8.2.6 below. The Contract Price includes Contractor Taxes and excludes Sales Taxes (other than those included in Contractor Taxes). Contractor shall be reimbursed for the full amount of Sales Taxes (other than those included in Contractor Taxes) properly paid by Contractor or its Subcontractors in accordance with this Agreement on a monthly basis at the time. 12/2/11

12/30/11 5r

Contractor receives payments pursuant to Section 8.5, subject to the documentation requirements of Section 8.6.

8.2.2 <u>Subcontractors</u>. Excluding any Sales Taxes that are determined by a Governmental Authority not to be eligible for exemption under Applicable Laws, Owner will not have any responsibility whatsoever with respect to taxes assessed against, paid, or deemed to be paid by Subcontractors.

8.2.3 <u>Taxes Related to Employment of Persons</u>. Contractor accepts full and exclusive liability for the payment of any and all contributions or taxes that are measured by wages, salaries, or other remunerations paid to Persons employed by Contractor or its Subcontractors for the Work, or which arise by virtue of their employment, and which now or hereafter are imposed by any Governmental Authority. Such contributions or taxes shall include those for unemployment insurance, social security insurance, workers' compensation, old age retirement benefits and other payroll taxes of any kind. Contractor shall comply with Applicable Law relating to such subjects and shall fully reimburse Owner for any of such taxes or contributions that Owner may be required to pay.

8.2.4 <u>Sales and Use Taxes</u>. Certain items of Equipment, Materials and Consumables purchased in the performance of the Work may be exempt from Sales Taxes in accordance with KRS 139.470 and KRS 139.480 and any successor provisions or similar provisions. Owner is responsible for notifying Contractor in a timely manner as to the nature of items that are exempt from Sales Taxes and, upon Contractor's request, Owner will promptly provide Contractor with tax exemption certificates which, when properly utilized by Contractor, will allow Contractor to purchase such items from Subcontractors without payment of Sales Taxes. Contractor shall present any such certificate to the Subcontractor on a timely basis as needed to qualify for the applicable exemption.

8.2.5 <u>Indemnity and Assessment</u>. Owner shall defend and indemnify and hold Contractor harmless from and against any and all claims for, and resulting liability for Sales Taxes and related penalties and interest, and any dispute resolution costs and attorneys' fees that may be asserted on all items which Contractor purchased under resale or other exemption certificates provided by Owner and for which Sales Taxes are later assessed, **provided** that any resulting liability is not caused by the act or omission of Contractor. With Contractor's consent, Owner, at its own expense, will have the right to direct the basis on which any tax assessment will be paid or contested and to control any contest leading to the settlement of assessed taxes. Notwithstanding anything to the contractor or Owner's interests regarding any tax assessments and/or litigation. Contractor shall cooperate with Owner in connection with any audit or contest.

8.2.6 <u>Other Taxes</u>. Contractor shall be responsible for any taxes on Construction Aids used in the performance of the Work. Owner shall be responsible for property taxes, if any, on Equipment, Materials and Consumables purchased by Contractor on behalf of Owner from and after delivery to the Job Site.

8.2.7 Tax and Accounting Information. Contractor shall provide assistance as reasonably requested by Owner or its tax consultant(s), in confirming eligibility and qualification for exemptions from Sales Taxes (and any other exemptions) to the relevant Governmental Authorities. From time to time during the term of this Agreement and within thirty information (30) Days of a request therefor, Contractor shall provide Owner with information, including regarding quantities, descriptions, costs, and allocations of property acquired in connection with the Work as reasonably requested by Owner in connection with the preparation of Owner's tax returns, Owner's defense of its tax treatment of such items, for the purpose of satisfying regulatory requirements or as otherwise required in connection with calculating, obtaining exemption from, or rebate of. Sales Tax. Contractor agrees to participate in any Sales Taxes exemption or other exemption or rebate programs identified by Owner, to complete and deliver the applicable documentation to obtain tax-exemption for purchases made by Contractor, including for Equipment and Materials, and to pass any Sales Tax savings or rebates through to Owner.

Progress Payments. Owner shall pay Contractor for the Work in monthly 8.3 installments based upon the full and verified progress not to exceed amounts established in the Milestone Payment Schedule (based on completed Work) set forth in Exhibit B. Notwithstanding anything herein to the contrary, in no event may Contractor submit an Application For Payment that would exceed the aggregate payment amount specified for such month in Exhibit B. The Parties will adjust the payment Project Schedule set forth in Exhibit D as may be necessary and agreeable to prorate any change in the Contract Price over the installments yet to be made. If there is a substantial change in the Project Schedule, a corresponding change in the Milestone Payment Schedule set forth in Exhibit B will be made.

Day of each calendar **Application For Payment.** 8.4 month, Contractor shall furnish Owner with an Application For Payment for Work completed through the last Day of the previous calendar month, accompanied by the substantiating data required by Section 8.6. Prior to delivery of the Full Notice to Proceed to Contractor and Contractor's written acknowledgement thereof, the aggregate compensation for which Contractor may submit a monthly Application For Payment will not exceed the limit for such month as set forth in Exhibit B.

Payment of Substantiated Amount. Owner shall pay Contractor the amount of 8.5 each Application For Payment which has been substantiated by Owner, less Retainage and other amounts properly withheld hereunder. Owner shall make such payment after receipt of the complete Application For Payment, subject to the provisions of this Article 8. If Owner fails to make a payment in respect of an Application For Payment when due or improperly withholds amounts due to Contractor, interest will accrue on such overdue amounts at the Agreed Rate from the date such amount was due to have been paid to and until (but not including) the date it is paid. Contractor understands and agrees that Owner's payment determination under this Article 8 may be made in conjunction with the Financing Parties and that approval of the Financing Parties may be required prior to making payment in accordance with this Article 8, subject to payment being due after receipt of the completed Application For Payment. Payments made to Contractor in respect of Work 12/29/111

12/30/11

performed by Subcontractors constitute trust funds for such Subcontractors and shall not be commingled by Contractor with other funds of Contractor or its Affiliates.

8.6 Supporting Documentation. Each Application For Payment submitted by Contractor must be accompanied by the following, all in form and substance satisfactory to Owner:

- (i) a duly executed or acknowledged Contractor's certification stating that:
 (a) all Subcontractors have been paid amounts properly due under their respective subcontracts and identifying all Major Subcontractors with whom Contractor has entered into subcontracts; (b) the applicable Work has been performed in accordance with and complies with this Agreement;
 (c) it has reviewed all financial information contained in the invoice and it is true, correct and complete; and (d) no Liens or Claims have been filed or commenced in connection with the Work;
- (ii) a detailed description of the Work performed during the period as to which the Application for Payment relates; photographic and other documentation of the status and completion of the Work; reasonable documentation demonstrating achievement of milestones and otherwise demonstrating the earned value of the Work;
- (iii) duly executed partial or final lien waivers, as appropriate, in the forms set forth in Exhibit F-5 from Contractor and from all Major Subcontractors that are to receive payment. The final Application For Payment must be accompanied by final and full waivers of claims and Liens from Contractor and, to the extent not previously provided, Major Subcontractors entitled to receive payment in connection with the performance of the Work;
- (iv) a copy of the status report pursuant to Section 4.23.1 for the previous month;
- (v) a copy of all invoices pursuant to which Contractor has paid sales or use taxes for which it seeks reimbursement; and
- (vi) such other information, documents, and other materials: (a) reasonably required by Owner or this Agreement or (b) as may be required by the laws or customs of the jurisdiction in which the Job Site is located in order to protect the owners of the Ghent Generating Station Site from Liens or other liabilities.

8.7 Withholding to Protect Owner from Loss. Owner may, without prejudice to other rights of Owner hereunder, withhold payment on an Application For Payment or any other amount due to Contractor or a portion thereof (or draw on the Retainage if sufficient funds to withhold are not then available) to the extent such payment is disputed by Owner or because of:

12/20/11
- (i) Contractor's failure to carry out the Work in accordance with this Agreement or any material breach of this Agreement;
- (ii) other amounts due to Owner from Contractor under this Agreement, including liquidated damages then due and owing;
- (iii) the existence of defective, deficient or nonconforming Work not yet corrected by Contractor whether or not payment for such Work pursuant to Section 8.5 has been previously made. Contractor may invoice Owner for such amounts withheld in the next regular Application For Payment made after correction or completion of such Work;
- (iv) an amount equal to party to complete outstanding Punch List Items. Amounts withheld for completion of Punch List Items may be invoiced by Contractor in the Application For Payment immediately following satisfactory completion of such Punch List Items; or
- (v) Liens filed or Claims commenced by any Person that has performed a portion of the Work unless Contractor has furnished an acceptable bond (as determined in accordance with this Agreement) to protect Owner therefrom.

8.8 Final Payment.

8.8.1 <u>Reconciliation</u>. As a condition of final payment hereunder, Contractor shall have submitted a statement summarizing and reconciling all previous Applications For Payment, payments by Owner, Change Orders, and the Retainage. Subject to the provisions of this Agreement,

Owner shall pay Contractor all remaining amounts due. Notwithstanding anything to the contrary contained herein, the final payment will not become due and payable until: (i) a Certificate of Final Completion has been executed by Owner; (ii) Owner has received all warranties, Operating and Maintenance Manuals, schematics, Design Documents, performance testing data, As-Built Drawings and such other items as are required by this Agreement; (iii) all Permits or other approvals required of Contractor have been submitted to Owner; and (iv) the conditions of Section 8.8.2 and 8.8.3 have been properly completed. The receipt of final payment constitutes a waiver by Contractor of all claims against Owner (and its property) not previously made in writing by Contractor, except that nothing herein may be construed to imply a waiver of any right to any amount which is the subject of a written protest in accordance with Article 23 at the time final payment is made. Owner shall make final payment to Contractor

or have otherwise

2-29.1.

occurrea,

8.8.2 <u>Release</u>. As a condition of final payment and surrender of the Retainage, Contractor shall submit to Owner a release and an affidavit, in the form set forth **Exhibit F-5**, that releases all claims of Contractor under the Agreement except for claims

12/30/W Sr.

specifically identified in the release and represents that all indebtedness connected with the Work for which Owner or its property might in any way be responsible has been paid, waived or otherwise satisfied; but if any such indebtedness has not been satisfied, except for claims specifically identified in the release, Contractor may satisfy this obligation if it furnishes a bond reasonably satisfactory to Owner to indemnify Owner against any such item of responsibility or obligation. If any Claim of any kind or nature whatsoever is filed in respect of the Work and such Claim or any Lien arises from or is alleged to arise from any failure of Contractor or any Subcontractor to pay the indebtedness connected with the Work, Contractor shall indemnify, defend and hold Owner harmless for amounts that Owner must pay, in defending or discharging such Claim or Lien, including all costs, reasonable attorneys' fees, charges and interest. This provision will survive any expiration or termination of this Agreement.

Satisfaction of Obligation. Notwithstanding any provision to the contrary 8.8.3 in this Agreement, Owner and Contractor acknowledge and agree that Contractor will not be entitled to final payment of the Contract Price or release of the Retainage unless and until Contractor has achieved Final Completion.

Disputed Applications For Payment. If there is any dispute about any amount 8.9 which is requested by Contractor or which is claimed by Owner to be due and payable by Contractor, the amount not in dispute will be promptly paid in accordance with the provisions hereof, and any deduction of a disputed amount that is not specifically agreed to by Contractor or Owner, as applicable, and which is then determined by arbitration, litigation, or by mutual agreement, to have been improperly withheld will be promptly paid by Owner or Contractor, as applicable, together with interest from the date such amount otherwise would have been payable to the date of payment at the Agreed Rate.

Payment of Subcontractors. Contractor shall promptly pay each Subcontractor 8.10 when due the amount to which such Subcontractor is entitled. Contractor shall, by an appropriate agreement with each Subcontractor, require each Subcontractor to make payments to its sub-subcontractors on the same basis. Owner has no obligation to pay or to see to the payment of any monies to any Subcontractor except as may otherwise be required by Applicable Law, in which event Contractor shall immediately reimburse Owner therefor, with interest thereon at the Agreed Rate.

22

12/30/11

1230/11 Cn





CONTRACT CHANGES

10.1 Owner-Initiated Changes.

10.1.1 <u>Change Order Requests</u>. Owner may, from time to time, without invalidating this Agreement, order changes in (including additions to or deletions of) the Work or changes in the time for or sequence of completion of all or any portion of the Work ("Changes"), by notification in writing to Contractor (such notice, which will be substantially in the form of Exhibit F-3 hereto, a "Change Order Request"), and the Contract Price, Project Schedule, or other applicable provision of this Agreement will be adjusted accordingly in a Change Order. Each Change Order Request will be accompanied by a description of the Changes requested.

10.1.2 Contractor Response. Contractor shall, within fifteen (15) Days after receipt of a Change Order Request (unless such time period is extended by mutual agreement of Contractor and Owner or as otherwise herein provided), provide Owner for its review and approval, with a written description of any adjustment to the Contract Price, the Project Schedule or any other effect on the Work or the terms of this Agreement resulting from the Changes set forth in the Change Order Request. Any adjustment arising out of a Change will be negotiated by Owner and Contractor on a completely open-book basis and all such adjustments as to Contract Price shall be made on a basis consistent with the basis of the Contract Price build-up set forth in Exhibit C. Contractor, among other things, shall provide Owner with an itemization or estimate of: (i) total job hours (home office and construction by craft) and associated salaries, wages, premiums and benefits; (ii) estimated quantities, qualities (as required) and direct costs of Equipment and Materials; (iii) any bids received from any Subcontractors actually contacted by Contractor in connection with such Changes; (iv) all other direct costs associated with the performance of the Work, including travel and living, reproduction, printing, consultants, Subcontractors, costs of Contractor-owned equipment at Contractor's usual and customary rates; (v) applicable contingencies for the foregoing cost items appropriate and customary for the degree of uncertainty in the estimate: and (vi) the impact, if any, on the Project Schedule (the "Contractor Response"). Itemized proposals are also required for Subcontractor costs.

12/2/11

12/30/1 52

Contractor and Owner shall use their respective good faith efforts to agree on a lump sum, fixed-price for such Change. If Owner and Contractor cannot agree on the price for such Change, Owner may, at its option, direct Contractor to proceed on an openbook time and materials basis with no contingency (with all items of cost covered by Exhibit R being charged at the amounts and rates set forth on Exhibit R),

10.1.3 Owner's Response. Owner shall, within twenty (20) Days after receipt of Contractor Response (the "Owner Response Period"): (1) notify Contractor as to whether it agrees or disagrees with such Contractor Response, and of Owner's position regarding the effect of the Changes and the Change Order Request on the Contract Price, Project Schedule, or other applicable provision of this Agreement and (ii) embody the agreed upon changes in the Work, the Contract Price, or the Project Schedule in a document substantially in the form of Exhibit F-3 to be executed by the Parties (a "Change Order"). During the Owner Response Period, the Owner Representative and the Contractor's Representative shall make themselves available and shall use reasonable efforts to meet or otherwise confer to discuss the Change Order Request, Contractor Response and the estimates therein contained, and to answer any questions or clarify any information provided with respect thereto. Owner or the Owner Representative may request the Contractor's Representative to provide additional reasonable information or further information and data to the extent Contractor has failed to provide such information and data required to be provided or if there are errors, mistakes, or omissions in any information or data previously provided as part of the estimates contained in the Contractor Response. If Contractor fails to provide data or information required to be provided as part of the Contractor Response or fails to correct any errors or mistakes in such information or data, which prevents Owner from properly analyzing such data, or information, the Owner Response Period will be extended by that period of time commencing on the date Contractor receives notice of such failure, error, or mistake and ending on the date the correct data and information is received by Owner.

10.1.4 Owner Authorization. Contractor is not required to perform any Change until a Change Order has been issued therefor or Owner has expressly authorized or directed Contractor in writing to perform a Change prior to such approval, which it is entitled to do. If Owner so authorizes or directs Contractor to proceed with the Change set forth in a Change Order Request prior to issuing a Change Order therefor (the "Owner Authorization"), Owner shall, as part of such Owner Authorization: (i) acknowledge in writing to Contractor that it will issue a Change Order therefor, upon agreement on the effect of the Changes on the Contract Price and Project Schedule and (ii) agree to pay Contractor in accordance with the invoicing procedures contained herein any undisputed portion of Contractor's proposed change in the Contract Price resulting from such Changes. Upon receiving such Change Order or such Owner Authorization, Contractor shall perform the approved or authorized Changes in accordance with and subject to all of the terms of this Agreement. Contractor may not suspend, in whole or in part, performance of the Work during any dispute over any Changes set forth in the Change Order Request or an Owner Authorization or during the review and negotiation of any Change Order based thereon (or any adjustment to the Contract Price or Project



Schedule to be set forth therein) unless directed to do so by Owner. Notwithstanding anything to the contrary herein, if directed pursuant to an Owner Authorization issued in accordance with the provisions hereof to proceed with Work as to which a dispute exists about whether such Work is in or out of the scope of Work required of Contractor hereunder, pending resolution of such dispute, Contractor shall (without waiving any rights or remedies with respect to such Change or dispute) proceed with such Work.

10.2 Contractor Change Notice. If Contractor wishes to make a claim for an adjustment of the Contract Price, an adjustment of the Project Schedule, or Change in any other applicable provision under this Agreement (on account of the occurrence of an Excusable Event), Contractor must give notice ("Contractor Change Notice") in accordance with Article 9. This Contractor Change Notice must be given by Contractor before proceeding to execute any additional Work, except in an emergency endangering life or property in which case Contractor may act to prevent threatened damage, injury, or loss. Contractor shall document and otherwise support any Contractor Change Notice on the same basis and within the thirty (30) and sixty (60) Day periods specified in Article 9. Owner shall respond to all such Contractor, setting forth Owner's position, and, if appropriate, issuing a Change Order or Owner Authorization based thereon. All Contractor Change Notices must include sufficient documentation for Owner to be able to make a well-informed decision.

10.3 Emergencies. Any request for a Change claimed by Contractor on account of a need to take action in light of an imminent threat to the health and safety of individuals or property will be determined by the Parties based upon the facts of each such incident.

10.4 Resolution. In the event of a failure to agree to any adjustment of this Agreement, including, to the Contract Price or Project Schedule as required by the terms of this Agreement as a result of a Change Order Request or a Contractor Change Notice, either Party may demand resolution of such issues in accordance with Article 23.

10.5 Contractor Caused Delays. To the extent any delay or suspension is caused by Contractor or any of its Subcontractors, no adjustment will be made to the Contract Price, Project Schedule or other provision of this Agreement.

ARTICLE 11

TEST AND INSPECTIONS

2-29/11/2

11.1 Testing. Contractor shall conduct, arrange, or obtain (at its sole expense) all inspections, tests, including the Performance Guarantee Tests required to be performed to meet its obligations under this Agreement (including all certification testing and associated reports required by Applicable Law or applicable Codes and Standards that must be conducted by a qualified independent party), tests that are necessary for the proper execution and completion of the Work. Except as provided in Section 6.12, Contractor shall provide for or arrange for all testing personnel (excluding operating personnel to be provided by Owner in accordance with Section 5.9). The Performance Guarantee Tests must be conducted: (i) within sixty (60) Days after completion of hot commissioning on a date selected by the Parties; (ii) when all Systems

12/30/11

are operating as designed (without temporary equipment); (iii) when the Ghent CCR Transport System is operating in full automatic mode; (iv) when commencing performance testing is consistent with Good Utility Practices; and (v) pursuant to the Performance Guarantee Test Procedures and in strict accordance with Applicable Law (except, that with respect to noise or emission standards or limitations, the Performance Guarantee Tests shall be conducted instead in strict accordance with the sound and emission standards set forth in Exhibit G). A draft of the proposed Performance Guarantee Test Procedures shall be prepared by Contractor and delivered to Owner not less than one hundred twenty (120) Days prior to the Scheduled Mechanical Completion Date. Owner shall review such draft and provide written comments to Contractor with thirty (30) Days of receipt of the draft Performance Guarantee Test Procedures. Contractor and Owner shall cooperate and diligently work to complete an agreed final version of the Performance Guarantee Test Procedures no later than thirty (30) Days prior to the Scheduled Mechanical Completion Date, Contractor and Owner shall cooperate and in good faith finalize Performance Guarantee Test Procedures in accordance with Exhibit G.

Witnessing Tests and Inspection. Contractor shall perform all inspection, pre-11.2 Performance Testing, expediting, quality surveillances, and traffic services as necessary for the performance of the Work. Contractor's responsibilities under this Section 11.2 include inspecting and testing such Work, including Equipment and Materials, as is customarily inspected or tested in accordance with Professional Standards, including inspecting work in progress at intervals appropriate to the stage of construction, fabrication, or shipment off of the Job Site as necessary to ensure that such Work is proceeding in accordance with this Agreement and the Project Schedule. All third party inspections, tests, or approvals must be performed by qualified organizations acceptable to Contractor and Owner. If Applicable Law requires any Work to specifically be inspected, tested, or approved, Contractor shall assume full responsibility therefor and furnish to Owner the required certificates of inspection, testing, or approval. Not later than sixty (60) Days prior to the first expected delivery of Equipment or Materials to the Ghent Generating Station Site, Contractor will supply to Owner a quality surveillance plan for all Equipment and Materials that will be inspected by Contractor. A chart, accompanying each monthly progress report delivered by Contractor shall specify the date, time, and location of factory tests, inspections, and witness points of which Contractor is then aware of and intends to witness with respect to Equipment and Materials or other work to be provided or performed by its Subcontractors in the forty five (45) Day period following the Month in which each such progress report is delivered to Owner. If and when Contractor obtains new information about such factory tests, inspections, or witness points or other factory tests, inspections, or witness points that was not available to it when the monthly status report was delivered, it will promptly advise Owner and update such information as necessary to allow Owner a reasonable opportunity to attend such event. Contractor shall give reasonable notice of changes to such dates, times, and locations to allow Owner to make arrangements to attend, it being understood that tests, inspections, or witness points to be performed outside the United States will require additional notice. Owner and its invitees have the right, but not the obligation, to attend any inspections, tests, or approvals of the Work. In no event shall Contractor's obligation to provide notice regarding certain tests, inspections, or approvals or Owner's and its designees' right to attend tests, inspections, or approvals limit, delay, or modify Contractor's obligation to perform all tests, inspections, or obtain approvals required by this Agreement. Successful completion of factory or other off-Job Site tests is a precondition to shipment of such Equipment to the Job Site or other Contractor storage facility, unless otherwise agreed by Owner. Contractor shall

12/30/11 SL thereafter implement such plan and reperform and demonstrate that such test has been passed. Should Contractor fail to give proper notice under this Section 11.2, at Owner's option, Contractor shall reperform or reinspect any such test or inspection as to which Contractor failed to give proper notice if Owner has reviewed the test results (which are to be provided to Owner within seven (7) Days of such failure or as soon thereafter as test results are available to Contractor) and gives notice to Contractor questioning the validity, accuracy, or completeness thereof. Attendance by Owner at any such test or inspection despite the lack of proper notice will waive Owner's right to cause Contractor to so retest or reinspect.

Defects During Performance Guarantee Tests. Without limiting the 11.3 requirements of Section 6.12, after completion of any Performance Guarantee Test or any retest, Owner and Contractor shall consult concerning the results of such test and Owner shall advise Contractor in writing of any defects, deficiencies, or needed adjustments in the Work that they have discovered or observed during the performance thereof. If Contractor is notified of such defects, deficiencies, or adjustments, Contractor shall immediately commence and promptly: (i) complete corrective measures to rectify such defects or deficiencies and any other defects or deficiencies of which it is aware (including, replacement of any defective parts) and (ii) make any necessary adjustments, in each case, at Contractor's expense. As soon as reasonably practicable but in no event more than twenty-four (24) hours after the completion of such Performance Guarantee Tests (or as soon thereafter as such reports are first available to Contractor) following the applicable test or inspection, Contractor shall forward copies of all test or inspection results together with such other information Owner reasonably requires in relation thereto.

Retests. Prior to any retest of the Ghent CCR Transport System, Contractor shall 11.4 give reasonable notice advising Owner that all Defects and deficiencies have been corrected and all necessary adjustments have been made. Such notice must identify the date upon which the Ghent CCR Transport System will be ready for such retesting. Within five (5) Days, Owner shall review the adjustments or corrections made by Contractor and will determine whether a retest is warranted. Following a favorable determination by Owner, except as provided in Section 6.12, Contractor shall promptly reperform the retest. After each such retest, Owner and Contractor shall consult concerning the results thereof, and Owner shall advise Contractor in writing of any additional or remaining defects or deficiencies that must be corrected by Contractor. If the Performance Guarantee Tests are not successfully completed, they must continue to be repeated in accordance with this Agreement.

ARTICLE 12

CORRECTION OF WORK

Correction of Work. Prior to the Commercial Operation Date, Contractor shall, 12.1 at the earliest practical opportunity, correct, repair, or replace any portion of the Work that is defective or does not conform to the requirements of this Agreement, without regard to the stage of completion of the Work or the time or manner of discovery of the defect or nonconformance. If other portions of the Work are adversely affected by or are damaged by such defective or nonconforming Work, Contractor shall, at its sole cost and expense and at the earliest practical opportunity, correct, repair or replace such affected or damaged Work. Contractor shall bear the 2/22/11

12/30/11

cost of correcting such defective or nonconforming Work, including additional testing and inspections and compensation for any Design or engineering services and expenses made necessary thereby.

12.2 Urgent Repairs. If by reason of any accident, failure, or event occurring to, in, or in connection with the Work or any part thereof either during the execution of the Work or during any period of warranty hereunder, any remedial or other work or repair is in the opinion of Owner urgently necessary and Contractor is unable or unwilling at once to do such work or repair, Owner may, with its own forces or other contractors, do such work or repair. If the work or repair so done is Work which Contractor was liable to do at its own expense under this Agreement, all costs and expenses incurred by Owner in so doing less any insurance proceeds received as a result of such event shall be paid by Contractor to Owner on demand. Owner, as soon after the occurrence of any such emergency as may be reasonably practicable, shall notify Contractor thereof in writing.

12.3 Damage to Existing Facilities. To the extent that Contractor or its Subcontractors cause any physical damage or loss to any equipment, structure, or portion of the Work (including Ghent CCR Transport System) as to which care, custody, and control and risk of loss has passed to Owner or to the Existing Facilities (but not Work for which care, custody, and control and risk of loss has not yet passed to Owner) in the course of the performance of the Work,

Owner otherwise releases Contractor from liability for such damage or loss. Payments to be made by Contractor under this Section 12.3 shall be made to Owner within fifteen (15) Business Days of Owner's demand therefor.



Str. H

12/30/11 52



12/30/11 42

PROTECTION OF PERSONS AND PROPERTY

Safety Programs. Contractor shall be responsible for initiating, maintaining, and 14.1 supervising all safety precautions and programs in connection with the performance of this Agreement, including a fitness for duty policy and other appropriate precautions and programs for areas in and around the Ghent Generating Station Site. Within one hundred twenty (120) Days of the Effective Date, Contractor shall prepare and deliver to Owner a plan to ensure proper health, safety, and environmentally sound practices are employed and enforced in the performance of the Work. At a minimum, such plan shall require Contractor to comply, and shall cause all Subcontractors to comply, with those rules, regulations and procedures set forth in Exhibit H. The efficacy or implementation of such plan will not relieve Contractor of its obligations under this Agreement. If Owner becomes aware of any Work or the performance of any Work, that it reasonably believes constitutes a threat to the health or safety of persons. property, or the environment, then, without limiting any other rights of Owner hereunder, Owner may (but will not be obligated to) immediately suspend the performance of the Work and thereafter promptly advise Contractor of the cause therefor. Such suspension may be maintained until such cause is removed. All costs related to such suspension and any other adverse impact on Contractor or the Work attributable thereto shall be the responsibility of Contractor and no relief under this Contract shall be provided. Owner, in its reasonable opinion, may exclude from the Ghent Generating Station Site any individual whose conduct is prejudicial to safety, health, protection of persons, property, or the environment, or is found or suspected to be in violation or in disregard of the requirements of this Article, this Agreement, or Applicable Law.

14.2 Applicable Law. Contractor shall give notices and comply with Applicable Law bearing on the safety of Persons, property and the environment or their protection from damage, injury or loss, including all standards of the U.S. Occupational Safety & Health Administration applicable to the Work.

14.3 Safety Precautions. Contractor shall take all reasonable precautions for safety of, and shall provide all reasonable protection to prevent damage, injury or loss to:

- (i) employees and Subcontractors or other individuals performing the Work and all other Persons who may be affected thereby, including other individuals on the Ghent Generating Station Site;
- (ii) the Work and Equipment, Materials, Consumables, and Construction Aids, whether in storage on or off of the Job Site, under the care, custody or control of Contractor or Subcontractors; and

12/29/14

12/3:11

(iii) other property at the Ghent Generating Station Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, waterways, roadways, structures and utilities, it being agreed that Contractor will be liable for the cost of damages incurred by Owner and caused by Contractor or its Subcontractors.

14.4 Community Relations. Contractor recognizes the introduction of Contractor's workforce (and those of its Subcontractors) has the potential to disrupt the local community. Contractor will proactively alert those individuals for whom it is responsible on the Ghent Generating Station Site to exercise due caution entering and leaving the Ghent Generating Station Site and to otherwise conduct themselves in a manner consistent with good community relations.

14.5 Security. Contractor shall take all precautions and measures as may be necessary to secure the Job Site and other portions of the Ghent Generating Station Site on which Work is being performed at all hours, including evenings, holidays and non-work hours. Contractor is not entitled to rely on security provided by Owner. Contractor shall coordinate its Job Site security functions with Owner's existing security functions so as not to detract from, or impose upon, existing security measures at the Ghent Generating Station Site. Contractor shall erect, maintain, or undertake, as required by existing conditions and the performance of this Agreement, all reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notifying Owner and users of adjacent sites and utilities. Such precautions may include the provision of security guards and/or fencing.

14.6 Dangerous Materials. When use or storage of dangerous materials or equipment or unusual methods are necessary for execution of the Work, Contractor shall exercise utmost care and carry on such activities only under the supervision of properly qualified personnel. Explosives shall not be used on the Job Site or the Ghent Generating Station Site, without the prior written consent of Owner.

14.7 Safety Personnel. Contractor shall assign at least one full-time Job Site safety officer who shall be responsible for introducing, administering, and monitoring procedures to promote safe working conditions on the Job Site (and other areas of the Ghent Generating Station Site where Work is to be performed) and compliance with Applicable Law. Contractor shall provide an appropriate medical facility at the Job Site.

14.8 Loading. Contractor must not load or permit any part of the construction, the Job Site or other portions of the Ghent Generating Station Site to be loaded so as to endanger the safety of Persons or property.

14.9 Notices to Owner. Contractor shall immediately (and in each case, but in no event later than a period of time equal to one-half the amount of time Owner has to report any incident to a Governmental Authority) notify Owner by telephone or messenger of any and all potential lost time accidents, recordable injuries (as defined under OSHA), and any property damage arising out of or in connection with the Work. Contractor shall provide Owner with a

12/2A/M

12/30/11

written report, giving of full details and statements of any witnesses within twenty-four (24) hours of the occurrence of the event.

14.10 Code of Business Conduct. Contractor hereby acknowledges receipt of the PPL Corporation Standards of Integrity and agrees to comply therewith as it may be amended from time to time.

14.11 Hazards and Training. Contractor shall furnish adequate numbers of trained, qualified, and experienced personnel and appropriate safety and other equipment in first-class condition, suitable for performance of the Work. Such personnel shall be skilled and properly trained to perform the Work and recognize all hazards associated with the Work. Without limiting the foregoing, Contractor shall participate in any safety orientation of Owner (or any Owner Affiliate) familiarization initiatives related to safety and shall strictly comply with any monitoring initiatives as determined by Owner. Contractor acknowledges that it has inspected all equipment, structures, and property of Owner to determine the existence of hazards incident to working thereon or thereabouts, and has adopted suitable precautions and methods for the protection and safety of its employees and the property.

14.12 Drug and Alcohol. No individual will perform any of the Work while under the influence of any illegal or controlled substance or alcohol. No alcohol may be consumed four (4) hours prior to any individual's performance of the Work or anytime during the workday. An individual will be deemed under the influence of alcohol if a level of .02 percent blood alcohol or greater is found. In addition to the requirements of the drug and alcohol testing program set forth in Exhibit H, Contractor shall: (i) institute a random drug and alcohol testing program covering all individuals that will perform any of the Work; (ii) promptly, upon the written request of Owner, perform drug and/or alcohol tests on all individuals that will perform any of the Work; and (iii) perform drug and alcohol tests on any individual that will perform any of the Work under either of the following circumstances: (a) where the individual's performance either contributed to an accident or cannot be completely discounted as a contributing factor to an accident which involves off-site medical treatment of any individual or (b) where Owner determines in its sole discretion that there is reasonable cause to believe such individual is using drugs or alcohol or may otherwise be unfit for duty. Individuals tested in accordance with clause (a) or (b) above will not be permitted to perform any Work until the test results are established. Contractor shall be solely responsible for administering and conducting drug and alcohol testing, as set forth herein, at Contractor's sole expense. As applicable and in addition to any other requirements under this Agreement, Contractor shall develop and strictly comply with any and all alcohol and drug testing requirements required by Applicable Law. Notwithstanding anything to the contrary herein, if the rules and regulations of Owner pertaining to the Ghent Generating Station Site are at any time more stringent than the requirements of this Agreement, Contractor will comply and cause its Subcontractors to comply with such more stringent rules and regulations.

14.13 Dust Emissions. During execution of the Work, the Contractor shall ensure that no visible dust and no fugitive dust be allowed to leave the Job Site.

ARTICLE 15

12/29/11

12130/11 5%

SEPARATE CONTRACTORS AND **ACTIVITIES BY OWNER**

Separate Work. Owner reserves the right to perform either with its own forces 15.1 or through other contractors and subcontractors construction or operations related to the Ghent CCR Transport System or any other construction or other work at the Job Site or the Ghent Generating Station Site.

Contractor shall use reasonable best efforts to arrange the 15.2 Integration. performance of the Work so that the Work and the work of Owner's forces or any of its separate contractors are properly integrated, joined in an acceptable manner and performed in the proper sequence without any disruption or damage to the Work, the work or business operations of Owner or any work of Owner's forces or its separate contractors.

Coordination. Contractor shall provide for coordination of the activities of 15.3 Contractor's, and its Subcontractors' forces with the activities of Owner's forces and each of its separate contractors and Owner, as applicable in accordance with the coordination plan established pursuant to Section 4.20.

15.4 Use of Job Sife. Contractor shall afford all separate contractors reasonable opportunity for storage of their materials and equipment and for performance of their work on the Job Site. Owner shall direct all separate contractors to cooperate with Contractor and to avoid actions that could unreasonably interfere with the activities of Contractor.

15.5 Deficiency in Work of Owner and Separate Contractors. Without otherwise limiting Contractor's obligations under this Agreement, if part of Contractor's Work depends for proper execution or results upon construction or operations by Owner or another separate contractor of Owner, Contractor shall, prior to proceeding with that portion of the Work, promptly report to Owner any discrepancies or defects in such other construction or operations that would render it unsuitable for proper execution and results by Contractor. The Parties shall resolve in good faith any such discrepancies or defects or any disagreements relating thereto, and Owner shall correct or cause the separate contractor to correct its defects and deficiencies. Failure of Contractor to promptly report discrepancies or defects of which it has or upon reasonable investigation should have had knowledge shall constitute an acknowledgment by Contractor to Owner that Owner or separate contractor's completed or partially completed construction or operations are fit and proper to receive Contractor's Work, except as to discrepancies and defects not then reasonably discoverable.



12/30/11 in



12/30/11





REPRESENTATIONS AND WARRANTIES

17.1 Contractor. Contractor hereby represents and warrants the following to Owner, on and as of the Effective Date, which representations and warranties shall survive the execution and delivery of this Agreement, any termination of this Agreement and the final completion of the Work:

- (i) it is able to furnish the tools, Equipment, Materials, Consumables and Construction Aids, labor, supervision and demolition, Design and construction services required to complete the Work and perform its obligations hereunder, and has sufficient experience and competence to do so;
- (ii) it is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware;

12/2/1/1

į

t

- (iii) it is authorized to do business in the Commonwealth of Kentucky and properly licensed by all Governmental Authorities having jurisdiction over Contractor or the Work;
- (iv) Contractor is well acquainted with the Ghent Generating Station Site, including the general and local conditions, as well as other conditions that may influence the performance of the Work, including the requirements of Applicable Law, the other construction activities that will be performed on the Ghent Generating Station Site during performance of the Work, the condition of Existing Facilities on the Ghent Generating Station Site that will be utilized, integrated into, interconnected with or modified in the performance of the Work and building standards and trade practices affecting the Work;
- (v) no litigation, investigation or proceeding of or before any arbitrator or Governmental Authority is pending against or, to the knowledge of Contractor, threatened against or affecting Contractor or any of its properties, rights, revenues assets or the Work (a) which could reasonably be expected to have a material adverse effect on the properties, business, prospects, operations or financial condition of Contractor or (b) which could reasonably be expected to have a material adverse effect on the ability of Contractor to perform its obligations under this Agreement;
- (vi) this Agreement has been duly authorized, executed, and delivered by it and constitutes the legal, valid, and binding agreement of Contractor, enforceable against Contractor in accordance with its terms; and
- (vii) the Guarantee has been duly authorized, executed and delivered by Guarantor and constitutes the legal, valid and binding agreement of Guarantor, enforceable against Guarantor in accordance with its terms.

17.2 Owner. Owner hereby represents and warrants the following to Contractor, on and as of the Effective Date, which representations and warranties shall survive the execution and delivery of this Agreement, any termination of this Agreement and the final completion of the Work:

- (i) it is duly organized, validly existing and in good standing under the laws of the state of its formation and is duly qualified to do business in the Commonwealth of Kentucky;
- (ii) no litigation, investigation or proceeding of or before any arbitrator or Governmental Authority is pending against or, to its knowledge, threatened against or affecting it or any of its properties, rights, revenues, assets: (a) which could reasonably be expected to have a material adverse effect on the properties, business, prospects, operations or financial condition of it or (b) which could reasonably be expected to have a

12/22/11

material adverse effect on its ability to perform its obligations under this Agreement; and

(iii) this Agreement has been duly authorized, executed, and delivered by it and constitutes the legal, valid, and binding agreement of it, enforceable against it in accordance with its terms.



12/24/11

12/30/11 En



HAZARDOUS SUBSTANCES

19.1 Hazardous Substances.

19.1.1 Encountering Hazardous Substances. In order to give the Owner an opportunity to discover and, as appropriate remediate, any Pre-Existing Hazardous Substances, Contractor shall give Owner at least 30 days and no more than 60 days

advance written notice before it commences disturbing any areas of the Job Site it has not previously disturbed. If, in the course of performance of the Work, Contractor either spills, leaks, or releases (including threatened releases) Hazardous Substances into the environment or encounters Hazardous Substances (or a substance it reasonably believes is a Hazardous Substance) on the Job Site, except as contemplated by Section 19.1.3, or any portion of the Ghent Generating Station Site on which Work must occur, in such quantities or at such levels that may require investigation or remediation pursuant to Applicable Law, Contractor shall immediately suspend the Work in the area affected (except as provided below) and immediately orally report the condition to an individual on the Emergency Notification List and to the Owner Representative followed as soon as reasonably possible by a written notice to the Owner Representative. In any such event, the obligations and duties of the Parties are as follows:

- (i) Owner may direct Contractor to take appropriate immediate mitigating action;
- (ii) to the extent such condition involves: (a) a Hazardous Substance introduced to the Job Site or the Ghent Generating Station Site by Contractor, its Subcontractors or any Person for whom either may be responsible; (b) a Hazardous Substance contained in or packed with Equipment; (c) the negligent or intentional spill, leak, release, or threatened release by Contractor, its Subcontractors or any Person for whom either may be responsible of a Hazardous Substance used by Owner in the operation or maintenance of the Existing Facilities or the Ghent CCR Transport System or a portion thereof; provided that such Hazardous Substance was properly contained and labeled (or other adequate warning had been given of its existence); (d) a Contractor responsibility as provided in Section 19.1.3; or (e) the negligent or intentional exacerbation (or any exacerbation if Contractor failed to give the required notice under the first sentence of this Section 19.1.1) by Contractor of a Pre-Existing Hazardous Substance, then any investigation, response, removal, cleanup, or other remedial action required (1) to restore the status quo ante and (2) by Applicable Law or any Governmental Authorities (collectively, "Environmental Action") shall be performed by Contractor;
- (iii) if the condition does not involve a Hazardous Substance in such quantities and/or at such levels that may require investigation or remediation pursuant to Applicable Law, Contractor shall, promptly after receiving written notice from Owner authorizing Contractor to recommence activities in the subject area, resume the portion of the Work that had been suspended;
- (iv) if such condition involves a Pre-Existing Hazardous Substance, except to the extent as provided in clause (ii) above, then any Environmental Action shall be performed by Owner at its expense. Owner shall handle, remove and dispose of such materials in compliance with Applicable Law; and

12/30/11 En

except as otherwise required by Applicable Law, any Environmental (\mathbf{v}) Action, notification and other communication with third parties, including Governmental Authorities, and reports and documentation related to such Environmental Action shall be undertaken by Owner.

Contractor shall use diligent efforts to avoid any adverse effect on, or impediment to, the efforts undertaken by Owner, its agents, or independent contractors in connection with any Environmental Action or other remedial work Owner deems appropriate at the Job Site or the Ghent Generating Station Site during the term of this Agreement. The Parties acknowledge and agree that Contractor shall not commence or continue any construction activities on any portion of the Job Site or the Ghent Generating Station Site on, in, or under which Environmental Actions or other remedial actions are to be (or are being) performed until such actions are to the point where construction activities will not interfere with such actions, as evidenced by appropriate certifications from the applicable environmental engineer or remediation contractor and any required approvals of any applicable Governmental Authorities. Contractor agrees to use good faith diligent efforts to continue the unaffected portions of the Work and to adjust and schedule its activities at the Job Site or the Ghent Generating Station Site so as to minimize, to the extent reasonably practicable, any adverse effect on the cost and progress of the Work resulting from the performance of any remedial actions.

19.1.2 Contractor Obligations. Contractor shall not generate, dispose, bring, transport, or store (and shall prohibit Subcontractors from generating, disposing, bringing, transporting, or storing) Hazardous Substances to or on the Job Site or the Ghent Generating Station Site, and shall not utilize (and shall prohibit Subcontractors from utilizing) any construction materials or equipment (whether or not totally enclosed) containing asbestos, polychlorinated biphenyls, benzene, lead or urea formaldehyde; provided, however, that Contractor (and Subcontractors) may use and store (if properly containerized and labeled) in quantities reasonably necessary to perform the Work the following, but only in accordance with Applicable Law: gasoline, diesel fuel, fuel oil(s), gravel(s), lube oil(s), greases, sealant(s), combustible gases, form oil(s), solvent(s), adhesives, and all other materials, that are normally used or consumed in or during construction or testing of the work and its constituent Systems and Components. Contractor shall not bring or store any other Hazardous Substances to the Ghent Generating Station Site without the specific prior written authorization from Owner. Contractor shall provide Owner with Material Safety Data Sheets in English, ("MSDSs") properly completed covering any Hazardous Substance brought to the Ghent Generating Station Site and furnished by Contractor (or its Subcontractors). Contractor shall maintain on the Job Site, at all times, complete records and inventories, including MSDSs of materials described in this Section 19.1.2 that are being used by it or its Subcontractors, or any Persons for whose actions it is responsible on the Ghent Generating Station Site. Contractor shall be responsible for the management, prompt removal, cleanup and off-site disposal of Hazardous Substances brought to or generated at the Ghent Generating Station Site by Contractor, any Subcontractor or any Person for whose actions Contractor or any Subcontractor is responsible, including Hazardous Substances contained or packed in Equipment, except to the extent such Hazardous Substances are spilled, leaked, or released (including threatened releases) by Owner (or 2/29/11 any of its subcontractors) provided Contractor or any Subcontractor has utilized reasonable efforts to properly label and store such Hazardous Substances. In this regard, Contractor shall comply, and shall cause its Subcontractors to comply, with all Applicable Law and the Hazardous Substances Management Plan. Contractor shall have ownership of, and title to, all contaminated media introduced or created in performing its obligations under Section 19.1.1(ii) and this Section 19.1.2, and shall have sole responsibility in responding to such conditions including complying with reporting obligations, providing for access restrictions and warnings, manifesting and any other obligations under Applicable Law.

Owner shall determine whether reporting is required under Applicable Law and shall initiate the reporting of conditions on the Job Site if required. Unless Owner provides written authorization, Contractor acknowledges and agrees that it shall not report, or cause any other person to report, any information regarding environmental conditions to any Governmental Authority, except, as required by Applicable Law. Contractor shall use its best efforts to afford Owner an opportunity to present all objections and defenses Owner or Contractor may have prior to the making of such report by Contractor. Contractor retains its obligation to report to Owner any conditions created by activities of Contractor, its Subcontractors, or other Persons whose actions Contractor or any Subcontractor is responsible in the course of activities pursuant to this Agreement. Contractor shall be responsible for emergency notification to Owner, in accordance with Section 4.16, as well as additional immediate and follow-up reporting with respect to any spill or release of a Hazardous Substance into the environment at the Job Site or the Ghent Generating Station Site that occurs after the Effective Date arising out of its performance of the Work.

19.1.3 Hazardous Substances Identified by Contractor, Contractor acknowledges that the process of integrating the Ghent CCR Transport System into each of the Units requires the identification and removal of asbestos, lead paint, and possibly other Hazardous Substances from Unit structures that will be affected by the Work. Remediation of such Hazardous Substances will be the responsibility of Owner, but will be subject to the conditions set forth in this Section 19.1.3. Contractor shall perform such inspections of each of the Units as necessary prior to commencing any Work to integrate any portions of the Ghent CCR Transport System into the Unit or its structures. Contractor shall identify each specific area where Hazardous Substances (i) are or may be present; (ii) would be disturbed by performance of the Work involved in integrating the Ghent CCR Transport System into the each of the Units; and (iii) that need to be removed. Contractor shall carefully outline each specific area where remediation is to occur with brightly-colored spray paint. Contractor shall confine the marking of areas to the minimum area necessary to allow the Work to safely proceed. Such marking must clearly and distinctly identify and define the precise areas to be remediated. Contractor shall give Owner reasonable advance notice that it will be marking such areas and shall deliver a written notice to Owner when it has finished marking and is ready for Owner to commence remediation. Owner will have no less than thirty (30) Days to complete such remediation. Upon completion of such remediation, Owner will notify Contractor that it may proceed with the planned Work for such areas. If, during the course of its Work in such areas, Contractor discovers additional areas where Hazardous Substances require

12/30/11

2/24/11

removal that it did not mark as required hereunder, Contractor shall mark such areas as provided in this Section 19.1.3 and provide written notice to Owner to remove Hazardous Substances in such areas. Owner will use commercially reasonable efforts to remediate such additional areas as soon as possible but in every case within a reasonable period of time. The cost of all such remediations will be borne by Owner; provided, however, if Owner is requested by Contractor to: (a) accelerate any remediation work to be performed by Owner under this Section 19.1.3 or (b) remediate additional areas that Contractor did not mark at the time it marked other areas within the same general vicinity, Contractor shall be responsible for any incremental costs incurred on account of such acceleration or having to remobilize a third party contractor to perform such additional remediation. Notwithstanding anything in this Agreement to the contrary, Contractor is not entitled to any adjustment of this Agreement, by Change Order or otherwise, on account of Hazardous Substances (all of which constitute Pre-Existing Hazardous Substances) identified or required to be identified by Contractor in accordance with this Section 19.1.3.

19.1.4 Labeling. Contractor and its Subcontractors shall properly store, label and dispose of Hazardous Substances brought to the Ghent Generating Station Site by Contractor, its Subcontractors, and others for whom they are responsible, and train their employees in the safe usage and handling of Hazardous Substances, including any training that is required either by the Hazardous Substances Management Plan or Applicable Law.

ARTICLE 20

INDEMNIFICATION

Contractor's Indemnity. To the fullest extent permitted by Applicable Law, 20.1 Contractor shall indemnify, defend, and hold harmless Owner, Owner Engineer, and their respective successors, assigns, officers, directors, members, employees, agents, Affiliates, and representatives (collectively, the "Owner Indemnitee(s)"), from and against any and all claims, causes of action, proceedings, demands or suits (collectively, "Claims") and any and all judgments, liabilities, losses, expenses, damages, fines or penalties, including court costs, reasonable attorneys' fees and costs (whether incurred as the result of a third party claim or a claim to enforce any indemnity obligation of Contractor), and pre- and post-judgment interest (collectively "Liabilities"), to the extent arising from or in connection with: (i) Claims of third parties arising from the intentional wrongful acts or the negligent acts or omissions of Contractor, its Subcontractors, agents, or anyone directly or indirectly employed by them or anyone for whose acts they may be responsible, in connection with the Work or its other obligations under this Agreement without regard to the legal theory underlying such Claims or Liabilities, including strict liability; (ii) Claims by any Governmental Authority arising from violations of Applicable Law by Contractor, its Subcontractors, agents, or anyone directly or indirectly employed by them or anyone for whose acts they may be responsible, and rectification of the causes of such violations; (iii) Claims by any Governmental Authority for taxes that are the responsibility of Contractor or any of its Subcontractors or any of their respective agents or employees with respect to any payment for the Work made to or earned by Contractor or any of its Subcontractors or any of their respective agents or employees under this Agreement; 12/29/11 12/29/11

12/30/11 en

provided, however, this indemnity shall not apply to the extent any such Claims or Liabilities arise or result from the negligent or intentionally wrongful acts or omissions of Owner, its subcontractors, and their respective employees, representatives, and agents. This indemnification, defense, and hold hamless obligation shall survive the termination or expiration of this Agreement until the expiration of the applicable statutes of limitation therefor. In claims against any Owner Indemnitee under this Section 20.1 by an employee of Contractor, a Subcontractor, anyone employed by them or anyone for whose acts they may be liable, the indemnification obligation under this Section 20.1 is not limited by a limitation on the amount or type of damages, compensation or benefits payable by or for Contractor, a Subcontractor or any other above referenced Person under compensation acts, disability benefit acts, or other employee benefit acts.

Owner's Indemnity. To the fullest extent permitted by Applicable Law, Owner 20.2 shall indemnify, defend, and hold harmless Contractor, its Affiliates and their respective successors, assigns, officers, directors, employees, agents, and representatives (the "Contractor Indemnitec(s)"), from and against any and all Claims and Liabilities to the extent, arising from or in connection with Claims of third parties relating to the performance by Owner of its duties and obligations under this Agreement; provided, however, this indemnity shall not apply to the extent any such Claims or Liabilities arise or result from the negligent or intentionally wrongful acts or omissions of Contractor, its Subcontractors or any party for whom either may be liable. This indemnification, defense, and hold harmless obligation shall survive the termination or expiration of this Agreement until the expiration of the applicable statutes of limitation therefor. In claims against any Contractor Indemnitee under this Section 20.2 by an employee of Owner, anyone employed by it or anyone for whose acts it may be liable, the indemnification obligation under this Section 20.2 is not limited by a limitation on the amount or type of damages, compensation or benefits payable by or for Owner or any other above referenced Person under compensation acts, disability benefit acts, or other employee benefit acts.

20.3 Defense of Claims or Actions.

20.3.1 Notice. If any Claim is made or brought by any third party with respect to which an Owner Indemnitee or a Contractor Indemnitee entitled to indemnity under any indemnification provision of this Agreement, (individually, an "Indemnified Party" and collectively, the "Indemnified Parties") believes it is entitled to indemnification, the Indemnified Party shall give written notice of such Claim (a "Claim Notice") and a copy of the Claim, process and any legal pleading with respect thereto (if and to the extent available to the Indemnified Party) to the party that is required to provide indemnification under this Agreement (individually, an "Indemnifying Party" and collectively, the "Indemnifying Parties") promptly, but in each case within ten (10) Days of being served or otherwise informed of such Claim, process or legal pleading. Failure to give such notice in a timely manner will not diminish the indemnification obligations of the Indemnifying Parties under this Agreement except to the extent the failure or delay in giving such notice results in actual and material prejudice to the Indemnifying Party. Any Party seeking indemnification hereunder shall deliver to the Party from which indemnification is sought a detailed description (including the nature of the Claim and the amount thereof) of each Claim.

2/2/11

12/30/11 É.

20.3.2 Assumption of Defense. The Indemnifying Parties shall have the right to assume the defense of any such Claim. If the Indemnifying Parties wish to assume the defense of such Claim, such assumption shall be evidenced by written notice to the After such notice, the Indemnifying Parties shall engage Indemnified Parties. independent legal counsel of reputable standing selected by such Indemnifying Parties and reasonably acceptable to the Indemnified Parties, to assume the defense and may contest, pay, subject to Section 20.3.5, settle or compromise any such Claim on such terms and conditions as the Indemnifying Parties may determine. If the Indemnifying Parties assume the defense of any such Claim, the Indemnified Parties shall have the right to employ their own counsel, at their own expense; provided, however, if the Indemnified Parties have reasonably concluded and specifically notified the Indemnifying Parties that there may be specific defenses available to the Indemnified Parties which are in conflict with those available to the Indemnifying Parties, then the Indemnified Parties will be entitled to retain independent counsel at the Indemnifying Party's expense to assume the defense of the Indemnified Parties.

20.3.3 <u>Independent Counsel</u>. If the Indemnifying Parties do not agree in writing to assume the defense of such Claim, the Indemnified Parties may engage independent counsel of reputable standing selected by them to assume the defense and may contest, pay, settle or compromise any such Claim on such terms and conditions as the Indemnified Parties may determine; **provided**, **however**, that the Indemnified Parties shall not settle or compromise any Claim without the prior consent of the Indemnifying Parties if such Indemnifying Parties acknowledge in writing their liability for any Liabilities incurred or required to be paid in respect of such Claim and provide adequate assurances of their ability to satisfy any such Liabilities.

20.3.4 <u>Cooperation</u>. The Indemnified Parties and the Indemnifying Parties shall cooperate in good faith in connection with any common defense.

20.3.5 <u>Settlement</u>. The Indemnifying Parties shall not, except with the consent of the Indemnified Parties, enter into any settlement or consent to entry of any judgment that: (i) does not include as an unconditional term thereof the giving by the Person or Persons asserting such Claim to all Indemnified Parties of an unconditional release from all liability with respect to such Claim; (ii) includes a statement as to an admission of fault, culpability or failure to act by or on behalf of any Indemnified Party; or (iii) imposes any conditions, future obligations or limitations on any of the Indemnified Parties; **provided**, **however**, that for settlements with any Governmental Authority, the consent of the Indemnified Parties shall not be unreasonably withheld or delayed and in making such determination, the Indemnified Party shall exercise its reasonable business judgment (without taking into account the indemnity afforded hereby) consistent with usual and customary settlement terms, conditions and practices for the applicable Governmental Authority.

12/30/11 A



INSURANCE

Contractor (and its Subcontractors) shall provide and maintain the insurance specified in **Exhibit I** in accordance with the terms and provisions thereof.

ARTICLE 22

TITLE & RISK OF LOSS

22.1 Transfer of Title. Transfer of title to the Work (or any portion thereof) will pass to Owner upon the earlier of delivery to Owner or the Job Site, incorporation into the Work or the Unit, or upon payment of the amount properly due under an Application For Payment covering such Work, notwithstanding any amount retained and other amounts withheld by Owner in accordance with the terms of this Agreement; provided, that Contractor shall cause title to be transferred to Owner at shipment of the Work to Owner or the Job Site to the extent that Owner reasonably requests. The transfer of title does not relieve Contractor of its obligation to provide and pay for all transportation and storage in connection with the Work. Contractor shall transfer title to Owner to the Work, including Equipment and Materials, free and clear of all Liens except for those Liens that have been created by Owner. Passage of title will not affect the allocation of risk of loss. Subject to Article 16, title to documents, submittals, reports, Design Documents (whether hard copy, electronic or other medium) shall pass to Owner on receipt.

22.2 Title Warranty. Contractor warrants that upon passage of title of the Work in accordance with Section 22.1, Owner will have good and indefeasible title to such Work, and not be subject to any Lien or other defect in title. If any Work is replaced under Article 12 or 13, Owner's title shall vest in the replacement Work upon the earliest of payment therefor, its arrival at the Job Site or incorporation into the Ghent CCR Transport System or the Existing Facilities. In the event of any nonconformity with this warranty, Contractor, at its own expense, upon written notice of such failure, shall indemnify and hold Owner harmless from the consequences of and defend the title to such Work and, if necessary, shall promptly replace such Work and any other affected portion of the Work, and such obligation shall survive the expiration, cancellation or termination of this Agreement.

22.3 Risk of Loss. Except as otherwise provided herein, care, custody and control of the Work and the risk of loss or damage to the Work will pass to Owner on Commercial Operation, and Owner hereby releases Contractor from bearing such risk with respect to such Work after such date. With respect to Work performed or provided from and after Commercial Operation, risk of loss shall pass to Owner on the earliest to occur of delivery, installation or completion. Owner shall assume care, custody, and control of the Work and the risk of physical loss or damage thereto from and after such date, except as provided above. Contractor shall replace, repair, or reconstruct the Ghent CCR Transport System, including Equipment and Materials intended for the use of or necessary to the completion of the Work and furnished by Contractor, its Subcontractors or any other Person that is lost, damaged, or destroyed prior to transfer of care, custody, and control and risk of loss of the Work to Owner. Contractor shall be responsible to assure safe delivery of all Equipment, Materials, and other items to the Job Site.

22.4 Contractor Tools. Risk of loss or damage to the equipment, tools, and Construction Aids of Contractor, its employees or Subcontractors and their employees will at all times remain with Contractor, Subcontractors and their respective employees. Contractor hereby releases Owner and will require its Subcontractors to release Owner from and against any liability for loss or damage in respect thereof. If any casualty to such equipment, tools, or Construction Aids occurs, the loss as a consequence thereof will not be the responsibility of Owner hereunder. Construction equipment brought to the Ghent Generating Station Site or the Job Site or used thereon by Contractor or Subcontractors must be adequately insured against casualty.

ARTICLE 23

DISPUTE RESOLUTION

23.1 Resolution. An authorized representative of a Party may submit a claim, dispute or other controversy arising out of, or relating to, this Agreement which an authorized representative of a Party does not believe can be resolved by the Parties' authorized representatives (hereinafter collectively referred to as a "Dispute") to a Senior Officer from each Party for resolution by mutual agreement between the Senior Officers. Any agreed determination by the Senior Officers shall be final and binding upon the Parties. However, if the Senior Officers do not arrive at a mutual decision as to the Dispute within thirty (30) Days (or such longer time as the Parties agree) after notice to each Party of the Dispute, such Dispute will, if aggregate amount claimed

be settled by arbitration in accordance with the terms and provisions set forth in Section 23.2 hereof. If the Dispute is not in the Arbitration Range, either Party may pursue any other available remedy at law or in equity. For purposes of this Agreement, the term "Senior Officer" means the chief executive officer, president or any senior vice president of a Party.

23.2 Arbitration Proceedings.

23.2.1 Demand for Arbitration. All arbitration proceedings shall take place in Louisville, Jefferson County, Kentucky and shall be conducted in accordance with Construction Industry Rules then in effect of the American Arbitration Association. Notice of the demand for arbitration shall be filed with the other Party and shall be made within a reasonable time after such Party is permitted to arbitrate the Dispute as provided herein ("Notice"). The Notice shall specify the name and address of an arbitrator designated by such Party, the nature of the dispute and the amount involved. No demand for arbitration may be made or permitted after the date when the institution of legal or equitable proceedings based on such Dispute would be barred by the applicable Kentucky statute of limitations.

23.2.2 <u>Selection of Arbitrator</u>. Within twenty-one (21) Days after receipt of the Notice, the Party that received such Notice shall respond (the "Response") by written notice specifying the name and address of the arbitrator designated by it. If a Party fails to deliver its response within such twenty-one (21) Day period, the arbitrator specified in the Notice shall be the sole arbitrator of the dispute. Within fourteen (14) Days after

12/29/11

12/30/11

receipt of the Response, the two arbitrators shall appoint a third arbitrator. Such third arbitrator shall have prior judicial experience. All arbitrators shall be in all cases neutral Persons with no financial or personal interest in the result of the arbitration or any present professional relationship with the Parties or their counsel. The arbitrators shall endeavor to conduct the arbitration proceedings expeditiously in order to be able to render a decision within thirty (30) Days of selection of the third arbitrator. The decision of the third arbitrator shall control if no majority decision can be reached.

23.2.3 <u>Consolidation</u>. No arbitration arising under this Agreement shall include, by consolidation, joinder or any other manner, any Person not a Party to this Agreement, unless: (i) such Person is substantially involved in a common question of fact or law; (ii) the presence of such Person is required if complete relief is to be accorded in the arbitration; and (iii) such Person has consented to such inclusion.

23.2.4 <u>Binding Nature</u>. The agreement herein among the Parties to arbitrate under certain circumstances will be specifically enforceable in any court of competent jurisdiction. In rendering their decision and award, the arbitrators may not add to, subtract from, or otherwise modify the provisions of this Agreement and must apply the substantive law of the Commonwealth of Kentucky. Any decision rendered by the arbitrator(s) pursuant to any arbitration must be in writing, must explain the basis on which the decision or award is based, must be delivered to both Parties and must be final and binding upon the Parties, and judgment may be entered upon it in accordance with Applicable Law in any court of competent jurisdiction.

23.2.5 <u>Costs and Expenses</u>. The arbitrators, at their discretion, in egregious cases, shall have the authority to award the prevailing Party recovery of all or any portion of the costs of the arbitration, including reasonable attorneys' fees and charges.

23.2.6 <u>Discovery</u>. The Parties have the right to conduct reasonable discovery. The right to conduct discovery shall be granted by the arbitrators in their sole discretion with a view to avoiding surprise and providing reasonable access to necessary information or to information likely to be presented during the course of the arbitration.

23.2.7 <u>Cross Examination</u>. Direct testimony may be admitted by sworn affidavit; **provided** that the opposing Party is given the right to cross examine any witness whose testimony is so admitted.

23.2.8 <u>Arbitration Notices</u>. Communications under this Article 23 may be given in the manner provided in Section 25.5.

23.3 Continuation of Work. Pending final resolution of any Dispute, Contractor shall proceed diligently with the performance of its duties and obligations under this Agreement, and Owner shall continue to make undisputed payments in accordance with such Agreement.

Star V

12/30/11

TERMINATION

24.1Termination for Convenience.

24.1.1 Termination. Owner may terminate this Agreement, in whole or in part, without cause upon written notice to Contractor, specifying the extent to which this Agreement is terminated and the date on which such termination is to be effective. If this Agreement is so terminated, Contractor shall cease performance of the Work (or terminated portion thereof) on the date specified in such notice. Upon such termination, Contractor and its Subcontractors shall place no further subcontracts, including lease and rental agreements, or purchase orders, for Equipment, Materials, Consumables, facilities, or services, including craft labor, except as may be necessary for completion of the portion of the Work not terminated by Owner. Upon request by Owner, Contractor shall promptly provide Owner a listing of all subcontracts, including lease and rental agreements, and purchase orders, which pertain to the performance of the terminated Work, and shall furnish Owner with complete copies thereof. Contractor shall, at Owner's request, preserve and protect the Equipment, Materials, and Consumables purchased for or committed to the terminated Work (whether completed or in progress and whether delivered to the Job Site or on order), the facilities on the Job Site, and Work in progress and completed Work (whether at the Job Site or at other locations), pending Owner's instructions. With respect to the terminated Work, Contractor shall promptly make every reasonable effort to cancel associated subcontracts, lease and rental agreements, and purchase orders upon terms satisfactory to Owner, or Contractor shall take such other action with respect to same as may be directed by Owner. If this Agreement is so terminated, Contractor, as its sole and exclusive remedy hereunder, shall be entitled to an amount equal to: (i) the amounts set forth in Exhibit B for milestones properly completed by Contractor; (ii) with respect to any partially completed milestone, amounts determined by multiplying the percentage of properly completed Work with respect to such milestone by the respective value of such milestone as set forth in Exhibit B; (iii) reimbursement for: (a) all cancellation charges necessarily incurred by Contractor in relation to its Subcontractors; (b) an amount equal to other reasonable termination related actual costs necessarily incurred by Contractor; and (c) and reasonably incurred actual costs of demobilization, less: (1) amounts previously paid to Contractor with respect to the Work and (2) amounts properly withheld under this Agreement. In no event shall the amount required to be paid to Contractor hereunder as a consequence of termination together with amounts that Owner has previously paid to Contractor under this Agreement exceed the Contract Price. Contractor shall calculate amounts due pursuant to this Section 24.1, which amounts are subject to audit by Owner on a completely open-book basis.

24.1.2 Assumption. Notwithstanding anything to the contrary herein, Owner, in lieu of the payment of any cancellation charge in relation to Contractor's Subcontractors pursuant to Section 24.1.1, may, upon request, assume all of Contractor's obligations under any purchase order or other agreement with a Subcontractor. Contractor, upon Owner's request, shall promptly provide Owner with (i) a copy of any purchase order or 2221

12/30/11 En

other agreement with a Subcontractor and (ii) an estoppel certificate stating all known unsatisfied Liabilities under any such purchase order or other agreement. Thereafter, upon Owner's request, Contractor shall assign all of its right, title and interest in any such purchase order or other agreement with a Subcontractor to Owner. Such assignments shall be in form and substance reasonably satisfactory to the Parties, and at no additional cost to, Owner. Such agreements shall be in full force and effect upon such assignment. Contractor covenants that a provision substantially similar to this Section 24.1.2 shall be inserted in each such agreement with a Subcontractor to preserve the rights of Owner under this Section 24.1.2.

24.1.3 <u>Mitigation</u>. If any cancellation payment is due to Contractor from Owner, Contractor shall use its reasonable best efforts to minimize the amount of such cancellation payment.

24.2 Termination by Owner for Cause.

24.2.1 <u>Default by Contractor</u>. The occurrence of any one or more of the following, and, with respect to clauses (ii), (iv), (vi) or (vii), the continuation of the same for thirty (30) Days after Contractor's receipt of written notice thereof from Owner, will constitute a default by Contractor under this Agreement (a "Contractor Default"); provided that with respect to clauses (ii), (iv), (vi) or (vii), if such Contractor Default cannot reasonably be cured within such thirty (30) Day period, such cure period shall extend for a total of ninety (90) Days, so long as Contractor diligently pursues a cure throughout such period:

- (i) Contractor or Guarantor experiences an Insolvency Event;
- (ii) the material breach of any representation or warranty made by Contractor herein;
- (iii) Contractor attempts to assign, convey or transfer this Agreement or any interest or right herein without Owner's prior written consent, except as otherwise permitted by this Agreement;
- (iv) Contractor fails to materially observe or perform any other covenant, agreement, obligation, duty or provision of this Agreement;
- (v) the interest of Contractor passes to any Person otherwise than as permitted herein;
- (vi) Contractor fails to make prompt payment to Subcontractors of amounts due for labor, Equipment, Materials, Consumables, or Construction Aids, or unreasonably disregards laws or ordinances or the lawful requirements of any competent authority or the instructions of Owner consistent with this Agreement;
- (vii) Contractor fails, neglects, refuses or is unable at any time during the course of the Work, except in the event of an Excusable Event, to provide

ample Equipment, Materials, Consumables, or Construction Aids, or labor to perform Work in accordance with the Project Schedule or to provide a recovery plan to satisfaction of Owner, including failure to successfully complete the Performance Guarantee Tests;

- (viii) Contractor fails to comply with Article 21 (Insurance) or the Parent Guarantee, for any reason, ceases to be valid or enforceable in any way;
- (ix) Contractor fails to achieve Commercial Operation within ninety (90) Days of the Guaranteed Commercial Operation Date; or
- (x) Final Completion does not occur on or before the Guaranteed Final Completion Date.

24.2.2 Owner's Remedies. Upon the occurrence of a Contractor Default, Owner may, without prejudice to any other right or remedy Owner may have under this Agreement: (i) terminate this Agreement; (ii) take possession of the Job Site and of all Equipment, Materials, Consumables, and Information owned or held by Contractor; (iii) finish the Work by whatever method Owner may deem expedient; and (iv) draw on the Retainage or withhold amounts due to Contractor to make payments therefor. If the unpaid balance of the Contract Price exceeds the cost of finishing the Work, then Contractor will be paid for all Work properly performed by Contractor to the date of termination (which amount will in no event exceed the difference between the unpaid portion of the Contract Price and Owner's cost of completing the Work). However, if the cost of finishing the Work (including the cost of arranging for completion of the Work and performing such Work on an accelerated basis to preserve as nearly as possible adherence to the Project Schedule) and performing Contractor's other obligations under this Agreement exceeds the unpaid balance of the Contract Price, Contractor shall immediately pay the difference to Owner on demand upon receipt of an invoice from Owner. Owner will be entitled to withhold further payments to Contractor until the Work has been completed and a determination that Contractor is entitled to further payments. Any amounts not paid hereunder when due will bear interest at the Agreed Rate. Owner may in its discretion employ such other Persons to finish the Work by whatever method or means as Owner in its sole discretion may deem expeditious; provided that the balance of Work and obligations will be completed utilizing reasonable methods of construction management and construction consistent with the Project Schedule, as adjusted. If this Agreement is terminated pursuant to this Section 24.2.2 and is later determined that no Contractor Default had occurred, such termination shall be deemed to be a termination pursuant to Section 24.1.1.

24.3 Termination by Contractor for Cause.

24.3.1 <u>Default by Owner</u>. The occurrence of any one of more of the following matters, and, with respect to clauses (ii) or (iii), the continuation of the same for thirty (30) Days after Owner's receipt of written notice thereof from Contractor, will constitute a default by Owner under this Agreement (an "Owner Default"); provided that with respect to clauses (ii) or (iii), if such Owner's Default cannot reasonably be cured within

Ner!

12130/1

such thirty (30) Day period, such cure period shall extend for a total of ninety (90) Days, so long as Owner diligently pursues a cure throughout such period:

- (i) Owner experiences an Insolvency Event.
- the breach of any material representation or warranty made by Owner (ii)herein:
- Owner fails to observe or perform any material covenant, agreement, (iii) obligation, duty or provision of this Agreement, including timely payment, but excluding any such failure that is compensable pursuant to Article 9 hereof.

24.3.2 Contractor's Remedies. Upon the occurrence of an Owner's Default, but subject to Section 23.3, Contractor may suspend its performance of the Work (for a maximum of fifteen (15) Days in the aggregate) or terminate this Agreement. If this Agreement is so terminated by Contractor for an Owner Default, Contractor, as its sole and exclusive remedy hereunder, shall be entitled to receive an amount calculated in accordance with Section 24.1.

Suspension of the Work. Owner may, without cause, order Contractor to 24.4 suspend the Work or extend the Project Schedule in whole or in part for such period of time as Owner may determine. Any such suspension or extension will commence as soon as reasonably possible, but no later than the second (2nd) Business Day after Contractor's receipt of written notice thereof from Owner. Such suspension or extension shall continue for the period specified in Owner's notice to Contractor. Contractor will resume any suspended Work within three (3) Business Days of Owner's written notice directing the same. Without limiting Owner's rights under Section 24.1, should a suspension of the entire Work which is ordered by Owner pursuant to this Section 24.4 (but excluding any suspension caused by a Force Majeure) continue for three hundred sixty (360) or more consecutive Days, either Party may thereafter terminate this Agreement by written notice to the other Party and the rights and remedies of Contractor will be the same as those which are expressed in Section 24.1 hereof in the event of termination for convenience by Owner. In the case of any suspension of the entire Work under this Section 24.4, except to the extent an acceleration of the Work is authorized by Owner, the Project Schedule will be extended by a period equal to the suspension period, and adjusted by Owner, in consultation with Contractor, to account for same. In the case of any suspension or extension of a portion of the Work under this Section 24.4, the Project Schedule shall be extended only to the extent the suspension or extension can be shown on the Project Schedule to affect the critical path. This analysis will be based on the most current Project Schedule in effect on the Day of suspension and take into account the critical path for completing the entire Work and the noncritical path activities suspended or extended. Owner shall pay Contractor in accordance with the terms of payment set forth in this Agreement for the Work completed prior to the time of any suspension and for the direct reasonable costs that result from Contractor's compliance with the suspension notice. Owner shall pay Contractor the direct reasonable costs (including escalation) associated with the restart of suspended Work under this Section 24.4 and shall resume payments to Contractor in accordance with the terms of payment under this Agreement thirty (30) Days after the restart of the Work. It is a precondition to all claims by Contractor for NAN

12/30/11 n
extension or compensation under this Section 24.4, that the same be made within ninety (90) Days after the suspension period has ended, or said claim is barred.

ARTICLE 25

MISCELLANEOUS PROVISIONS

Governing Law. This Agreement is governed by, and construed in accordance 25.1with, the laws of the Commonwealth of Kentucky without reference to the conflict of laws rules thereof. Each of the Parties hereby agrees that any legal proceedings that may arise under this Agreement shall be brought in the United States District Court for the Western District of Kentucky, located in Louisville (and if such court does not have jurisdiction over a matter at controversy between the Parties, any state court located in Louisville, Kentucky). Accordingly, each of the Parties hereby submits to the jurisdiction of the United States District Court for the Western District of Kentucky, located in Louisville (and if such court does not have jurisdiction over a matter at controversy between the Parties, any state court located in Louisville, Kentucky) for purposes of all legal proceedings that may arise under this Agreement. Each of the Parties irrevocably waives, to the fullest extent permitted by Applicable Law, any objection which it may have or hereafter have to the personal jurisdiction of such court or the laying of the venue of any such proceeding brought in such a court and any claim that any such proceeding brought in such a court has been brought in an inconvenient forum. Each of the Parties to this Agreement hereby consents to process being served in any such proceeding by the mailing of a copy thereof by certified mail, postage prepaid, to its address specified in Section 25.5 (as it may be changed or provided herein). EACH OF THE PARTIES HEREBY KNOWINGLY, VOLUNTARILY, AND INTENTIONALLY WAIVES ANY RIGHTS IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION OR ARISING OUT OF, UNDER, OR IN CONNECTION WITH. ANY OF THE TRANSACTION AGREEMENTS OR ANY OTHER DOCUMENTS ENTERED INTO IN CONNECTION THEREWITH, OR ANY COURSE OF CONDUCT, COURSE OF DEALING, STATEMENTS (WHETHER VERBAL OR WRITTEN), OF THE PARTIES THERETO.

25.2 Entire Agreement. This Agreement represents the entire agreement between Owner and Contractor with respect to the subject matter hereof, and upon issuance of the Full Notice to Proceed by Owner shall supersede all prior negotiations, binding documents, representations or agreements, whether written or oral. This Agreement may be amended or modified only by a written instrument signed by Owner and Contractor as appropriate.

25.3 Successors and Assigns. Contractor may not assign, convey, or transfer this Agreement, or any part thereof, without Owner's prior written consent. This Agreement is binding upon, and inures to the benefit of, the successors and permitted assigns of the Parties. Owner may assign, novate or declare any trust of the whole of any part of this Agreement and any benefit, interest, right or cause of action arising under this Agreement to an Affiliate or to a third party with comparable technical and financial abilities.

25.4 Contractual Relationship. Nothing contained in this Agreement may be construed as creating a contractual relationship of any kind: (i) between Owner and any Subcontractor (except as provided in Section 3.6 and Section 13.4 hereof) or (ii) between any

12/30/11

Persons other than Owner and Contractor. Contractor is an independent contractor and all of its agents and employees shall be subject solely to the control, supervision, and authority of Contractor. Owner and Contractor disclaim any intention to create a partnership or joint venture. Contractor may not act for or have any power or authority assume any obligation or responsibility on behalf of Owner.

25.5 Notices. All notices pertaining to this Agreement must be in writing, signed by a duly authorized representative of the Party giving such notice and will be deemed given when received by personal delivery, recognized express courier, or electronic mail (immediately followed by recognized express courier) to the other Party at the address designated below:

If to Owner:

Kentucky Utilities Company 820 W. Broadway Louisville, KY 40202 Attn: Scott Straight Telephone: 502-627-2701 E-mail: <u>scott.straight@lge-ku.com</u>; and

820 W. Broadway Louisville, KY 40202 Attn: Joe Clements Telephone: 502-627-2760 E-mail: joe.clements@lge-ku.com; and

820 W. Broadway Louisville, KY 40202 Attn: Ronald Gregory Telephone: 502-627-4531 E-mail: <u>ronald.gregory@lge-ku.com</u>; and

220 W. Main St. Louisville, KY 40232 Attn: General Counsel Telephone: 502-627-3450 E-mail: john.mccall@lge-ku.com

If to Contractor:

TIC – The Industrial Company 785 Greens Parkway – Suite 125 Houston, TX 77067 Attn: Troy White, Project Sponsor Telephone: 832-232-5763 E-mail: troy.white@ticus.com; and

2/29/11

TIC – The Industrial Company 785 Greens Parkway – Suite 125 Houston, TX 77067 Attn: Vicki Sheppard Telephone: 832-323-5721 E-mail: Vicki.sheppard@ticus.com; and

TIC – The Industrial Company 188 Inverness Drive West – Suite 700 Englewood, CO 80112 Attn: Pat Barney, General Counsel Telephone: 303-325-9663 E-mail: pat.barney@ticus.com

Addresses may be changed by a Party effective upon receipt of notice of such address change.

25.6 Rights Cumulative. Except as otherwise expressly provided or limited in this Agreement: (i) rights and remedies available to Owner or Contractor as set forth in this Agreement are cumulative with and in addition to, and not in limitation of, any other rights or remedies available to such Parties at law or in equity and (ii) any specific right or remedy conferred upon or reserved to Owner or Contractor in any provision of this Agreement will not preclude the concurrent or consecutive exercise of a right or remedy provided for in any other provision hereof. Notwithstanding the foregoing, the rights and remedies of Owner remain subject to the liability limitations set forth in Section 7.6.

25.7 Incorporation by Reference. The recitals set forth on the first page of this Agreement are hereby incorporated into this Agreement by this reference and expressly made a part of this Agreement.

25.8 No Waiver. No course of dealing or failure of Owner or Contractor to enforce strictly any term, right or condition of this Agreement may be construed as a waiver of such term, right or condition. No express waiver of any term, right or condition of this Agreement will operate as a waiver of any other term, right or condition.

25.9 Agreed Rate. Unpaid amounts will bear interest at an annual (365 or 366 Days, as appropriate) rate equal to the lesser of: (i) two percent (2%) in excess of the Prime Rate and (ii) the maximum rate permitted by Applicable Law (the "Agreed Rate").

25.10 Financing Cooperation. Contractor shall provide such reasonable assistance and cooperation to Owner as may be necessary for Owner to secure Financing for the Ghent CCR Transport System, including developing and providing information regarding the Ghent CCR Transport System reasonably available to Contractor; making presentations to potential Financing Parties, their consultants and representatives; and responding to any questions or requirements asked or imposed by any Financing Parties. Contractor hereby consents to the collateral assignment of this Agreement to the Financing Parties. All obligations of the Parties are expressly subject to the execution of final agreements required to close the Financing for the Ghent CCR Transport System contemplated by this Agreement except Owner's obligation to pay

NR

12/30/11

for Work performed and costs incurred. Contractor also agrees to enter into a consent to collateral assignment with the Financing Parties regarding this Agreement, which consent will contain such provisions as are typically provided to the Financing Parties, including giving the Financing Parties financial information of Contractor reasonably satisfactory to Financing Parties, copies of certain notices delivered to Owner hereunder, and affording the Financing Parties an independent right to cure any Owner's Defaults hereunder. Contractor shall also provide customary inside counsel legal opinions as required by the Financing Parties.

25.11 Audit.

Access to Records. Contractor (and its Subcontractors) shall maintain: 25.11.1 (i) complete and accurate financial books and records to allow compliance with Sections 8.2, Section 20.4, or which relate to any cost-based (i.e., Work not covered by lump-sum prices), components of the Work billed under this Agreement or relating to the quantity of units billed under any unit pricing agreed to by the Parties; and (ii) complete and accurate books and records relating to Contractor's obligations with respect Hazardous Substances, health and safety, environmental management, emergency response, testing and inspection of Equipment and Materials, NERC Requirements, Performance Guarantee Tests, and the requirements of Sections 25.23 and 25.24 (all the foregoing hereinafter referred to as "Records") for a minimum of five (5) years following Final Completion. All such Records must be open to inspection and subject to audit and reproduction during normal working hours by Owner or its respective authorized representatives to the extent necessary to adequately permit evaluation and verification of any invoices, payments, time sheets, or claims based on Contractor's actual costs incurred in the performance or delivery of Work under this Agreement. For the purpose of evaluating or verifying such actual or claimed costs, Owner and its respective authorized representatives will have access to Records at any time, including any time after final payment by Owner to Contractor pursuant to this Agreement. Owner and its respective authorized representatives shall have access, during normal working hours, to all necessary Contractor facilities. Contractor shall provide adequate and appropriate work space to conduct audits pursuant to this Section 25.11.1. Owner shall give Contractor reasonable notice of the date and time they, any individual Owner or any agent of either of them intend to commence any audit. If the audit should disclose an overbilling, Contractor will pay Owner's invoice rectifying such overbilling within thirty (30) Days of receipt. The rights of Owner set forth in this Section 25.11.1 shall survive the termination or expiration of this Agreement.

25.11.2 <u>Unit Prices</u>. When Contractor's invoice includes compensation for Work performed on unit price basis, Contractor shall submit Contractor's determination of units of Work performed determined in accordance with the provisions of this Agreement, and substantiated by documents. Upon verification by Owner of such documents, Owner will advise Contractor in writing of either Owner's acceptance of Contractor's determination of units or Owner's determination of such units. If Contractor believes that Owner has incorrectly determined the units of Work performed, Contractor will comply with the provisions of Article 23. All undisputed amounts will be due and payable in accordance with this Agreement.

12/24/11

When Contractor's invoice includes 25.11.3 Time and Materials. compensation for Work performed on a time and material basis or similar basis of compensation, Contractor shall submit to Owner a detailed breakdown in a form reasonably acceptable to Owner of the costs of labor, Equipment, Materials, Consumables, Construction Aids, and markups or fees (if any) due for such month to Contractor. All costs reflected in such breakdown shall be net of applicable credits. The term "Applicable Credits" means any and all refunds, rebates, credits, discounts, or similar amounts (including any interest thereon) received by Contractor or to which Contractor is or will be entitled in connection with its performance of the Work. Contractor shall pay, and require each of its Subcontractors to pay, all invoices received in connection with performance of the Work as required by the terms of their respective contracts. Upon receipt of such breakdown, Owner shall, in good faith, make the determination of the amount due Contractor, and unit prices (if applicable) shall govern. The determination of Owner shall be final and conclusive, subject to Contractor's right to dispute such determination pursuant to Article 23. Each invoice shall be accompanied with partial releases of Liens in form substantially in the form of Exhibit F-5.

25.11.4 <u>Lump Sum</u>. In connection with price adjustments under Article 10 to Work priced on a lump-sum basis, any of Contractor's prices specified in this Agreement with respect to such adjustments (such as the following items to the extent that the amounts therefor are set forth in Exhibit C: price lists, unit rates, markups and multipliers, and the fixed elements of prices including Contractor owned equipment, insurance, benefits and burden rates, small tools, expendable, equipment operating and maintenance costs) are not subject to audit; provided, however, the determination of price adjustments under Article 10 for all purposes under this Agreement shall be on a completely open-book basis. The determination of costs for time and materials Work performed under this Agreement shall be on a completely open-book basis and, with respect to those items covered by Exhibit R, the rates and charges set forth in Exhibit R shall be the sole basis for any Contract Price increase related thereto. Contractor agrees to provide any cost information required by NERC, provided that such information shall be used solely for the purpose of fulfilling NERC requirements

25.12 Survival. Articles 7, 13, 16, 17, 18, 19, 20, 23 and 25 and Sections 4.39, 8.8.2, and 22.2 of this Agreement and all other Articles and Sections thereunder providing for indemnification, or limitation of or protection against liability of either Party, survives the termination, cancellation or expiration of this Agreement.

25.13 No Third Party Beneficiaries. The provisions of this Agreement are intended for the sole benefit of Owner and Contractor and, except to the extent specifically identified herein, there are no third party beneficiaries other than assignees contemplated by the terms herein.

25.14 Non-Recourse. Anything to the contrary notwithstanding, the obligations of Owner under this Agreement (i) are individual and special obligations of Owner and do not constitute obligations of (and no recourse shall be had with respect thereto to) any shareholder of Owner, any of its Affiliates, or any shareholder, partner, member, officer, director, commissioner or employee of any such Person and (ii) no action shall be brought or maintained against any

X.g.

such partner, Affiliate companies, or any shareholder, partner, member, officer, director, commissioner or employee of any thereof. The Parties acknowledge and agree that any provision of this Agreement that limits, excludes or protects against any liability of Owner, grants indemnity or confers rights or remedies that are intended to benefit (and be enforceable by) Owner, shall be applicable to Owner whether or not Owner is specifically referenced in such provision.

25.15 Parent Guarantee. Owner's obligation to make any payment to Contractor hereunder is subject to the receipt of the Parent Guarantee executed by the Guarantor in the form attached hereto as Exhibit F-8, guaranteeing the full and timely payment and performance of all of Contractor's obligations under this Agreement. Contractor covenants that the Parent Guarantee will remain in full force and effect to and until the obligations of Contractor under this Agreement have been fulfilled.

25.16 Provisions Required by Law. Any term or condition required to be contained in this Agreement as a matter of Applicable Law that is not so contained herein is deemed to be incorporated in this Agreement as though originally set forth herein.

25.17 Severability. If any provision of this Agreement, or the application thereof to any Person or circumstance, is to any extent held invalid or unenforceable by a court of competent jurisdiction or pursuant to arbitration as provided herein, the remainder of this Agreement, and the application of such provision to Persons or circumstances other than those as to which it is specifically held invalid or unenforceable, will not be affected thereby, and each and every remaining provision of this Agreement will be valid and binding to the fullest extent permitted by Applicable Law; provided, however, the Parties agree to negotiate in good faith and shall reform this Agreement to as closely as possible resemble the original intent and allocation of risks and benefits.

25.18 Joint Effort. Preparation of this Agreement has been a joint effort of the Parties and the resulting document (or any portion) is not to be construed more severely against one of the Parties than against the other.

25.19 Counterparts. This Agreement may be executed in two (2) or more counterparts, each of which is deemed to be an original, but all of which together shall constitute one and the same instrument.

25.20 Approvals Not To Relieve Contractor. No approval, consent or failure to disapprove, inspect or failure to inspect, or comment on, any matter by or the submission of any drawing or document to, or acquiescence on the part of, Owner, including any Turnover Acknowledgment, will relieve Contractor of any liability for any of its obligations under this Agreement or otherwise.

25.21 Consultants. At their option, Owner and the Financing Parties may retain the services of others, including engineers and financial consultants, to assist Owner and Financing Parties in monitoring the conduct of the Work by Contractor. All rights of access and protection, including Contractor's obligation to indemnify, defend and hold harmless Owner and Financing Parties, will be afforded equally to any of their agents and consultants.

12/20/11

25.22 Equal Employment Opportunity. To the extent applicable, Contractor shall comply with all of the following provisions, which are incorporated herein by reference: (i) Equal Opportunity regulations set forth in 41 CFR \S 60-1.4(a) and (c), prohibiting employment discrimination against any employee or applicant because of race, color, religion, sex, or national origin; (ii) Vietnam Era Veterans Readjustment Assistance Act regulations set forth in 41 CFR § 60-250.4 relating to the employment and advancement of disabled veterans and Vietnam era veterans; (iii) Rehabilitation Act regulations set forth in 41 CFR §60-741.4 relating to the employment and advancement of qualified disabled employees and applicants for employment; (iv) the clause known as "Utilization of Small Business Concerns and Small Business Concerns Owned and Controlled by Socially and Economically Disadvantaged Individuals" set forth in 15 USC § 637(d)(3); and (v) the subcontracting plan requirement set forth in 15 USC § 637(d).

25.23 Minority, Women, Disadvantaged and Local Business Enterprises. Owner has a "Supplier Diversity Policy" to provide the maximum opportunity for Minority Business Enterprises ("MBEs"), Women Business Enterprises ("WBEs"), Disadvantaged Business Enterprises ("DBEs"), and union and non-union local contractors ("LCs") to participate as subcontractors for goods and services. As such, every attempt should be made by Contractor to include MBEs, WBEs, DBEs, and LCs, on subcontract bid lists for the Work. To the extent practicable, this requirement shall be passed down to Subcontractors. Contractor shall provide a supplier diversity program for the Work. Elements of such a program shall at a minimum include: (i) a goal setting process for identifying MBE/WBE/DBE/LC opportunities for involvement in the Work; (ii) an outreach process to identify and attract possible MBE/WBE/DBE/LC business interest in the Work, including working with local organizations such as Kentuckiana Minority Business Council; (iii) a pre-qualification process to assess the suitability of interested MBE/WBE/DBE/LCs; (iv) a bidding process inclusive of suitable MBE/WBE/DBE/LCs on subcontract bid lists; and (v) a monitoring process to provide statistical reporting on opportunities and utilization. Contractor shall report to Owner with respect to its compliance with this Section 25.23 in accordance with Exhibit F-9 (Second Tier Procurement Program).

25.24 Local Involvement; Reporting.

25.24.1 Local Workers. Contractor shall make a diligent good faith effort to hire, to the maximum practical extent, qualified Local Hires for and in connection with the performance of the Work in the order of priority in which geographic areas are listed in the definition of Local Hires.

25.24.2 Local Content. Contractor shall make a diligent good faith effort to include, to the maximum practical extent, qualified contractors on subcontract bid lists for portions of the Work that are to be subcontracted by Contractor or its Subcontractors in the order of priority in which geographic areas are listed in the definition of Local Spend.

Reporting. On a monthly basis commencing on the Effective Date, 25.24.3 Contractor shall provide a written report substantially in the form set forth in Exhibit F-9, detailing statistical data relating to its workforce and those of its Subcontractors, including information relating to MBEs, WBEs, DBEs, LCs, Local 56 Spend, WBE Spend and MBE Spend. Contractor will not be obligated to provide confidential individual employee information under this Section 25.24.3.

25.25 Patent Limitations. Notwithstanding anything in this Agreement to the contrary, Contractor's obligations and liabilities with respect to patents in connection with the Work provided under the terms (as of the Effective Date of this Agreement) of the Ghent CCR Transport Equipment Contracts shall be limited to only United States patents to the extent that the patent terms (as of the Effective Date of this Agreement) of the applicable Ghent CCR Transport Equipment Contract are so limited.

· · ·

12/29/11

IN WITNESS WHEREOF the Parties have caused this Agreement to be executed by their duly authorized representatives as of the Effective Date.

Kentucky Utilities Company

By: Club Kell DAN for Joe Charts Title: Cantrust Administratur 12/30/11

TIC - The Industrial Company By: <u>Aug</u> Mr. Title: <u>VILE PRESIDENT</u>

12/30/11

2 2 V

As of December 19, 2011

Milestone Payment Schedule Summary - Exhibit B

Planned Pasjected Hergy Hergy																											EPC Ag
			1	Grem denotes prinausly invoiced BA Yelson denotes BAGE SCOPE payme Ped denotes scheduled mileitanet mi	ASE SCOPE payment ministeres, next ministeres included in correct ma and pet invoiced,	tentra invoira							Statu.						 Providence 	Bue fonces previously brack # SHARGE DRDER, prymeri microsek, Drage dentes DHARE ONDER, prymeri adecaras horace o carrent Parple dentes previously inviced DESCUTION RELATED prymer and	i mentifi i indice. stanet.						
	s Scheekula Artivity	Description	TIC WYDICE	IDRU EPC EP	-C Kerth	brovice Grous Reference browiest Look	Schudula 4 Athuby	Duc/pie>	FGD Bypreduct Deeptimen	ling System Kenthy Cumulation	Hends H	nveise Grana	Schedula	Description	Pos Carreyor Systems	Cumulation	Wenth	konnica Grons Raferancy Invalcad And		Bastiption		Centrelying Systems Menthly Curr	nduba	New 20	Involce	Genes T	STAL
						Release Provide The	Alberty	TOTAL BELOW #	+				TOTAL BE							TOTAL BELOWY							
									<u> </u>																		
														·····													
	-							Award of Purchase Order					AvadelCo	ntad													
							_													······································							
								Receipt of Main Malmaly	+																		
					_		-						Pioturemtr	ulal Maiar Materiala						Research of Parent Guarnley Programment al Malor Malorials							
		stuance al JITP											12/DE/IT Recepted A	oppored General Ananopment Drawings						Recept of Approved General Arrangement Drawing							
	ht	THE RANKY INFORTACI												······································	+					·····							
	l±	stue ste coprénsion plun mue protech purplese order stue CCR realistie handing spers for comment																			1						
	U U	labilate an innoving a guarnersi Jabilate civile a timerk case																									
	1 1/1	ssue pri-éhyineered building spet ssue Level III project schedule																			+						
		ssue bansformet spect Rechical Distribution Environent Study Altreanse DAE Equipment Milestons, Payment														_					+						
		le and anti-handing equipment contract																									
		Achiere ple daving contractor spue underground utilies drewings					_																				
	lu	nue electrical one-lines nue PADs					1						<u> </u>								+						
	<u>k</u>	ward pre-engine reed buildings					+	<u>+</u>	<u> </u>				Drivery of b	lains Components	+				 		+						
	R	Receive underground ploing materials Receive underground electrical ductoant materials Lari stear & gubbing pipe rack foundations											<u> </u>	······································	<u> </u>					<u>, , , , , , , , , , , , , , , , , , , </u>	++						
	111	ileri siez & gubbing pipe rack foundations sure By ash bullding foundation di parings Diblain appt pval for underground fire foco mömittal																		<u> </u>	<u> </u>						
	S	icomit GA drawings							<u> </u>																		
	51	Conveyor foundations clearing & grubbing complete Nart pipe rock foundations					-																				
	lts	itart CCR tarea underground electrical true electricalization tol building sorres for owner review							<u> </u>							-					++						
	SL	tari DCR area relating wal foundations			_															······································							
		iturknaf sleef om såe																		······································							
	55	ssue sufimerged chain Eght conveyor foundation drawe start CCR, wea equipment foundations	nga																								
	1	ward pre-packaged electricalization bellouiding																									
		nue geptien processing plant formdaken drævings do contractor mobiles																			<u> </u>						
		Complete CCR Bandormer foundstons							F											······································							
	Co	Complete XFAIR foundations at the new lands for conveyor foundations complete																	-								
	Fr	inish existing plant East concrete foundations																									
	1 100	Starl mechanical equipment erection at CCR area (exc) anveyors] Jobize pre-enoneered building contractor							<u> </u>							_				Delvery of Fly Ach Conveying System							
	51	tart erfibig plant piping & electrical instatetion AE Epsilpment Milestone Phyment					-					·····		······································					1								
		inch esching plant West concrete foundations																		Debyry el Boltom Ath Handling System							
	Si Si	itari erecing gyssum processing plani building Inish CCR area relaring wat foundations																									
		itari ereting gyptum slorece thed												·····						······································							
		ine rack structural steel installation complete Velver By ash handing equipment (except portal rectaim	ner)					·						·····				·			1						
	51	itig slos Ital etecting bet conveyors CCP area																									
	54	el GPP pump CCR area																									
	Se Se	el grown processing vacuum pumps stids					-												+								
	0	AE Esubment Hilestone Payment																			1						
	Se Se	sh water return sump complete et instrument air comprisitors fr ach balding 384 complete					-		F====							++			† {		1						
		ilo star lower & bridge eraction complete																									
	Sk Un	to toofs completed																									
	51	el submerged Bahl conveyors el vacann receivers						+	F=====										11								
		iel dolti wash slädy Watrazi Santal bulding delivered						+	F							++			+		+						
	En	reci bollom ash bunker cover tobli restamer debrerv																	1		1						
		el gyptum vzcuum bell libris														1				·····							
	Se Ba	ackfeed Power										·			1				+								
	St 	ieddred Power Init 2/3 hert fines pus pe pump building somplete tart erecting maintenancelding building			T				F====					·····		<u> </u>			<u>t </u>		<u>+</u> [
	Set	ná 2/2 hert lines pus pe pump bulding somplete tet erszting mánten anceldoro buldino vyrozm processny plani buldiny erecton complete					1	+	F======						1	++			1		+						
	54 83 97 97 97 97 97 97 97 97 97 97 97 97 97	na 20 hert fores puese pump bulding complete lart ereting mänitenansektoris bulding syssum pieressang planitediten eretion compikte anachstering Acheles Stutied auchstering Acheles Stutied													1				<u>t</u>								
	54 83 97 97 97 97 97 97 97 97 97 97 97 97 97	na 20 hert bres pus pe pump budding complete ball ett dyng mainten ancel daros bulders wynum processing plant budding er ection complete commissiona Activity is Statied							F											Nethanical Completion Balton Ath Handha Catan							
	Star	nk 20 her den seues enne buking semelek alle retten minimanet klan buking spann beretang plant buking ereten som kla annstruke Statist annstruke statist annstruke ereten in plant ereten ereten											Mechanical C	Completion					- L.	Nechanical Completion Boltom Ash Handing System Mechanical Completion Taylor Is Gheni Unit 2 Mechanical Completion Taylor Isheni Unit 1							
	Star	na 20 hert fores puese pump bulding complete lart ereting mänitenansektoris bulding syssum pieressang planitediten eretion compikte anachstering Acheles Stutied auchstering Acheles Stutied											Nechanical C	Carraly Ion						Mechanical Completion Try-In for Obent Unit 2 Mechanical Completion Try-In for Obent Unit 1 Commercial Completion Boltom Ach Handing System							
	5 m 6 m 7 m 8 m 8 m 9 m 9 m 9 m 9 m 9 m 9 m 9 m 9	nk 20 her den seues enne buking semelek alle retten minimanet klan buking spann beretang plant buking ereten som kla annstruke Statist annstruke statist annstruke ereten in plant ereten ereten						Successfully: Penting by India Derrikinal Test					Mechanical C							Mechanical Completion Te-In for Ghent Unit?							
	5 m 6 m 7 m 8 m 8 m 9 m 9 m 9 m 9 m 9 m 9 m 9 m 9	ng 27 Berling gunge samp hadiga gangkita Una trangga mahanandahan hadiga uganta tenes tanggalah hadiga yartaan gangkita mahananana Akatari and pada cedana yartaan mahananana katari mahananana katari and pada cedana yartaan mahananana katari and pada cedana yartaan mahanananananananananananananananananana						Successfully Parama Ru Inda Operational Traj												Mechanical Completon Tre-In for Gheni Unit 2 Mechanical Completon Tre-In for Gheni Unit 1 Commercial Completon Bottom Ach Handing System Commercial Completon Tre-In for Gheni Unit 2							
	5 m 6 m 7 m 8 m 8 m 9 m 9 m 9 m 9 m 9 m 9 m 9 m 9	ng 27 Berling gunge samp hadiga gangkita Una trangga mahanandahan hadiga uganta tenes tanggalah hadiga yartaan gangkita mahananana Akatari and pada cedana yartaan mahananana katari mahananana katari and pada cedana yartaan mahananana katari and pada cedana yartaan mahanananananananananananananananananana																		Mechanical Completon Tre-In for Gheni Unit 2 Mechanical Completon Tre-In for Gheni Unit 1 Commercial Completon Bottom Ach Handing System Commercial Completon Tre-In for Gheni Unit 2							
	5 m 6 m 7 m 8 m 8 m 9 m 9 m 9 m 9 m 9 m 9 m 9 m 9	ng 27 Berling gunge sump hadige genetika Una transpanninan materialah hadiga pertakan gang gapata tersega pantah materialah hadiga pertakan genetika materialah hadiga pertakan genetikan wahika Pentan aral pada centana pertakan materialah pentana materialah pentana																		Mechanical Completon Tre-In for Gheni Unit 2 Mechanical Completon Tre-In for Gheni Unit 1 Commercial Completon Bottom Ach Handing System Commercial Completon Tre-In for Gheni Unit 2							
	5 m 6 m 7 m 8 m 8 m 9 m 9 m 9 m 9 m 9 m 9 m 9 m 9	ng 27 Berling gunge sump hadige genetika Una transpanninan materialah hadiga pertakan gang gapata tersega pantah materialah hadiga pertakan genetika materialah hadiga pertakan genetikan wahika Pentan aral pada centana pertakan materialah pentana materialah pentana																		Mechanical Completon Tre-In for Gheni Unit 2 Mechanical Completon Tre-In for Gheni Unit 1 Commercial Completon Bottom Ach Handing System Commercial Completon Tre-In for Gheni Unit 2							
	5 st (b) (b) (c) (c) (c) (c) (c) (c) (c) (c	ng 27 Berling gunge sump hadige genetika Una transpanninan materialah hadiga pertakan gang gapata tersega pantah materialah hadiga pertakan genetika materialah hadiga pertakan genetikan wahika Pentan aral pada centana pertakan materialah pentana materialah pentana																		Manuscal Complete Tar-Is for Charal Uby 1 Command Complete Tar-Is for Charal Uby 1 Command Complete Tar-Is for Charal Uby 1 Command Complete Tar-Is for Charal Uby 2 Command Complete Tar-Is for Charal Uby 3 Legendrad Complete Tar-Is for Charal Uby 4 Command Complete Tar-Is for Charal Uby 4							
		M2 20 Berling guege some hadden godets all renden minimations by dee ingenen insert some på det hadden er er er den ennelske andre berenden andre berenden andre berende inder berende inder berende må 1 beite consoles																		Mannes Compilen Techte Chen Und Generale Compilet Techte Chen Und Generale Compileten Battern Adv Hanfing Spann Generale Compileten Techter Genet Und Generale Compileten Techter Ghent Und Angeweich Compileten Techter Ghent Und 4							

KU Ghent CCR Transport Project EPC Agreement

GH Transport - TIC Rollup 28-Dec-11

BOFO w/o OFE Equipment OFE Equipment (13-Dec-11) WesTech Original Contract #511453	
WesTech Original Contract #511453	
Beumer Original Contract #514629	
CO #1 - Truck Loading Station	
Pending CO - EPC Pipe Conveyor Loads	
ASH	
Original Contract #517454	
Pending CO for Finish Coatings KU's Original Clarifications (11-Nov-11)	
#3 - Eliminate electrical vault	
#6 - Addition of retention basins	
#7 - Sump discharge to Gypsum Stack	
#12 - Gypsum storage bldg fully enclosed	
#16 - Walkways and lightning on pipe racks	
#23 - Demo of Augusta Bldg	
#39 - Trench cootings	
#45a - Savings on Pipe Conveyor foundations	
#58 - Electrical MCC work north of US 42	
#60 - DCS Climate Controlled rooms	
#78 - SA & IA at head end of Pipe Conveyor	
#81 - Per December 09, 2011 Letter	
#82 - Elevated pipe rack south of US 42	
KU's Revised Clarifications (28-Nov-11)	
21 - Additional DCS Stations	
42a - Option 2 Electrical Feed	
42b - Additional 480v Equipment	· ·
75 - Coatings in Fly Ash Contract	x
78 - SA and IA at Head End of Conveyor	
Value Engineering (17-Nov-11)	
#1 - Rotate Bottom Ash Building	
#4 - Flop gates in-lieu of traveling heads	
#12 - Wash water to ATB #2	
Additional Scope (09-Dec-11)	
TC 13 Fly Ash Electrical Building Mods	
TC 03 Silo Changes due to Equipment	
Clarification 81 SA and SW Scope Reduction	
TC 25 WesTech Scope Break	•
Additional Scope (13-Dec-11)	
TC 25 WesTech Revision #1	
Marshalling Cabinets	
Additional Scope Insurance (21-Dec-11)	-
Additional Scope (22-Dec-11)	·
Spare starters in MCC's	· ·
TC 52 Marshalling Cabinets	-

JB/h zg-Der-11

EXHIBIT D

PROJECT SCHEDULE & KEY DATES

Item

Date

Mobilization Date

Scheduled Mechanical Completion

Guaranteed Commercial Operation

Guaranteed Final Completion



Tie-In Scheduled Outages



12/30/11 GN

Acceptable Equipment and Materials Suppliers

The following sub-suppliers have been pre-approved by the Buyer/Owner to provide the corresponding components. If the Seller proposes to substitute a sub-supplier not on this list, then the sub-supplier's credentials shall be submitted to the Buyer/Owner for approval. Sufficient data must be provided to the Buyer/Owner in a timely manner to prove that the proposed equipment is of an equivalent quality and has proven operation in similar service conditions, including capacity, to that provided by the pre-approved sub-suppliers. If an offshore sub-supplier is proposed, the Seller shall provide information proving satisfactory domestic stockpiling of any required spare parts. Final decision concerning acceptance of an alternate sub-supplier shall be solely at Buyer/Owner's discretion.

Where multiple sub-suppliers are identified, those indicated in bold text are preferred by the Buyer/Owner and should be used for base bid purposes. Where no sub-supplier is named, the Seller shall select based on previous successful service in similar utility power plant applications.

12/30/11 GN

15/m 29-Dec"11

EXHIBIT F1

APPLICATION FOR PAYMENT
DATE: CONTRACTOR:
REQUEST FOR PAYMENT #: CONTRACT #:
FOR THE MONTH ENDING:
MILESTONES ACHIEVED
ADDITIONAL DESCRIPTION:
I. GROSS BILLINGS:
TOTAL THROUGHTOTAL EARNEDTOTAL EARNEILAST PERIODTHIS PERIODTO DATEA: BASE CONTRACTTO DATE
B: CHANGE ORDERS
TOTALS:
II. THIS PERIOD'S BILLING
A.TOTAL VALUE OF MILESTONES ACHIEVED THIS PERIOD:
B. AGGREGATE PAYMENT AMOUNT SPECIFIED BY THIS MONTH IN EXHIBIT H
C. SUBTOTAL (LESSER OF A OR B):
D. SALES TAX:
E. TOTAL DUE THIS MONTH:
RESPECTFULLY SUBMITTED BY:

Abhu 29-Dec-11

,

12/30/11

EXHIBIT F2

CERTIFICATE OF (MECHANICAL COMPLETION, COMMERCIAL OPERATION, TIE-IN OR FINAL COMPLETION)

The following form shall be used for Certificates of Mechanical Completion, Commercial **Operation and Final Completion:**

CERTIFICATE OF

[Fill in blank with: MECHANICAL COMPLETION, COMMERCIAL OPERATION or FINAL COMPLETION, as appropriate]

This constitutes the Certificate of [Fill in blank with: Mechanical Completion, Commercial Operation or Final Completion, as appropriate] as contemplated by the Engineering, Procurement and Construction Agreement, entered into as of the _____ day of ______ 2011 (the "Agreement"), by and between Kentucky Utilities Company, a Kentucky corporation ("Owner"), and _____, a _____, a corporation ("Contractor").

[The following sentence applies to Certificates of Commercial Operations, only.] For purposes of Section 6.7 of the Agreement, the date on which Owner received the notice from the Contractor pursuant to which this Certificate is being issued is

Executed on this _____ [date] by the Companies on behalf of the Owners.

Kentucky Utilities Company

By

Title:

ABA 1 29-Dec=11

	ibit F-3 CHANGE ORDER	
	File No.:	
Project:	Date:	
Project No.:		
Contractor:	Change Order No.:	
Contract For:	Contract No.;	
Change Order Title:		
The Agreement is hereby changed as follo Change Description:	ws:	
Adjustment to Schedule (including change	e(s) to key dates listed in Exhibit D):	
Adjustment to Payment Schedule:		
Adjustment to Performance Guarantees:		
Other Agreement Adjustments (specify):		
	Original Contract Price	
	Prior Change Orders Amount	
	Contract Price Prior to this Change Order	
	-	
	This Change Order Amount	
	Net Contract Price Amount	
	er, the price and time extension in this Change Order is ful irect and indirect, incurred in connection with the condition e markups and fees.	as giving rise to
his Change Order including all applicable This Change Order, when executed, co	irect and indirect, incurred in connection with the condition	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co Agreement, except as modified above or b	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co Agreement, except as modified above or b Acceptance for and on behalf of TIC	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro y a previous Change Order, shall apply hereto.	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co Agreement, except as modified above or b Acceptance for and on behalf of TIC By:	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro y a previous Change Order, shall apply hereto. Acceptance for and on behalf of Kentucky Utilities	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co Agreement, except as modified above or b Acceptance for and on behalf of TIC By:	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro- y a previous Change Order, shall apply hereto. Acceptance for and on behalf of Kentucky Utilities By:	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co Agreement, except as modified above or b Acceptance for and on behalf of TIC By: Date:	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro- y a previous Change Order, shall apply hereto. Acceptance for and on behalf of Kentucky Utilities By: Date:	ns giving rise to
this Change Order including all applicable This Change Order, when executed, co	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro- y a previous Change Order, shall apply hereto. Acceptance for and on behalf of Kentucky Utilities By: Date: Title:	ns giving rise to
this Change Order including all applicable. This Change Order, when executed, example a state of the second state of the secon	irect and indirect, incurred in connection with the condition e markups and fees. onstitutes a modification to the Agreement, and all pro- y a previous Change Order, shall apply hereto. Acceptance for and on behalf of Kentucky Utilities By: Date: Title:	ns giving rise to ovisions of the

12/30/11 M

EXHIBIT F4

INTEGRATED PROJECT SCHEDULE

Preliminary Project Schedule

- 1. Exhibit D of the Agreement contains a preliminary Project Schedule.
- A. Project Schedule:
 - 1. An expanded critical path method (CPM) construction project schedule shall be submitted per Exhibit X. Base the schedule on the preliminary Project Schedule in Exhibit D of the Agreement and incorporate review comments and other feedback. Submit an electronic file of the schedule to Owner for review.
 - 2. The CPM Project Schedule shall be developed using Primavera P6 and submitted at least monthly in resident format.
 - 3. The Project Schedule shall show the Work in a graphic format suitable for displaying scheduled and actual progress.
 - a. Prepare schedules as a horizontal bar chart with separate bar for each major portion of the Work or operation.
 - b. The Project Schedule shall indicate major phases and key items of the Work including: starting date, engineering, procurement, delivery, construction, major milestones, dates of Mechanical Completion, start-up, commissioning, testing, Commercial Operation and Final Completion.
 - c. Breakdown Work into significant construction activity entries, with dates Work is expected to begin and be completed. Sequence of listings shall be in the chronological order of the start of each item of Work.
 - d. Scale and spacing shall allow room for notations and revisions.
 - e. Sheet Size: Minimum 11 x 17. 4.
 - 4. The number and types of activities comprising the Project Schedule shall be subject to Owner approval. At least 90% of all activities shall be for durations of less than two weeks. No single schedule activity shall be greater than 30 days.
 - 5. The Project Schedule shall show durations, calculated early and late start and finish dates, and float for each activity. Plug dates shall not be used.
 - 6. Contractor shall not change the milestone dates of the accepted Project Schedule without prior concurrence of Owner.
 - 7. The Project Schedule shall be resource loaded.
 - 8. The Project Schedule shall use unique activity identifiers which correlate to the same activities in the work breakdown structure (WBS) associated with the schedule of values used for invoicing.
 - 9. Coordinate Project Schedule with Milestone Payment Schedule, submittal schedule, procurement schedule, progress reports, and payment requests.

JBth 29-Doc-11

Ghent CCR Transport Exhibit F5 Kentucky Utilities

EXHIBIT F5

LIEN WAIVERS AND GENERAL RELEASE

FORM OF CONDITIONAL PARTIAL LIEN WAIVER TO BE PROVIDED BY MAJOR SUBCONTRACTORS:

Ghent CCR Transport Project

In accord with [insert subcontract name and paragraph(s) reference] the undersigned, for and in consideration of the payments made by ______, (the "Company"), to the undersigned, for labor employed in and/or equipment or materials furnished for the construction of the above-referenced Project pursuant to the above-referenced subcontract, hereby certifies as follows:

1. Upon receipt of the sum of ______, the undersigned will have received payment in full, less retainage, for labor employed in, equipment or materials furnished and/or performance of work for the construction of the Project through the ______ day of ______, 201X and except for receipt of said sum and as an inducement to the Company to make payment of the same, and except for retainage and the following matters [OUTSTANDING CLAIMS TO BE IDENTIFIED], the undersigned hereby affirms that there are no outstanding claims against the Company in connection with any labor employed, work done, equipment or materials furnished or performance of work by the undersigned pursuant to the above-referenced subcontract through such date.

2. Contingent upon receipt of the sum referenced in paragraph 1, the undersigned does hereby waive, release and quit claim in favor of the Owner of the Project, any owner of the site on which the Project is located, each and every party acquiring title to and interest in and/or making a loan on the Project and the title company or companies examining and/or insuring title to the Project and any and all of their successors and assignees (collectively the "Beneficiaries"), all rights that presently exist or hereafter may accrue to the undersigned to assert a lien, whether contractual, statutory or constitutional, upon the land and improvements comprising the Project which rights have arisen or shall arise out of or in connection with labor employed, equipment or materials furnished or performance of work pursuant to the above-referenced subcontract and amendments thereto, but only for labor employed and/or equipment and materials furnished and/or work performed through the ______ day of ______, 201X, and excepting the claims listed in paragraph 1 above (if any).

3. The undersigned has not assigned any claim against the Company, or any of the other Beneficiaries, nor any lien or right to perfect a lien with respect to the Project, and the undersigned has the right, power and authority to execute this document.

1 ABH

12/30/11 G1 4. The undersigned warrants that all laborers and subcontractors employed by it in connection with the Project, and all suppliers or materialmen from which it has acquired equipment or materials incorporated into the Project and any lien or bond claimant relating to the undersigned's work in connection with the Project have been paid all amounts due and owing through the _____ day of ______, 201X and that none of such laborers, subcontractors, suppliers, materialmen, or claimants has any unsatisfied or unbonded lien or related claim against the Project through such date, except in cases of Owner's non-payment. The undersigned further warrants that, to the best of its knowledge and belief, all applicable due and outstanding taxes, fees, contributions and benefits relating directly or indirectly to the undersigned's work and which are the responsibility of the undersigned have been paid in full.

5. No security interest has been given or executed by the undersigned for or in connection with any materials, appliances, machinery, fixtures or furnishings placed upon or installed in the Project. This Affidavit and Partial Release of Liens is for the benefit of the Company and the Beneficiaries.

This Affidavit and Partial Release of Lien is an independent covenant and will operate and be effective with respect to work performed, labor employed and/or equipment and materials furnished and/or performance of work under the above-referenced subcontract and any related supplemental contract or contracts for extra or additional work on the Project performed by the undersigned through the _____ day of ______, 201X.

IN WITNESS WHEREOF, this Affidavit and Partial Release of Lien has been executed on this _____ day of _____, 201X.

WITNESS:

(Name)

Subscribed and sworn to me this _____ day of _____, 201X.

Notary Public

My Commission Expires:

(Notarial Seal)

2/B/L

FORM OF CONDITIONAL PARTIAL LIEN WAIVER TO BE PROVIDED BY CONTRACTOR:

Ghent Generating Station CCR Transport Project

In accord with Section 8.6(iii) of the Engineering, Procurement and Construction Agreement, dated as of [_______, 20XX] ("Agreement"), by and between ______("Contractor") and Kentucky Utilities Company (the "Owner"), the Contractor, for and in consideration of the payments to be made by the Owner to the Contractor, for Work performed, labor employed and/or for Equipment furnished in connection with the construction of the Facility pursuant to the above-referenced Agreement, hereby certifies as follows. Capitalized terms used and not defined herein shall have the respective meanings set forth in the Agreement:

1. In conjunction with this lien waiver, Contractor is submitting an invoice to Owner in the sum of ______with respect to Work performed, labor employed and/or Equipment provided in connection with the construction of the Facility for the period the _____ day of , 201X through to the _____ day of _____, 201X.

2. Contingent upon receipt of the sum referenced in paragraph 1 in accordance with the Agreement, the Contractor does hereby waive, release and quit claim in favor of the Owner and the Financing Parties and any and all of their successors and assignees (collectively the "Beneficiaries"), all rights that presently exist or hereafter may accrue to the Contractor to assert a lien, whether contractual, statutory or constitutional, upon the Facility and/or all or any portion of the Ghent Generating Station Site or any improvements thereon which rights have arisen or arise out of or in connection with the performance of the Work pursuant to the Agreement, but only for Work performed, labor employed and/or Equipment furnished through the _____ day of _____, 201X, except for the following claims:

3. The Contractor has not assigned any lien or right to perfect a lien against the Facility and/or all or any portion of the Ghent Generating Station Site or any improvements thereon, and the Contractor has the right, power and authority to execute this document.

4. No security interest has been given or executed by the Contractor for or in connection with any materials, equipment, appliances, machinery, fixtures or furnishings placed upon, provided for or installed in the Facility and/or on all or any portion of the Ghent Generating Station Site. This Affidavit and Partial Release of Liens is for the benefit of the Beneficiaries.

3 ABA

12/20/11 w

This Affidavit and Partial Release of Lien is an independent covenant and will operate and be effective with respect to Work performed and labor employed and/or Equipment furnished under the Agreement and any related supplemental contract or contracts for extra or additional work performed by the Contractor in connection with the Facility for the period the _____ day of _____, 201X to the _____ day of _____, 201X.

IN WITNESS WHEREOF, this Affidavit and Partial Release of Lien has been executed on this ______ day of ______, 201X.

WITNESS:

(Name)

Subscribed and sworn to me this _____ day of _____, 201X.

Notary Public

My Commission Expires:

(Notarial Seal)

+ JB/L 24 April

12/30/11 w

FORM OF FINAL AND FULL LIEN WAIVER TO BE PROVIDED BY MAJOR SUBCONTRACTORS:

Ghent Generating Station Project

In accord with [insert subcontract name and paragraph(s) reference] the undersigned, for and in consideration of the payments made by ______ (the "Company"), to the undersigned, for labor employed in and/or equipment and materials furnished for the construction of the above-referenced Project pursuant to the above-referenced subcontract, hereby certifies as follows:

1. Upon receipt of the sum of ______, the undersigned will have received final payment in full for all labor employed in and/or equipment and materials furnished and/or performance of work in connection with the construction of the Project, and except for receipt of said sum and as an inducement to the Company to make payment of the same, and, except for the following matters [OUTSTANDING CLAIMS TO BE IDENTIFIED], the undersigned hereby affirms that there are no outstanding claims against the Company in connection with the Project in connection with any labor employed, equipment and materials furnished and/or performance of work by the undersigned pursuant to the above-referenced subcontract.

2. In consideration of the payments made to date and upon receipt of the sum of ________, the undersigned does hereby waive, release and quit claim in favor of the Owner of the Project, any owner of the site on which the Project is located, each and every party acquiring title to and interest in and/or making a loan on the Project and the title company or companies examining and/or insuring title to the Project and any and all of their successors and assignees (collectively, the "Beneficiaries"), all rights that presently exist of hereafter may accrue to the undersigned to assert a lien, whether contractual, statutory or constitutional, upon the land and improvements comprising the Project which rights have arisen or shall arise out of or in connection with labor employed, equipment and materials furnished and/or performance of work pursuant to the above-referenced subcontract and amendments thereto, except for the claims identified in paragraph 1 above (if any).

3. Contingent upon the receipt of the sum of ______, and except with respect to claims identified in paragraph 1 above, the undersigned does hereby forever release, waive, and discharge the Company, the Project, the Owner of the Project and the other Beneficiaries from any and all causes of action, suits, debts, accounts, damages, encumbrances, judgments, claims, and demands whatsoever, in law or equity which the undersigned and/or its successors and/or assignees ever had, or ever will have against any of them by reason of labor employed in, equipment and materials furnished and/or the performance of work relating to the construction of the Project; and the undersigned hereby agrees to indemnify and hold the above parties harmless from any and all damages, costs expenses, demands, suits, and legal fees, directly or indirectly relating to any of the foregoing claims or liens.

4. The undersigned has not and will not assign any claim against the Company, or any of the Beneficiaries, nor any lien or right to perfect a lien against the Project, and the undersigned has the right, power, and authority to execute this Affidavit, Waiver and Release.

Exh_F5_Lein Waivers and General Release

5 ABIL

12/30/1 h

5. The undersigned warrants that all laborers and subcontractors employed by it, and all suppliers or materialmen from which it has acquired equipment or materials incorporated into the Project and any lien or bond claimant relating to the undersigned's work have been paid all amounts due and owing through the date hereof and that none of such laborers, subcontractors, suppliers, materialmen, or claimants has any claim, demand, or lien against the Project through the date hereof. The undersigned further warrants, to the best of its knowledge and belief that all due and outstanding applicable taxes, fees, contributions and benefits relating directly or indirectly to the undersigned's work and which are the responsibility of the undersigned have been paid in full.

6. No security interest has been given or executed by the undersigned for or in connection with any equipment, materials, appliances, machinery, fixtures, or furnishings placed upon or installed in the Project.

This Affidavit, Waiver of Lien and Release shall be an independent covenant and shall operate and be effective with respect to labor employed, equipment and materials furnished and/or performance of work under the above-referenced subcontract and any related supplemental contract or contracts for extra or additional work on the Project performed by the undersigned. This Affidavit of Waiver and Release is for the benefit of the Company and the Beneficiaries.

IN WITNESS WHEREOF, this Affidavit, Waiver of Lien and Release has been executed on this _____ day of _____, 201X.

WITNESS:

(Name of Company)

Subscribed, sworn to, and acknowledged before me this _____ day of ____, 201X.

Notary Public

My Commission Expires: _______(Notarial Seal)

6 ABIN

FORM OF FINAL AND FULL LIEN WAIVER TO BE PROVIDED BY CONTRACTOR:

Ghent Generating Station Site

In accord with Section 8.6(iii) of the Engineering, Procurement and Construction Agreement, dated as of [_______, 20XX] ("Agreement"), by and between ______ ("Contractor") and Kentucky Utilities Company ("Owner") the Contractor, for and in consideration of the payments made by the Owner to the Contractor for Work performed, labor employed in and/or Equipment furnished for the construction of the Facility pursuant to the above-referenced Agreement, hereby certifies as follows. Capitalized terms used and not defined herein shall have the respective meanings as set forth in the Agreement:

1. Upon receipt of the sum of ______, the Contractor will have received final payment in full for all Work performed, labor employed in and/or Equipment furnished in the construction of the Facility, and except for receipt of said final payment and [OUTSTANDING CONTRACTOR CLAIMS TO BE IDENTIFIED], and as an inducement to the Owner to make such final payment, the Contractor hereby affirms that there are no other outstanding claims against the Owner in connection with this Project. The above excludes claims for unreimbursed Sales Taxes for which the Contractor is entitled to be reimbursed under the Agreement and amounts for which the Contractor on the date of this Affidavit.

2. In consideration of the payments made to date and upon receipt of the sum of ______, the Contractor does hereby waive, release and quit claim in favor of the Owner and the Financing Parties and any and all of their successors and assignees (collectively, the "Beneficiaries"), all rights that presently exist of hereafter may accrue to the Contractor to assert a lien, whether contractual, statutory or constitutional, upon the Facility and/or all or any portion of the Ghent Generating Station Site or improvements thereon, which rights have arisen or arise out of or in connection with the performance of the Work pursuant to the Agreement, except for the claims identified in paragraph 1 above.

3. Contingent upon the receipt of the sum of ______, and except with respect to claims identified or referenced in paragraph 1 above, the Contractor does hereby forever release, waive, and discharge the Beneficiaries from any and all causes of action, suits, debts, accounts, damages, encumbrances, judgments, claims, and demands whatsoever, in law or equity which the Contractor and/or its successors and/or assignees ever had or ever will have against any of them by reason of non-payment for Work performed, labor employed in and/or Equipment furnished relating to the construction of the Facility; and, in accordance with and subject to the terms of the Agreement, the Contractor hereby agrees to indemnify and hold the above parties harmless from any and all damages, costs expenses, demands, suits, and legal fees, directly or indirectly relating to any of the foregoing claims or liens.

4. The Contractor has not and will not assign any claim against the Beneficiaries, nor any lien or right to perfect a lien against the Facility and/or all or any portion of the Ghent Generating Station Site or improvements thereon, and the Contractor has the right, power, and authority to execute this Affidavit, Waiver and Release.

Exh_F5_Lein Waivers and General Release

161/h

5. The Contractor warrants that all laborers and subcontractors employed by it, and all suppliers or materialmen from which it has acquired Equipment incorporated into the Facility and any lien or bond claimant relating to the Contractor's work have been paid all amounts due and owing through the date hereof and that none of such laborers, subcontractors, suppliers, materialmen, or claimants has any unsatisfied or un-bonded lien against the Facility and/or all or any portion of the Ghent Generating Station Site or improvements thereon, through the date hereof. The Contractor further warrants that all due and outstanding applicable taxes, fees, contributions and benefits relating directly or indirectly to the Contractor's work and which are the responsibility of the Contractor under the Agreement have been paid in full.

6. No security interest has been given or executed by the Contractor for or in connection with any materials, equipment, appliances, machinery, fixtures, or furnishings placed upon or installed in Facility and/or all or any portion of the Ghent Generating Station Site or improvements thereon.

This Affidavit, Waiver of Lien and Release shall be an independent covenant and shall operate and be effective with respect to Work performed, labor employed and/or Equipment furnished under the Agreement and any related supplemental contract or contracts for extra or additional work performed by the Contractor in connection with the Facility. This Affidavit of Waiver and Release is for the benefit of the Beneficiaries.

IN WITNESS WHEREOF, this Affidavit, Waiver of Lien and Release has been executed on this _____ day of _____, 201X.

WITNESS:

(Name of Company)

Subscribed, sworn to, and acknowledged before me this _____ day of _____, 201X.

Notary Public

My Commission Expires:

(Notarial Seal)

8 79-Da-11

12/30/11 M

GHENT GENERATING STATION DIGGING PERMIT

æ

.

<u>*</u>...

EVISED: 8-17-05		PERMIT NUMBER:	· ·
This form is to be completed and approved before	ore beginning a	my digging at the Ghent Generation Statio.	n.
lequestor's Name & Company:		· .	
5 ¹⁰			_
)ate Of Request:			
late Work Is Planned To Start:			
escription Of Work To Be Done & Specific A	reas Involved:		a at here
	· · · · · · · · · · · · · · · · · · ·		
			•••
cetch Or Drawing Attached? (Y/N)	• •		
CONSTRUCTION OFFICE APPROVA	LS:	• • • • • • • • • • • • • • • • • • • •	
			*
		······································	· • •
ime:	Date:	ч	
HENT STATION APPROVALS:	* *		
ent Planning Office	Date:		
12-347-4052 ot 4053)	·	· · · ·	
hers, If Required (To Be Determined By Ghent	Station's Plann	ung Office Representative):	
	Date:		
		<u></u>	•
TE PERMISSION GIVEN TO DIG:			,
ME OF PERSON NOTIFIED:			
ies Fd: Ghent Planning Office: 1213olii AL	•	FGD Construction Office:	HBAH zy-Dectil

HOT WORK PERMIT

S10P1 Avoit/hot/work/orseekemalternative/sater/method/til/possible:

This *Hot Work Permit* is required for any temporary operation involving open flames or producing heat and/or sparks. This includes, but is not limited to: brazing, cutting, grinding, soldering, torch-applied roofing and welding.

e se SPa	
Instructions 1. Firesofiety supervisor: A. Verify precautions listed at right (or do not proceed with the work). B. Completo and retain Part 1. (Part 1A is for quality assurance documentation, if necessary.) C. Issue Part 2 to person performing hot work.	Required Procautions Checklist Available sprinkters, hose streams and extinguishers are in service/operable. Itat work equipment in good working condition. Requirements within 35 ft. (11 m) of hot work Flammable hould, dust, lint and olly denosits removed.
Hot work by Employee Contractor Job number	Combustible floors wet down, covered with damp sand or floors sive the floors wet down, covered with damp sand or fire-resistive sheets. Remove other combustible material where possible. Otherwise, protect
Location/Luilding and Ilon Nature of job Nature (prist) and signature of person performing hot work	 with FM Approved walding peds, blankets and curtains, fire-resistive tarpaulins or metal shiolds. All wall and floor openings covered. FM Approved welding pads, blankets and curtains installed under and around work. Project or shut down ducts and conveyors that might carry sparks to distant combustible material.
I verify the above location has been exemined, the precautions checked on the Required Precautions Checklist have been taken to provent fire, and permission is authorized for this work. Name (priot) and signature of literalety sepervisor/operations supervisor	Itot work on walls, ceilings or roofs Construction is noncombustible and without combustible covering or insulation Combustible material on other side of walls, ceilings or roofs is moved away Hot work on onclosed equipnient Enclosed equipment cleaned of all combustible material. Construction is proged of flammable liquid/vepor. Pressurized vessels, piping and oquipment removed from service.
Permit Date Time B.m. Expires Date Date Date Date Date Date Date Date	 Pressured vascus, prima and organism removed non-service, isolated and vented. Fire watch/hot work area monitoring Fire watch will be provided during and for 60 min, after work, including any break activity. Fire watch is supplied with suitable extinguishers, and where practical, a charged small hose.
To order additional hot work permits or other FM Global resources, order online 24 hours a day, seven days a week, at www.fmglobalcatalog.com. FM 610 b81' F2630 (REV. 6/07) Printed in USA (6/07) © 2003-2007 FM Global All rights reserved.	 Fire Watch is trained in use of equipment and in sounding elarm. Fire watch may be required in adjoining ereas, above and below. Monitor hot work area for an additional three (3) hours after the 60-min. fire watch. Other precautions takon:

12/30/11

EXHIBIT F8

PARENT GUARANTEE OF CONTRACTOR'S OBLIGATION

This Parent Guarantee (this "Guarantee") is made as of the ______ day of ______, 2011 by Tic Holdings, Inc., a Colorado corporation ("Guarantor"), to and in favor of Kentucky Utilities Company, a Kentucky corporation ("KU").

WITNESSETH:

WHEREAS, TIC – The Industrial Company, a Delaware corporation ("Contractor"), has entered into the Engineering, Procurement and Construction Agreement dated as of ______, 2011 (the "Agreement") for the supply of a Coal Combustion Residual (CCR) Transport system for installation and incorporation into the electric power generating unit specified thereby, along with certain ancillary or appurtenant equipment, materials and related technical services, as more particularly described therein; and

WHEREAS, Contractor is a wholly-owned subsidiary of Guarantor; and

WHEREAS, it is a condition to KU's payment obligations under the Agreement that Guarantor shall execute and deliver to KU a guarantee of performance and payment by Contractor in the form of this Guarantee;

NOW, THEREFORE, for and in consideration of the foregoing premises, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, GUARANTOR COVENANTS AND AGREES AS FOLLOWS:

AGREEMENT

1. Unless otherwise defined herein, all capitalized terms used herein and defined in the Agreement shall have their respective meanings as therein defined. All references to the Agreement made herein shall mean and refer to the Agreement as amended from time to time.

2.

Guarantor hereby irrevocably and unconditionally guarantees to KU, its successors (a)and permitted assigns, as primary obligor and not merely as a surety, (i) the full and prompt payment when due of all of Contractor's payment obligations under the Agreement and (ii) the full and prompt performance when due of all of Contractor's representations, warranties, covenants, duties and agreements contained in the Agreement. All obligations, representations, warranties, covenants, duties and agreements described in clauses (i) and (ii) above are collectively referred to herein as the "Obligations." If at any time Contractor fails, neglects or refuses to timely or fully perform any of the Obligations as expressly provided in the terms and conditions of the Agreement, then, upon receipt of written notice from KU specifying the failure, Guarantor shall perform, or cause to be performed, any such obligation, responsibility, or undertaking as required pursuant to the terms and conditions of the Agreement, including without limitation all payment obligations under the Agreement. It is expressly understood and agreed by Guarantor that to the extent Guarantor's obligations hereunder relate to Obligations which require performance other than the payment of money, KU may proceed against Guarantor to effect specific performance thereof (to the extent such relief is available) or for payment of damages resulting from Contractor's nonperformance. With respect to any claim, action or proceeding against Guarantor in connection with this Guarantee and except as otherwise provided herein, Guarantor shall be entitled to assert

12 30(11 Exh_F8_Parent Guarantee.doc

1 JB/h 1 24-Dac-11

those defenses, to the extent based on or arising pursuant to the terms and conditions expressly contained in the Agreement or as may otherwise be available at law or in equity, that Contractor would be entitled to assert in its own right had such claim, action or proceeding been asserted or instituted against Contractor directly.

(b) Notwithstanding anything to the contrary contained herein, Guarantor's aggregate liability for any and all demands made under this Guarantee shall not exceed the Contract Price. This Guarantee will terminate and be of no further force or effect upon the expiration of the Warranty Period expiration date other than with respect to claims made by KU under this Guarantee prior thereto.

3. This Guarantee is a continuing, absolute, irrevocable and unconditional guarantee by Guarantor of the Obligations. The obligations of Guarantor under this Guarantee shall remain in full force and effect and shall not be affected, reduced, modified or impaired, and Guarantor shall have no right to terminate this Guarantee or to be released, relieved or discharged, in whole or in part, from its payment or performance obligations under this Guarantee, by reason of any of the following:

(a) any amendment, supplement or modification to, waiver of, consent to or departure from, or failure to exercise any right, remedy, power or privilege under or in respect of, the Agreement, the Obligations, or any other agreement or instrument relating thereto to which Contractor is a party;

(b) the voluntary or involuntary liquidation, dissolution, sale or other disposition of all or substantially all of Contractor's assets, or the insolvency, bankruptcy, reorganization, dissolution or liquidation of, or any similar occurrence with respect to, or cessation of existence of, or change of ownership of Contractor, or any rejection of any of the Obligations in connection with any bankruptcy, insolvency or similar proceedings in which Contractor is the debtor or the insolvent party (with or without notice to or knowledge of Guarantor) or any disallowance of all or any portion of any claim against Contractor by KU, its successors and permitted assigns in connection with any such proceeding;

(c) any lack of genuineness, validity, regularity or value of the Agreement, any of the Obligations or any other agreement or instrument relating thereto to which Contractor is a party and which has been entered into by the parties in accordance with the terms of each respective agreement;

(d) any lack of legality or enforceability of the Agreement, any of the Obligations or any other agreement or instrument relating thereto to which Contractor is a party and which has been entered into by the parties in accordance with the terms of each respective agreement;

(e) any failure on the part of Contractor to perform or comply with any term of the Agreement;

(f) the assignment or transfer of this Guarantee, the Agreement or any other agreement or instrument referred to in the Agreement or applicable to Contractor or the Obligations by KU to a Person to which the Agreement is permitted to be assigned;

(g) any change in the ownership of any equity interest in Guarantor (including any such change that results in Guarantor no longer owning (directly or indirectly) an equity interest in Contractor;

(h) the merger or consolidation of Contractor into or with any other Person, or Contractor's loss of its separate corporate identity or Contractor ceasing to be an affiliate of Guarantor;

2 AB1/2 29-0cc-11

(i) any action by KU under the Agreement granting indulgence or extension of time for payment to any Person; or

(j) except as set forth in Section 2(b), any other event, circumstance, act or omission whatsoever which might in any manner or to any extent vary the risk of Guarantor or otherwise constitute a legal or equitable discharge of a surety or guarantor.

Without limiting the foregoing, Guarantor hereby consents and agrees that the following actions may be undertaken from time to time without notice to Guarantor: (x) the Agreement may be amended in accordance with its terms to increase or decrease the obligations of KU or Contractor thereunder; and (y) KU and Contractor may compromise or settle any unpaid or unperformed Obligations or any other obligation or amount due or owing, or claimed to be due or owing, under the Agreement.

4. Guarantor hereby unconditionally waives the defenses under this Guarantee of promptness, diligence, presentment, demand for payment, protest, notice of dishonor, notice of default, notice of acceptance, notice of intent to accelerate, notice of acceleration, notice of the incurring of the Obligations created under or pursuant to the Agreement and all other notices whatsoever (provided that Guarantor shall be entitled to written notice from KU specifying Contractor's failure in accordance with Section 2(a) above).

5. Guarantor agrees that this is a guarantee of payment and performance and not merely a guarantee of collection. The liability of Guarantor under this Guarantee shall not be conditional or contingent upon the pursuit of any remedy against Contractor or any requirement that KU exhaust any right, power or remedy or proceed against any other person prior to any action against Guarantor under the terms hereof. This Guarantee shall continue to be effective or be reinstated, as the case may be, if at any time any payment to KU by Contractor under the Agreement is rescinded or must otherwise be returned by KU for any reason, including, without limitation, upon the insolvency, bankruptcy or dissolution of Contractor or otherwise, all as though such payment had not been made.

6. Notwithstanding any payment or payments made by Guarantor hereunder, Guarantor shall not be entitled to exercise or enforce any subrogation rights arising by reason of payments made under this Guarantee, nor shall the Guarantor seek or be entitled to seek any contribution or reimbursement from Contractor in respect of payments made by Guarantor hereunder until all amounts owing to KU by Contractor on account of the Obligations and all amounts owing hereunder are indefeasibly paid in full. Once the Guarantor has so paid all amounts to KU in full under this Guarantee, then, and at such time, KU will reasonably assist Guarantor in exercising and enforcing any and all subrogation rights Guarantor may have, *provided* that the reasonable costs of providing such assistance shall be reimbursed by Guarantor to KU.

7. Guarantor agrees that payment or performance of any of the Obligations or other acts which toll any statute of limitations applicable to the Obligations or the Agreement shall also toll the statute of limitations applicable to Guarantor's liability under this Guarantee. No delay or omission on the part of KU, its successors or permitted assigns, in the exercise of, or failure to exercise, any right or remedy shall operate as a waiver thereof, a waiver of any other rights or remedies or a release of Guarantor from any Obligations hereunder and no single or partial remedy shall preclude any further exercise thereof or the exercise of any other right or remedies provided herein are cumulative and not exclusive of any rights and remedies provided by law or otherwise to enforce this Guarantee or the Obligations.

8. Guarantor additionally represents and warrants to KU as follows:

1213011 M Exh_F8_Parent Guarantee.doc

3 ABAK 201-Doc-11

(a) Guarantor is a corporation duly organized, validly existing, authorized to do business and in good standing under the laws of the State of Colorado.

(b) Guarantor has the requisite corporate power and authority to own its property and assets, transact the business in which it is engaged and to enter into this Guarantee and carry out its obligations hereunder. The execution, delivery, and performance of this Guarantee have been duly and validly authorized and no other corporate proceedings on the part of Guarantor or its affiliates are necessary to authorize this Guarantee or the transactions contemplated hereby.

(c) No authorization or approval or other action by, and no notice to or filing with, any governmental authority or other regulatory body or third party is required for the due execution, delivery and performance by Guarantor of this Guarantee.

(d) This Guarantee, when executed, shall constitute a valid and binding agreement of Guarantor and is enforceable against Guarantor in accordance with the terms of this Guarantee, except as may be limited by bankruptcy or insolvency or by other laws affecting the rights of creditors generally and except as may be limited by the availability of equitable remedies.

(e) As of the date hereof, the execution, delivery, and performance of this Guarantee does not and will not (i) result in a default, breach or violation of (A) the certificate or articles of incorporation or bylaws of Guarantor or (B) to the best of Guarantor's knowledge after due inquiry, any provision of any applicable law, rule, statute, legislation or order of any governmental authority, or (ii) constitute an event which would permit any person or entity to terminate rights or accelerate the performance or maturity of any indebtedness or obligation of Guarantor, the effect of which would materially affect Guarantor's ability to meet its obligations under this Guarantee, or (iii) constitute an event which would require any consent of a third party or under any agreement to which Guarantor is bound, the absence of which consent would materially and adversely affect Guarantor's ability to meet its obligations under this Guarantee.

9. No amendment of any provision of this Guarantee shall be effective unless it is in writing and signed by Guarantor, KU and any permitted assignee of KU's rights hereunder, and no waiver of any provision of this Guarantee, and no consent to any departure by Guarantor therefrom, shall be effective unless it is in writing and signed by KU and any permitted assignee of KU's rights hereunder.

10. This Guarantee is a continuing guarantee and shall (a) remain in full force and effect until its expiration in accordance with Section 2(b) above, (b) be binding upon Guarantor and its successors and (c) inure to the benefit of and be enforceable by KU and its successors and permitted assigns. Guarantor may not assign its rights or delegate its duties without the written consent of KU.

11. This Guarantee shall be governed by and construed in accordance with the internal substantive laws of the Commonwealth of Kentucky, excluding rules governing conflicts of laws. Each of Guarantor and, by its acceptance hereof, KU, (a) irrevocably submits to the exclusive jurisdiction of the United States District Court for the Western District of Kentucky, located in Louisville (and if such court does not have jurisdiction over a matter at controversy between Guarantor and KU, any state court located in Louisville, Kentucky), (b) waives any objection it may now or hereafter have to the laying of venue of any of the aforesaid courts in any action or proceeding involving this Guarantee and (c) agrees that service of process in any action or proceeding in such courts may be effected by mailing a copy thereof by registered or certified mail to them at their address provided in Section 12 below. EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT

Exh F8 Parent Guarantee.doc

4 Ablin 29-Dec-11

OF OR RELATING TO THIS GUARANTEE OR THE TRANSACTIONS CONTEMPLATED HEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY).

12. Any notices or other communication to be given hereunder shall be given in writing, sent by (a) personal delivery, (b) internationally recognized expedited delivery service, (c) registered or certified United States mail, postage prepaid, or (d) facsimile (followed by registered or certified United States mail, postage prepaid) as follows:

If to Guarantor:

TIC Holdings, Inc. 2211 Elk River Road Steamboat Springs, Colorado 80477 Attention: Pat Barney, General Counsel

With a copy (which shall not constitute notice) to:

TIC – The Industrial Company 188 Inverness Drive West, Suite 700 Englewood, Colorado 80120 Attention: Cam Chandler, Legal Counsel

If to KU:

Kentucky Utilities Company 820 West Broadway Louisville, Kentucky 40202 Attention: Scott Straight, Director of Project Engineering Email: scott.straight@eon-us.com

With a copy (which shall not constitute notice) to:

Kentucky Utilities Company 220 West Main Street Louisville, Kentucky 40202 Attention: John McCall, General Counsel Email: john.mccall@eon-us.com

or to such other address or to the attention of such other individual as hereafter shall be designated in writing by the applicable party sent in accordance herewith. Any such notice or communication shall be deemed to have been given either at the time of personal delivery or, in the case of delivery service or mail, as of the date of receipt at the address and in the manner provided herein, or in the case of facsimile, upon receipt.

13. In the event that any of the provisions, or portions or applications thereof, of this Guarantee are held to be unenforceable or invalid by any court of competent jurisdiction, KU and Guarantor shall negotiate an equitable adjustment in the provisions of this Guarantee with a view toward effecting the purpose of this Guarantee, and the validity and enforceability of the remaining provisions, or portions or applications thereof, shall not be affected thereby.

5 JBK-24-Dec-11

IN WITNESS WHEREOF, Guarantor, intending to be legally bound hereby, has caused this Guarantee to be executed and effective as of the date first written above upon the signature of its duly-authorized officer appearing below:

.

TIC HOLDINGS, INC.

By:	
Nam	e:
Tifle	:

12/30/11 w Exh_F8_Parent Guarantee.doc

6 ABth 24-Da-11

EXHIBIT F9

Second Tier Procurement Program

1.0 Summary

Attached are Contractor forms for data submittal regarding the use of Minority Business Enterprises (MBE), Women Business Enterprises (WBE), Disadvantaged Business Enterprises (DBE), union and non-union local contractors (LC), union contractors (UC), and non-union contractors (NUC). Contractor does not have to use the exact form; however, all the information requested must be submitted. Expenditures for MBEs, WBEs, DBEs, LCs, UCs, NUCs must be supplied separately; do not combine these expenditures. Where a supplier meets more than one category, identify the categories and whether or not the values for each category include the expenditures. Ghent CCR Transport Supplier forms submitted to the Contractor under the Ghent CCR Transport Supplier Contract shall be submitted to the Owner.

2.0 Definitions and Terms of Management and Ownership

- 2.1 WBE: Woman Business Enterprise: A certified business enterprise that is at least 51% owned by a woman; or, in the case of any publicly owned business, at least 51% of the stock of which is owned by one or more women, and whose management and daily business operations are controlled by one or more women.
- 2.2 MBE: Minority Business Enterprise: A certified business enterprise at least 51% owned by a minority individual or group, or in case of any publicly owned business, at least 51% of the stock of which is owned by one or more minority groups, and, and whose management and daily business operations are controlled by one or more of these individuals. KU shall presume that minority includes, but is not limited to African Americans, Hispanic American, Asian Pacific Americans, Native American, Subcontinent Asian American, and other groups.
- 2.3 DBE: Disadvantaged Business Enterprise: A certified business enterprise at least 51% owned by a disadvantaged individual or group, or in case of any publicly owned business, at least 51% of the stock of which is owned by one or more disadvantaged groups, and, and whose management and daily business operations are controlled by one or more of these individuals. KU shall presume that disadvantaged includes, but is not limited to Veteran Owned, Service Disabled Veteran Owned and other groups found to be disadvantaged by the U.S. Small Business Administration.
- 2.4 LC: union and non-union local contractor: A business enterprise with primary operations within the Commonwealth of Kentucky or the Louisville Kentucky, Evansville Indiana, or Cincinnati Ohio Metropolitan Statistical Area as defined by the U.S. Census Bureau.
- 2.5 UC: union contractor: A business enterprise with primary operations utilizing union labor.
- 2.6 NUC: non-union contractor; A business enterprise with primary operations utilizing nonunion labor.

fBth-

3.0 Expenditure Definitions

- 3.1 Direct expenditures are those materials and services acquisitions from MBEs/WBEs/DBEs/LCs/UCs/NUCs directly attributable to the Agreement. Direct expenditures shall be reported at 100% of cost.
- 3.2 Indirect expenditures are those materials and service acquisitions from MBEs/WBEs/DBEs/LCs that can not be identified or apportioned to any specific customer. For example -- "overhead" items such as paper, computing expenses, or office maintenance incurred by your company. Also included would be any and all materials and services purchased from MBEs/WBEs/DBEs/LCs that are used in the direct production of your product or service. These "indirect expenditures" should be proportionately allocated and reported to Owner as outlined in the example below:

"MBE Indirect Allocation" formula <u>Agreement Price</u> = Allocation Factor Total YTD Sales

(Total YTD MBE^{*} Indirect Expenditures X Allocation Factor) = Agreement MBE^{*} Allocation

Example: Contractor's total sales are \$50 million; Agreement Price is \$3 million, and total indirect MBE expenditures are \$5,000,000.

 $\frac{\$3,000,000}{\$50,000,000} = 0.06$

5,000,000 X 0.06 = \$300,000 (KU's MBE Allocation)

For both direct and indirect expenditures, list the names of the business enterprise, addresses, and dollar amounts.

*Note: A separate and similar calculation should also be made for WBEs, DBEs and LCs.

4.0 Expenditure Reporting

Submit the requested information with each monthly invoice per Article 16.19 of the Agreement (Notices).

13/h-

12/30/11

m

(Company Name)

MBE/WBE/DBE/LC Indirect Expenditure Report KU Second Tier Program _____, 20__

KU Contract/Purchase Order Number (if applicable): Indirect MBE Expenditure Allocation: Use Factor from Previous Page Sales To KU \$ \$_____ Total MBE Expenditures **KU** Allocation \$ Indirect WBE Expenditure Allocation: \$ Sales To KU \$ Total WBE Expenditures KU Allocation \$ Indirect DBE Expenditure Allocation: Sales To KU \$ \$_____ **Total WBE Expenditures** \$ KU Allocation Indirect LC Expenditure Allocation: Sales To KU \$ -----\$ Total WBE Expenditures \$ LG&E and KU Services Co,

<u>Note:</u> See indirect expenditure allocation formula explanation in Exhibit F9 of the Agreement (Second Tier Reporting Program)

Submitted By

Title

Date

Bth 24-Decul

12/30/11 Ar
MBE/WBE/DBE/LC/UC/NUC Direct Expenditure Report
Second Tier Program

_____, 20___

Contractor:	Report Date:
	Contract/Purchase Order No (if applicable):
	Submitted By: Title:
Telephone:	

				(4)	(5)
(1) WORK DESCRIPTION	(2) SUBCONTRACTING/SUBSUPPLIER NAME CONTACT PERSON & TELEPHONE NO.	(3) TAX ID NO.	MBE/WBE DBE/LC UC/NUC	ORGINAL PROPOSED EXPENDITURES	ACTUAL PAYMENT
		· · · · · · · · · · · · · · · · · · ·			
		TOTAL			\$

REFER TO GUIDELINES ATTACHED TO THIS FORM

KU Guidelines for Second Tier (Direct Expenditure) Reporting

Contractor should complete requests 1 through 6, reporting its use of Minority Business Enterprises (MBEs), Women Business Enterprises (WBEs), Disadvantaged Business Enterprises (DBEs), union and non-union local contractors (LCs), union contractors (UCs), and non-union contractors (NUC) in direct connection with the goods/services provided under the Agreement.

Contractor should submit reports with the monthly invoice in a format equivalent to the attached MBE/WBE/DBE/LC/UC/NUC Direct Expenditure Report per

Article 16.19 of the Agreement (Notices).

Contractor should provide the following Second-Tier information when submitting its monthly invoice on MBE/WBE/DBE/LC/UC/NUC expenditures either on

the attached "Direct expenditure" form, or in an equivalent format: Work Description: a brief statement of the goods/services provided by the MBE/WBE/DBE/LC/UC/NUC

- Subcontractor/Sub-supplier Name Contact Person & Telephone No.: the name of a contact within the MBE/WBE/DBE/LC/UC/NUC who can 1.
- 2. verify expenditures made by GHENT CCR TRANSPORT Suppler.
- Tax ID Code: if available. 3.
- MBE/WBE/DBE/LC/UC/NUC: Designate whether company is an MBE/WBE/DBE/LC/UC/NUC 4.
- Original Proposed Expenditures: initial estimate of anticipated expenditures with MBE/WBE/DBE/LC/UC/NUC Actual Payment: total of payments made to the MBE/WBE/DBE/LC/UC/NUC through the current reporting period 5.
- 6.

131/L 5-29-Noc-11

EXHIBIT F-10

FORM OF PERFORMANCE LETTER OF CREDIT

BENEFICIARY: Kentucky Utilities Company, a Kentucky corporation 820 West Broadway, Louisville, Kentucky 40202 Telephone: (_______. Fax: (_______. E-mail: ______@lge-ku.com Irrevocable Standby Letter of Credit No. Re:

Issue Date: _____, 20____

We, /Bank Name/, hereby establish in favor of the above-named Beneficiary (the "Beneficiary") this Irrevocable Standby Letter of Credit No. _____ (this "Letter of Credit") at the request of and for the account of TIC – The Industrial Company, a Delaware corporation, with its principal place of business at 2211 Elk River Road, Steamboat Springs, Colorado 80487 ("TIC"), effective immediately and expiring on the date set forth in numbered Paragraphs 5 and 6 hereof.

This Letter of Credit, we are informed, is issued pursuant to the terms of the Engineering, Procurement and Construction Agreement, dated as of ______, 2011, by and between TIC and the Beneficiary, as the same may be amended from time to time (the "Agreement").

1. Stated Amount. The aggregate amount of funds available under this Letter of Credit shall be U.S. \$ _____ United States Dollars), less the aggregate amount of all previous drawings under this Letter of Credit (the "Stated Amount"). The Stated Amount may be reduced at any time upon our receipt of a Reduction Certificate executed by the Beneficiary in the form of Attachment E.

Drawings. A drawing hereunder may be made by the Beneficiary on any Business Day on or prior 2. to the date this Letter of Credit expires by delivering to /Bank Name/, at any time during its business hours on such Business Day, at /_____ / or at such other address as may be designated by written notice delivered to the Beneficiary as contemplated by numbered **Paragraph 9** hereof, an executed original (or, if applicable, a duplicate original issued in accordance with Paragraph 14) of this Letter of Credit together with (i) a draw certificate executed by an authorized officer of the Beneficiary in the form of Attachment A hereto (the "Draw Certificate"), completed and purportedly signed by such authorized officer and (ii) your draft in the form of Attachment B hereto (the "Draft"), completed and signed by such authorized officer. Partial drawings are permitted. Draw Certificates and Drafts under this Letter of Credit may be presented by you by means of the above documents sent by overnight delivery or ___/ (or at such courier to /Bank Name/, at its address set forth above, Attention: / other address as may be designated by written notice delivered to you as contemplated by numbered Paragraph 9 hereof).

3. Time and Method for Payment. We hereby agree to honor a drawing hereunder made in compliance with this Letter of Credit by transferring in immediately available funds the amount specified in the Draft delivered to us in connection with such drawing to such account at such bank in the United States as you may specify in your Draw Certificate at the opening of business on the first (1st) Business Day succeeding the date of presentation of the conforming Draw Certificate. All amounts to be paid under this Letter of Credit shall be made without set-off or counterclaim by us of any kind.

4. Non-Conforming Demands. If a demand for payment made by you hereunder does not, in any instance, conform to the terms and conditions of this Letter of Credit, we shall give you prompt notice, no later than one (1) Business Days following our receipt, that the demand for payment was not effected in 12/301 accordance with the terms and conditions of this Letter of Credit, stating the reasons therefor and that we will upon your instructions hold any documents at your disposal or return the same to you. Upon being notified that the demand for payment was not effected in conformity with this Letter of Credit, you may correct any such non-conforming demand prior to the date this Letter of Credit expires.

5. Expiration. This Letter of Credit shall automatically expire at the close of business on the earlier of one year from the Issue Date specified above (or such later date to which this Letter of Credit may be extended in accordance with the provisions of numbered Paragraph 6 hereof), or the date on which we receive a Cancellation Certificate in the form of Attachment C hereto executed by an authorized officer of the Beneficiary upon receipt of which this Letter of Credit shall be cancelled.

6. **Rollover.** This Letter of Credit shall be deemed automatically extended without amendment for additional periods of one (1) year each from the present or any future expiration date, unless at least sixty (60) days prior to any such expiration date we notify you by registered mail or overnight courier at the above address (or such other address as may be designated by you as contemplated by numbered **Paragraph** 9 hereof and as amended accordingly) that we elect not to consider this Letter of Credit extended for any such additional one (1) year period. In the event that thereafter a letter of credit in replacement of this Letter of Credit and satisfactory to you shall not have been delivered to you by the date that is thirty (30) days prior to the expiration date of this Letter of Credit, you may draw upon this Letter of Credit pursuant to numbered Paragraph 2 hereof in the full amount then available for drawing hereunder. In no event shall this Letter of Credit be automatically extended beyond , 20 (the "Outside Expiration Date"); provided, that, if TIC's letter of credit obligations shall remain in effect under the Agreement as of the Outside Expiration Date and you shall not have received by the date that is thirty (30) days prior to the Outside Expiration Date a letter of credit in replacement of this Letter of Credit satisfactory to you, you may draw upon this Letter of Credit pursuant to numbered Paragraph 2 hereof in the full amount then available for drawing hereunder.

7. Business Day. As used herein, "Business Day" shall mean any day on which commercial banks are not authorized or required to close in the Commonwealth of Kentucky.

8. Governing Law. Except as otherwise expressly stated herein, this Letter of Credit shall be governed by the International Standby Practices, International Chamber of Commerce Publication No. 590 ("ISP 98") and any amendments or revisions thereto. As to matters not addressed by ISP 98, this Letter of Credit shall be governed by and construed in accordance with the laws of the Commonwealth of Kentucky, without giving effect to any choice of law rules which may direct the application of the laws of another jurisdiction.

9. Notices. All communications to you in respect of this Letter of Credit shall be in writing and shall be delivered to the address (or, as applicable, the fax number) shown for you herein or such other address (or applicable fax number) as may from time to time be designated by you in a written notice to us and as amended accordingly. All documents to be presented to us hereunder and all other communications to us in respect of this Letter of Credit, which other communications shall be in writing, shall be delivered to the address (or, as applicable, fax number) for us indicated on the signature page hereof, or such other address (or applicable fax number) as may from time to time be designated by us in a written notice to you.

10. Irrevocability. This Letter of Credit is irrevocable.

11. Transferability. This Letter of Credit is transferable in its entirety, but not in part and may be successively transferred. Only the issuing bank is authorized to act as the transferring bank with respect to the Letter of Credit represented hereby. We shall not recognize any transfer of this Letter of Credit until this original Letter of Credit (or, if applicable, any duplicate original issued pursuant to Paragraph 14 below), together with any amendment(s) and an appropriately completed and executed Transfer Certificate in the form of Attachment D hereto, is received by us and our transfer fee of $\frac{14}{10}$ of 1% (Minimum \$250.00) of the

amount transferred is paid to us. The correctness of the signature and title of the person signing the transfer form must be certified by a duly-authorized officer acting in the name and on behalf of the Beneficiary or otherwise be authenticated by your bank. Upon and after any such transfer under this Letter of Credit, all Drafts and related documentation deliverable hereunder must be executed by the transferee as the Beneficiary hereof. This Letter of Credit may not be transferred to any person with which U.S. persons are prohibited from doing business under U.S. foreign assets control regulations or other applicable U.S. laws and regulations.

12. Fees. All fees associated with this Letter of Credit are for account of TIC, except transfer fees to be paid by Beneficiary associated with Attachment D of form F-10.

13. Complete Agreement. This Letter of Credit sets forth in full our undertaking, and such undertaking shall not in any way be modified, amended, amplified or limited by reference to any document, instrument or agreement referred to herein, except as provided in numbered Paragraph 8 hereof, Attachments A, B, C, D and E hereto and the notices referred to herein and any such reference shall not be deemed to incorporate herein by reference any document, instrument or agreement except as set forth above. No amendment or modification of this Letter of Credit may be made without the prior written consent of the Beneficiary and TIC except when the amount of this Letter of Credit is increased.

14. **Replacement Instrument**. If the original of this Letter of Credit has been lost, stolen, mutilated or destroyed, we will promptly issue to the Beneficiary a duplicate original of this Letter of Credit upon our receipt of our acceptable indemnity letter signed by the Beneficiary.

Sincerely,

[BANK NAME]

(Bank Address)

(City, State, Zip Code)

(Telephone number)

(Authorized Name and Title)

(Authorized Signature)

12130111 R

ATTACHMENT A

FORM OF DRAW CERTIFICATE

The undersigned hereby certifies to /Bank Name/ (the "Issuer"), with reference to Letter of Credit No. (the "Letter of Credit") issued by Issuer in favor of the Beneficiary (capitalized terms used and not defined herein shall have the respective meanings set forth in the Letter of Credit), as follows:

- (1) The undersigned is the ______ of Kentucky Utilities Company (Beneficiary) and is duly authorized by the Beneficiary to execute and deliver this Certificate on behalf of the Beneficiary;
- (2) Beneficiary hereby makes demand against the Letter of Credit by presentation of the Draft accompanying this Draw Certificate, for payment of (U.S.\$_____), such amount not being in excess of the Stated Amount;
- (3) Beneficiary is entitled to draw the amount set forth in clause (2) hereof because the conditions for a drawing pursuant to the Agreement or numbered **Paragraph 6** of the Letter of Credit have been met; and
- (4) You are hereby directed to make payment of the requested drawing to:

[Insert Wire Instructions]

/Beneficiary's Name/

Ву:	
Name:	
Title:	

Address: _____

12/30/11

ATTACHMENT B

DRAWING UNDER THE LETTER OF CREDIT NO.

Date:

ON: The first Business Day immediately succeeding the date of presentation

PAY TO: /Beneficiary's Name/

\$_____U.S.

FOR VALUE RECEIVED AND CHARGED TO THE ACCOUNT OF LETTER OF CREDIT NO.

[Beneficiary's Name]

By:	
Name:_	
Title:	

Address: _____

,2/30/11

12/30/11

ATTACHMENT C

CANCELLATION CERTIFICATE

Letter of Credit No.

The undersigned hereby certifies to /Bank Name/ ("Issuer"), with reference to Letter of Credit No. issued by Issuer to Kentucky Utilities Company (Beneficiary) (the "Letter of Credit") that either (i) the terms of the Agreement require the return of this Letter of Credit or (ii) the Agreement has been terminated and TIC has satisfied all of its outstanding obligations and paid all amounts remaining due under the Agreement following such termination. Pursuant to numbered Paragraph 5 thereof, the Letter of Credit shall expire upon Issuer's receipt of this certificate. Attached hereto is the Letter of Credit marked "Canceled". Capitalized terms used and not defined herein shall have the respective meanings set forth in the Letter of Credit.

[Beneficiary's Name]

By:	
Name:_	
Title:	

Address:	
Address:	

13/30/1

Page 6 of 9

ATTACHMENT D

TRANSFER CERTIFICATE

_____, 20-__

[BANK NAME AND ADDRESS]

Re: Letter of Credit No. -----

We request you to transfer all of our rights as Beneficiary under the Letter of Credit referenced above to the Transferee(s), named below:

Name of Transferee(s)

Address(es)

By this transfer, all our rights as the transferor, including all rights to make drawings under the Letter of Credit, go to the Transferee(s). The Transferee(s) shall have sole rights as beneficiary(ies), whether existing now or in the future, including sole rights to agree to any amendments, including increases or extensions or other changes and all references to "Beneficiary" in the Letter of Credit, any drawing certificate in the form of Attachment A or the other Attachments to the Letter of Credit shall be deemed to mean the Transferee(s). All amendments will be sent directly to the Transferee(s) without the necessity of consent by or notice to us. Capitalized terms used and not defined herein shall have the respective meanings set forth in the Letter of Credit.

We enclose the original Letter of Credit and any amendments. Please indicate your acceptance of our request for the transfer by endorsing the Letter of Credit and sending it to the Transferee(s) with your customary notice of transfer.

For your transfer fee of \$_____.

* Enclosed is our check for \$_____

You may debit my/our Account No.

NAME OF TRANSFEROR

NAME OF AUTHORIZED SIGNER AND TITLE

AUTHORIZED SIGNATURE

The signature and title above conform with those shown in our files as authorized to sign for the Beneficiary. Policies governing signature authorization as required for withdrawals from customer accounts shall also be applied to the authorization of signatures on this form. The authorization of the Beneficiary's signature and title on this form also acts to certify that the authorizing financial institution (i) is regulated by a U.S. federal banking agency; (ii) has implemented anti-money laundering policies and procedures that comply with applicable requirements of law, including a Customer Identification Program (CIP) in accordance with Section 326 of the USA PATRIOT Act; (iii) has approved the Beneficiary under its anti-money laundering compliance program; and (iv) acknowledges that *JBank NameJ* is relying on the foregoing certifications pursuant to 31 C.F.R. Section 103.121 (b)(6).

NAME OF BANK

AUTHORIZED SIGNATURE AND TITLE

PHONE NUMBER

ATTACHMENT E

REDUCTION CERTIFICATE

[Bank Name] [Address 1] [Address 2] Attention:

Irrevocable Standby Letter of Credit No.

The undersigned hereby certifies to *[Bank name]* ("Issuer"), with reference to the above-specified Inrevocable Standby Letter of Credit issued by Issuer to Beneficiary (the "Letter of Credit"), that, as of and from the reduction date specified below, the Stated Amount is hereby reduced as set forth below:

Reduction Date

Reduction Amount \$xx,xxx.xx Revised Stated Amount \$xx,xxx,xxx.xx

/Beneficiary's Name/

Ву:	
Name:	~
Title:	

Address: _____

12/30/W

EXHIBIT G

Guarantees and Performance Guarantee Test Procedure

Table of Contents

1.	INTRODUCTION	2
1.1	General	2
1.2	Contractor Requirements	2
1.3	Performance Guarantee Test and Conditions	4
1.4	Performance Guarantee Test Interruptions	5
1.5	Corrections	5
1.6 1.6.1 1.6.2		6 6 6
2.	PERFORMANCE GUARANTEE TESTS	7
2.1	Contractor's Preliminary Performance Guarantee Testing	7
2.2 2.2.1 2.2.3		8 8 9

13/1-29-Da-41 1

1. Introduction

1.1 General

Performance Guarantees shall be as specified herein and shall be measured during the final Performance Guarantee Test under normal operating conditions and no special cleaning, adjustments, or other preparations will be allowed. All Performance Guarantees are to be tested simultaneously in one Performance Guarantee Test. All Performance Guarantees Guarantees are make-right obligations.

Contractor shall be responsible for carrying out the Performance Guarantee Tests which are those tests that are conducted to demonstrate that the Performance Guarantees have been achieved. Performance Guarantee Tests shall be completed after satisfactory achievement of Mechanical Completion as outlined in the Agreement.

1.2 Contractor Requirements

The Contractor is responsible for all testing required by this Exhibit and the Agreement to demonstrate that the Ghent CCR Transport System has been designed and constructed and is capable of performing according to the requirements of the Agreement

The Contractor shall be responsible to provide all required Performance Guarantee Test connections, equipment, instrumentation, or testing Subcontractors for Performance Guarantee Testing.

Contractor shall certify to the Owner that furnished Equipment and Systems are ready for Performance Guarantee Tests.

If the results of any Performance Guarantee Test results are inconclusive due to bad data or inadequate data, the Performance Guarantee Tests shall be repeated.

If malfunction of the Equipment and/or Systems furnished by Contractor under this Agreement negates the Performance Guarantee Test, or if the Equipment and/or Systems furnished under this Agreement do not meet the performance requirements as determined by the Performance Guarantee Tests, the cost of additional testing will be at Contractor's expense.

Contractor will provide an independent third party laboratory/testing contractor/testing agency ("Third party Testing Contractor) to administer all Performance Guarantee Tests and to demonstrate Performance Guarantees have been achieved. Performance Guarantee Testing is to be performed as soon as practical after Mechanical Completion is achieved . If during Performance Guarantee Tests, all Performance Guarantees are not met simultaneously, Contractor shall make revisions to its Work during the next scheduled

J-BAL EQ-Dec-11

12/30/11 W outage or at an earlier time convenient to Owner. If Performance Guarantees are still not achieved, additional repairs and retesting shall be done by Contractor. All costs for repairs (including engineering, equipment, material, installation and retesting) shall be borne by Contractor.

3 29. Davil



-4 29-Dec-11

12/30/11 W

Kentucky Utilities



12/30/11

5



6 ZQ-Dec-11



ABAL 29-Dar-M

7





12/30/11 m

8 29-Dec-1



9 2-4-Dec-11

EXHIBIT H

HEALTH AND SAFETY DOCUMENTS

Table of Contents

1.0	Health and Safety Documents	2
1.1	Passport Program Process Steps	2
1.1	.1 Step 1 – Certification	2
1.1	.2 Step 2 – Passport Training	2
	.3 Step 3 – Attestation Form	3
	.4 Step 4 – Site Specific Orientation	4
	.5 Step 5 – Contractor Reporting Requirements	4
1.2	Administration	5
1.3	Passport Revocation and Reinstatement Guidelines	5
APP	ENDIX B LG&E AND KU SERVICES COMPANY & AFFILIATES	6
APF	PENDIX C CONTRACTOR / SUBCONTRACTOR SAFETY POLICY	6
	ntractor On-site Health and Safety Representative	10
	Health and Safety Management Plan	12
	Hazard Analysis	14
	izard Analysis Requirements	14

T 24-Dec-M

1

1.0 Health and Safety Documents

Contractor is required to cooperate with Owner on all matters of health and safety. To enhance the welfare of all who visit and work in and around Owner's facilities, an enhanced contractor safety program has been developed. Building on internal and external best practices, a cross-functional team has developed the "Passport Program." The key components of the program are outlined below.

All personnel working on Site shall undergo both the Ghent Generating Station Contractor Orientation course and the Ghent Generating Station Ammonia Awareness course on their first arrival at Site. The approximate time to complete these courses is one hour.

Contractor shall also comply with Owner's Safety Manual, which is available upon request. As part of Contractor's requirements, several reports are required, which are detailed in the Passport Program and Safety Manual.

Safety is a core value at Kentucky Utilities. To enhance the welfare of all who visit and work in and around Kentucky Utilities facilities, an enhanced Contractor safety program has been developed. Building on internal and external best practices, a cross functional team has developed the "Passport Program." The Passport Program is designed to cover industrial workers. The key components of the program are outlined below.

1.1 PASSPORT PROGRAM Process Steps

1.1.1 Step 1 – Certification

All Kentucky Utilities Contractors will be certified prior to performing work.

1.1.2 Step 2 – Passport Training

All industrial workers employed by a certified Contractor must complete an 8-hour training program designed to enlighten employees on the importance of safety and the hazards associated with working in an industrial environment. This training will also identify additional specific OSHA, EPA and DOT compliance training that may be needed in certain situations. The Passport training, however, will not serve to fulfill any of the compliance training required by the above listed agencies. It will be the responsibility of the Contractor to provide any compliance training required for their employees.

There will be two options available to Contractors with regard to the Passport Training:

2 ABIL

Option $1 - \text{Train-the-Trainer} - \text{Kentucky Utilities will provide a curriculum and conduct train-the-trainer sessions at appropriate intervals for Contractor's key safety/ training personnel. For those Contractors choosing this option, a resume for each prospective trainer must be submitted and must include the following information:$

• Training delivery and development experience

,

- Knowledge of OSHA, DOT, and EPA Standards applicable to the work for which Contractor is responsible
- Health and safety knowledge and experience in managing a Health and Safety program
- Experience in the utility and power generation industry
- Experience in the type of work to be performed by Contractors

By virtue of their attendance and ability to pass the closing examination, these key personnel would then be approved to provide training to Contractor's employees to meet the requirements of a "Passport."

NOTE: Kentucky Utilities reserves the right to reject any Contractor employee as a potential trainer if any of the following occurs:

The above referenced information regarding experience and qualifications is not submitted

The information submitted does not adequately indicate the prospective trainer's ability to perform the duties of a trainer for the Passport program.

The prospective trainer does not complete the required train-the-trainer session, including successfully pass the final examination.

Option 2 – External Provider - External providers of the Kentucky Utilities Passport safety training program will also be assessed and certified by Corporate Safety in accordance with Option 1. This will allow certified Contractors to seek Passport training for their employees from an external provider.

1.1.3 Step 3 – Attestation Form

Contractors will be required to attest to the fact that each employee who will be working on the Site has received the required Passport training and is current on all required compliance training for the work that employee will be performing. Although Kentucky Utilities will be looking for confirmation that compliance training has been completed, it is not a requirement that the Contractor provide training records for all individuals, and Kentucky Utilities will not monitor compliance training delivered by Contractors to their employees. However, Passport training audits will be performed to ensure the adequacy

-J-BIL Zu-Dec.~M

of the training provided. If an incident does occur, Kentucky Utilities may request individual training records as part of their accident and incident investigation process.

Upon successful completion of the required safety training, the Contractor will enter the employee's name, date of birth and training information into the Kentucky Utilities Contractor Health & Safety System. An electronic notification will be sent to the appropriate Kentucky Utilities representative for Passport authorization. Upon approval (on-line), the Contractor will be notified electronically that the passport has been approved and that the Contractor can print and issue a Passport card to their employee. The card will have an identification number printed on it that will associate the worker with his or her records in the Database. The Contractor employee must carry this card and valid photo ID at all times while on Site.

The Passport does not serve as security clearance for an employee. The Passport merely attests to the fact that the Contractor employee has completed all required training. Site access will be handled in accordance with local site access procedures. For long-term Contractors, a photo ID with a magnetic strip may also be issued to a Contractor employee for security purposes. For all other Contractor employees, a sign-in sheet may be utilized to track individuals on Site.

1.1.4 Step 4 – Site Specific Orientation

Each Contractor employee must attend a site specific orientation training identifying parking directions, security procedures, site map, emergency evacuation procedures, emergency contact names, medical facility locations, specific alarms, and site-specific hazardous materials. A separate orientation will be required for each generation site for which the Contractor employee works. This orientation will normally occur on the first day of work on the job Site.

1.1.5 Step 5 – Contractor Reporting Requirements

All accidents, injuries, dangerous occurrences and near misses shall be reported as soon as possible to the nominated representative of Kentucky Utilities. In any case, all injuries shall be reported within 1 hour.

Contractors shall also report statistical information to Kentucky Utilities on a monthly basis. This is to include all sub contractors working for contractor on the Site. The information required is:

- Number of hours worked at each Kentucky Utilities job site
- Number of Fatalities, Lost Workday Cases and OSHA Recordable Injuries for each job site.

The preceding month's statistical information shall be entered into the Kentucky Utilities Contractor Health & Safety Database by the Contractor no later than the 5^{th} day of the month.

- ABILL Za-Dec-M

1.2 Administration

- All affected Contractor employees must have a passport.
- The Passport is valid for 12 Months. Refresher training options will be developed and provided annually
- The cost of training will be the responsibility of the Contractor.
- The Contractor is responsible for ensuring that all of the above requirements are met for every individual worker utilized in work at Kentucky Utilities facilities. This includes all subcontractors utilized by Contractor. Contractor will be responsible for ensuring that such subcontractor employees have met all of the requirements regarding issuance of a Passport and for ensuring that all reporting requirements outlined in Step 5 above are fulfilled with regard to the subcontractor.
- Kentucky Utilities reserves the right to revoke an individual's Passport. See Passport Revocation and Reinstatement Guidelines below.
- Contract crew safety audits will be performed to assess effectiveness of and compliance with intent of Passport Program. These audits will be conducted by Site Safety, Site Contract Proponents, and Managers.
- Corporate Health & Safety reserves the right to audit contractors for appropriate compliance and passport training documents.

1.3 Passport Revocation and Reinstatement Guidelines

Kentucky Utilities reserves the right to revoke an individual's Passport. A Passport can be revoked for:

- Failure to comply with safety rules, procedures or programs
- Failure to comply with drug and alcohol rules or testing requirements
- Creation of an unsafe condition that has potential to result in death or serious injury
- Any reason deemed appropriate by the Responsible Manager

It is NOT mandatory for a Passport to be revoked following a report of an accident, injury or near-miss. If an investigation finds sufficient evidence to support revocation, a report and recommendation will be sent to the Responsible Manager for review and action.

If a Contractor wishes to appeal a revocation decision, this should be made in writing to the Responsible Manager.

A Passport can be reinstated after the Contractor has satisfied the Responsible Manager that the individual will conduct themselves appropriately on the Site.

5 - y-Dec-11

If an individual's Passport is revoked for a second time, the individual will not be allowed to reapply for a Kentucky Utilities Passport within 5 years without consent of a General Manager.

2.0 Contractor Safety Policy

Appendix B LG&E and KU Services Company & Affiliates (KU, LG&E) LG&E and KU Services Company

Appendix C Contractor / Subcontractor Safety Policy

PURCHASE ORDER #:_	CONTRACT JOB #:	
NAME OF CONTRACTO	R:	
SCOPE	OF	Work:
Work		Location:
CONTACT NAME:	WORK ORDER #:	

1. Contractor / Subcontractor Safety Policy

General

LG&E and KU Services Company, is committed to safety excellence and in providing a safe and healthful work environment for anyone working on our property. The personal safety and health of each employee, contractor and the safety of the general public are of primary importance to LG&E and KU Services Company. Accordingly, there is no job so important that safety policies and procedures or legal obligations are compromised.

This Policy does not replace the Contractor's/subcontractor's ("Contractor") existing safety and health program(s), provided that their program(s) meet or exceed these and any additional site specific minimum requirements. Contractor's employees not following this Policy will be subject to removal from the job site.

The Contractor is required to comply with all federal and state safety laws and all provisions of the LG&E and KU Services Company, Health & Safety Manual. The Contractor is responsible for conducting its work and activities safely. LG&E and KU Services Company expect and require that you continuously update your employees with respect to safety issues



relevant to the work and to take immediate corrective action when your employees violate safety rules or procedures.

It is the responsibility of Contractors' construction managers, superintendents, safety representatives and foremen/supervisors to ensure workers under their supervision maintain safe work areas and perform their tasks in a safe manner. It is also the responsibility of each worker to follow every precaution and LG&E and KU Services Company safety rule and Policy to protect them and their fellow workers.

Contractors are responsible for ensuring that any subcontractors working under their purview are held to the same performance expectations, and therefore this Policy, as the contractor themselves.

2. <u>Scope</u>

General

This Policy applies to all construction activities performed for LG&E and KU Services Company by Contractor's employees or employees of the Contractor's subcontractors. Construction activities may originate from construction contracts, service contracts, purchase orders, or in-house work orders. This Policy is in addition to the requirements of the General Services Agreement or other contract under which the Contractor is performing construction activities.

3. General Safety Requirements

- 1. Contractors will comply with all applicable federal and state regulations and the LG&E and KU Services Company safety rules and programs relevant to the work performed.
- 2. Contractors will ensure that any and all subcontractors working under their purview comply with all applicable federal and state regulations and the LG&E and KU Services Company safety rules and programs relevant to the work performed.
- Contractors are responsible for their employees and any and all subcontractors working for them. Contractors are responsible for ensuring that the subcontractors follow all provisions of this document. Contractors are responsible for providing their employees, and subcontractors with all information provided by LG&E and KU Services Company regarding:
 - * Occupational health and safety;
 - Federal, state and local environmental regulations including LG&E and KU Services Company environmental compliance policies and procedures;

- JBAL Z9-D-02-W

- * Exposure to atmospheric health, serious physical or chemical hazards; and
- * Precautionary measures and procedures for performing the work.
- 4. All Contractors' employees, and any subcontractor employees, shall receive training under the LG&E and KU Services Company Contractor Health and Safety Passport Program.
- 5. The LG&E and KU Services Company Policy prohibits the Contractor's employees, agents or representatives from:
 - Consuming or possessing alcohol while on the LG&E and KU Services Company job sites, including the parking lots;
 - * Reporting to perform work on the LG&E and KU Services Company job sites with unauthorized drugs on his/her person or while under the influence of drugs or alcohol;
 - * Intentionally dumping unauthorized chemicals/materials into a sewer, waterway or on the ground;
 - * Mishandling LG&E and KU Services Company waste;
 - * Allowing employees to perform work that involves operating heavy equipment or working at elevations when using prescribed medication that can cause drowsiness or otherwise impair the employee's ability to perform the work in a safe manner.
- 6. The following conduct is prohibited by the Contractor at and about LG&E and KU Services Company property:
 - * Theft, horseplay, gambling, sabotage or attempted sabotage.
 - * Threatening, intimidating or abusing employees, customers, vendors or guests
 - of LG&E and KU Services Company.
 - * Fighting, creating, or inciting a disturbance.
- 7. LG&E and KU Services Company have a smoke-free policy in all buildings and vehicles.
- 8. Attendance at job site safety meetings is required of the Contractor at the discretion of the LG&E and KU Services Company authorized representative. At least one representative of the Contractor will attend such job safety meetings.
- 9. Any Contractor's employee, who appears sick, extremely tired, or otherwise unable to perform his/her job in a safe manner will be reported to the Contractor's supervision for evaluation and possible removal from the job site.
- 10. Contractors are responsible for establishing control measures to protect their employees, subcontractors or workers under their control, from exposure to hazards (chemical, atmospheric health and physical) present at the job site.

- JBML zg-Dec-11

- 11. The Contractor must provide electrical ground fault protection for employees using construction power (temporary branch circuits to include extension cords) through the use of approved ground fault circuit interrupters (GFCI). Additionally, Contractors must provide ground fault protection when using permanent facility power and using cord and plug equipment in wet or damp locations. Applies to 120-volt single phase 15 and 20-ampere receptacle outlets.
- 12. Contractor employees will work in full pants and shirts appropriate for the task being performed and in compliance with appropriate regulations. Shorts and tank tops are not allowed unless otherwise specified. (Some jobs will require wearing long sleeve shirts.)
- 13. Contractors shall not transport employees in the cargo bed of a truck or trailer.
- 14. All Contractors must receive authorization from the LG&E and KU Services Company authorized representative, before performing work in areas posted as "DANGEROUS OR HAZARDOUS."
- 15. Employees of Resident Contractors, defined as those Contractors with an annual contract and who provide day-to-day services for LG&E and KU Services Company, shall be required to have a negative drug pre-test when hired and before reporting to work at an LG&E and KU Services Company site. They shall also be required by the Contractor to participate in a drug and alcohol testing program that randomly tests 50% of their employees annually, while working on an LG&E and KU Services Company site.
- 16. If a Contractor brings "transient" workers on site for "plant outages", "project work" or "major construction", the transient workers shall be required to have a negative drug pre-test when hired and within 7 days before reporting to work at an LG&E and KU Services Company site. The transient contractors are added to testing pool with 100% annual random testing for the duration of the assignment. If a contractor sends one of their workers to another LG&E and KU Services Company site with no interruption of service, no pre-work drug test is required. The worker remains in the 100% annual random testing pool. If a worker reports to another LG&E and KU Services Company site with an interruption in service of thirty days or more, the worker shall be required to have a negative drug test before reporting to work at that site. The worker remains in the 100% annual random testing pool.

4. Specific Safety Requirements

Contractor Safety Qualification

Contractor selection and ultimate certification shall include an evaluation of the Contractor's prior safety performance, current written safety programs, safety training,

- JB/h 29-1)ec-1

and qualifications of key Health & Safety (H&S) personnel to assure LG&E and KU Services Company that the Contractor is capable of meeting its safety performance goals. Employees of certified Contractors and any subcontractor employees shall undergo "Passport Training" for those designated as Industrial Workers prior to performing work at an LG&E and KU Services Company facility. This by no means will replace regulated compliance training for the work the contractor employee will be performing.

Subcontractor Safety Qualifications

Subject only to the specific exception stated below, any and all subcontractors used by a Contractor to perform work for LG&E and KU Services Company shall meet or exceed the following criteria:

- a) The subcontractor's incident rates for the three (3) most recent calendar years do not exceed, in any one (1) year, the industry average, based on NAISC (or SIC), as published by the Bureau of Labor Statistics;
- b) The subcontractor has not experienced any employee fatality identified within any of the three (3) most recent calendar years' statistics.
- c) The subcontractor has not received any citation, from OSHA, the Kentucky Public Service Commission or any other state agency regulating utilities in the most recent three (3) calendar years; and
- d) The subcontractor has a current Workers Compensation Insurance Experience Modification Rate (EMR) less than or equal to 1.0.

LG&E and KU Services Company may, at the sole option of such company, provided written authorization for the use of a subcontractor not meeting the above criteria; provided that such authorization must specifically identify how the subcontractor fails to meet the criteria and state additional protective measures the Contractor shall put in place in order to use such subcontractor. Such authorization may be withdrawn at any time for any reason.

The criteria stated above are minimum standards and Contractors using subcontractors shall seek out subcontractors with the highest safety performance available.

CONTRACTOR ON-SITE HEALTH AND SAFETY REPRESENTATIVE

The Contractor shall appoint a qualified on-site Health and Safety Representative, accepted by the LG&E and KU Services Company authorized representative, with the

- ABIL 29-Dec-M

authority to enforce all of the safety requirements of this Policy, including implementation of the Contractor's Injury and Illness Prevention Program.

LG&E and KU Services Company authorized representative and H&S will make a riskbased decision as to the qualification level of the Contractor H&S representative. Requirements may range from a full-time on-site safety professional (Certified Safety Professional) to a craft supervisor or "person in charge" with competency as measured by experience training.

Whenever the Contractor has any employees or subcontractors on the job site, the Contractor must have a designated representative on the construction worksite that is knowledgeable of the project's hazards and has full authority to act on behalf of the Contractor. The Contractor's designated representative must make periodic observations of the construction worksite to identify and correct any instances of noncompliance with the project health and safety requirements.

QUALIFICATION EVALUATION

Based on the level of H&S qualification determined necessary by LG&E and KU Services Company, the Contractor shall submit documentation, for review and acceptance by LG&E and KU Services Company in support of the proposed designated representative. Suggested qualifications may include, but are not limited to:

- Professional certifications (CSP, CIH, ASP, etc.).
- Curriculum detailing work experience and EH&S responsibilities on projects of similar scope for the previous five years, at a minimum.
- Evidence of construction safety training such as the 10-hour or 30-hour OSHA training.
- Proof of "Competent Person" (as defined below) or "Qualified Person" (as defined below) status attained by the proposed on-site H&S representative.

Contractor Health and Safety Representative Responsibilities

The Contractor H&S Representative shall:

- Assist in the development of the contractor's safety plan and job site management system.
- Support training of contractor personnel.

11 J-BAL 29-Dec-VI.

- Evaluate the Contractor's safety process continuously.
- Attend any pre-job meetings to discuss their site-specific safety plan.
- Conduct and formally document job briefings.
- Assist in the identification of jobs requiring a hazard analysis.
- Assist in evaluating potential subcontractors in accordance with this Policy.

COMPETENT PERSON

Each Contractor shall provide to LG&E and KU Services Company a written list of those persons designated as a Competent Person, who shall be available at the work site and capable of identifying existing and predictable hazards in the surroundings or working conditions which are unsanitary, hazardous, or dangerous to workers, and who has authorization to take prompt corrective measures to eliminate them. Persons shall be responsible for conducting periodic observations of the job sites, materials and equipment, and shall maintain the accident prevention program. Contractor shall ensure that each Competent Person listed has been trained in the following areas as applicable:

- Asbestos
- Cranes
- Confined Space
- Demolition
- Excavations
- Fall Protection
- Industrial Trucks
- Ladders
- Scaffold
- Steel Erection
- Tower Climbing

5. HEALTH AND SAFETY MANAGEMENT PLAN

Prior to commencement of contract work, the Contractor shall develop and submit to the LG&E and KU Services Company authorized representative a written Health & Safety (H&S) Management Plan on how the contract work will be completed without endangering the health and safety of those performing the work or anyone else working in the general area. The H&S Management Plan will be developed for the following higher risk contracts, including projects:



- All construction projects (new site and refurbishment)
- ✤ Contracts with an estimated value of \$250,000 and over
- Long term contracts (12 months and over)
- * Contracts for which the Contractor will use subcontractors.
- * Contracts that provide a service by performing high risk* activities.
- * Any other contracts at the discretion of the contract manager.

*High risk activities include but are not limited to:

- * Electrical work requiring an Electrical Work Permit
- * Asbestos removal
- * Cooling tower maintenance
- * Demolition
- * Hot work in hazardous area
- Permit Required Confined Spaces
- * Scaffolding
- * Tank cleaning or testing
- Welding in hazardous areas
- * Working at heights
- Work on telecommunications towers
- * Work involving excavations to a depth of more than 4 feet
- Work involving the use of explosives
- Work on or near pressurized gas pipes
- * Work over or adjacent to water
- ✤ Work involving diving

The Health and Safety Management Plan shall contain at a minimum:

- The name of the On-site Health and Safety representative who is responsible for the implementation of their safety plan.
- LG&E and KU Services Company policy on environment, safety and health.
- LG&E and KU Services Company policy on substance abuse and testing policies if applicable.
- * How and when each Contractor will conduct their job briefings.
- Provisions for conducting and documenting weekly job site safety audit/inspections by manager/supervisor level personnel.
- Training methods used to meet OSHA training requirements, and to ensure that safety program requirements are communicated to all Contractor personnel.

- * Incident reporting, first aid, and emergency procedures.
- List of all Competent Persons overseeing those tasks in which OSHA requires such person(s), such as excavation, asbestos abatement and scaffolding.

Subcontractors shall be held to the same level of performance as the Contractor's written H&S Management Plan. The Contractor shall submit written documentation for its subcontractors that demonstrates how their subcontractors shall meet compliance with the site safety plan.

6. HAZARD ANALYSIS

Contractor shall complete a "<u>Contractor Safety Management / Project Specific Hazard</u> <u>Analysis</u>" * and a "<u>Contractor Hazard Mitigation Plan</u>"*. These documents shall be submitted to the LG&E and KU Services Company authorized representative prior to the initiation of any work. In addition, a "<u>Quality Assurance Closure Form for Contractors</u>"* shall be completed and submitted to the LG&E and KU Services Company authorized representative at the completion of the project.

* The Hazard Analysis, Mitigation Plan and Closure Form is presented during the Contractor Passport Train-the-Trainer session.

All Contractor and subcontractor personnel scheduled to work in the activities identified, shall receive safety training in those activities prior to working on them. (A safety toolbox meeting would be an acceptable forum to meet this requirement). The Contractor shall maintain proof of employee training, and shall make available such proof upon request. Note: This by no means shall replace their regulatory compliance training.

HAZARD ANALYSIS REQUIREMENTS

A hazard analysis shall be written based on the following conditions:

- All major outage work
- Special Projects
- Jobs with the highest injury or illness rates
- Jobs with the potential to cause severe or disabling injuries or illness, even if there is no history of previous accidents
- Jobs complex enough to require written instructions
- At the discretion of the LG&E and KU Services Company authorized representative

7. Engineered Protective Systems

14 JAML 24-Decill

The Contractor shall submit for review to the LG&E and KU Services Company authorized representative such safety system that is required by regulation to be designed by a registered professional engineer. This review is solely to verify that the Contractor has had the required protective systems prepared and stamped by a registered professional engineer.

LG&E and KU Services Company review of any documents showing the design or construction of protective systems for worker and property protections shall not relieve the Contractor of its obligations to comply with applicable laws and standards for the design and construction of such protective work. Contractor shall indemnify and hold harmless LG&E and KU Services Company and their engineering personnel from any and all claims, liability, costs, actions and causes of action arising out of or related to the failure of such protective systems. The Contractor shall defend LG&E and KU Services Company, its officers, employees and agents including without limitation engineer personnel, in any litigation or proceeding brought with respect to the failure of such protective systems.

The cost of required safety engineering services required for safety and protective systems shall be borne solely by the Contractor and shall be deemed to have been included in the amount bid for the work as stated in the contract.

8. Safety Training and Education

Contractor shall ensure that its workforce is compliant trained and qualified to perform the work. Contractor shall ensure that all subcontractor employees demonstrate the same level of competence.

Site Orientation

All Contractors / subcontractors shall undergo an LG&E and KU Services Company "site specific" training/orientation prior to engaging in work activities at a generating station. In addition, Contractors that conduct work at LG&E and KU Services Company generation facilities that process ammonia shall also undergo an ammonia awareness training/orientation prior to conducting work.

Contractor employees conducting work in a substation must first complete a Substation Entry training program.

Contractors Pre-job Orientation

Contractor shall require and administer a pre-job orientation to its employees and all subcontractor employees prior to engaging in work activities. Contractor shall maintain on the work site a detailed outline of the orientation and a signed and

15 ABIL 29-Dec-N

dated roster of all employees who have completed the orientation. The orientation shall address the following elements at a minimum:

- Employee rights and responsibilities
- Authority and responsibility to issue Stop Work Order
- Alcohol and drug abuse policy
- Contractor's disciplinary procedures
- First aid and medical facilities
- Hazard recognition and procedures for reporting or correcting unsafe conditions or practices
- Procedures for reporting accidents and incidents
- Hazard Communication Program
- Access to employee exposure monitoring data and medical records
- Protection of the environment, including air, water, and storm drains from construction pollutants
- Location of and access to reviewed Health & Safety Management Plan, Project Specific Hazard Analysis, and Hazard Mitigation Plan.
- Location and contents of required postings

Daily Job Briefings

Contractors shall ensure that all of their personnel (employees and subcontractors) on the job site receive the daily Job Briefing before they start each job. Job Briefings shall discuss, at a minimum, the hazards associated with the job; work procedures involved; special precautions; energy source controls; and personal protection equipment requirements. This job briefing shall be conducted by the contractor's person in charge. Should the scope of the work change, than another job briefing shall be conducted.

9. Emergency Procedures

An emergency is any situation that poses an immediate threat to life or property. Each Contractor shall maintain one person currently qualified in CPR and First Aid on site at all times. Refer to the site orientation, or the LG&E and KU Services Company authorized representative for specific information for handling of a life threatening or other serious injury, fire, etc. Following the occurrence of an emergency, the contractor shall ensure that all proper incident reports are completed and distributed, and that the LG&E and KU Services Company authorized representative is notified immediately.

Incident Reporting

In the event a job site accident occurs, the Contractor shall immediately implement controls and restrictions on the accident site to ensure the site remains undisturbed until released by the LG&E and KU Services Company authorized representative. All accidents shall be reported to the LG&E and KU Services Company authorized representative immediately after the site is

-16 ABIL 24-DecrM


secured. A written incident report shall be furnished within the same day of the incident. A job site accident would include, but not be limited to a fire, explosion, equipment failure, release or exposure to toxic liquids, fumes or vapors, etc.

Near Miss / Injury-free Event

It is the responsibility of the Contractor, to complete all near miss investigations, and to report these occurrences with recommendations / implementation of corrective actions. The report is to be submitted to the LG&E and KU Services Company authorized representative within 24 hours.

Medical Treatment Event

The Contractor shall report all accidents (either occupational injury or illness) requiring medical treatment, as soon as possible, but no later then the end of the work shift, to the LG&E and KU Services Company authorized representative along with a copy of the first report of the injury. Serious injuries (defined as an injury that would require off site medical attention) shall be reported within 15 minutes, even during off shifts. (Review project specific emergency notification procedures.)

Fatality

It is the responsibility of the Contractor to immediately notify LG&E and KU Services Company should a fatality occur. It is the responsibility of the Contractor to notify the Kentucky Occupational Safety & Health, Division of Compliance within the appropriate Kentucky notification periods.

Stop Work Order

A stop work order must be given when imminent danger is identified or where significant damage to equipment or property or environmental degradation could occur if the operation continued. Any employee of a Contractor that observes an imminent-danger situation is responsible for stopping the work and reporting it to their supervisor. When a stop work order is issued, only those areas of a construction project immediately involved in the identified hazardous situation are to be included in the order.

Immediately after stopping work, the person issuing the order, or their supervisor, must report to the LG&E and KU Services Company authorized representative of their action. Work shall not resume until the LG&E and KU Services Company authorized representative has agreed that the imminent danger has been eliminated.

10. Hazard Specific Requirements

The Contractor will ensure that their employees (and all subcontractor employees) are properly equipped and trained to comply with the LG&E and KU Services Company standards and federal and state regulations; including but not limited to the following:

TI ABIL

Asbestos

Blasting and the use of explosives Chemical Safety/Hazard Communication Commercial Diving Operations Confined Space Entry Control of Energy Sources (Lockout/Tagout) Crane Operations, including rigging Electrical Fall Management (personal fall arrest systems, scaffolding, walking & work surfaces, ladders and floor & wall openings) Hazardous Waste and Chemical Spills Hot Work Personal Protective Equipment (PPE) Powered Industrial Trucks

Trenching

11. Enforcement

The Contractor is responsible for the health and safety of its employees and any subcontractor employees under their control. Enforcement of this Policy, as well as other recognized safety requirements, is the responsibility of the Contractor. The evaluation does not constitute acceptance of the Contractor's safety programs or work practices nor, in any way relieve a Contractor of full responsibility for meeting all appropriate OSHA regulations to ensure the safety of its employees.

Whenever there is a jurisdictional question of which standard will apply, the most stringent safety practice will take precedence. The Contractor must document exceptions and attach them to this form. Contractors and their employees who do not follow this Policy are subject to removal from the worksite as well as being banned from future LG&E and KU Services Company projects/contracts.

LG&E and KU Services Company reserve the right to evaluate the safety of Contractor's work practices to determine if they meet LG&E and KU Services Company standards and state/ federal regulations. In addition to the audit rights under the applicable contract LG&E and KU Services Company reserve the right to audit any and all documents (job briefings, audits, etc.) at anytime during the course of the work.

- ABML zg-Dec-M

12. LG&E and KU Services Company Safety and Health Issues

Contract work may involve use, handling, storage, or work in vicinity of *hazardous chemicals or materials*. (Concerns are Hazard Communication...spill prevention/response).

Contractor may perform work (operation, maintenance or emergency response function) as necessary.

Contractor may perform hot work (e.g. welding, torch cutting, brazing, etc.)

Contract may require Contractor to work in or near confined spaces.

Contract work may require using/working under clearance procedures for the control of hazardous energy (*lockout/tagout*).

Contract may involve work on an uncontrolled hazardous substance site, Superfund site, or other contaminated site that could trigger Hazardous Waste Operations and Emergency Response (*HAZWOPER*) planning and training requirements. (Ref: CERCLIS List)

Contract work may involve application, handling or disturbance of *lead*, *cadmium* and/or *zinc chromate containing materials*. An example would be the removal of *toxic surface coatings* (i.e. paint).

Contract work may involve handling, disturbance, abatement or work around asbestos containing materials (ACM).

Contract work may involve application of pesticides, herbicides, etc.

13. Hazardous Chemical Communication

The following is a list of Hazardous Chemicals and atmospheric contamination that may be encountered at LG&E and KU Services Company sites. It should in no way be deemed as the only contamination that could be encountered at LG&E and KU Services Company sites. Always be aware of the contamination that could be encountered and become familiar with their Material Safety Data Sheets.

-19 JBML 29-Dec-M

.

Chemical Name	<u>Formula</u>	Trade Name	<u>Description/ Target</u> <u>Organ</u>
Anhydrous Ammonia	NH3 (99-10))%)	Liquid colorless gas or compressed liquid with extremely pungent odor. Targets eyes, skin and respiratory system.
Arsenic	AS	Organic Arsenic	Targets skin, kidneys, liver and resp. system.
Asbestos		Hydrated Mineral	Fibers found in
			insulation, gaskets, packing, vinyl asbestos flooring, roofing, and other materials. Targets respiratory system. Can cause lung cancer.
Carbon Dioxide	CO2	Carbonic Acid Gas	Targets respiratory system and
		Dry ice	cardiovascular system
Carbon Monoxide	CO	Flue gas/Monoxide gas.	e Colorless, odorless Targets lungs, blood, can be immediately fatal.
Chromium Hexavalent	Cr(VI)	Hexavalent Chromit	
		respiratory tract, ski	n and eyes. Irritant.
Hydrogen Sulfide	H ₂ S	Sewer gas Hydrosulfuric Acic	Colorless gas with strong rotten egg odor, quick loss of sense of smell, can

20

_________ 29-Dec-11

be immediately fatal.

Hydrogen	H ₂	Liquid Gas	Colorless, odorless, targets eyes, skin respiratory system
Lead	Pb	Lead metal	Heavy soft gray metal. Targets eyes, kidneys and blood.
Ozone	O ₃	Triatomic Oxygen	Colorless, targets eyes and respiratory sys.
Sulfur Dioxide	SO ₂ respiratory	Sulfuric Acid sys.	Targets eyes, skin,

and

_AB//h za-Dec-11

Exh_H_Health Safety and Passport 12/30/11 an

21

14. Definitions

Competent Person: means one who is capable of identifying existing and predictable hazards in the surroundings or working conditions which are unsanitary, hazardous, or dangerous to workers, and who has authorization to take prompt corrective measures to eliminate them.

Qualified Person: is one who, by possession of a recognized degree, certificate, or professional standing, or who by extensive knowledge, training, and experience, has successfully demonstrated their ability to solve or resolve problems relating to subject matter, the work, or the project.

I have read the LG&E and KU Services Company Contractor Safety Policy as outlined above and I understand and agree to abide by the requirements set forth therein; and confirm this by signing below.

CONTRACTOR SENIOR MANAGER:_____

Тітье:

DATE:_____

12130/11

ar.

3.0 CONTRACTOR/SUBCONTRACTOR SAFETY AND HEALTH

QUESTIONNAIRE AND CHECKLIST

THIS QUESTIONNAIRE IS REQUIRED FOR ALL CONTRACTORS AND SUBCONTRACTORS PRIOR TO STARTING WORK

The Company is committed to providing a safe and healthy workplace for employees and Contractors/Subcontractors. To qualify to perform work, the Contractor/Subcontractor shall provide the following information and agree to obtain the following information from all subcontractors utilized.

Contracto	or/Subcon	tract	or Nam	e:				Date:						
Contracte	ed Activity	(ple	ase de	scribe):										
Contrac	tor/Subcor	ntrac	tor Rep	presentative:										
Please	provide	а	brief	description	of	the	work	activities	and	location(s)	undertaken	by	your	company.

The following information must be from the facilities providing labor. We are not interested in overall statistics at a national or international level. Describe the area to which this questionnaire applies._____

In the table below, provide the three most recent full years of history for the area or region this questionnaire applies. In addition, attach copies of applicable OSHA 300 Logs (showing the actual injuries, etc. – not the summaries) and verification of your EMR/discount rate information.

ITE	DESCRIPTION	20	20	20	
M A	Interstate Experience Modification Rate (EMR)				

12/30/11	Exh_H_Health Safety and Passport
her-	

______B/L 29-0cc-11 Ghent CCR Transport Exhibit H

B Recordable Incident Rate (RIR)		a with a solution lobor please		
C Lost Time Incident Case Rate (LTICR)(only incidents that resulted in days away from work) — …		Using the OSHA 300 Logs from the facilities providing labor, please document the following:		
D Lost Workday injury and Illness Case Rate (LWDCR)(includes days away from work, job transfers and job restrictions) — … <	в	Recordable Incident Rate (RIR)	 	
E Number of Injuries and Illnesses (Total Line Entries of 300 Log)	с	Lost Time Incident Case Rate (LTICR)(only incidents that resulted in days away from work)	 	
E Number of Injuries and Illnesses (Total Line Entries of 300 Log)	D	Lost Workday Injury and Illness Case Rate (LWDCR)(includes days away from work, job transfers and job restrictions)	 	
G Number of Job Transfer or Restriction Cases (Column 1 of 300 Log)	E		 	
H Number of Injury Related Fatalities (Column G of 300 Log)	F		 	—
I Employee Hours Worked/Year (If unknown use # of employees x 2080)	G		 	
(If unknown use # of employees x 2080)	Н	Number of Injury Related Fatalities (Column G of 300 Log)	 	
(If unknown use # of employees x 2080)			 	
(If unknown use # of employees x 2080)			 -	
(If unknown use # of employees x 2080)			 	
	I	Employee Hours Worked/Year (If unknown use # of employees x 2080)		
J Total Number of Employees	J	Total Number of Employees		

- JB1/L Z9-Dec-11

К	NAISC or Standard Ind	ustrial Classification (SIC)				
L	(B) Rate = E x 200,000 ÷ Hours	(C) Rate = F x 200,000 ÷ Hours	(D) Rate =	: (F + G) :	x 200,00	0 ÷ Hours

	Question	Y/N	Comments
1.	Does your company have a written safety and health program? Please attach a copy with this submission.		
2.	Does your company have a written Hazard Communication Program?		
3.	 Does your company have a written environmental compliance assurance program? Does your company have a written DOT Operator Qualification Plan? Please attach a copy with this submission for review. Note: Plan must meet or exceed LG&E and KU Services Company Gas Distribution Operator Qualification Plan. 		

12(30/11 Er

26

4.	Does your company use subcontractors? (This Questionnaire is required for all Subcontractors)		
	If you do use sub-contractors, do you qualify subcontractors based on their ability to address safety, health, and environmental requirements?		
	Do you verify that subcontractors meet regulatory requirements?		
	Does your subcontractor have a DOT Operator Qualification Plan or are they qualified under your plan. If they have their own plan then please submit a copy for review		
5.	Are all documents, pertaining to this questionnaire, available for auditing? If no, please explain		
6.	Who in your company is responsible for coordinating your safety and health program?		
	Name/Job		
	Title:		
	Phone # ()		
	Is safety and health a full time responsibility for this position?	 	
7.	Has your company received any citations from a regulatory agency during the last three years?	-	
	If yes, describe citation(s)		

_____JB1/4 29-Dec-11

8.	Does your company perform safety audits/review?
9.	If yes, are safety audits documented? Who reviews the safety audit/review and how often?
	Job
	Title:

- AB1/L 24-Dec-11

0.	Does vour cor	npany provide/require the follow	wing?		
	Eye Protection 1910.133)	(ANSI-Z41.1)(
	Fall Protection 1910.66)	(ANSI-Z41.1)(29 CFR 192	6.501 or		
	Foot Protection	(ANSI-Z41.1)(29 CFR 1910).136)		
	Hand Protection 1910.138)	(ANSI-Z41.1)(29	CFR		
	Hard Hats	(ANSI-Z89.1)(29 CFR 1910).135		
	Hearing Protection 1910.95)	(ANSI-Z41.1)(29	CFR		
	Respiratory Prot	ection (ANSI-Z41.1)(29	CFR		
	1910.134	•			
		latony required Personal Protoc	stiv o		
11.	Equipment, what	latory required Personal Protect other PPE is required or suppli	ed?		

Exh_H_Health Safety and Passport

28

12/30/11 m

12.	Describe how you will meet the requirements for first aid and medical provision under this contract.	
13.	Does your company have scheduled, documented employee safety meetings?	
	If yes, how often?	
14.	Who conducts the safety meetings?	
	Job Titles:	
15.	What managers/supervisors participate in the safety meetings?	
	Job Titles:	
16.	Are meetings reviewed and critiqued by managers/supervisors?	

_____AB/L_____ zy-Doc-11

12/30/11 En

17.	Does your company hold on-site (tailgate/toolbox) safety meetings?		
	If yes, how often?		
	Who conducts these safety meetings?		
	Job Titles:		
	Is documentation available?		
18.	Does your company have a written policy regarding drug screening or testing of your employees?		
	If Yes, please provide a copy of your plan to The Company representative.		
1.9.	Does your drug-testing program conform to DOT requirements?		
	Comments:		
	If yes, which set of DOT regulations does your drug-testing program designed to satisfy?		
	Pipeline and Hazardous Material Safety Administration PHMSA		
	Federal Motor Carrier Safety Administration FMCSA		

- JB/L 29-0ec-11

20.	Does your company have policy requiring written accident/incident reports (spills, injuries, property damage, etc.)?	
21.	Does your company conduct accident/incident investigating?	
	If yes, please attach a brief outline of procedures	
22.	Does your company document, investigate and discuss near miss accidents?	
	If yes, is documentation available?	
23.	Are accident/incident reports reviewed by managers/supervisors?	
24.	Indicate the circumstances in which your company's employees may be subject to drug screening. Employment Random Probable Cause Post Accident Periodic Other	·

--- JBML 29-Dec-11

32

Ghent CCR Transport Exhibit H_____

or NA." (Estimated Percentage of Employees should reflect the percentage of employees providing labor who have received training).

ease respond to all items with "Yes, No, o		PROGRAM	EST.%	FREQUENCY
PROGRAMS/TRAINING	REFERENCE SOURCE	DOCUMENTED AND WRITTEN Y/ N/ NA		OF TRAINING FOR INDIVIDUAL EMPLOYEES
Asbestos Class IV	OSHA 29 CFR 1926.1101			
(Awareness) Asbestos Class III	OSHA 29 CFR 1926.1101			
Asbestos Class I and II	OSHA 29 CFR 1926.1101			
Confined Space Entry	OSHA 29 CFR 1910.146(g)			
Cranes	OSHA 29 CFR 1926 Subpart CC			
Natural Gas Operations	DOT 49 CFR 192, Subpart N			
DOT HM-126\f Hazmat Employee	DOT 49 CFR 172.704	4		
Generation, Transmission, and Distribution Standard	OSHA 29 CFR 1910.269			
Electrical Safety	1910.332	FR		
Emergency Evacuation	OSHA 29 CFR 1910.38(a)			

- JB/h zq-Da-11

Exh_H_Health Safety and Passport

-

12/30/11 ac

				<u> </u>	1	
Excavations	OSHA 1926.651	29	CFR			
Fall Protection	OSHA 29 CI	FR 192	26.500	 	 	
First Aid/CPR	OSHA 1910.	29 CF 151(b)		 	
Forklifts	OSHA 1910.178(l)	29	CFR	 		
Hazard Communications	OSHA 11910			 		
Hazwoper - Awareness Level	OSHA 1910.120	29	CFR	 		
Hazwoper 8 Hour	OSHA 1910.120	29	CFR			
Hazwoper 24 Hour	OSHA 1910.120	29	CFR		 	
Hazwoper 40 Hour	OSHA 1910.120	29	CFR			
Hazwoper Supervisor 8 Hour	OSHA 1910.120	29	CFR			
Hearing Conservation	OSHA 1910.95	29	CFR			
Incipient Fire Fighting		A 29 (0.157				
Lead Worker		4 29 C 26.62(
Lead Supervisor	See	e Abov	ve			

- JB/L Z9-Dec-11

Lockout/Tag out Authorized Person	OSHA 29 CFR 1910.147(c)(7)		
Lockout/Tag out Affected Person	See Above		
New Employee Orientation	OSHA 29 CFR 1910.119(g)		
Personal Protective Equipment	OSHA 29 CFR 1910.132(f)		
Process Safety Management	OSHA 29 CFR 1910.119		
Respiratory Protection	OSHA 29 CFR 1910.134		
Scaffolding	OSHA 29 CFR 1926.454		
Substance Abuse	DOT 46 CFR 16.401 & 391.119		

Signature _____

Title_____

Date: _____

EXHIBIT I

INSURANCE

INSURANCE OBTAINED BY CONTRACTOR

A. Contractor shall procure, pay premiums, and provide its own insurance coverage during the performance of the Work. At a minimum Owner requires Contractor to obtain the following coverages:

Contractor shall at all times during the period in which the Agreement is in force, provide and maintain insurance of the type and with limits as stated in this Exhibit I.

Contractor is responsible for maintaining, at its own cost, any other insurance it deems necessary.

The insurance specified may be provided in a policy or policies, primary and excess.

Contractor shall (i) provide and maintain the following insurance and (ii) cause each such insurance policy, except with regard to Workers' Compensation and Professional Liability, to name Owner and Owner's Engineer as additional insured and waive rights of subrogation against Owner, and (iii) submit evidence of such coverage (including evidence of Owner as additional insured and waiver of subrogation) to Owner prior to performing any Work on the Ghent Generating Station Site.

(a) Workers' Compensation and Employer's Liability Policy, which shall include:

JBlh na Dre-V

12/30/11 al

Ghent CCR Transport Exhibit 1



12/30/11 ac

21/B/14 21/24-Dec=11



3 /B/h 24-Dec-11

Exhibit_1_Insurance 122311.docx

12/30/11 The

	이 같은 것 같은
	그는 그는 것 같은 것 같
·	· · · · · · · · · · · · · · · · · · ·
· · · ·	- 2011년 1월 19일 - 11일 - 11일 - 11일 - 11 - 11일 - 1
	에는 것은
	(1996년) 1997年(1998년) 1997년 1977년 - 1978년 1977년 1977년 1977년 1977년 1977년 1977년 1977년 1977년 1977년 1977년 1987년 1977년 1978년 1978년 1978년 1978년 1977년 197
	그는 그는 그는 것이 아이지 않는 것이 같은 것이 같은 것을 방법을 받았다.
	- 2011년 1월 19일 - 19일 - 19일 - 1 - 19일 - 19g - 19g - 19g - 1
	- 2011년 1월 1997년 1월 1 1997년 1월 1997년 1월 199 1997년 1월 1997년 1월 19
	이 같은 것은 것을 받았다. 것은 것은 것을 것을 수 있는 것을 것 같은 것은
•	

4 J.B.M. 29-Da-11

Exhibit_I_Insurance 122311.docx



1



5/B/L-5/24-Dec-11

Exhibit_1_Insurance 122311 docx

12/30/11 M

Ghent CCR Transport Exhibit 1 Kentucky Utilities



6 JBM 24-Dec-11

Exhibit_1_Insurance 122311.docx



Thursday, December 08, 2011

JBM-Za-Daa-11

12/30/11 64

EXHIBIT M MEETINGS AND PROGRESS REPORTS

1.0 Meetings and Progress Reports Details

1.01 <u>SUMMARY</u>:

- A. This section outlines the meeting, scheduling and reporting requirements for the Agreement. The Contractor will provide support for these activities as applicable including attending project meetings.
- B. Project Meetings:
 - 1. Preliminary Conference.
 - 2. Engineering Coordination Meetings.
 - 3. Weekly Model meeting
 - 4. Onsite Weekly Progress meeting
- C. Schedules and Reports:
 - 1. Initial Coordination Submittals.
 - 2. Work Progress Schedule.
 - 3. Work Progress Reports.
 - 4. Delivery Schedule.
- D. Related Work Specified Elsewhere:
 - 1. Submittals: In accordance with Exhibit X
 - 2. Equipment and Materials

1.02 <u>PROJECT MEETINGS</u>:

- A. Preliminary Conference:
 - 1. Contractor will conduct a meeting within 20 days after the Effective Date of Agreement, to review items stated in the agenda and to establish a working understanding between the parties as to their relationships during performance of the Work. The conference shall be attended by:
 - a. Contractor.
 - b. Representatives of principal Subcontractors and Suppliers.
 - c. Owner's representative(s).
 - 2. Meeting Agenda:
 - a. Projected fabrication/construction schedules.
 - b. Project coordination.
 - c. Procedures and processing of:
 - (1) Substitutions.
 - (2) Submittals.
 - (3) Change Orders.
 - (4) Applications for Payment.
 - d. Procedures for testing.
 - 3. Location of Meeting: Owner's office or by teleconference.
 - 4. Reporting: Contractor will prepare and distribute minutes of the meeting to each party represented.
- B. Engineering Coordination Meetings:
 - 1. Contactor will schedule and conduct a meeting at least monthly for coordination during Contractor's equipment engineering and design phase of the Work. Meetings shall be attended by:
 - a. Contractor representative(s) including engineering personnel.

TO-Dec-11

- b. Representatives of principal Subcontractors and Suppliers.
- c. Owner's representative(s).
- 2. Meeting Agenda:
 - a. Review of action items.
 - b. Facility design interfaces.
 - c. Equipment and Material procurement status.
 - d. Engineering/fabrication/manufacturing schedules.
 - e. Requests for information (RFIs).
- 3. Location of Meetings: Project Site or teleconference.
- 4. Reporting: Contractor will prepare and distribute minutes of the meetings to each party represented.
- 1.03 SCHEDULES AND REPORTS:
 - A. Initial Coordination Submittals: Within the time period(s) defined in accordance with Exhibit X, Contractor shall submit to Contractor and Owner for review and acceptance:
 - 1. A preliminary Work progress schedule.
 - 2. A preliminary schedule of Submittals, as stated in accordance with Exhibit X.
 - 3. Certification of insurance.
 - B. Work Progress Schedule:
 - 1. After submittal of preliminary Work progress schedule, submit to Contractor and Owner a detailed Work progress schedule within the time period(s) defined in accordance with Exhibit X. Base the schedule on the preliminary Work progress schedule and incorporate review comments and other feedback.
 - 2. The schedule shall show the Work in a graphic format suitable for displaying scheduled and actual progress.
 - a. Prepare schedules as a horizontal bar chart with separate bar for each major portion of the Work or operation.
 - b. The schedule shall also show the Work broken down into major phases and key items with the dates Work is expected to begin and be completed. Sequence of listings shall be in the chronological order of the start of each item of Work.
 - c. Scale and spacing shall allow space for notations and revisions.
 - d. Sheet size: Minimum 11 x 17.
 - 3. Provide sub-schedules to define critical portions of entire schedules.
 - 4. Coordinate Work progress schedule with Work progress reports and delivery schedule.
 - 5. Contractor will review and comment on Work progress schedule:
 - a. Contractor shall print and distribute copies of the accepted schedule to Owner, Suppliers, and other parties required to comply with scheduled dates.
 - 6. Contractor shall not change the accepted milestone dates in the Work progress schedule without prior written approval of Owner.
 - 7. Submit to Owner an updated schedule at least once monthly. Schedule shall show actual progress and any proposed changes in the schedule of remaining Work.
 - C. Work Progress Reports:
 - 1. Submit monthly a report on actual Work progress. More frequent reports may be required should the Work fall behind the accepted schedule.
 - 2. Work progress reports shall consist of marked copies of prints made from the accepted Work progress schedule, and a narrative report which shall include but not be limited to the following:
 - a. A description of current and anticipated delaying factors, if any.
 - b. Impact of possible delaying factors.
 - c. Proposed corrective actions.



- 3. Work reported complete but not readily apparent to Owner must be substantiated with supporting data.
- 4. Should operations fall behind accepted schedule to an extent that completion of Work within the Contract Time appears doubtful, Contractor shall, at no change in Contract Price, take corrective action to get back on schedule.
- D. Delivery Schedule:
 - 1. Within 120 days after the Effective Date of this Agreement, Owner and Contractor shall agree on a delivery schedule for all Equipment and Materials to be furnished for which the delivery time has not been previously agreed or specified.
 - 2. Contractor shall notify Owner at least two Days in advance of any equipment deliveries.

2.0 Meeting Notes Template

Ghent CCR Transport

Document No.: Date Issued:

Meeting Notes

 Meeting
 Telecom
 Conference Report

 Distribution:
 Date, Time & Place:

-3 AB/L 30-Dec-11

- 4 BIL 30-Dee-11

Subject:	Participants:
Originated By:	Recorded By:
Action Item	Assigned to/Status

3.0 **Progress Report Table of Contents**

1.0 EXECUTIVE SUMMARY

- 2.0 DESCRIPTION OF SERVICES
 - A. ENGINEERING ACTIVITIES
 - 1. Architectural
 - 2. Civil
 - 3. Control Systems
 - 4. Electrical
 - 5. Mechanical
 - 6. Plant Design
 - B. PROCUREMENT
 - C. SAFETY & HEALTH
 - D. CONSTRUCTION
 - E. COMMUNITY RELATIONS
 - F. LABOR RELATIONS
 - G. STARTUP
- 3.0 ENVIRONMENTAL
- 4.0 SCHEDULE
- 5.0 ISSUES AND CONCERNS
- 6.0 APPENDICES
 - A. Payment Schedule
 - B. Scope Change Register
 - C. Critical Schedule Activities
 - D. Project Schedule Milestone Dates

- E. Percent Complete Curves:
 - > Total Project
 - > Engineering
 - > Construction
 - ➤ Startup
- F. Bulk Commodity Curves:
 - > Concrete
 - ➢ Structural Steel
 - > Above Ground large Bore Pipe
 - > Above Ground Small Bore Pipe
 - > Cable Tray
 - > Above Ground Conduit
 - > Cable
 - > Terminations
- G. System Turnover Curves:
 - > Construction to Startup
 - ➢ Startup to Owner
- H. Staffing Curve
- 1. Purchase Order and Sub-Contract Award Status
- J. Owner Inspection Summary
- K. Supplier Quality Shop Visits
- L. Project Schedule
- M. Progress Photos



EXHIBIT O

OWNER PROVIDED ITEMS

The following items will be provided by Owner to support the Work and, as applicable, in accordance with the interface requirements (e.g. – location, pressures, and flow rates) as identified in Exhibit A.

The personnel, services, information and other items specified in the Project Schedule and/or Technical Specifications (Exhibit A), (e.g., including utilities available, O&M staff available, inter-tie for energization available, requirements for electrical tie-ins, etc.) will be provided no later than the dates stated in the Project Schedule and/or Exhibit A (or, if Contractor is behind schedule, on such later date as may be reasonably required by Contractor for the performance of the Work). The furnishing of personnel, services, information and other items not specified in the Project Schedule shall be at such time and manner as may be reasonably required by Contractor for the performance of the Work to support the Project Schedule.

Exhibit P sets forth the responsibilities of the Parties with respect to the Permits required for the Work, including Permits relating to matters addressed in this Exhibit O.

As used below, "reasonable quantities," means the aggregate quantities that should reasonably be anticipated to be required for the construction and commissioning of the Ghent CCR Transport System.

- 1. Electrical Power
 - a. Owner will provide reasonable quantities of electrical power supply as requested by Contractor for Contractor's use during construction at locations designated by Owner. Power at the head end of the pipe conveyor and along the pipe conveyor route is the responsibility of the contractor.
 - b. Owner will provide back-feed electrical power as necessary for the Work, in reasonable quantities requested by Contractor, no later than February 11, 2013.
 - c. Owner will complete any upgrades indicated as being required by the ETAP study (i.e., the study to be performed by Contractor to determine the degree, if any, of upgrades necessary on the Owner's side of the electrical supply tie-in points in order to supply the load required by the Ghent CCR Transport System) by the later of January 14, 2013 or 293 days after Contractor delivers the ETAP study to Owner.
- 2. Water
 - a. Owner will provide reasonable quantities of service water to support construction in quantities requested by Contractor at quality and pressure as described in Attachment A, available at the point(s) agreed by the Parties.

78/1-29-Dec-11

- b. Owner will provide reasonable quantities of service water in quantities requested by Contractor at quality and pressure as described in Attachment A to support start-up, commissioning, testing and operation of the Ghent CCR Transport System.
- c. Owner will provide reasonable quantities of potable water for construction purposes and purposes specified in Exhibit A in quantities requested by Contractor at quality and pressure described in Attachment A.

3. Waste Disposal

- a. Owner will provide a suitable on-site location for disposal of excess excavated material and demolished paving (asphalt & concrete) at an agreed location within 1 mile of the Job Site. A temporary spoil area on the Ghent Generating Station Site will be agreed upon between the Parties for material that is to be reused as fill.
- b. Wastewater discharge facilities available at the points agreed by the Parties.
- 4. Owner O&M Personnel

Owner will supply to Contractor a utility norm of personnel (working Ghent's operations normally scheduled shifts) for the start-up and commissioning, testing, checkout, training, and operation and maintenance of the Ghent CCR Transport System.

5. Owner Security Personnel

Owner will provide a security person at the main gate of the Ghent Generating Station Site. Owner is not responsible for patrols or Contractor access gates.

6. Pipe Conveyor Truck Loading Area Access

Owner will provide reasonable access to the pipe conveyor truck loading area at the Ghent Generating Station for Contractor and its Subcontractors in connection with the performance of the Work.

EXHIBIT P

PERMITS

Contractor shall assist Owner in obtaining and maintaining the necessary permits required in the name of Owner and shall obtain and maintain all other permits and licenses as required by the Agreement. Critical permits required for the development, design, engineering, financing, procurement, construction and operation of the Project, include but are not limited to those detailed below.

Owner's Responsibility:



Contractor' Responsibility:

Any statutory or non-statutory clearance/permits required to perform the Work, including but not limited to:



- JB//-29-Des-11

Exh_P_Permits

Project : Ghent CCR Transport Project Location : Ghent, Kentucky, KY		Current as of : Expiration Date :	Jan 1, 2011 Dec 31, 2011		Current as of : Expiration Date :	Jan 1, 2012 Dec 31, 2012		urrent as of : piration Date:	Jan 1, 2013 Dec 31, 2013	
Classification	Base <u>Rate/Hr</u>	Overtime <u>Rate/Hr</u>	Subsistence <u>Rate/Hr</u>	Base <u>Rate/Hr</u>	Overtime <u>Rate/Hr</u>	Subsistence <u>Rate/Hr</u>	Base <u>Rate/Hr</u>	Overtime <u>Rate/Hr</u>	Subsistence <u>Rate/Hr</u>	
Senior Operations Manager Senior Project Manager										
Senior Construction Manager										
Project Manager										2
Construction Manager										
Proj Superintendent I										
Proj Superintendent II										그 전 가 것 것
Proj Superintendent III										
Proj Superintendent IV										
Senior Project Engineer										
Proj Coord & Proj Engr I										
Coord/Field Engr II										
Coord/Field Engr III										
Coord/Field Engr IV										
Office Manager										
Safety Manager										
Material Manager										
QC Manager										
QC Inspector										
Instr Technician Level 1										
Instr Technician Level 2										
Instr Technician Level 3										
General Foreman										
Craft Foreman										
Leadman										
Licensed Journeyman										
Heavy Crane Operator										
Craftsman Level 1										
Craftsman Level 2										
Craftsman Level 3										
Helper Level 1										
Helper Level 2										
Helper Level 3										a depart 2 fps
Laborer										
Safetyman										
Warehouseman										
Toolman Field Clark										
Field Clerk	كوريد يكاريما ويناوي بالمهر تمع				and the second	and the second				

EXHIBIT "R" - LABOR RATES

Classificatio

...

Subsistence will be invoiced at the rates above plus applicable payroll taxesfor only those employees brought onto the jobsite from outside of the local area. Subsistence will be invoiced at the above hourly rate for all AB/K ZQ-Da-W

12/30/1 m

Notes:

hours worked.

TIC generally pays salaned employees on a per-week basis, and hourly employees on a per-day basis.

Additional hours (to equate to the full daily or weekly rate) will be invoiced in accordance with jobsite policies for special circumstances such as weather delays.

Overtime is billed for all hours over 40 per week and for all hours worked on holidays recognized by the owner.

Safety Material and consumables are included in the above hourly unit labor rates.

Materials will be billed at invoice cost plus tax and freight, plus overhead and profit of 15.00 percent.

Subcontractors will be billed at invoice cost plus 15.00 percent.

Rev Date: 7/1/11 Computer: File: 11.0069.12

B/L 29-Dec-11

EXHIBIT S

SITE AND SITE CONDITIONS

Table of Contents

1.0	Site Location	2
2.0	Site Areas	2
3.0	Site Description	3
4.0	Site Infrastructure	3
5.0	Basic Meteorological Data	3
6.0	Existing Construction Buildings	6

7.0 NERC Reliability Standards

- ABILL Zq-Decill
1.0 Site Location

The Ghent Generating Station Site (hereinafter in this Exhibit S, "Site") is located in Carroll County, Kentucky, adjacent to the Ohio River, approximately 50 miles southwest of Cincinnati, OH and 59 miles northeast of Louisville, Kentucky.





Drawing GH0-SK-C-00001 shows the general limits of the Site. Ghent CCR Transport 1 shows Landfill site. Ghent CCR Transport 2 shows CCR Transport access gates, access roads, laydown areas, and CCR Transport area.

- JBML 29-Doc-11

-3 29-Doc-1

The areas shown on the layout drawings shall also be used in respect of site construction purposes, for office requirements, stores and fabrication areas, workshops, canteen, Contractor parking and toilet facilities, and other necessary facilities.

3.0 Site Description

During construction of the Ghent CCR Transport System, the facility will remain open and operating. Operation of the plant requires periodic deliveries of coal by both, truck and rail, and continuous disposal of plant effluent. The Contractor shall conduct its operations in a manner that will not impede or adversely affect operation of the facility, delivery of coal, or disposal of plant effluent. Existing railroad tracks, roadways, and disposal facilities shall be kept open and usable, except during short, pre-scheduled and agreed upon durations. The Contractor shall note, other contractors will be performing work on-site during the construction period. The Contractor shall coordinate work to prevent any interference with plant operations or other contractors work.

4.0 Site Infrastructure

Site access is available via US 42. There is limited rail connection to the Site via CSX.

Construction power supply connection will be provided from an existing source of KU's choosing excluding the head end of the pipe conveyor. A suitable connection for construction power in offices shall be agreed with Owner.

Construction water supply shall tie in to existing services on Site and shall be agreed with Owner. Contractor shall take precautions to prevent contamination of Owner's potable water supply.

No permanent Construction sewage drainage is available at the Site and temporary sewage disposal arrangements will be the responsibility of the Contractor.

5.0 Basic Meteorological Data

Where additional data for design is required Contractor shall be responsible for determination of appropriate limiting conditions. Contractor is advised that this data is not up to date and is recommended to gather additional recent data.

Louisville Meteorological Data for Precipitation (inches): Averages 1960 – 1990, Extremes 1948 – 1995.

12/30/11 in

	Rainfall	Rainfall	Rainfall	Rainfall	Snowfall	Snowfall
	mean	high	nin	1day max	mean	max
January	2.86	-	0.45	3.00	5.90	28.40
February	3.30	9.02	0.76	3.66	5.00	15.90
March	4.66	14.91	1.03	6.97	3.10	22.90
April	4.23	11.10	0.76	4.08	0.20	1.60
May	4.62	11.57	1.37	4.60	~	-
June	3.46	10.11	0.49	5.12	-	-
July	4.51	10.05	0.99	5.09	~	-
August	3.54	8.79	0.23	3.12	-	
	Rainfall	Rainfall	Rainfall	Rainfall	Snowfall	Snowfall
	mean	high	min	1day max	mean	max
September	3.16	10.49	0.27	4.30		
October	2.71	6.47	0.39	2.64	-	2.40
November	3.70	9.12	0.72	3.58	1.00	13.20
December	3.64	8.86	0.65	2.77	2.20	9.30
Annual	44.39	59.80	30.38	6.97	17.40	43.10

Louisville Meteorological Data for Temperature (°F): Averages 1960 – 1990, Extremes 1948 – 1995.

	Monthly .	Averages		Daily Extremes		
	Max	Min	Mean	Max	Min	
January	40.3	23.2	31.7	77	-22	
February	44.8	26.5	35.7	77	-19	
March	56.3	36.2	46.3	86		
April	67.3	45.4	56.3	91	22	
May	76	54.7	65.3	95	31	
June	83.5	62.9	73.2	102	42	
July	87	67.3	77.2	105		
August	85.7	65.8	75.8	101	46	
September	80.3	58.7	69.5	104	33	
October	69.2	45.8	57.6	92	23	
November	56.8	37.3	47.1	84	-1	
December	45.1	28.6	36.9	76	-15	
Annual	66	46	56.1	105	-22	

- ABAL 29-Dec-11

Month	Mean	Max	Min
January	75.3	100	23
February	68.9	100	6
March	66.0	100	5
April	63.4	100	17
May	67.7	100	23
June	72.7	100	27
July	69.5	100	15
August	69.9	100	25
September	68.1	100	5
October	70.8	100	23
November	71.1		23
December	77.2		2.1

Louisville Meteorological Data for Humidity: 1993 - 2002.

Louisville Meteorological Data for Pressure (mbar): 1993 - 2002.

Month	Mean	Max	Min
January	1,019.8	1,043.8	994.4
February	1,019.7	1,039.4	990.7
March	1,017.9	1,042.1	989.6
April	1,015.7	1,032.3	994.4
May	1,015.2	1,029.6	998.1
June	1,015.3		995.4
July	1.016.2	_	1,005.2
August	1.017.2		990.0
September	1,016.9		997.8
October	1,019.0	1 .	
November	1,020.0	1 1	
December	1,019.8		

Louisville Meteorological Data for Wind Speed and Stability Class: 1984 – 1992.

Wind Speed Category (miles/hour)						
Calm	1-3	4-7	8-12	13-18	19-24	> 24
0.3	0.2	0.6	0.0	0.0	0.0	0.0
0.5	1.3	3.2	2.1	0.0	0.0	0.0
0.3	0.4	3.5	6.2	1.2	0.1	0.0
0.7	1.1	9.8	19.2	14.1	2.0	0.3
0.0	0.0	7.6	5.8	0.0	0.0	0.0
4.1	4.3	11.1	0.0	0.0	0.0	0.0
	Calm 0,3 0.5 0.3 0.7	Calm 1-3 0.3 0.2 0.5 1.3 0.3 0.4 0.7 1.1 0.0 0.0	Calm 1-3 4-7 0.3 0.2 0.6 0.5 1.3 3.2 0.3 0.4 3.5 0.7 1.1 9.8 0.0 0.0 7.6	Calm 1-3 4-7 8-12 0.3 0.2 0.6 0.0 0.5 1.3 3.2 2.1 0.3 0.4 3.5 6.2 0.7 1.1 9.8 19.2 0.0 0.0 7.6 5.8	Calm 1-3 4-7 8-12 13-18 0.3 0.2 0.6 0.0 0.0 0.5 1.3 3.2 2.1 0.0 0.3 0.4 3.5 6.2 1.2 0.7 1.1 9.8 19.2 14.1 0.0 0.0 7.6 5.8 0.0	0.3 0.2 0.6 0.0 0.0 0.0 0.5 1.3 3.2 2.1 0.0 0.0 0.3 0.4 3.5 6.2 1.2 0.1 0.7 1.1 9.8 19.2 14.1 2.0 0.0 0.0 7.6 5.8 0.0 0.0

6.0 Existing Construction Buildings

No existing buildings are available at the Site for construction use. Any temporary construction buildings needed for the Contractors use will have to be temporary and provided by the Contractor.



-6 JBML Za-Dec-11

tem	Value Engineering Items	
tem	Value Englicering items	
1	Rotate the bottom ash building 90 degrees.	
	Provide a portal reclaimer using a straight boom rather than a	
2	knuckle boom.	
3	For the Gypsum shed, possibly only enclose 3 sides.	
	n the second stars are for enter to lies of travellar	
4	For the conveyor system, use flop gates in lieu of traveling heads.	
4	For the ash handling battery/UPS system, use power from	
5	existing plant.	
	For the battery/UPS system, use a sealed lead-acid battery	
	type system similar to a telecommunication UPS system for	
6	the landfill area and gypsum area UPS systems	
-	Review the retaining wall design on the north along the creek	
7	and the southwest.	
	Direct bury 15KV and 25KV cable to landfill area routed	
8	alongside pipe conveyor	
	Combine the Unit 1 & Unit 2 Ash handling electrical systems	
9	and the Unit 3 & Unit 4 Ash handling electrical systems	
	I William a channer shift for pipion routh of 43 on the part and of	
10	Utilize a sleeper rack for piping south of 42 on the east end of Gypsum Slurry Rack (Clarification Item # 82)	
10	Look at the requirements for walkways along pipe racks and	
	the associated service air, service water, and receptacles to	
11	be located along walkways.	
12	Allow the wash water to go to ATB2.	
	Delete Restoration of spare Bottom Ash Pipe Lines From Unit	
13	1 & 3 Boilers To SFC Area	
	Utilize carbon steel piping in lieu of Basalt piping on straight	
	run sections of piping on spare bottom ash piping from Unit 1	
	& 3 Boilers to Submerged Flight Conveyor. Basalt abrasion	
	spools would be used at changes of direction similar to what is being supplied by Diamond Power on Fly Ash and Bottom	
12 -	Ash Contract.	
13.ə		

115/h-30-Dec-4

EXHIBIT U

TRAINING

Contractor will develop and implement a Job Site-based comprehensive training program for Owners experienced operating and maintenance (O&M) personnel to understand the overall integrated operation and maintenance of the Ghent CCR Transport System. O&M personnel will be required to pass the training course prior to operating or maintaining the Ghent CCR Transport System.

The training program will consist of Job Site classroom training and instruction for 64 Owners' O&M personnel, followed by on-the-job training through Commercial Operation. The level of detail and pace of this training program will be based on the assumption that Owners' personnel being trained have some knowledge and experience with power plants and power plant components and require only orientation to the specific operational requirements of the new Equipment.

Training for Equipment provided in the Ghent CCR Transport Contract will be provided by the each equipment supplier. The EPC Contract shall provide training for the balance of plant Equipment. Training shall be coordinated and scheduled in conjunction with the EPC Supplier's training to provide an overall comprehensive training program for the Ghent CCR Transport system.

Within 6 months after Effective Date, Contractor will prepare and submit the training program and preliminary schedule to Owners for review. It is expected that classroom training will begin approximately 8 weeks before and complete no later than the start of the final Tie-In Outage of the Ghent CCR Transport system, and Owners should ensure the availability of all nominated personnel requiring training at that time. Contractor shall gain approval from Owner for all training developed and implemented.

Classroom Training

Contractor's training materials will be based on system descriptions, each containing an operations section with a basic description of systems and equipment operations. Contractor's training materials will also provide a basic description of several modes of integrated plant operations. The level of detail provided in these materials will address any detailed step-by-step sequence action required by the operators. An alarm response section will address those alarms for which the appropriate operator response may not be obvious.

Lesson Plan Outline

The student's lesson text will incorporate the following elements:

- Course Objectives what the student is required to learn
- System Overview what the system does and how it works

13/h. 29-Dec-11

- Major Component Description functional description
- Flow Path and Controls major flow paths, instrumentation, protective devices, controls, and interlocks
- Principles of Operation
- Startup, Normal Operation, Shutdown, and Infrequent operation procedures
- Alarm Responses
- Support Systems systems needed for operations
- Hazards and Safety Features
- Routine Maintenance
- Emergency Procedures
- Figures and Tables (as required)

Contractor's trainers will present those portions of the classroom training that are prepared by Contractor (approximately 50 percent of the training). The remainder of the classroom training will be presented by the original equipment manufacturers (OEMs), major equipment and system vendors, or subcontractors. Contractor will arrange for, schedule, and coordinate the classroom training presented by OEMs/subcontractors.

The classroom training provided by the vendors/subcontractors will cover the following equipment and systems:

- Ghent CCR Transport System
- Gypsum Byproduct Dewatering Belt System
- Bottom Ash Dewatering Systems
- Pipe Conveyor System
- Fly Ash Conveying System
- Electrical Equipment
- Distributed Control System
- Auxiliary Systems

Student Evaluations

Up-to-date records reflecting each student's progress, including test results and attendance, will be submitted to Owners upon completion of each training course for review and evaluation.

Materials

Contractor will provide training materials, manuals, slides, films, and other instructional material as may be necessary for training.

29-Da-11

Training aids, such as films, slides, computer presentations, computer-simulated process interactive videos, and software packages/materials, along with supplies necessary to support, maintain, and successfully supplement the training program, will be furnished and then turned over to Owners at completion.

Facilities

Training will be conducted at the Ghent Generating Station Training building or other mutually agreed-on-site location.

On-The-Job Training

Contractor considers on-the-job training an essential part of an overall training program. During the commissioning phase, Owners O&M staff will work closely with the Contractor Startup team and the OEM technical advisors. This will give the O&M staff an opportunity to convert knowledge gained in the classroom to practical experience in the plant by assisting with hands-on activities such as preoperational system valve lineups. In addition, Contractor will involve Owners operations personnel during the development of the operating procedures and allow active participation in writing detailed sequences of actions required for basic integrated operations, such as cold start to full load, hot restart, normal shutdown, and response to plant abnormal conditions.

3 AB//L 29-Dec-11

1.

EXHIBIT V

OPERATING AND MAINTENANCE MANUALS

1.0 Operating and Maintenance Manuals Submittal Instructions

- In addition to electronic Submittals specified in Exhibit X, Equipment instruction books and operating manuals prepared by the each equipment supplier and the PJFF Supplier shall include the following:
 - a. Index and tabs.
 - b. Instructions for installation, start-up, operation, inspection, maintenance, parts lists and recommended spare parts, and data sheets showing model numbers.
 - c. Matrix of all regularly scheduled maintenance requirements.
 - d. Applicable drawings.
 - e. Warranties and guarantees.
 - f. Name and address of nearest manufacturer-authorized service facility.
 - g. All additional data specified.
- 2. Information listed above shall be bound into hard-back binders. Four manuals are required, two equivalent to heavy duty Bok-Hinge Split Prong, McBee Swing Hinge or Crane Binder Technologies post type binders or approved equal and two in standard binders. Sheet size shall be 8-1/2" x 11". Binder color shall be black. Capacity shall be a minimum of 1-1/2 inches, but sufficient to contain and use sheets with ease.

a. Provide the following accessories:

- i. Label holder.
- ii. Business card holder.
- iii. Sheetlifters.
- iv. Horizontal pockets.
- b. The following information shall be imprinted, inserted, or affixed by label on the binder front cover: See Appendix 01330-C for details.
 - i. Owner's name.
 - ii. Owner's facility or plant name.
 - iii. Equipment item name.
 - iv. Volume number (if applicable).
 - v. Contract number.
 - vi. Manufacturer's name and address.

c. The following information shall be imprinted, inserted, or affixed by label on the binder spine:

- i. Equipment item name.
- ii. Owner's name and Owner's facility or plant name.
- iii. Manufacturer's name.
- iv. Contract number.
- v. Volume number (if applicable).

d. Submit mockup of cover and spine for Owner's review.

ABAT-

2.0 Typical Instruction Book or Operating Manual Cover and Spine Layout



(Spine)

(Cover)

NOTES:

- 1. Choose either "Instruction Book" or "Operating Manual."
- 2. All lettering shall be a block style font, imprinting color to contract with binder color specified.
- 3. Cover lettering shall be point sizes indicated in column to right of cover illustration.
- 4. Spine letter shall be 14-point minimum.
- 5. *Volume number required only if instructions are contained in more than one volume.

ABIL 29-Dec-U

Ghent CCR Transport System

Unit of Property Description

Transport System

Electrical Systems Instrumentation and Controls Instrument Air Compressor Electrical/Control Building Maintenance/Shop Building Transfer Tower Concrete Foundations (by quantity) Transformer Foundations (by quantity) **Underground Utilities** Underground Electrical Ductbanks **Underground Piping** U1 Fines Purge Pump Building U2/3 Fines Purge Pump Building U4 Fines Purge Pump Building **CCR Underground Electrical CCR Equipment Foundations CCR** Equipment **CCR** Transformers CCR Retaining Wall Foundations **CCR Retaining Wall** CCR Conveyor Belts: Conveyors Belts/Buckets Rollers Motor Gear Reducer Supporting Steel

Pipe Rack Foundations Pipe Rack

Pipe Conveyor Foundations Pipe Conveyor

AB/L 29-Dec-11

Fly Ash	Fly Ash Handling Equipment Pump Motors Pug Mills Fly Ash Building Foundations Fly Ash Building 1/2 Fly Ash Building 3/4 Fly Ash Pipe Rack Silo "A" Silo "B" Marketable Silo Silo Stair Tower (2) 10 inch pressure piping 12 inch pressure piping 14 inch pressure piping 10 inch vacuum piping 12 inch vacuum piping 12 inch vacuum piping 2 Exhausters 2 Blowers 2 Filter Separators 4 Pressure Vessels
Bottom Ash	Submerged Flight Conveyor Submerged Flight Conveyor Foundation Submerged Flight Conveyor Building Pump Motors Ash Water Return Sump Ash Bunker Cover
Gypsum Dewatering	Gypsum Storage Shed Vacuum Belt Filter Vacuum Pumps Skids Cake Wash Skids Cloth Wash Skids Gypsum Processing Plant Foundation Gypsum Processing Plant Building Gypsum Processing Plant Sump Portal Reclaimer

JB/h 29-Dec-11

EXHIBIT X

,

Submittals, Reviews, and Hold Points

TABLE OF CONTENTS

1.0	Submittal Schedule	2
2.0	Review	2
3.0	Hold Points	2

JiB/K 24-Dec. 11

GHENT CCR Transport Project 172271.70.0100

Engineering, Procurement, and Construction

RFQ Issue 02Jun2011

٦

		TABLE 00451.2 - Schedule	of Sub	mittals				
Schod	ule of Subr	nittals and Applicable Liquidated Damages						T
	T	Intrais and Approable Exquidated Duringgeo			Submittal Dates		· · · · · · · · · · · · · · · · · · ·	Owner
Item	Reference	Submittal Item	Calenc	dar Days	Event	Due Date	Liquidated	Review
No.				,			Damages	Required Yes/No
	Ical Submi	Itals		L			Yes/No	Tesmu
Gene	ral	Plant Startup Procedures, Including sample Turnover	60	Before	Startup of first system		No	Yes
1		Packages	00	201010				
2	<u> </u>	Training		Before	Turn-over of each system			
3		Equipment O&M Manuals		Before	Start of Training		No	No
4		Ghent CCR Transport System Operations and Maintenance	30	Before	Start of Training		No	Yes
1	tructural	Manuals		I.,	1		L	
5		Site survey	15	After	Survey completed		No	No
6		Dewatering and settlement records			Final / As-Buill		No	No
7		Proposed dewatering system including: • Calculations • Piezometer locations • Dewatering Plan Layout • Installation Details • Filler materials • Backup system	10	Before	Sile Disturbance		No	Yes
8		Pile design details including pile lengths, catculations, reinforcing, uplift details, reinforcing steel and concrete strengths, concrete mix design, and concrete compressive test results.	20	Before	Pile installation begins		No	Yes
9	<u> </u>	Individual Pile Driving Records, including all test results		After	Pile Installation complete		No	No
	lations			r			·····	T
10		Concrete placement plan for special structures	15	Before	Concrete placement		No No	No No
11		Results of testing and test reports.	7 90	After After	Tesls Effective Date	L	No	Yes
12		Plant building and equipment foundations key plan with dimensions or coordinates, standard details and general notes	90	Aller	LIECTIVE DATE		,10	103
13		Preliminary foundation drawings	120	After	Effective Date		No	Yes
truct	ural Steel			T		r	· · · · · · · · · · · · · · · · · · ·	
14	T	Structural Steel Material Certification			Upon request		No	No No
15		High Strength Bolls Material Certification			Upon request Upon request		No No	No
16		Direct Tension Indicators Material Certification Preliminary Structural Steel Drawings for designed buildings	90	After	Effective Date		No	Yes
17		and structures (including pipe rack)						
Aateri	ial Handling					1		
18		Preliminary CCR handling flow diagrams	60	After	Effective Date Effective Date		No No	Yes Yes
19		Preliminary CCR conveyor plans and elevations	60	After		I		1 103
3ulidi 20	ngs	Final location, arrangement, and size of the buildings	120	After	Effective Date	I	No	Yes
20		Preliminary building floor plans and elevations	60	After	Effective Date		No	Yes
22		Building architectural notes, standard details, schedules for doors, windows, openings and miscellaneous hardware	120	After	Effective Date		No	No
	rground Piple	19 U/G piping systems drawings and details	90	Afler	Effective Date	1	No	Yes
23 rote	tive Coaling		I	- <u>L</u>				
24		Product data sheets	90	After	Effective Date		No	No
25		Surface preparation/coaling procedures	90	After	Effective Date		No	No
26	1	Lining/coating inspection/test procedures	90	After	Effective Date		No	No
27		Coating system application/repair procedures	90	After	Effective Date		No No	No Yes
28		Color scheme /coaling samples	120	After After	Effective Date Completion of work		No	No
29		Painting certification		1 Anei		L		
30	anical	Plot plan / general arrangement	60	After	Effective Date	1	No	Yes
31		Protipian's general analysiment Preliminary P&ID's and System Descriptions for EACH system, including - All major mechanical systems - HVAC and plumbing	120	Afler	Effective Date		No	Yes
32	1	Sump/Tank Capacities In gallons and hours	60	After	Effective Date		No	No
33		Major Equipment List	150	After	Effective Date		No	Yes Yes
34		Valye List	150	Afler	Effective Date Effective Date		No No	Yes
35		Pipeline List	150	After After	Effective Date		No	Yes
36		Mechanical Device List	100	Aller		1	No	Yes
37 38		Tie-in list Facility access arrangement drawings (walkways, platforms, ladders, and stalrways)	120	After	Effective Date	· · · · · · · · · · · · · · · · · · ·	No	Yes
40		Preliminary Piping arrangement composite drawings	120	Afler	Effective Date		No	Yes
41		Demolition Drawings	120	After	Effective Date		No	Yes
42		Pipe Support Drawings	160	Aîter	Effective Date	<u> </u>	No	Yes

AB/h-Page 1 of 3 29-Dcc-11

Source: 00451, 2006, v.1.0 12/30/11 W

GHENT CCR Transport Project 172271.70.0100

Engineering, Procurement, and Construction

RFQ Issue 02Jun2011

chedule of Submittals and Applicable Liquidated Damages			Submittal Dates					
tem No.	Reference Document	Submittal Item	Calend	ar Days	Event	Due Date	Liquidated Damages	Review Required
	Document	Major equipment specifications, including, but not limited to the	T	Belore	Equipment Procurement		No	Yes
43		Major equipment specifications, metoding, but not infinited to the following:						
		- Bulk Material Handling Equipment						
		- Fly Ash Silos						
		- Slurry Pumps						
		- General Service Pumps						
		- Tanks - Fire Protection Systems						1
		- Agitalors	1					
		- Compressed Air Equipment						
		Vendor equipment drawings, including		Before	Shipment to jobsite		No	No
44		- detailed equipment arrangement drawings						1
	1	- performance curves and data						
		- shop testing plan, schedule, and test results						
		- lubrication requirements - utility requirements (air, water, etc.)						
	ļ	- motor nameplate data						
		- maintenance / pull space requirements						
		- operating instructions / requirements						Yes
45		Fire protection system drawings, Including	120	After	Effective Date		No	105
40		- hydraulic calculations						
		- hydrant layout				1		
		- exlinguisher layout						
		- hose station layout - fire protection system(s) layout						
		- fire panel details, location, arrangement						
		Decido submittal for submittal lo insurer						1
	1	- Final conformed to construction drawings and documentation						
					Eller Date		No	No
46		Field Erected Tank drawings, Including	120	After	Effective Date			
		- outline drawings						
	1	- nozzle schedule			1			
		- calculations						
		- tank inspection reports Valve data, including valve drawings / cut sheats, lest reports,		Before	Shipment to jobsite		No	No
47		operator Information, Cv curves						
10		Control Valve date, Including valve drawings / cut sheets,		Before	Shipment to jobsite		No	No
48		calculations, test reports, flow data (Cv, Cg, or Cs),						
		instrumentation		1				
		Safety and Relief Valve data, including valve drawings / cut		Belore	Shipment to jobsite		No	No
49	1	sheets, calculations, test reports						
		HVAC Systems design information and drawings, Including	150	After	Effective Date		No	No
50		- ductwork and equipment arrangement						
		- system calculations						
		- vendor drawings						
Flant	rical							Yes
51		Preliminary Overall one-line diagrams	90	After	Effective Date		No No	Yes
52		Preliminary Electrical Load List	150	After	Effective Date		No	Yes
53		Transformer & Medium Voltage Meter and Relay One-Line		Before	Equipment Procurement		140	103
		Diagrams						Yes
54		Low Voltage Switchgear Meter and Relay One-Line Diagrams		After	Equipment Procurement		No	1 105
- 54					C. I was Discoursed		No	Yes
-55		Three Line Diagrams		After	Equipment Procurement Equipment Procurement		No	Yes
56		Modifications to Existing One-Line, Metering/Relaying, and/or		Alter	Edublietik Lincorement			
		Three-Line Diagrams		Before	Equipment Procurement		No	Yes
57		Preliminary Load Flow and motor starting calculation		Belore	Equipment Procurement		No	Yes
58		Preliminary Short circuit calculation		Before	Equipment Procurement		No	No
59		Preliminary Transformer sizing		Before	Equipment Procurement		No	No
60		Preliminary Baltery Sizing Grounding and lightning protection design calculation		Before	Equipment Procurement		No	No
61				Before	Equipment Procurement		No	No
62		Cable Sizing Underground duct bank cable sizing		Before	Equipment Procurement		No	No
63		Underground duct bank cable sizing		Belore	Equipment Procurement		No	No
64		Transformer specifications		Before	Equipment Procurement		No	Yes Yes
66		Medium and Low Voltage Switchgear Specifications		Before	Equipment Procurement		No	Yes
67		Motor Control Center Specifications		Before	Equipment Procurement		No	Yes
68		DC System Specifications	1	Before	Equipment Procurement		No	Yes
69		UPS Specification		Before	Equipment Procurement		No	Yes
70		Electrical Tie-In List	L	Before	Release for Construction		No	No
7		Circuit and Raceway Schedule as a sort-able data file (Refer to	1	Before	Release for Construction	'		
1 '	·	Client Format Standards)			Release for Construction		No	No
7:	2	Completed motor data sheets for each individual motor	ļ	Before	Release for Construction		No	Yes
7:		Lightning protection drawings	 	Before	Release for Construction		No	Yes
7.		Grounding Drawings		Before	Release for Construction		No	Yes
7	5	Lighting/paging/communication/security drawings		Before	Release for Construction		No	No
	6	Cable Tray Drawings	1					

Page 2 of 3 29-De--11

12/30/11

Schedule and Progress Controls Tables

GHENT CCR Transport Project 172271.70.0100

Engineering, Procurement, and Construction

RFQ Issue 02Jun2011

1

		1 Submittai item		Submittal Dates					
item No,	Reference Document			lar Days	Event	Due Date	Liquidated Damages	Review Require	
77		Raceway Drawings		Before	Release for Construction		No	No	
78		Schematic Diagrams and termination lists		Before	Release for Construction		No	No	
79		Heat tracing drawings		Before	Release for Construction		No	No	
80		Protective relay settings and device coordination calculation		Before	Equipment Energization		No	Yes	
81		Relay instruction and service manuals, device setting summary, setting sheets, protective device software files, coordination curves, CT ratio correction factor and excitation curves		Before	Equipment Energization		No	No	
83		Transformer tap selection calculation		Before	Equipment Energization		No	No	
84		Engineering Detailed Design Documents listed above.			Each Revision		No	No	
85		Manufacturers Design Documents			Each Revision		No	No	
	ientation & C								
	ientation & C		180	After	Effective Date		No	Yes	
86 87		Control room arrangement drawing Overall control system architecture drawing - for Owner Approval	180	Alter	Effective Date		No	Yes	
88		Overall control system architecture diagram		Before	Release for Construction		No	Yes	
89		VO List - Hardware Cutoff		Before	Release for Fabrication		No	No	
90		VO List - Software Culofi		Before	Release for Configuration		No	No	
92		Logic discriptions		Before	Release for Configuration		No	Yes	
92		Operator graphic sketches		Before	Release for Configuration		- No	Yes	
93		Preliminary Instrument list		Before	Equipment Procurement		No	Yes	
95		Flow element calculation data sheets showing beta ratio, dP, flow		Before	Equipment Procurement		No	No	
00		Instrument Installation details		Before	Release for Construction		No	Yes	
96	ļ	Instrument factory calibration sheets		Upon	Shipment to jobsite		No	No	
97		Engineering Detailed Design Documents listed above.			Each Revision		No	No	
98		Manufacturers Design Documents			Each Revision		No	No	
99	L			I	1				
100	Q100 in Allach A	uirements - Q002 in Attachment A Quality Manual, controlled copy	30	After	Effective Date		No	No	
101	Q100 in Atlach A	Certification Letter or Certificate of Authorization (copy), if certified by a registered agency, e g, ASME Certificate of Authorization	30	Afler	Effective Date		No	No	
102	Q100 in Attach A	Subsupplier listing	30	After	Effective Date		No	Yes	
103	Q100 in Allach A	Inspection and test plan	50	After	Effective Date		No	Yes	
104	Q100 in Allach A	Nolification of inspection/test (for Owner hold/witness points)	14	Before	Tesl/Inspection		No	No	
Sener		eguirements - Q100 in Altachement A							
105	Q100 In Attach A	Welding Procedure Specifications (WPS) with applicable Procedure Qualification Records (PQR)	30	Before	Start of Fabrication		No	Yes	
106	Q100 in Atlach A	Welder and welding operator qualification records		Upon	Request		No	Yes	
107	Q100 in Altach A	Procedures for storing, issuing, and reconditioning of electrodes, wires, and fluxes	30	Before	Start of Fabrication		No	Yes	
108	Q100 in Altach A	Repair procedures associated with a nonconformance report	5	After	Discovery of Repair		No	Yes	
109	Q100 in Allach A	Post-weld heat treatment procedures	30	Before	Start of Fabrication Start of Fabrication		No	Yes	
110	Q100 in Allach A	Solution annealing heat treatment procedure, if required	30 30	Before	Start of Fabrication		No	Yes	
111	Q100 in Allach A	CMTR or equivalent material test reports for filler materials for walding P-No. 5B or 15E (9Cr-1Mo-V) alloy materials	30	Before	Start of Fabrication		No	No	
112	Q100 in Allach A	Visual Inspectors' qualifications and certificates	30	Before	Start of Fabrication		No	Yes	
113	Q100 in Altach A	Nondestructive examination procedures Root side purging procedures	30	Before	Start of Fabrication		No	Yes	
114	Q100 in Altach A		5	After	Discovery of Nonconf.		No	Yes	
115	Q100 in Allach A Q100 in	Nonconformance reports ASME/NBIC Data Reports	30	After	Fabrication Completion		No	Yes	
116									

Page 3 of 3 2-4-0 ec-11

1

12/30/11 W

1.0 Submittal Schedule (Contained within Table 00451.2)

2.0 Review

Contractor shall allow 10 Business Days in Owner's office for reviewing original Submittals and 10 Days in Owner's office for reviewing resubmittals.

3.0 Hold Points

The following list of activities shall be treated as Hold Points:

- A. Backfill Placement Compaction Testing
- B. Rebar Placement
- C. Concrete Placement
- D. Insulation Placement over Ductwork (Weld Inspections)
- E. Paint Primer and Final Coat
- F. Structural Steel Connection Final Bolt Up
- G. Embedded Grounding and Conduits (Prior to Concrete Pour)
- H. Electrical Devices (Transformers, Switchgear, MCC, Controls) Prior to Energization

ABILL 29-Dec-11



KENTUCKY UTILITIES COMPANY CONTRACTNo. 517454

Bottom Ash Handling & Fly Ash Conveying Systems

This contract (this "Contract") is entered into, effective as of October 31, 2011 (the "Effective Date"), between Kentucky Utilities Company, a Kentucky corporation, an affiliate of LG&E and KU Services Company (hereinafter referred to as "KU" or "Company"), whose address is 820 West Broadway, Louisville, KY 40220, and Diamond Power International Inc, a Delaware corporation (hereinafter referred to as "Contractor"), whose address is 2600 Bast Main Street, P.O. Box 415, Lancaster, Ohio, 43130, Contractor is wholly owned subsidiary of The Babcock & Wilcox Power Generation Group, Inc., a Delaware corporation ("BWPGG"). BWPGG has entered into a General Services Agreement (the "GSA") with LG&E and KU Services Company, a Kentucky corporation. For purposes of this Contract, the terms of the GSA shall apply as if Contractor was a party to the GSA rather than BWPGG and the terms of this Contract (other than Exhibit 2, the GSA) constitute a Purchase Order under the GSA. The parties herelo agree as follows:

1.0 GENERAL

Contractor shall provide the following: <u>Bottom Ash Handling System & Fly Ash Conveying System</u> (individually "System" and collectively the "Systems"), and other work as more specifically described in Articles 2.0 and 3.0 hereof (hereinafter referred to as the "Work") and COMPANY shall compensate the Contractor for the Work, under all the terms and conditions hereof.

2.0 DESCRIPTION OF WORK

- 2.1 Except as otherwise expressly provided herein, Contractor shall supply all labor, supervision, materials, equipment, tools and warehousing, and shall pay all expenses, necessary or appropriate in the performance of the Work.
- 2.2 No materials containing asbestos shall be supplied or used in the performance of Work.
- 2.3 Without limitation, Contractor shall meet all requirements set forth in the Lead Construction Standard 29 CFR 1926.62.
- 2.4 The Work shall include but not be limited to the following:
 - 2.4.1 The Work specified in Exhibit 1 and 1a.

1|Page

3.0 SPECIFICATIONS, EXHIBITS AND DRAWINGS

All Work shall be performed in strict accordance with the following specifications, exhibits and drawings which are incorporated herein by reference.

3.1 Exhibits

EXIJIBIT NO.	TITLE
Exhibit No. 2	General Services Agreement "GSA"
Exhibit No. 3	Bottom Ash Handling Project Cost and Deliverables
Exhibit No. 3a	Fly Ash Conveying Project Cost and Deliverables
Exhibit No. 4	LG&E AND KU SERVICES COMPANY
· · ·	Contractor/Subcontractor Safety Policy
Exhibit No. 5	LG&E AND KU SERVICES COMPANY Drug Testing
	Program
Exhibit No. 6	LG&E AND KU SERVICES COMPANY Contractor Code
	of Conduct
Exhibit No. 7	Daily Activity Report
Exhibit No. 8	Passport Overview
Exhibit No. 9	Chauge Order/Work Release
Exhibit No. 10	Project Specific Hazard Analysis
Exhibit No. 11	Hazard Mitigation Plan
Exhibit No. 12	Contractor Quality Assurance Closure Form
Exhibit No. 13	Contractor Safety Questionnaire
Exhibit No. 14	Assignment and Assumption Agreement
Exhibit No. 15	Cable Tab Records Submittal Specification
Exhibit No 16	Electronic Submittal Spec for Engineered Dwgs
Exhibit No. 17	Electronic Submittal Spec for Vendor Docs
Exhibit No. 18	Bottom Ash Handling Technical Clarification Log
Exhibit No. 18a	Fly Ash Conveying 'Teclunical Clarification Log
Exhibit No. 19	Job Performance Monitoring Tool
Exhibit No, 20	Parent Guarantee
Exhibit No. 21	KU Ghent Cross reference lube oil list
Exhibit No. 22	Fly Ash Conveying Spare Parts
Exhibit No. 23	Bottom Ash Handling Spare Parts
Exhibit No. 24	Form for Letter of Credit (LOC)

3.2 DRAWINGS (See Exhibit 1 and Exhibit 1a)

4.0 NOT USED

5.0 TERM

This Contract shall become effective upon Effective Date and continue through Final Completion, subject to Article 21 of the GSA (Exhibit 2).

2|Page

AB/1-29-Dec-4

12130/V

6.0 PERFORMANCE SCHEDULE

-- -----

- 6.1 Contractor shall commence performance of the Work on the Effective Date and shall complete Work not later than the dates listed below in Table 1 (Bottom Ash Handling) and Table 2 (Fly Ash Conveying).



2BML 29-Denill

12/30/11 m



COMPANY intends to assign this Contract to a general contractor (the "General Contractor") who will integrate the Work with work performed by the General Contractor and its other subcontractors into an integrated installed coal combustion residuals transport system (the "System") at the Company's Ghent Generating Station. Alternatively, the Company reserves the right to complete the System itself with the participation by subcontractors. In either case, with respect to the System, the following terms have the meanings assigned them below:

> "Mechanical Completion" means when, with respect to the System, all of the following have occurred; (i) all components of the System have been furnished and installed, checked for alignment, lubrication, rotation, and hydrostatic and pneumatic pressure integrity; (ii) turnover acknowledgement of all components of the System has been given by Company; (iii) all components of the System have been installed, cleaned and statically tested and cold commissioning has been completed; (iv) all initial tests of the System have been completed and passed and all instruments in the System have been calibrated; (v) an initial punch list has been established and mutually agreed upon by Company and General Contractor (or subcontractors, as applicable); (vi) the performance guarantee test procedures for the System have been submitted to and accepted by Company.

> **"Commercial Operation" will have been achieved when all of the following have occurred: (i) Mechanical Completion has occurred; (ii) all work with respect to the System is complete except punch list items; (iii) all performance guarantee tests on the System have been successfully performed, and all necessary and desirable system adjustments identified during the start-up and testing process have been made; (iv) all performance guarantees on the System have been simultaneously achieved in a single performance guarantee test; (v) all obligations of General Contractor (or subcontractors, as applicable) required to have

> > -

.....

12/30/11 ar 4 Page

ABIL 29-Dec-M

been performed as of Commercial Operation have been properly discharged; (vi) final versions of the operating and maintenance manuals approved by Company have been delivered; (vii) all other submittals required to be submitted prior to or as of Commercial Operation have been submitted; (viii) all special tools have been delivered to Company.

***"Final Completion" will be deemed to have occurred when all of the following have occurred in respect of the System: (i) Commercial Operation has been achieved, (ii) all guarantee tests on the System have been conducted demonstrating compliance with all guarantee points; (iii) all as-built drawings, design documents, station manuals and other documents required to be delivered to Company have been so delivered, (iv) liquidated damages for which General Contractor (or the subcontractors, as applicable) is Hable and other amounts owed to Company with respect to the System, if any, have been paid to Company (v) all work with respect to the System has been completed other than work that requires future performance (e.g., warranty work), (vi) General Contractor (or the subcontractors, as applicable) has demobilized from the Work site leaving it clean and free of debris.

- 6.1.1 Contractor shall coordinate Contractor-supplied material with COMPANY representative and deliver materials to the job site as required.
- 6.1.2 Contractor shall notify COMPANY's representative at least one full working day prior to working any Saturday, Sunday or COMPANY holiday.
- 6.2 If Company reasonably determines that Contractor has fallen behind its schedule for completing the Work and that there is a material risk that Contractor will not meet the scheduled dates for the milestones set forth in Section 6.1 above, Company may give notice to this effect to Contractor. Upon the giving of such notice, Contractor shall have 10 days to develop and implement a plan to ensure it will meet all scheduled milestone dates and to deliver such plan to Company. If no such plan is delivered in the 10 days, or if the Company reasonably determines that such plan does not adequately ensure meeting all scheduled milestone dates, Company may terminate the Contract for cause.

7.0 STANDARD TERMS AND CONDITIONS

The terms of LG&E AND KU SERVICES COMPANY's General Services Agreement (attached as Exhibit 2) executed by BWPGG on November 19, 2010, and the Contractor Code of Business Conduct and Contractor Safety Policies are hereby incorporated by reference herein and are thereby made a part of this Contract.

7.1 ASSIGNMENT AND ASSUMPTION AGREEMENT

Company reserves the right to assign this Contract to the General Contractor pursuant to the Assignment and Assumption Agreement set forth in Exhibit 14 attached hereto and made a part hereof. Upon request of Company, Contractor shall execute an acknowledgement of such an assignment and assumption, provided, that the assignment and assumption shall be effective and enforceable regardless of whether or when Contractor executes such acknowledgement or otherwise expresses any consent thereto. Upon such assignment and assumption, Company shall be released of its obligations under this Contract and Contractor shall look only to the General Contractor to fulfill the obligations designated as being those of Company hereunder. Contractor also acknowledges that General Contractor may in turn assign this Contract to another entity or to the Company and such assignments shall not require any consent from Contractor.

7.2 ORDER OF PRECEDENCE

The body of this Contract and the Exhibits hereto are to be considered complementary and what is required by one will be binding as if required by all. Contractor has included within the Contract

5|Page

J-B//L-29-Dec-11

17,13011

Price the cost to complete the Work in its entirety and to fulfill its other obligations hercunder. In addition, the Work includes all that should be properly included and all that would be customarily included within the general scope and magnitude of the Work in order to achieve the obligations. The failure to specifically list a requirement in one document, once such requirement is specifically listed in another, does not imply the inapplicability of such requirement and Contractor shall provide or perform, as appropriate, as part of this Contract all Work or items required to conform to the Technical Specification (Exhibits 1 and 1a), the other Exhibits and the standards herein contained. In the event of a conflict between the body of this Contract and the Exhibits, the body of this Contract governs. In the event of a conflict in an Exhibit or between or among Exhibits, the requirement most favorable to Company will take precedence, except as may be otherwise determined by Company. Each Modification (defined as any contract amendment or change order) will take precedence over that part of this Contract (including, as applicable, any prior Modification) which it supersedes.

8.0 CONTRACTOR DRUG AND ALCOHOL TESTING

8.1 Plant Outage, Plant Project, or Major Construction Work: The work under this Contract is considered "Plant Outage, Plant Project, or Major Construction Work". In accordance with the revised I.G&E AND KU SERVICES COMPANY Contractor / Subcontractor Safety Policy effective 1-31-08, all contractor employees working on-site for "Plant Outage, Plant Project, or Major Construction Work" shall be required to complete a negative drug and alcohol pre-test within seven (7) days before reporting to work on-site. The Contractor shall complete random drug and alcohol testing on ten percent (10%) of their on-site workforce assigned to this work site within thirty (30) days of the start of Work and shall test ten percent (10%) subsequently on an on-going monthly basis for the duration of their on-site "project" work assignment. LG&E AND KU SERVICES COMPANY Corporate Health and Safety will be auditing Contractor compliance with these requirements. Testing costs are the responsibility of the Contractor.

9.0 SPECIFIC REPORTING REQUIREMENTS

Contractor shall promptly submit the schedules and reports set forth below:

9.1 In accordance with Exhibit 3 (Project Cost and Deliverables).

6]Page

JB1L 29-Dec-M

12130/11



71Page ABAL 29-Dec-11

12130/11 M

	- 이미 전 이미
	- 「「「」「」「」「」「」「」」「」」「」」「」」「」」「」」「」」「」」「」」
	그는 것 같은 것 같
	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
· · · · · · · · · · · · · · · · · · ·	
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	그는 것 같은 것 같
	에는 것이 있는 가 가 있는 것이 없이 없이 없이 있는 것이 있는 것이 없이 없이 있는 것이 없이 없이 없이 없이 있 같이 것이 같이 것이 있는 것이 없이
	2. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.
	「「「」」「「」」」、「「」」「「「」」」、「」」」「「」」」「「」」、「」、「
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
and the second	· · · · · · · · · · · · · · · · · · ·
	- 이 이 이 이 이 이 이 이 이 이 이 이 이 이 이 이 이 이 이
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	· · · · · · · · · · · · · · · · · · ·
	ことが、 「「「「「「」」」 「「」」 「」」 「」」 「」」 「」」 「」」 「」」
1	
	그는 것 같아요. 가지 않는 것 같아요. 가지 않는 것 같아요. 이 가지 않는 것 않는
e f	그는 그는 그는 그는 그는 그는 것이 있는 것이 가지 않는 것이 같은 것이 많이 많이 많이 많이 많이 많이 나갔다. 귀엽錄.
	· · · · · · · · · · · · · · · · · · ·
and the second	이 가지 않는 것 같은 것 같
4	이 가지 않는 것 같은 것 같
	· · · · · · · · · · · · · · · · · · ·
	그는 것 같은 것 같
· ·	

10.2 SPECIAL INVOICING INSTRUCTIONS

10.3.1 See the Article titled "Invoices and Effect of Payments" in the GSA.

10.3.2 Invoices shall be prepared in one original and one copy distributed as follows:

7	Original:	Kentucky Utilitics Cor 820 West Broadway Louisville, KY 40202 Attn: Jeff Heun	որույ	
	Сору:	Kentucky Utilities Con 820 West Broadway Louisville, KY 40202 Attn: Judy Disney	ipany	
	İnyoice Information	Project # Company Contact Contractor contact	511453 JeftYHeum Mike Gibson	

8|Page

JBAL 24-Dec-11

12/3014 M



CONTRACTUAL NOTICES 11.0

See the Article titled "Notices" in the GSA for provisions governing contractual notices. In addition, a copy of all notices to COMPANY shall be sent to:

11.1	Kentucky Utilities Company address:	820 West Broadway Louisville, KY 40202 Attn: Charles Ransdell (502) 627-3299 <u>charles.mnsdell@lgc-ku.com</u>
11.2	Contractor's Address:	Diamond Power International, Inc. 2600 East Main Street, P.O. Box 415 Lancaster, Ohio, 43130

ENTIRE AGREEMENT 12,0

This Contract, including all specifications, exhibits and drawings listed in this Contract and the GSA, constitutes the entire agreement between the parties relating to the Work and supersedes all prior or contemporaneous oral or written agreements, negotiations, understandings and statements pertaining to the Work or this Contract.

lazotti@a-s-li.com

9112 age /BAL-29-Dec-11

12130/11

The parties hereto have executed this Contract on the dates written below, but it is effective as of the date first written above,

Kentucky Utilities Company BY: CLEMENTS Jae NAME (Print): CONTRACTS TITLE: MANAGES DATE: 1 10 Diamond Power-Juffigrational, Inc. Sign Here BY: Course Л NAME (Print); 2051 TITLE: 201 DATE: -END- .

10|Page

ABAL 29-Da-11

COMMERCIAL SCHEDULE A LABOR HOURLY RATES

Description	Straight Time	Overtime	Premium Time
Field Service Representative			

- 1. The above rates may be used for agreed to Change Orders and Standby Time.
- 2. Except as may be expressly provided otherwise elsewhere in this Contract, the rates set forth above are inclusive of all direct wage rates, fringe benefits, labor allowances, payroll taxes, insurance, small tools which cost Contractor less than \$1,500 per tool, temporary construction facilities, consumables, expendables, overhead profit and all other costs and expenses incurred by Contractor in performing the Work and this Contract.
- 3. The rates will only apply to actual hours worked or standby time, as agreed and attested to by a COMPANY Representative.
- 4. Overtime rates will be paid for hours worked in excess of 8 hours per day when worker is on 5 8 hour day schedule; for hours worked in excess of 10 hours per day when worker is on 4 10 hour day schedule. Overtime will be paid for hours in excess of forty (40) hours per week. All overtime must be pre-authorized by COMPANY
- 5. Premium time will be paid for all hours worked on and the following holidays.

New Years Day Memorial Day Good Friday Independence Day Labor Day Thanksglving Christmas Eve Christmas Day

AS/1/2 29-Dec-11



KENTUCKY UTILITIES COMPANY CONTRACT No. 514629

Pipe Conveyor System

This "Contract" is a Statement of Work within the meaning of the General Services Agreement (the "GSA") between (i) LG&E and KU Services Company, a Kentucky corporation, and (ii) Beamer Corporation, a Delaware corporation (hereinafter referred to as "Contractor"), whose address is 4435Main Street, Suite 750, Kansas City, Missouri, 64111. This Contract is entered into, effective as of October 7, 2011 (the "Effective Date"), between Kentucky Utilities Company, a Kentucky corporation, an affiliate of LG&E and KU Services Company (hereinafter referred to as "KU" or "Company"), whose address is 820 West Broadway, Louisville, KY 40220, and Contractor. The parties hereto agree as follows:

1.0 GENERAL

Contractor shall perform the following: Engineer, provide, deliver, and install a <u>Pipe Conveyor System</u> as more specifically described in Articles 2.0 and 3.0 hercof (hereinafter referred to as the "Work") and COMPANY shall compensate the Contractor for the Work, under all the terms and conditions hereof.

2.0 DESCRIPTION OF WORK

- 2.1 Except as otherwise expressly provided herein, Contractor shall supply all labor, supervision, materials, equipment, tools and warehousing, and shall pay all expenses, necessary or appropriate in the performance of the Work.
- 2.2 No materials containing asbestos shall be supplied or used in the performance of Work.
- 2.3 Without limitation, Contractor shall meet all requirements set forth in the Lead Construction Standard 29 CFR 1926.62.
- 2.4 The Work shall include but not be limited to the following:
 - 2.4.1 Pipe Conveyor Spec 172150.77.0430 dated June 16th, 2011.

3.0 SPECIFICATIONS, EXHIBITS AND DRAWINGS

All Work shall be performed in strict accordance with the following specifications, exhibits and drawings which are incorporated herein by reference.

1|Page

9-15/16 29-Doc-11

^{3.1} Exhibits

EXHIBIT NO.	<u>Revision</u> <u>No.</u>	TITLE
EXHIOR ING. 2		Otherar services Agreement OSA
Exhibit No. 3		Project Cost and Deliverables
Exhibit No. 4		LG&E AND KU SERVICES COMPANY Contractor/Subcontractor Safety Policy
Exhibit No. 5		LG&E AND KU SERVICES COMPANY Drug Testing Program
Exhibit No. 6		LG&E AND KU SERVICES COMPANY Contractor Code of Conduct
Exhibit No. 7		Daily Activity Report
Exhibit No. 8		Passport Overview
Exhibit No. 9		Change Order/Work Release
Eshibit No. 10		Project Specific Hazard Analysis
Exhibit No. 11		Hazard Mitigation Plan
Exhibit No. 12		Contractor Quality Assurance Closure Form
Exhibit No. 13		Contractor Safety Questionnaire
Exhibit No. 14		Assignment and Assumption Agreement
Exhibit No. 15		Cable Tab Records Submittal Specification
Exhibit No 16		Electronic Submittal Spee for Engineered Dwgs
Exhibit No. 17		Electronic Submittal Spec for Vendor Docs
Exhibit No. 18		Technical Clarification Log
Exhibit No. 19		Job Performance Monitoring Tool
Exhibit No. 20		Ghent Cross Reference Lube List

4.0 TEMPORARY FACILITIES AND UTILITIES

4.1 FURNISHED BY CONTRACTOR

Except as otherwise expressly set forth in this Contract, Contractor shall supply, install, properly maintain and remove all temporary facilities and utilities necessary for performance of the Work, including but not limited to:

- 4.1.1 All temporary buildings
- 4.1.2 All sanitary facilities, including janitorial services
- 4.1.3 First-aid facilities
- 4.1.4 Fuels and lubricants
- 4.1.5 Transportation facilities on and off site
- 4.1.6 Communication facilities
- 4.1.7 Compressed air and gases
- 4.1.8 Maintenance of Contractor's lay down, storage and work areas and roads within such areas
- 4.1.9 Rigging, scaffolding and all equipment for crection

2|Page

JBML-Eq-Dec-H

- 4.1.10 All cranes and other necessary equipment for lifting and moving equipment
- 4.1.11 All small tools & equipment necessary to perform Work
- 4.1.12 All standard expendable or consumable construction items and supplies including welding material and tools
- 4.1.13 Containers, ice, cups for drinking water
- 4.1.14 Cost of unloading, loading and storing all materials, equipment and supplies
- 4.1.15 Electrical power for small tools, etc., required in the actual work areas
- 4.1.16 Dumpsters and waste disposal

The type of facilities, move-in and move-out dates and locations on job sites shall be subject to and in accordance with the review and approval of COMPANY.

4.2 FURNISHED BY COMPANY

4.2.1 Temporary Construction Power at terminal point(s) determined by Company.

5.0 TERM

This Contract shall become effective upon Effective Date and continue through Final Completion, subject to the Article titled "Termination at COMPANY's Option" set forth in the attached GSA. COMPANY makes no promise or guarantee as to the amount of Work to be performed under this Contract.

6.0 PERFORMANCE SCHEDULE



6.1 Contractor shall commence performance of the Work on the Effective Date and shall complete Work not later than the dates listed below.

3[Page

1B11-129-000-M

12130/11 an COMPANY intends to assign this Contract to a general contractor (the "General Contractor") who will integrate the Work with work performed by the General Contractor and its other subcontractors into an integrated installed coal combustion residuals transport system (the "System") at the Company's Ghent Generating Station. Alternatively, the Company reserves the right to complete the System itself with the participation by subcontractors. In either case, with respect to the System, the following terms have the meanings assigned them below:

"Mechanical Completion" means when, with respect to the System, all of the following have occurred: (i) all components of the System have been furnished and installed, checked for alignment, lubrication, rotation, and hydrostatic and pneumatic pressure integrity; (ii) turnover acknowledgement of all components of the System has been given; (iii) all components of the System have been installed, cleaned and statically tested and cold commissioning has been completed; (iv) all initial tests of the System have been completed and passed and all instruments in the System have been calibrated; (v) an initial punch list has been established and mutually agreed upon by Company and General Contractor (or subcontractors, as applicable); (vi) the performance guarantee test procedures for the System have been submitted to and accepted by Company.

**"Commercial Operation" will have been achieved when all of the following have occurred: (i) Mechanical Completion has occurred; (ii) all work with respect to the System is complete except punch list items; (iii) all performance guarantee tests on the System have been successfully completed, and all necessary and desirable system adjustments identified during the start-up and testing process have been made; (iv) all performance guarantee test; (v) all obligations of General Contractor (or subcontractors, as applicable) required to have been performed as of Commercial Operation have been properly discharged; (vi) final versions of the operating and maintenance manuals approved by Company have been delivered; (vii) all other submittals required to be submitted prior to or as of Commercial Operation have been submitted; (viii) all special tools have been delivered to Company.

***"Final Completion" will be deemed to have occurred when all of the following have occurred in respect of the System: (i) Commercial Operation has been achieved, (ii) all guarantee tests on the System have been conducted demonstrating compliance with all guarantee points; (iii) all as-built drawings, design documents, station manuals and other documents required to be delivered to Company have been so delivered, (iv) liquidated damages for which General Contractor (or the subcontractors, as applicable) is liable and other amounts owed to Company with respect to the System, if any, have been paid to Company (v) all work with respect to the System has been completed other than work that requires future performance (e.g., warranty work), (vi) General Contractor (or the subcontractors, as applicable) has demobilized from the Work site leaving it clean and free of debris.

- 6.1.2 Contractor shall coordinate Contractor-supplied material with COMPANY representative and deliver materials to the job site as required.
- 6.1.3 Contractor shall arrange or coordinate material storage at the job site or offsite shop to avoid material theft or damage.
- 6.1.4 Contractor shall notify COMPANY's representative at least one full working day prior to working any Saturday, Sunday or COMPANY holiday.

4 | Page

JBML. 29-Da-W

6.2 If Company reasonably determines that Contractor has failen behind its schedule for completing the Work and has reasonable concerns whether Contractor will meet the scheduled dates for the milestones set forth in Section 6.1 above, Company may terminate the Contract for cause.

7.0 STANDARD TERMS AND CONDITIONS

7.1 ASSIGNMENT AND ASSUMPTION AGREEMENT

Company reserves the right to assign this Contract to the General Contractor pursuant to the Assignment and Assumption Agreement set forth in Exhibit 14 attached hereto and made a part hereof. Upon request of Company, Contractor shall exceute an acknowledgement of such an assignment and assumption, provided, that the assignment and assumption shall be effective and enforceable regardless of whether or when Contractor executes such acknowledgement or otherwise expresses any consent thereto. Upon such assignment and assumption, Company shall be released of its obligations under this Contract and Contractor shall look only to the General Contractor to fulfill the obligations designated as being those of Company hereunder. Contractor also acknowledges that General Contractor may in turn assign this Contract to another entity or to the Company and such assignments shall not require any consent from Contractor.

7.2 ORDER OF PRECEDENCE

The Body of this Agreement and the Exhibits hereto are to be considered complementary and what is required by one will be binding as if required by all. Contractor has included within the Contract Price the cost to complete the Work in its entirety and to fulfill its other obligations hereunder. In addition, the Work includes all that should be properly included and all that would be customarily included within the general scope and magnitude of the Work in order to achieve the obligations. The failure to specifically list a requirement in one document, once such requirement is specifically listed in another, does not imply the inapplicability of such requirement and Contractor shall provide or perform as appropriate as part of this Agreement all Work or items required to conform to the the other Exhibits and the standards herein contained. In the event of a conflict in an Exhibit or between or among Exhibits, the Requirement most favorable to Company will take precedence, except as may be otherwise determined by Company. Each Modification (defined as any contract amendment or change order) will take precedence over that part of this Agreement (including, as applicable, any prior Modification) which it supersedes.

7.3 LIMITATION OF LIABILITY

5 Page

JBML 29-Dec-M

8.0 CONTRACTOR DRUG AND ALCOHOL TESTING

8.1 Plant Outage, Plant Project, or Major Construction Work: The work under this Contract is considered "Plant Outage, Plant Project, or Major Construction Work". In accordance with the revised LG&F AND KU SERVICES COMPANY Contractor / Subcontractor Safety Policy effective 1-31-08, all contractor employees working on-site for "Plant Outage, Plant Project, or Major Construction Work" shall be required to complete a negative drug and alcohol pre-test within seven (7) days before reporting to work on-site. The Contractor shall complete random drug and alcohol testing on ten percent (10%) of their on-site workforce assigned to this work site within thirty (30) days of the start of Work and shall test ten percent (10%) subsequently on an on-going monthly basis for the duration of their on-site "project" work assignment. LG&E AND KU SERVICES COMPANY Corporate Health and Safety will be auditing Contractor compliance with these requirements. Testing costs are the responsibility of the Contractor.

9.0 SPECIFIC REPORTING REQUIREMENTS

Contractor shall promptly submit the schedules and reports set forth below:

- 9.1 Critical Path Schedule, as specified by COMPANY. The schedule shall show labor required by time interval and shall reflect percent of schedule completion by time interval. Schedule shall be updated monthly and three (3) copies furnished to COMPANY
- 9.2 A report every other week showing scheduled progress versus actual progress, giving details of how the Work will be completed in relation to the schedule.
- 9.3 Equipment construction schedule correlated to Work schedule.



6|Page

J-BML Z9-Dec.-M



JB11-29-Dce-11


J-BAL 29-Dec-11

12/20/11



11.0 CONTRACTUAL NOTICES

See the Article tilled "Notices" in the GSA for provisions governing contractual notices. In addition, a copy of all notices to COMPANY shall be sent to:

11,1	Kentucky Utilitics Company address:	820 West Broadway Louisville, KY 40202 Attn: Charles Ransdell (502) 627-3299 <u>charles.ransdell@lge-ku.com</u>
11.2	Contractor's Address:	Beumer Corporation

4435 Main St. Suite 750 Kansas City, MO 64111

12.0 ENTIRE AGREEMENT

This Contract, including all specifications, exhibits and drawings listed in this Contract and the GSA, constitutes the entire agreement between the parties relating to the Work and supersedes all prior or contemporancous oral or written agreements, negotiations, understandings and statements pertaining to the Work or this Contract.

The parties hereto have executed this Contract on the dates written below, but it is effective as of the date first written above.

Kentucky Utilities Company,	
BY: Spe Clement	
NAME (plini): Jae Creme NTS	
TITLE: MANAGUE, COUTCACTS	
DATE:	
Beumer Corporation	SignHere
BY:	Sign Here
NAME (Print): SIMON SHIP	•
TITLE: <u>VICE RESIDENT</u>	
Дать: <u>12 (240):27 2011 -</u>	

-END-

9|Page

ABIK-29-Dec-M

COMMERCIAL SCHEDULE A LABOR HOURLY RATES

Description	Straight Time	Overtime	Premium Time
SUPERINTENDENT	2		· · · · · · · · · · · · · · · · · · ·
FOREMAN			
OPERATOR			
LABORER			
CARPENTER			
WELDER / Millwright / Ironworker			
PIPE FITTER			
ELECTRICIAN			

- 1. The above rates may be used for agreed to Change Orders and Standby Time,
- 2. Except as may be expressly provided otherwise elsewhere in this Contract, the rates set forth above are inclusive of all direct wage rates, fringe benefits, labor allowances, payroll taxes, insurance, small tools which cost Contractor less than \$1,500 per tool, temporary construction facilities, consumables, expendables, overhead profit and all other costs and expenses incurred by Contractor in performing the Work and this Contract.
- 3. The rates will only apply to actual hours worked or standby time, as agreed and attested to by a COMPANY Representative.
- 4. Overtime rates will be paid for hours worked in excess of 8 hours per day when worker is on 5 8 hour day schedule; for hours worked in excess of 10 hours per day when worker is on 4 10 hour day schedule. Overtime will be paid for hours in excess of forty (40) hours per week. All overtime must be pre-authorized by COMPANY
- 5. Premium time will be paid for all hours worked on and the following holidays.

New Years Day Mémorial Day Good Friday Independence Day Labor Day Thanksgiving Christmas Eve Christmas Day

J-BAL 29-Dec-M

COMMERCIAL SCHEDULE B EQUIPMENT RATES

·	
Equipment Description	
Equipment Description	
Air Compressor - 185 CFM	
Pavement Breakers/ Air Drill and Hose	
Up to 9700 Watts	
Pump - 2" Centrifugal w/hoses	
Pump - 3" Centrlfugal w/hoses	
Mud Buggle I Marooka	
Truck - 3/4 ton	-
Truck - Mob./ Demob. Lowboy or Float	
Truck - Mob./ Demob. Rollback or Flatbed	
Welding Rig	
Telehandlers	
Track Loaders - 950/ 953 or equal	
Dozer - up to D-5 or equal	
Dozer -D6 or equal	
Excavator - 304 to 308 or equal	
Excavator- 312 or equal	
Excavator- 322 or equal	
Excavator - 325 or equal	
Excavator - 330 or equal	
Rubber Tire Backhoe	
Skid Steer - Less than 75HP	
Up to 8000#	
Up to 26' - Battery	
Up to 807 500#	
Trl-Axle	
Up to 150 HP Smooth or Sheepsfoot	

AB1/L 29-Dec-M

12/30/11

	newsprannen operation and the second
Plate Compactor	
Light Plant UP to 5000 Watts	
Mat (4 x 16)	

- 1. The above rates may be used for agreed to Change Orders and Standby Time.
- 2. The above rates are inclusive of all maintenance, repairs, fueling, lubrication, spare parts, insurance and taxes associated with the operation of the equipment and overhead and profit.
- 3. The rates are exclusive of all equipment drivers and operators which are set forth in Schedule A Labor Hourly Rates.
- 4.
- 5. Prior approval by COMPANY's Engineer or Representative is required for all equipment rentals in connection with changes in the Scope of Work.

/B/h 29-Dec-11



KENTUCKY UTILITIES COMPANY CONTRACT No. 511453

FGD Byproduct Dewatering System

This "Contract" is a Statement of Work within the meaning of the General Services Agreement (the "GSA") between (i) KENTUCKY UTILITIES COMPANY (formerly E.ON U.S. Services Inc. by name change, hereinafter referred to as "COMPANY"), a Kentucky corporation, and (ii) Westech Engineering, Inc., a <u>California</u> corporation (hereinafter referred to as "Contractor"), whose address is 3625 South West Temple, Salt Lake City, Utab, 84115. This Contract is entered into, effective as of July 21, 2011 (the "Effective Date"), between Kentucky Utilities Company, a Kentucky corporation, an affiliate of LG&E and KU Services Company (hereinafter referred to as "KU" or "Company"), whose address is 820 West Broadway, Louisville, KY 40220, and Contractor. The parties hereto agree as follows:

1.0 GENERAL

Contractor shall perform the following: <u>FGD Byproduct Dewatering System</u> as more specifically described in Articles 2.0 and 3.0 hereof (hereinafter referred to as the "Work") and COMPANY shall compensate the Contractor for the Work, under all the terms and conditions hereof.

2.0 DESCRIPTION OF WORK

- 2.1 Except as otherwise expressly provided herein, Contractor shall supply all labor, supervision, materials, equipment, tools and warehousing, and shall pay all expenses, necessary or appropriate in the performance of the Work.
 - 2.2 No materials containing asbestos shall be supplied or used in the performance of Work.
 - 2.3 Without limitation, Contractor shall meet all requirements set forth in the Lead Construction Standard 29 CFR 1926.62.
 - 2.4 The Work shall include but not be limited to the following:
 - 2.4.1 FGD Byproduct Dewatering System Spec 172150.65.3240

3.0 SPECIFICATIONS, EXHIBITS AND DRAWINGS

All work shall be performed in strict accordance with the following specifications, exhibits and drawings which are incorporated herein by reference.

3.1 Exhibits

EXILIBIT NO.	<u>Revision</u> <u>No.</u>	Тпів
Exhibit No. 2		General Services Agreement "the Standard Terms"
Exhibit No. 3		Schedule of Submittals and Applicable Liquidated Damages
Exhibit No. 4		LG&E AND KU SERVICES COMPANY Contractor/Subcontractor Safety Policy
Exhibit No. 5		LG&E AND KU SERVICES COMPANY Drug Testing Program
Exhibit No. 6		LG&E AND KU SERVICES COMPANY Contractor Code of Conduct
Exhibit No. 7		Daily Activity Report
Exhibit No. 8		Passport Overview
Exhibit No. 9		Change Order/Work Release
Exhibit No. 10		Project Specific Hazard Analysis
Exhibit No. 11		Hazard Mitigation Plan
Exhibit No. 12		Contractor Quality Assurance Closure Form
Exhibit No. 13		Contractor Safety Questionnaire
Exhibit No. 14		Assignment and Assumption Agreement
Exhibit No. 15		Cable Tab Records Submittal Specification
Exhibit No 16		Electronic Submittal Spec for Engineered Dwgs
Exhibit No. 17		Electronic Submittal Spec for Vendor Does
Exhibit No. 18		Clarifications and Exceptions
Exhibit No. 19		Job Performance Monitoring Tool
Exhibit No. 20		Westech Engineering 1010125 dated June 23, 2011

4.0 TEMPORARY FACILITIES AND UTILITIES

4.1 FURNISHED BY CONTRACTOR

Except as otherwise expressly set forth in this Contract, Contractor shall supply, install, properly maintain and remove all temporary facilities and utilities necessary for performance of the Work, including but not limited to:

- 4.1.1 All temporary buildings
- 4.1.2 All sanitary facilities, including janitorial services
- 4.1.3 First-aid facilities
- 4.1.4 Fuels and lubricants
- 4.1.5 Transportation facilities on and off site
- 4.1.6 Communication facilities
- 4.1.7 Compressed air and gases

2|Page

AB112 29-Dec-11

- 4.1.8 Maintenance of Contractor's lay down, storage and work areas and roads within such areas
- 4.1.9 Rigging, scaffolding and all equipment for erection
- 4.1.10 All cranes and other necessary equipment for lifting and moving equipment
- 4.1.11 All small tools
- 4.1.12 All standard expendable or consumable construction items and supplies including welding material and tools
- 4.1.13 Containers, ice, cups for drinking water
- 4.1.14 Cost of unloading, loading and storing all materials, equipment and supplies
- 4.1.15 Electrical power for small tools, etc., required in the actual work areas
- 4.1.16 Dumpsters and waste disposal

The type of facilities, move-in and move-out dates and locations on job sites shall be subject to and in accordance with the review and approval of COMPANY.

5.0 TERM

This Contract shall become effective upon award and continue through Final Completion, subject to the Article titled "Termination at COMPANY's Option" set forth in the attached Standard Terms. COMPANY makes no promise or guarantee as to the amount of Work to be performed under this Contract.

6.0 PERFORMANCE SCHEDULE

6.1 Contractor shall commence performance of the Work on contract award and shall complete Work not later than the dates listed below.



COMPANY intends to assign this Contract to a general contractor (the "General Contractor") who will integrate the Work with work performed by the General Contractor and its other subcontractors into an integrated installed coal combustion residuals transport system (the "System") at the Company's Ghent Generating Station. Alternatively, the Company reserves the right to complete the System itself with the participation by subcontractors. In either case, with respect to the System, the following terms have the meanings assigned them below:

3 Page

12/30/1 M

"Mechanical Completion" means when, with respect to the System, all of the following have occurred: (i) all components of the System have been furnished and installed, checked for alignment, lubrication, rotation, and hydrostatic and pneumatic pressure integrity; (ii) turnover acknowledgement of all components of the System has been given; (iii) all components of the System have been installed, cleaned and statically tested and cold commissioning has been completed; (iv) all initial tests of the System have been completed and passed and all instruments in the System have been calibrated; (v) an initial punch list has been established and mutually agreed upon by Company and General Contractor (or subcontractors, as applicable); (vi) the performance guarantee test procedures for the System have been submitted to and accepted by Company.

**"Commercial Operation" will have been achieved when all of the following have occurred; (i) Mechanical Completion has occurred; (ii) all work with respect to the System is complete except punch list items; (iii) all performance guarantee tests on the System have been successfully completed, and all necessary and desirable system adjustments identified during the start-up and testing process have been made; (iv) all performance guarantees on the System have been simultaneously achieved in a single performance guarantee test; (v) all obligations of General Contractor (or subcontractors, as applicable) required to have been performed as of Commercial Operation have been properly discharged; (vi) final versions of the operating and maintenance manuals approved by Company have been delivered; (vii) all other submitteds required to be submitted prior to or as of Commercial Operation have been submitted; (viii) all special tools have been delivered to Company.

***"<u>Final Completion</u>" will be deemed to have occurred when all of the following have occurred in respect of the System: (i) Commercial Operation has been achieved, (ii) all guarantee tests on the System have been conducted demonstrating compliance with all guarantee points; (iii) all as-built drawings, design documents, station manuals and other documents required to be delivered to Company have been so delivered, (iv) liquidated damages for which General Contractor (or the subcontractors, as applicable) is liable and other amounts owed to Company with respect to the System, if any, have been paid to Company (v) all work with respect to the System has been completed other than work that requires future performance (*e.g.*, warranty work), (vi) General Contractor (or the subcontractors, as applicable)has demobilized from the Work site leaving it clean and free of debris.

- 6.1.2 Contractor shall coordinate Contractor-supplied material requirement with COMPANY representative and deliver materials to the job site as required.
- 6.1.3 Contractor shall arrange or coordinate material storage at the job site or shop to avoid material theft or damage.
- 6.1.4 Contractor shall notify COMPANY's representative at least one full working day prior to working any Saturday, Sunday or COMPANY holiday.
- 6.2 If Company reasonably determines that Contractor has fallen behind its schedule for completing the Work and has reasonable concerns whether Contractor will meet the scheduled dates for the milestones set forth in Section 6.1 above, Company may terminate the Contract for cause.

4)Page

AB1/L 29-Dec-M

7.0 STANDARD TERMS AND CONDITIONS



7.1 ASSIGNMENT AND ASSUMPTION AGREEMENT

Company reserves the right to assign this Contract to the General Contractor pursuant to the Assignment and Assumption Agreement set forth in Exhibit 13 attached hereto and made a part hereof. Upon request of Company, Contractor shall execute an acknowledgement of such an assignment and assumption, provided, that the assignment and assumption shall be effective and enforceable regardless of whether or when Contractor executes such acknowledgement or otherwise expresses any consent thereto. Upon such assignment and assumption, Company shall be released of its obligations under this Contract and Contractor shall look only to the General Contractor to fulfill the obligations designated as being those of Company hereunder. Contractor also acknowledges that General Contractor may in turn assign this Contract to another entity or to the Company and such assignments shall not require any consent from Contractor.

7.2 ORDER OF PRECEDENCE

The Body of this Agreement and the Exhibits hereto are to be considered complementary and what is required by one will be binding as if required by all. Contractor has included within the Contract Price the cost to complete the Work in its entirety and to fulfill its other obligations hereunder. In addition, the Work includes all that should be properly included and all that would be customarily included within the general scope and magnitude of the Work in order to achieve the obligations. The failure to specifically list a requirement in one document, once such requirement is specifically listed in another, does not imply the inapplicability of such requirement and Contractor shall provide or perform, as appropriate, as part of this Agreement all Work or items required to conform to the Technical Specification (Exhibit 1), the other Exhibits and the standards herein contained. In the event of a conflict between the Body of this Agreement and the Exhibits, the Body of this Agreement governs. In the event of a conflict in an Exhibit or between or among Exhibits, the requirement most favorable to Owner will take precedence, except as may be otherwise determined by Owner. Each Modification will take precedence over that part of this Agreement (including, as applicable, any prior Modification) which it supersedes.

5 Page

12/30/11 Ar

8.0 CONTRACTOR DRUG AND ALCOHOL TESTING

8.1 Plant Outage, Plant Project, or Major Construction Work: The work under this Contract is considered "Plant Outage, Plant Project, or Major Construction Work". In accordance with the revised LG&E AND KU SERVICES COMPANY Contractor / Subcontractor Safety Policy effective 1-31-08, <u>all</u> contractor employees working on-site for "Plant Outage, Plant Project, or Major Construction Work" shall be required to complete a negative drug and alcohol <u>pre-test</u> within seven (7) days before reporting to work on-site. The Contractor shall complete random drug and alcohol testing on ten percent (10%) of their on-site workforce assigned to this work site within thirty (30) days of the start of Work and shall test ten percent (10%) subsequently on an <u>on-going monthly basis</u> for the duration of their on-site "project" work assignment. LG&E AND KU SERVICES COMPANY Corporate Health and Safety will be auditing Contractor compliance with these requirements. Testing costs are the responsibility of the Contractor.

9.0 SPECIFIC REPORTING REQUIREMENTS

Contractor shall promptly submit the schedules and reports set forth below:

- 9.1 Critical Path Schedule, as specified by COMPANY. The schedule shall show labor required by time interval and shall reflect percent of schedule completion by time interval. Schedule shall be updated monthly and three (3) copies furnished to COMPANY
- 9.2 A report every other week showing scheduled progress versus actual progress, giving details of how the Work will be completed in relation to the schedule.
- 9.3 Equipment construction schedule correlated to Work schedule.

10.0 COMPENSATION



6 Page

JBAL 29-Doc-4





12/30/11 M



11.0 CONTRACTUAL NOTICES

See the Article titled "Notices" in the Standard Terms for provisions governing contractual notices. In addition, a copy of all notices to COMPANY shall be sent to:

11.1 Kentucky Utilities Company address:

11.2 Contractor's Address:

820 West Broadway Louisville, KY 40202 Attn: Charles Ransdell (502) 627-3299 charles.ransdell@lge-ku.com

Westech Engineering, Inc. 3625 South West Temple Salt Lake City, UT (801) 290-1473 (Tel) (801) 265-1080 (Fax)

8|Page

JBML 29-000-11

12 30/11 M

12.0 ENTIRE AGREEMENT

÷

This Contract, including all specifications, exhibits and drawings listed in this Contract and the Standard Terms, constitutes the entire agreement between the parties relating to the Work and supersedes all prior or contemporateous oral or written agreements, negotiations, understandings and statements perfaining to the Work or this Contract.

The parties hereto have executed this Contract on the dates written below, but it is effective as of the date first written above.

Kentucky Utilities Company
BY: Ine Clemint
NAME (P)int): Joe CL. P. M. ENTS
TITLE: Mgg. CONTRACTS
DATE: 8/15/11

Westech Engineering, Inc

ВY:	Ciaign. mart
NAME (Print):	Grag N. Martin
ŢITLE:	Vice President
DATE:	11 August 2011

-END-

12/30/11 ar.

COMMERCIAL SCHEDULE A LABOR HOURLY RATES

Description	Straight	Overtime	Premium
	Time		Time
Field Service Technician*			
	an in an		

- 1. The above rates may be used for agreed to Change Orders and Standby Time.
- 2. Except as may be expressly provided otherwise elsewhere in this Contract, the rates set forth above are inclusive of all direct wage rates, fringe benefits, labor allowances, payroll taxes, insurance, small tools which cost Contractor less than \$1,500 per tool, temporary construction facilities, consumables, expendables, overhead profit and all other costs and expenses incurred by Contractor in performing the Work and this Contract.
- 3. The rates will only apply to actual hours worked or standby time, as agreed and attested to by a COMPANY Representative.
- 4. Overtime rates will be paid for hours worked in excess of 8 hours per day when worker is on 5 8 hour day schedule; for hours worked in excess of 10 hours per day when worker is on 4 10 hour day schedule. Overtime will be paid for hours in excess of forty (40) hours per week. All overtime must be pre-authorized by COMPANY
- 5. Premium time will be paid for all hours worked on and the following holidays.

New Years Day Memorial Day Good Friday Independence Day Labor Day Thanksglving Christmas Eve Christmas Day

JBHL 29-Dec-11

ASSIGNMENT AND ASSUMPTION AGREEMENT

This Assignment and Assumption Agreement (this "<u>Agreement</u>") is made and entered into this 29th day of December, 2011, by and between Kentucky Utilities Company, a Kentucky corporation ("<u>Assignor</u>"), and TIC – The Industrial Company, a Delaware corporation ("<u>Assignee</u>") (collectively, the "<u>Parties</u>" and individually a "<u>Party</u>").

WHEREAS, Assignor and Assignee are parties to an agreement dated as of 29 December, 2011 (as amended, modified and supplemented from time to time, the "<u>Agreement</u>"); and

WHEREAS, Assignor has entered into a Contract numbered 511453 dated as of July 21, 2011 (as amended, modified and supplemented from time to time, the "Equipment Contract") with Westech Engineering, Inc., a California corporation (the "Equipment Manufacturing Supplier"), for the supply a FGD Byproduct Dewatering system; and

WHEREAS, in accordance with the terms of the Agreement, Assignor desires to assign to Assignee, and Assignee desires to assume from Assignor, Assignor's rights and obligations under the Equipment Contract as and to the extent set forth in this Agreement; and

NOW, THEREFORE, in consideration of the mutual covenants herein contained, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

ARTICLE 1

ASSIGNMENT AND ASSUMPTION

Section 1.1 Effective as of the date hereof, Assignor hereby assigns to Assignee all of its right, title and interest in and to the Equipment Contract, and Assignee hereby assumes all of Assignor's obligations under the Equipment Contract.

1

RRP 1/6/12

12/30/11

ARTICLE II MISCELLANEOUS PROVIŠIONS

Section 2.1 This Agreement may not be amended, modified or supplemented except pursuant to an instrument in writing signed by the Parties. The failure of any Party to enforce any of the terms and conditions or to exercise any right or privilege under this Agreement shall not be construed as waiving any such term or condition or right or privilege and the same shall continue and remain in force and effect as if no such failure to enforce or exercise has occurred. No waiver shall be valid unless so stated in writing by the Party granting the waiver.

Section 2.2 If any provision of this Agreement is held invalid or unenforceable, all other provisions will not be affected, and with respect to the provision held invalid or unenforceable, the Parties will amend this Agreement as necessary to affect the original intent of the Parties as closely as possible.

Section 2.3 This Agreement shall be governed by and construed in accordance with the laws of the Commonwealth of Kentucky without regard to its rules of conflict of laws that would require the application of laws of a different jurisdiction.

Section 2.4 This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same agreement and shall become effective when counterparts have been signed by each of the Parties and delivered to the other Parties. Any counterpart may be executed by facsimile signature or any image transmitted by electronic mail (such as a pdf file) and such facsimile signature or image shall be deemed an original.

Signatures on Next Page

BK

12 (30/11) ZEP MA 1/6/12 IN WITNESS WHEREOF, each of the undersigned has caused this Agreement to be duly signed as of the date above written.

KENTUCKY UTILITIES COMPANY

d By: Name: Cherrie C Title: Contract Admint Strapel

TIC - The Industrial Company (ASSIGNEE)

By: Name: VILE PREGIDEN SUNY Lee Title:

ACKNOWLEDGEMENT:

Westech Engineering, Inc., a California corporation, hereby acknowledges and agrees as of the date hereof that the Equipment Contract has been assigned and assumed pursuant to the terms hereof.

Westech Engineering, Inc.

1213011

By: Kex 245 Name: Title: Pres!

ASSIGNMENT AND ASSUMPTION AGREEMENT

This Assignment and Assumption Agreement (this "<u>Agreement</u>") is made and entered into this 29th day of December, 2011, by and between Kentucky Utilities Company, a Kentucky corporation ("<u>Assignor</u>"), and TIC – The Industrial Company, a Delaware corporation ("<u>Assignee</u>") (collectively, the "<u>Parties</u>" and individually a "<u>Party</u>").

WHEREAS, Assignor and Assignee are parties to an agreement dated as of 29 December, 2011 (as amended, modified and supplemented from time to time, the "<u>Agreement</u>"); and

WHEREAS, Assignor has entered into a Contract numbered 517454 dated as of October 31, 2011 (as amended, modified and supplemented from time to time, the "Equipment <u>Contract</u>") with Diamond Power International, Inc., a Delaware corporation (the "Equipment Manufacturing Supplier"), for the supply of a Bottom Ash Dewatering and Fly Ash Conveying system; and

WHEREAS, in accordance with the terms of the Agreement, Assignor desires to assign to Assignee, and Assignee desires to assume from Assignor, Assignor's rights and obligations under the Equipment Contract as and to the extent set forth in this Agreement; and

NOW, THEREFORE, in consideration of the mutual covenants herein contained, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

ARTICLE 1 ASSIGNMENT AND ASSUMPTION

Section 1.1 Effective as of the date hereof, Assignor hereby assigns to Assignee all of its right, title and interest in and to the Equipment Contract, and Assignee hereby assumes all of Assignor's obligations under the Equipment Contract.

ARTICLE II MISCELLANEOUS PROVISIONS

Section 2.1 This Agreement may not be amended, modified or supplemented except pursuant to an instrument in writing signed by the Parties. The failure of any Party to enforce any of the terms and conditions or to exercise any right or privilege under this Agreement shall not be construed as waiving any such term or condition or right or privilege and the same shall continue and remain in force and effect as if no such failure to enforce or exercise has occurred. No waiver shall be valid unless so stated in writing by the Party granting the waiver.

Section 2.2 If any provision of this Agreement is held invalid or unenforceable, all other provisions will not be affected, and with respect to the provision held invalid or unenforceable, the Parties will amend this Agreement as necessary to affect the original intent of the Parties as closely as possible.

Section 2.3 This Agreement shall be governed by and construed in accordance with the laws of the Commonwealth of Kentucky without regard to its rules of conflict of laws that would require the application of laws of a different jurisdiction.

Section 2.4 This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same agreement and shall become effective when counterparts have been signed by each of the Parties and delivered to the other Parties. Any counterpart may be executed by facsimile signature or any image transmitted by electronic mail (such as a pdf file) and such facsimile signature or image shall be deemed an original.

Signatures on Next Page

12/30/1

IN WITNESS WHEREOF, each of the undersigned has caused this Agreement to be duly signed as of the date above written.

KENTUCKY UTILITIES COMPANY

che By: DOA for Name: charles Romadell Title: Comt rout Ad netotor

TIC - The Industrial Company (ASSIGNEE) By: Náme: Title:

ACKNOWLEDGEMENT:

2 (30)11 MA Diamond Power International, Inc., a Delaware corporation, hereby acknowledges and agrees as of the date hereof that the Equipment Contract has been assigned and assumed pursuant to the terms hereof.

Diamond Power International, Inc.

By: ESTEL Name: // re A MERCIAL MANAZER Title:



January 18, 2012

Simon Shipp

Beumer Corp.

4435 Main Street Suite 750

Kansas City, MO 64111

Re: Assignment and Assumption Agreement

LG&E KU Services Company Project Engineering 820 W. Broadway Louisville, KY 40202 www.lge-ku.com

Charles Ransdell Contract Administrator T 502-627-3299 F 502-217-2858 M 502-751-2598 charles.ransdell@lge-ku.com

Simon,

The attached document reflects the document I am placing in the mail to your attention for signature on all three (3) originals, and return of two (2) to my attention.

Let me know when you have the signatures secured and when the pertinent documents are returned.

Happy New Year.

Regards,

Respectfully,

Charles Ransdell [/] Contract Administrator Project Engineering

ASSIGNMENT AND ASSUMPTION AGREEMENT

This Assignment and Assumption Agreement (this "<u>Assignment</u>") is made and entered into this 30th day of December, 2011, by and between Kentucky Utilities Company, a Kentucky corporation ("<u>Assignor</u>"), and TIC – The Industrial Company, a Delaware corporation ("<u>Assignee</u>") (collectively, the "<u>Parties</u>" and individually a "<u>Party</u>").

WHEREAS, Assignor and Assignee are parties to an agreement dated as of 30 December, 2011 (as amended, modified and supplemented from time to time, the "<u>Agreement</u>"); and

WHEREAS, Assignor has entered into a Contract numbered 514629 dated as of October 7, 2011 (as amended, modified and supplemented from time to time, the "<u>Equipment</u> <u>Contract</u>") with Beumer Corporation, a Delaware corporation (the "Equipment Manufacturing Supplier"), for the supply of a Pipe Conveyor system; and

WHEREAS, in accordance with the terms of the Agreement, Assignor desires to assign to Assignee, and Assignee desires to assume from Assignor, Assignor's rights and obligations under the Equipment Contract as and to the extent set forth in this Assignment; and

NOW, THEREFORE, in consideration of the mutual covenants herein contained, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties agree as follows:

ARTICLE 1 ASSIGNMENT AND ASSUMPTION

Section 1.1 Effective as of the date hereof, Assignor hereby assigns to Assignee all of its right, title and interest in and to the Equipment Contract, and Assignee hereby assumes all of Assignor's obligations under the Equipment Contract.

ARTICLE II MISCELLANEOUS PROVISIONS

Section 2.1 This Assignment may not be amended, modified or supplemented except pursuant to an instrument in writing signed by the Parties. The failure of any Party to enforce any of the terms and conditions or to exercise any right or privilege under this Assignment shall not be construed as waiving any such term or condition or right or privilege and the same shall continue and remain in force and effect as if no such failure to enforce or exercise has occurred. No waiver shall be valid unless so stated in writing by the Party granting the waiver.

Section 2.2 If any provision of this Assignment is held invalid or unenforceable, all other provisions will not be affected, and with respect to the provision held invalid or unenforceable, the Parties will amend this Assignment as necessary to affect the original intent of the Parties as closely as possible.

Section 2.3 This Assignment shall be governed by and construed in accordance with the laws of the Commonwealth of Kentucky without regard to its rules of conflict of laws that would require the application of laws of a different jurisdiction.

Section 2.4 This Assignment may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same agreement and shall become effective when counterparts have been signed by each of the Parties and delivered to the other Parties. Any counterpart may be executed by facsimile signature or any image transmitted by electronic mail (such as a pdf file) and such facsimile signature or image shall be deemed an original.

Signatures on Next Page

IN WITNESS WHEREOF, each of the undersigned has caused this Assignment to be duly signed as of the date above written.

KENTUCKY UTILITIES COMPANY

B Ciement) Title: Mappager CONTRACTI

(ASSIGNEE) By:

Name: Sung Lee Title: Vice President

TIC – The Industrial Company

ACKNOWLEDGEMENT:

Beumer Corporation, a Delaware corporation, hereby acknowledges and agrees as of the date hereof that the Equipment Contract has been assigned and assumed pursuant to the terms hereof.

Beumer Corporation

By:

Name: SIMON SHIAP Title: VICE PRESIDENT.

KENTUCKY UTILITIES COMPANY

FINANCIAL EXHIBIT (807 KAR 5:001 SEC. 6)

March 31, 2012

(1) Amount and kinds of stock authorized.

80,000,000 shares of Common Stock, without par value 5,300,000 shares of Cumulative Preferred Stock, without par value -- authorized, but unissued 2,000,000 shares of Preferred Stock, without par value -- authorized, but unissued

(2) Amount and kinds of stock issued and outstanding.

Common Stock: 37,817,878 shares issued and outstanding, without par value, recorded at \$307,818,689.

(3) Terms of preference of preferred stock whether cumulative or participating, or on dividends or assets otherwise.

None

1992/199

9000000

117 - ANSI

enderstelserser

15

(4) Brief description of each mortgage on property of applicant, giving date of execution name of mortgagor, name of mortgagee, or trustee, amount of indebtedness authorized to be secured thereby, and the amount of the indebtedness actually secured, together with any sinking fund provisions.

Date of Execution: As of October 1, 2010 (Supplemental Indentures were executed on October 15, 2010 and November 1, 2010.) Mortgagor: Kentucky Utilities Company Trustee: The Bank of New York Mellon Amount of Authorized Debt: One quintillion dollars Amount of Debt Secured: \$1,850,779,405 Sinking Fund Provisions: None Pledged Assets: Substantially all assets of Kentucky Utilities located in Kentucky

ļ

(5) Amount of bonds authorized, and amount issued giving the name of the public utility which issued the same, describing each class separately, and giving date of issue, face value, rate of interest, date of maturity and how secured, together with an amount of interest paid thereon during the last 12-month period.

Secured by first mortgage lien on substantially all assets in Kentucky.

Kentucky Utilities Company

			Principal	Interest Expense					
				Year Ended					
Date of	Date of	Rate of		March 31,	March 31,				
Issue	Maturity	Interest	Authorized	2012	2012				
Pollution Control Bonds									
05/01/00	05/01/23	Variable	\$ 12,900,000	\$ 12,900,000	\$ 19,922				
02/01/02	02/01/32	Variable	20,930,000	20,930,000	150,526				
02/01/02	02/01/32	Variable	2,400,000	2,400,000	16,905				
02/01/02	02/01/32	Variable	7,200,000	2,400,000	207,152				
02/01/02	02/01/32	Variable	7,400,000	7,400,000	52,094				
07/01/02	10/01/32	Variable	96,000,000	96,000,000	16,895				
10/01/04	10/01/34	Variable	50,000,000	50,000,000	81,942				
02/23/07	10/01/34	Variable	54,000,000	54,000,000	89,148				
05/24/07	02/01/26	5.75%	17,875,000	17,875,000	1,027,813				
05/24/07	03/01/37	6.00%	8,927,000	8,927,000	535,620				
10/17/08	02/01/32	Variable	77,947,405	77,947,405	128,318				
			\$ 355,579,405	\$ 350,779,405	\$ 2,326,335				
First Mortga	ge Bonds								
11/16/10	11/01/15	1.63%	\$ 250,000,000	\$ 250,000,000	\$ 4,062,500				
11/16/10	11/01/20	3.25%	500,000,000	500,000,000	16,250,000				
11/16/10	11/01/40	5.13%	750,000,000	750,000,000	38,437,500				
			\$ 1,500,000,000	\$ 1,500,000,000	\$ 58,750,000				

1

(6) Each note outstanding, giving date of issue, amount, date of maturity, rate of interest in whose favor, together with amount of interest paid thereon during the last 12-month period.

There are no notes outstanding as of March 31, 2012.

(7) Other indebtedness, giving same by classes and describing security, if any with a brief statement of the devolution or assumption of any portion of such indebtedness upon or by person or corporation if the original liability has been transferred, together with amount of interest paid thereon during the last fiscal year.

None, other than current and accrued liabilities.

(8) Rate and amount of dividends paid during the five previous fiscal years, and amount of capital stock on which dividends were paid. (1)

Dividends on Common Stock, without par value (not based on rate per share)

2007	-
2008	-
2009	-
2010	\$50,000,000
2011	\$123,500,000

-12

(1) On November 1, 2010, PPL Corporation completed its acquisition of E.ON U.S. LLC, the Company's parent. Upon completion of the acquisition, E.ON U.S. LLC was renamed LG&E and KU Energy LLC. The 37,817,878 shares are currently owned by LG&E and KU Energy LLC. From May 1998 to October 31, 2010, the 37,817,878 shares were all owned by E.ON U.S. LLC (formerly LG&E Energy LLC) and all dividends declared by KU's Board of Directors were paid to E.ON U.S. LLC. Subsequent to October 31, 2010, all dividends declared by KU's Board of Directors were paid to LG&E and KU Energy LLC. During the 1st quarter of 2012 KU declared and paid a dividend on common stock of \$24,000,000.

(9) Detailed Income Statement, Balance Sheet and Statement of Retained Earnings

Monthly Financial and Operating Reports are filed each month with the Kentucky Public Service Commission. Attached are detailed Statements of Income, Balance Sheets and Retained Earnings for the Company for the period ending March 31, 2012.

Kentucky Utilities Company Balance Sheet as of March 31, 2012

Liabilities and Proprietary Capital

\$ 5,196,733,225.04		431,466,892.94	45,907,397.34	85,241,359.67	11,623,874.20	20,993,395.72		339,849,758.04	886.52	5,995,929.86	10,207,802.39 415,494,53	34,275,059.84	86,500,323.28		3.237.051.00	153.471.798.55	-	26,696,148.67		6,894,316.49	179,120.94	250,000.00	6,465,195.55		4,418,522,257.57	2,419,286,203.27	6,837,808,460.84	
Total Liabilities and Stockholders Equity	Total	Miscellaneous Long-1 etriti Liautituss Accum Provision for Postretirement Benefits	Other Deferred Credits	Asset Retirement Obligations	Customer Advances for Construction	Accumulated Deterred Income Laws	Deferred Credits and Other		Total	Miscellaneous Current and Accrued Liabilities	Dividends Declared	Therest Approach	Customer Deposits	Accounts Payable to Associated Companies	Accounts Payable	ST Notes Payable to Associated Companies	Current and Accrued Liabilities	Total Capitalization	Total Long-Term Debt	LT Notes Payable to Associated Companies	First Mortgage Bonds	nellection Control Bonde	Total Proprietary Capital	Unappropriated Undistributed Subsidiary Earnings	Other Comprehensive Income	Less: Common Stock Expense Paid-In Capital	Proprietary Capital Common Stock	
\$ 5,196,733,225.04	1,007,191,259.08	134,724,221.36	12,482,304.43 2,630,529.78	62,573,225.51	3,147,887.16	100,707,740.58 108,999,483.67	581,925,806.57		210,306,840.91	10,000,000	י.	26,028,639.20	15,423,195.66	23,057,677.96	03,030,JIII.20 35 561 734 97	00 550 513 JR		3,979,235,125.05	1,840,750,373.75		1,489,970,968.75	350,779,405.00	2,138,484,751.30	14,468,538.75	(5,681,776.49) 1,506,021,217.35	315,858,083.00	\$ 308,139,977.56 321,288.87	

\$ 5,196,733,225.04	Total Assets
431,466,892.94	Total
20,993,395.72 11,623,874.20 85,241,359.67 267,700,866.01 45,907,397.34	Deferred Debits and Other Unamortized Debt Expense Unamortized Loss on Bonds Accumulated Deferred Income Taxes Deferred Regulatory Assets Other Deferred Debits
339,849,758.04	Total
886.52	Prepayments Miscellaneous Current and Accrued Assets
415,494.53	Emission Allowances
34,275,059.84 10,207,802.39	Plant Materials and Operating Supplies
86,500,323.28	Materials and Supplies-At Average Cost
133,47,1,790,22 3,237,051.00	Accounts Receivable-Less ReserveAccounts Receivable from Associated Companies
19,049,263.40	Special Deposits Temporary Cash Investments
26,696,148.67	Current and Accrued Assets Cash
6,894,316.49	Total
6,465,195.55 250,000.00 179,120.94	Investments Electric Energy, Inc Ohio Valley Electric Company Nonutility Property-Less Reserve
4,418,522,257.57	Total

Page 4 of 6

Assets

Utility Plant Utility Plant at Original Cost...... Less: Reserves for Depreciation and Amortization......

ŝ

-

Kentucky Utilities Company Statement of Income March 31, 2012

-

	Year Ended 3/31/2012
Electric Operating Revenues	\$ 1,522,035,957.11
Rate Refunds	47
Total Operating Revenues	1,522,035,957.11
Fuel for Electric Generation	516,817,356.17
Power Purchased	103,829,269.81
Other Operation Expenses	234,009,083.64
Maintenance	126,205,679.78
Depreciation	184,687,593.56
Amortization Expense	7,505,149.44
Regulatory Credits	(6,011,854.42)
Taxes	(, , ,
Federal Income	(17,247,753.46)
State Income	1,899,319.18
Deferred Federal Income - Net	102,811,975.59
Deferred State Income - Net	11,097,503.08
Property and Other	29,144,074.49
Investment Tax Credit	
Loss (Gain) from Disposition of Allowances	(886.52)
Accretion Expense	2,934,108.65
Accretion Expense	
Total Operating Expenses	1,297,680,618.99
Net Operating Income	224,355,338.12
Other Income Less Deductions	
Amortization of Investment Tax Credit	2,800,110.74
Other Income Less Deductions	533,585.57
AFUDC - Equity	47,125.06
Total Other Income Less Deductions	3,380,821.37
Income Before Interest Charges	227,736,159.49
Interest on Long-Term Debt	61,076,334.12
Amortization of Debt Expense - Net	3,773,938.89
Other Interest Expenses	5,107,381.64
AFUDC - Borrowed Funds	(13,892.09
Total Interest Charges	69,943,762.56
Net Income	\$ 157,792,396.93

Kentucky Utilities Company Analysis of Retained Earnings March 31, 2012

÷

÷

4

	Y	ear Ended 3/31/12
Retained Earnings Balance at Beginning of Period	\$	1,463,485,376.42
Net Income for Period		157,792,396.93
Deduct:		101,192,0000
Common Dividends		
Common Stock Without Par Value		(116,500,000.00)
Adjust for Equity in Subsidiary		
Earnings for Year EE Inc		1,243,444.00
Dividends Received Current Year		
-EE Inc		-
Retained Earnings Balance at End of Period		1,506,021,217.35
Unappropriated Undistributed Subsidiary Earnings		14,468,538.75
Retained Earnings and Undistributed Subsidiary Earnings at End of Period	\$	1,520,489,756.10

.

Exhibit 6 Page 1 of 7

SECRETARY'S CERTIFICATE

I, Gerald A. Reynolds, do hereby certify that I am the duly qualified and acting General Counsel, Chief Compliance Officer and Corporate Secretary of Kentucky Utilities Company, a Kentucky and Virginia corporation, (the "Company"), and that as such officer, I have access to all original records of the Company and that I am authorized to make certified copies of Company records on its behalf. I further hereby certify that the attached resolutions are a true and correct copy of the resolutions of the Company adopted by unanimous written consent effective as of June 5, 2012, and that the same have not been altered, amended or repealed.

IN WITNESS WHEREOF, I have executed this Certificate this \underline{SH} day of June 2012.

Gerald A' Reynolds General Counsel, Chief Compliance Officer And Corporate Secretary

ACTION OF THE BOARD OF DIRECTORS OF KENTUCKY UTLITIES COMPANY TAKEN BY WRITTEN CONSENT IN LIEU OF A SPECIAL MEETING

June 5, 2012

ISSUANCE OF FIRST MORTGAGE BONDS

WHEREAS, the Board of Directors of the Company (the "Board") has determined that it is desirable and in the best interests of the Company to issue up to \$300,000,000 aggregate principal amount of long-term debt in the form of first mortgage bonds for the purposes of providing funds for anticipated capital expenditures, operational or financial needs and other general corporate purposes.

NOW, THEREFORE, BE IT RESOLVED, by the Board as follows:

The Offering

- (a) That the Board authorizes and approves the issuance and sale by the Company from time to time, in one or more series, and in any combination, of up to \$300,000,000 aggregate principal amount of long-term debt in the form of first mortgage bonds (any of such bonds, the "Debt Securities") in one or more underwritten public offerings, negotiated sales, or private placement transactions (such offerings, sales and transactions collectively referred to herein as the "Offering"), the net proceeds of such Debt Securities to be used for general corporate purposes, including construction and other capital expenditures, operational funding requirements, and repayment, refunding or refinancing of short- or long-term debt at maturity or otherwise.
- (b) That the Company be, and it hereby is, authorized to issue and offer for sale the Debt Securities through or to one or more underwriters, selling or placement agents, or other purchasers pursuant to an underwriting, purchase or similar agreement, on and subject to such terms and conditions as may be approved by the Authorized Officers (as defined below), provided that the interest rate on such Debt Securities shall not exceed 6-6.5% per annum.

Debt Securities or First Mortgage Bonds

(density)

international de la constante d

a second

a alata

- (c) That the Chief Executive Officer, President, Chief Financial Officer, Chief Administrative Officer, General Counsel, Chief Compliance Officer and Corporate Secretary, any Vice, President, the Treasurer, and the Controller of the Company (each, an "Authorized Officer" and, collectively, the "Authorized Officers") are, and each of them hereby is, authorized by and on behalf of the Company, to negotiate, enter into, execute and deliver one or more supplemental indentures, company orders and/or officer's certificates (the "Supplemental Indentures") pursuant to the Company's Indenture dated October 1, 2010 to The Bank of New York Mellon, as trustee (such Indenture, as heretofore supplemented and as to be further supplemented and amended by any such instrument the "Indenture") relating to the creation and issuance of, and establishing the designation, form, characteristics and terms of the Debt Securities, in such form or forms and having such terms as the Authorized Officers executing the same shall approve, and to perform all of the agreements and obligations of the Company under the Supplemental Indentures and Indenture and to consummate the transactions contemplated thereby; and that each Authorized Officer be, and hereby is, authorized to execute and deliver such other agreements, certificates and documents and to take such other actions in connection with the execution and delivery of any Supplemental Indenture or other instrument pursuant to the Indenture as such Authorized Officers deem necessary, advisable or appropriate; with such changes therein, additions thereto or omissions therefrom, as any Authorized Officer executing, acknowledging or delivering the same shall approve, such Authorized Officer's execution. acknowledgement and/or delivery thereof to be conclusive evidence of such approval.
- (d) That the Authorized Officers are, and each of them hereby is, authorized, empowered and directed, in the name and on behalf of the Company, to execute, acknowledge and deliver new securities representing the Debt Securities in substantially such form and containing such terms and conditions as such Authorized Officer shall approve, with such changes therein, additions thereto or omissions therefrom as such Authorized Officer executing, acknowledging or delivering the same shall approve, such Authorized Officer's execution, acknowledgement and delivery thereof to be conclusive evidence of such approval.
- (e) That the Authorized Officers are, and each of them hereby is, authorized, empowered and directed to fix and approve the terms and conditions on which the Debt Securities are to be issued and authenticated and the final terms of the Supplemental Indentures or any other instrument pursuant to the Indenture, including, without limitation, the rights of the holders thereof, the interest rate or rates, the maturity date or dates, the sinking fund, redemption or repurchase provisions and prices, the purchase price or prices, the offering date and terms and all other matters relating thereto, and to take all such other actions as any Authorized Officer deems necessary, advisable or appropriate to consummate the transactions contemplated by the Supplemental Indentures.
- (f) That a facsimile of the corporate seal of the Company may be imprinted on the Supplemental Indentures and/or Debt Securities, which facsimile is hereby acknowledged to be the corporate seal of the Company for the purposes of sealing the Debt Securities.
- (g) That the Authorized Officers are, and each of them hereby is, authorized to execute and deliver on behalf of the Company, in the event that all or a portion of the Debt Securities

Exhibit 6 Page 4 of 7

bear a fixed or variable rate of interest: (i) one or more interest rate lock or swap agreements or similar agreements with one or more underwriters, banks or other financial institutions or other counter-parties, including affiliated entities, providing for the hedging of the interest rates on such securities, and (ii) any other agreement, document or instrument that may be necessary or appropriate in connection with any such transaction.

Paying Agent and Security Registrar

(h) That The Bank of New York Mellon is hereby appointed to act as the initial paying agent and security registrar for the Debt Securities described herein in accordance with the provisions of the Indenture; provided that any Authorized Officer may take all actions necessary or desirable, on behalf of the Company, to provide for any additional or different paying agent or security registrar for any Debt Securities, if such Authorized Officer deems such provision to be desirable, such officer's determination to be conclusively evidenced by his execution of documentation effecting such appointment or change.

Offering Documents

: ; (i) That, in connection with the issuance and sale of the Debt Securities, the Authorized Officers are, and each of them hereby is, authorized, empowered and directed, in the name and on behalf of the Company to: (i) prepare, or cause to be prepared, one or more prospectuses, offering memoranda or other appropriate disclosure documents (including all exhibits, annexes and other documents relating thereto) in connection with such issuance and sale of the Debt Securities, including any supplement(s) or amendment(s) thereto (the "Offering Memorandum"); (ii) execute, as such Authorized Officers or Authorized Officer deem(s) necessary, advisable or appropriate, any and all, agreements, documents and instruments in connection with such issuance and sale; and (iii) take all such other actions as such Authorized Officers or Authorized Officer deem(s) necessary, advisable or appropriate of the Debt Securities, such Authorized Officer so issuance and sale; and (iii) take all such other actions as such Authorized Officers or Authorized Officer deem(s) necessary, advisable or appropriate in order to effect the issuance and sale of the Debt Securities, such Authorized Officer's authority and determination to execute such documents and instruments and to take such actions being conclusively evidenced by such execution or action, as the case may be.

Authorization of Underwriting, Purchase or Similar Agreements

- (j) That the Authorized Officers are, and each of them hereby is, authorized and directed, for and on behalf of the Company, to fix and approve the terms of an underwriting, purchase or similar agreement relating to the issuance and sale of the Debt Securities to one or more underwriters, selling or placement agents or other purchasers thereof (the "Purchase Agreement") to be entered into by and among the Company and such underwriters, agents or purchasers as may agree to become parties thereto, and the Authorized Officers be, and each of them hereby is, authorized to execute and deliver the same, in such form or with such changes therein, additions thereto or omissions therefrom as such Authorized Officers may approve, such execution and delivery by any such Authorized Officer to be conclusive evidence of such authorization and approval.
- (k) That the Authorized Officers are, and each of them hereby is, authorized to execute and file such instruments, make all such payments, and do such other acts and things as, in the opinion of any Authorized Officer, may be necessary or desirable in order to comply

Exhibit 6 Page 5 of 7

with the rules and regulations promulgated under the Securities Act of 1933, as amended; and to qualify the Company or any Debt Securities under the securities or "Blue Sky" laws of such states of the United States and other jurisdictions as may be necessary or desirable, and to take further necessary action for said purposes.

Regulatory Approval

(I) That the Authorized Officers of the Company are, and each of them hereby is, authorized and empowered to execute and file, or cause to be filed, on behalf of the Company, such applications, petitions or notices (including amendments or supplements thereto) with the Public Service Commission of the Commonwealth of Kentucky, the State Corporation Commission of the Commonwealth of Virginia, the Tennessee Regulatory Authority and any other federal, state, or local commission, court, agency or body having jurisdiction as may be required to obtain any approvals, consents, orders or rulings as such officers or counsel for the Company may deem to be necessary or desirable in connection with the transactions contemplated hereby, as may be required by law or as may be deemed to be proper or appropriate in their judgment or in the judgment of counsel for the Company in connection with the foregoing.

General

•

ļ

- (m) That the Authorized Officers of the Company are, and each of them is, hereby authorized and empowered, in the name and behalf of the Company, to execute and deliver such agreements and other documents relating to electronic deposit and delivery, cash management, information services and such other matters as they shall deem necessary or desirable to otherwise facilitate the offering, issuance, sale and delivery of the Debt Securities and receive and apply the proceeds therefrom.
- (n) That the officers of the Company are, and each of them hereby is, authorized in the name and on behalf of the Company and under its corporate seal or otherwise, to take or cause to be taken all such further actions and to execute and deliver or cause to be executed and delivered all such further instruments, agreements, certificates and other documents in connection with the Offering as such persons may deem necessary, advisable or appropriate in connection with the transactions contemplated thereby and hereby, and to incur all such fees and expenses as shall be necessary, advisable or appropriate in their judgment in order to carry into effect the purpose and intent of any and all of the foregoing resolutions.
- (o) That any acts of the officers of this Company, which acts would have been authorized by the foregoing resolutions except that such acts were taken prior to the adoption of such resolutions, are hereby severally ratified, confirmed, approved and adopted as acts in the name of and on behalf of this Company.
- (p) That the Authorized Officers are, and each of them hereby is, authorized and directed to take any and all further action to see that the intent of the above resolutions is carried forth.

This unanimous written consent may be executed in two or more counterparts, all of which taken together shall be deemed one and the same instrument.

Exhibit 6 Page 6 of 7

ACTION OF THE BOARD OF DIRECTORS OF KENTUCKY UTILITIES COMPANY TAKEN BY WRITTEN CONSENT IN LIEU OF A SPECIAL MEETING

June 5, 2012

EXPANSION OF REVOLVING CREDIT FACILITIES

WHEREAS, the Company is borrower under a \$400 million Revolving Credit Agreement, dated as of November 1, 2010, among the Company, the Lenders party thereto and Wells Fargo Bank, National Association, as administrative agent, as amended (the "Existing Credit Agreement"), which currently is scheduled to expire in October 2016, subject to the ability to expand the facility by up to \$100 million at the Company's request, and subject to the consent of the lenders; and

WHEREAS, the Company desires to increase the amount of borrowings available under its revolving credit facilities to a total aggregate amount of up to \$500 million, through modification of the Existing Credit Agreement or by entering into additional revolving credit facilities having similar terms, such expansions or alternative facilities having a combined aggregate principal amount not to exceed \$100 million (collectively, the "Revolving Credit Facilities"); and

WHEREAS, the Board has determined that it is in the Company's best interests to amend or modify, as appropriate, the Existing Credit Agreement or negotiate and enter into additional revolving credit agreements, as appropriate, so as to effect the Revolving Credit Facilities; and

NOW, THEREFORE, BE IT RESOLVED, by the Board as follows:

Credit Agreements or Amendments

(a) That the Chief Executive Officer, President, Chief Financial Officer, the Chief Administrative Officer, the General Counsel, Chief Compliance Officer and Corporate Secretary, any Vice President, the Treasurer, and the Controller of the Company (each, an "Authorized Officer" and, collectively, the "Authorized Officers") be, and each of them hereby is, authorized by and on behalf of the Company, authorized and empowered to negotiate, execute and enter into, on behalf of the Company, such forms of amended, modified, replacement or new promissory notes or credit agreements, whether relating to the Existing Credit Agreement or to new credit arrangements, with existing or new banks and financial institutions, that such Authorized Officer deems necessary or desirable to document and effect the Revolving Credit Facilities; together with such other agreements, instruments, notices, certificates and documents, on such ferms and

Exhibit 6 Page 7 of 7

conditions as the officer executing such documents deems appropriate, with such officer's execution of a definitive agreement to conclusively evidence such officer's approval and the approval of this Board of Directors.

(b) That the Authorized Officers be, and each of them hereby is, authorized by and on behalf of the Company to: (i) request advances (including issuance of letters of credit) under the Revolving Credit Facilities; (ii) delegate to any other officers or employees of the Company, either acting individually or jointly, authority to request advances (including issuances of letters of credit) under the Revolving Credit Facilities; and (iii) execute and deliver any other agreements and documents and take any and all other action as contemplated by the Revolving Credit Facilities or as such officer may deem necessary or desirable in connection with the making of advances (including issuances of letters of credit) on account of the Company pursuant to the Revolving Credit Facilities.

(c) That the Authorized Officers be, and each of them hereby is, authorized and directed to cause the preparation of, to approve, or consent to, and execute and deliver the necessary documents, instruments, agreements or certificates necessary to enter into the Revolving Credit Facilities.

General

(d) That the Authorized Officers of the Company be, and each of them hereby is, authorized and empowered to execute and file, or cause to be filed, on behalf of the Company, such applications, petitions or notices (including amendments or supplements thereto) with the Public Service Commission of the Commonwealth of Kentucky, the State Corporation Commission of the Commonwealth of Virginia, the Tennessee Regulatory Authority and any other federal, state, or local commission, court, agency or body having jurisdiction as may be required to obtain any approvals, consents, orders or rulings as such officers or counsel for the Company may deem to be necessary or desirable in connection with the transactions contemplated hereby, as may be required by law or as may be deemed to be proper or appropriate in their judgment or in the judgment of counsel for the Company in connection with the foregoing.

(e) That any and all actions heretofore taken by the Authorized Officers within the terms of the foregoing resolutions, including any actions taken in connection with applications to the Public Service Commission of the Commonwealth of Kentucky, the Virginia State Corporation Commission and the Tennessee Regulatory Authority or any other federal, state, or local commission, court, agency or body having jurisdiction as required to obtain any approvals, consents, orders or rulings as such officers or counsel for the Company deemed to be necessary or desirable in connection with the transactions contemplated hereby, be and the same are hereby in all respects approved, ratified and confirmed.

This unanimous written consent may be executed in two or more counterparts, all of which taken together shall be deemed one and the same instrument.